UNIVERSITY OF BERGEN



Department of Physics and Technology

MASTER OF SCIENCE THESIS IN THEORETICAL PARTICLE PHYSICS

Properties of S_3 -Symmetric Three-Higgs-Doublet Models

Author: Anton Kunčinas Supervisor: Prof. Jörn Kersten Co-supervisor: Prof. Per Osland

Abstract

Some aspects of the S_3 -symmetric three-Higgs-doublet models are analysed. A CP conserving potential with both real and complex vacuum configurations is considered. The S_3 -symmetric potential is presented in the irreducible representation, which the main part of the thesis is based on, and in the Higgs basis. Hidden symmetries of the S_3 -symmetric 3HDM potential are analysed: behaviour of the scalar potential under the subgroups of the U(3) group, which result in Goldstone bosons, and the discrete \mathbb{Z}_2 symmetry required for a stable dark matter candidate. Possible models capable of accommodating the dark matter candidate are presented. Two vacuum configurations, one real and one complex, which can possibly accommodate the dark matter candidate, are further analysed. One of the vacuum configurations results in massless scalars and therefore the concept of soft symmetry breaking is applied. A numerical check of two vacuum configurations is performed based on the theoretical and experimental constraints.

Acknowledgements

I would like to express my very great appreciation to Per Osland for his guidance in shaping of the research, his continuous encouragement, useful critiques of the research, and finding the missing "the". I am particularly grateful for the assistance given by Jörn Kersten and Margarida Nesbitt Rebelo. I would also like to thank Thomas Gajdosik, Darius Jurčiukonis and Vytautas Dūdėnas for discussions.

Material of the thesis was presented during:

- <u>Anton Kuncinas</u>, Per Osland, "Extended Standard Model Higgs Sector: Analysis of the S₃-Symmetric Three-Higgs-Doublet Model", Open Readings, Vilnius, Lithuania, March 2018 (Talk).
- <u>Anton Kuncinas</u>, Per Osland, "Extended Standard Model Higgs Sector: Analysis of the S₃-Symmetric Three-Higgs-Doublet Model", Faggruppemøte for Subatomær Fysikk og Astrofysikk, Bergen, Norway, September 2018 (Talk).
- <u>Anton Kuncinas</u>, Per Osland, "In Search of the Extended Standard Model Higgs Sector Dark Matter: From Theory to Numerical Analysis", Open Readings, Vilnius, Lithuania, March 2019 (Talk).

Abbreviations and Notations

2HDM - Two-Higgs-Doublet Model
3HDM - Three-Higgs-Doublet Model
CKM - Cabibbo–Kobayashi–Maskawa Matrix
CP - Charge-Parity
DM - Dark Matter
FCNC - Flavour-Changing Neutral Currents
IDM - Inert Doublet Model
NHDM - Multi-Higgs-Doublet Model
PMINS - Pontecorvo–Maki–Nakagawa–Sakata Matrix
SM - Standard Model
VEV - Vacuum Expectation Value

The trigonometric functions are denoted by: $s_{\theta} \equiv \sin(\theta),$ $c_{\theta} \equiv \cos(\theta),$ $t_{\theta} \equiv \tan(\theta).$

Note: Lagrangian \equiv Lagrangian density Natural units are used: $\hbar = c = 1$ _____

Contents

\mathbf{A}	bstra	act	i		
A	cknov	wledgements	iii		
\mathbf{A}	bbrev	viations and Notations	v		
1	Introduction				
2	The	$e S_3$ -Symmetric Scalar Potential	3		
	2.1	The Scalar Potential	3		
	2.2	Two Possible Choices: Real and Complex Vacua	4		
		2.2.1 Real Vacua	4		
		2.2.2 Complex Vacua	5		
	2.3	Identifying the Goldstone States	9		
	2.4	Dark Matter Candidates	15		
	2.5	The Higgs Basis Transformation	16		
		2.5.1 Choice of the Higgs Basis	21		
		2.5.2 The Mass-Squared Matrices	22		
3	The	C-III-c Model	25		
Ŭ	3.1	Model With Exact S_3 Symmetry	25		
	0.1	3.1.1 Freedom of the Basis Redefinition	$\frac{-5}{25}$		
		3.1.2 The Mass-Squared Matrices	26		
		3.1.3 The Mass-Squared Matrices in Different Bases	30		
	3.2	Soft Symmetry Breaking	31		
	-	3.2.1 The C-III-c- ν^2 Model	33		
		3.2.2 The C-III-c- μ_2^2 Model	34		
		3.2.3 The C-III-c- ν^2 - μ_2^2 Model	36		
	3.3	CP Violation in the Scalar Sector of the C-III-c Models $\ldots \ldots \ldots \ldots \ldots \ldots$	37		
4	The	$e C-III-c-\nu^2 Model$	41		
	4.1	1 1 Occupite Courdinating Terror of Manage	41		
	4.0	4.1.1 Quartic Couplings in Terms of Masses	44		
	4.2	Scalar-Gauge Boson Interactions	45		
	4.3	Scalar-Fermion Interactions	48		
	4.4	Scalar-Scalar Interactions	50		
	4.5	Constraints	60		
		4.5.1 Constraints From the Scalar Masses	60		
		4.5.2 The Standard Model Limit	61		
		4.5.3 Potential Stability	61		
		4.5.4 Perturbativity	63		
		4.5.5 Tree-Level Unitarity	64		
		4.5.6 Electroweak Oblique Parameters	67		

5	The	R-II-1a Model	69
	5.1	The Mass-Squared Matrices	69
		5.1.1 Quartic Couplings in Terms of Masses	73
	5.2	R-II-1a in the Higgs Basis	74
	5.3	Scalar-Gauge Boson Interactions	75
	5.4	Scalar-Fermion Interactions	75
	5.5	Scalar-Scalar Interactions	78
	5.6	Constraints	84
		5.6.1 Electroweak Oblique Parameters	85
6	Nur	nerical Analysis	87
	6.1	General Approach	87
	6.2	The C-III-c- ν^2 Model	89
	6.3	The R-II-1a Model	94
7	Sum	nmary and Conclusions	99
	7.1	Results	99
	7.2	Future Research	99
\mathbf{A}	Diff	erent Forms of the Scalar Potential	101
	A.1	The Derman Potential	101
	A.2	Matrix Form	102
в	Der	ivatives of the Potential With Respect to the Fields	103
	B.1	The First Derivatives	103
\mathbf{C}	Pote	ential in the Higgs Basis	107
	C.1	$\mathcal{R}_{\beta_z}\mathcal{R}_{\beta_r}$ Higgs Basis Rotation	107
	C.2	$\mathcal{R}_{\beta_y}\mathcal{R}_{\beta_z}$ Higgs Basis Rotation	112
Bi	bliog	raphy	117

Chapter 1

Introduction

The Standard Model (SM) of particle physics has been extensively tested for a few decades. The last missing piece, the Higgs boson, was discovered in 2012 with a combined mass of $m_h = 125.09 \pm 0.21(\text{stat.}) \pm 0.11(\text{syst.})$ GeV based on data from the ATLAS and CMS experiments [1–4]. This is undoubtedly a fascinating discovery in the field of particle physics and might be the final missing piece. Nevertheless, there is still no experimental verification that it is the only Higgs boson. Acknowledging the fact that the SM is the theory which describes an approximate observable world it is worth taking note of the fact that there might be physics beyond the SM. Some of the phenomena, which the SM does not account for are: neutrino oscillations, asymmetry of matterantimatter, dark energy, gravity, *etc.* A particular physical phenomenon, which we are interested in and does not fit the frame of the SM is the absence of any Dark Matter (DM) candidate. As a consequence, the SM fails to describe nearly 85% of the matter. Extension of the Higgs sector could resolve some of the issues and is a common practice when models beyond the SM are constructed. Thus, we propose and are motivated that such extension could potentially solve several problems.

The SM uses the minimal Brout-Englert-Higgs mechanism [5–8], and a single complex SU(2)doublet is considered. The simplest extension of the SM electroweak sector is the Two-Higgs-Doublet Model (2HDM) [9–15], where the second SU(2) doublet is added to the SM-like SU(2)doublet. Such extension predicts a rich scalar spectrum: two additional neutral states $h_{(1,2)}$, or three counting the SM-like Higgs boson $h_{\rm SM}$, and a charged state h^{\pm} . The second SU(2) doublet can be further constrained to result in a viable DM candidate. These are the so-called Inert Doublet Models (IDM) [15–19]. The 2HDM and IDM models have been extensively analysed and result in some interesting properties. There is no direct restriction on the amount of additional SU(2)doublets. Two SU(2) doublets can be combined with the SM-like one. Such extension results in a Three-Higgs-Doublet Model (3HDM). This model incorporates a rich spectrum of additional particles: the SM-like Higgs boson $h_{\rm SM}$ along with other neutral scalars $h_{1..4}$, and two additional charged scalars $h_{1,2}^{\pm}$ are present. Some of these particles can decouple from the visible matter, either a charged and two neutral scalars or two charged and four neutral states, see Refs. [20-23] for the current research on the 3HDM DM. The SM SU(2) doublet can also be extended by other structures, e.q., a singlet can be added. The singlet extended model is also capable of describing the DM [24–27]. Nevertheless, all these models are highly constrained by the $\rho \approx 1$ parameter [28, 29], which depends on the hypercharge of the Higgs structures.

Any model must specify interactions between the particles. In terms of the SM extended scalar sector, of interest are interactions between scalars and fermions, and scalars and gauge bosons. Of particular interest is the Yukawa Lagrangian. The extended scalar sector should be able to incorporate the experimental results: masses of the fermions, the Cabibbo–Kobayashi–Maskawa (CKM) matrix, and the Pontecorvo–Maki–Nakagawa–Sakata (PMNS) matrix. Generic Yukawa couplings might result in unacceptably large Flavour-Changing Neutral Currents (FCNC), which are not observed. Properties of the recently observed scalar particle are in agreement with those of the SM Higgs boson. Therefore, any non SM-like couplings involving the observed scalar are strongly constrained. The discovery of the Higgs boson motivates to constrain the extended scalar sector in a way that the SM-like Higgs boson couplings are in agreement.

We are interested in the 3HDM extension. Historically, the inspiration behind the 3HDM were three generations of fermions. The most general 3HDM scalar potential results in 54 real free parameters [30]. It is a tough task to analyse such models, especially since a lot of freedom is introduced. Thus, some additional symmetries may be imposed. A possible classification of the 3HDM based on additional symmetries was presented in Ref. [31]. One of the possibilities is to impose the discrete S_3 symmetry [32,33]. The S_3 -symmetric scalar potential have been classified in terms of the minimization conditions and vacuum configurations in Ref. [34]. The thesis is based on the classification of the aforementioned article.

The thesis is organised as follows. In chapter 2 the general S_3 -symmetric potential is presented and some of the properties of the potential are considered. In chapter 3 a specific complex vacuum configuration C-III-c is discussed, however, this configuration results in unrealistic states and the symmetry is softly broken. In the following chapter 4, one of the softly broken models C-III-c- ν^2 is further on analysed with tree-level couplings and constraints presented. Chapter 5 is devoted to yet another model, but this time the real vacuum configuration R-II-1a is considered. The two aforementioned vacuum configuration are numerically analysed in chapter 6 to answer the question if those contain possible DM candidates. Some technicalities are addressed in appendices: in Appendix A the scalar potential in different notations is presented, in Appendix B first derivatives of the scalar potential are considered, and in Appendix C some of the possible transformations between the generic and the Higgs basis can be found.

Chapter 2

The S_3 -Symmetric Scalar Potential

In this chapter the basic building block, *i.e.* the S_3 -symmetric scalar potential is presented. Mainly, the irreducible representation framework is used. We analyse possible vacuum configurations and try to verify consistency of vacuum configurations with those presented in Ref. [34]. In Ref. [34] it was mentioned that there exists a special direction of the scalar potential, when the $\lambda_4 = 0$ constraint is applied, which results in a continuous SO(2) symmetry. We, however, considered the mass-squared matrices of all of the possible vacuum configurations and found additional hidden symmetries, see section 2.3. Other interesting properties of the scalar potential are also considered in this chapter.

2.1 The Scalar Potential

 S_3 is a non-Abelian group and is the permutation group of three objects, in this case permutation of the three Higgs doublets { ϕ_1 , ϕ_2 , ϕ_3 }, where

$$\phi_i = \begin{pmatrix} \varphi_i^+ \\ \frac{1}{\sqrt{2}} \left(\rho_i + \eta_i + i\chi_i\right) \end{pmatrix}, \text{ for } i = \overline{1,3},$$
(2.1.1)

where φ_i^+ is a complex field, and $\tilde{\eta}_i$ and $\tilde{\chi}_i$ are real fields, and, in general, the vacuum ρ_i is a complex value.

 S_3 has two ¹D irreducible representations $\mathbf{1}_S$ and $\mathbf{1}_A$, and a ²D doublet irreducible representation **2**. We chose the following ³D representation:

2:
$$\binom{h_1}{h_2} = \begin{pmatrix} \frac{1}{\sqrt{2}} (\phi_1 - \phi_2) \\ \frac{1}{\sqrt{6}} (\phi_1 + \phi_2 - 2\phi_3) \end{pmatrix},$$

1: $h_S = \frac{1}{\sqrt{3}} (\phi_1 + \phi_2 + \phi_3),$ (2.1.2)

where S in h_S indicates that it is an S_3 singlet.

In the new basis, the h_i fields are defined in the following way:

$$h_{i} = \begin{pmatrix} h_{i}^{+} \\ \frac{1}{\sqrt{2}} (w_{i} + \tilde{\eta}_{i} + i\tilde{\chi}_{i}) \end{pmatrix}, \quad i = \overline{1, 2},$$

$$h_{S} = \begin{pmatrix} h_{S}^{+} \\ \frac{1}{\sqrt{2}} (w_{S} + \tilde{\eta}_{S} + i\tilde{\chi}_{S}) \end{pmatrix},$$
(2.1.3)

where the vacuum values w_i and w_S can be complex.

In both representations the Vacuum Expectation Values (VEV) should satisfy constraint:

$$v = \sqrt{\rho_1^2 + \rho_2^2 + \rho_3^2} = \sqrt{w_1^2 + w_2^2 + w_S^2} \simeq 246 \text{ GeV}.$$
 (2.1.4)

Transformation from one basis (2.1.1) to another (2.1.3) is given by the following S_3 ³D representation:

$$\begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix} = \begin{pmatrix} \frac{1}{\sqrt{2}} & -\frac{1}{\sqrt{2}} & 0 \\ \frac{1}{\sqrt{6}} & \frac{1}{\sqrt{6}} & -\frac{2}{\sqrt{6}} \\ \frac{1}{\sqrt{3}} & \frac{1}{\sqrt{3}} & \frac{1}{\sqrt{3}} \end{pmatrix} \begin{pmatrix} \phi_1 \\ \phi_2 \\ \phi_3 \end{pmatrix},$$
(2.1.5)

or equivalently:

$$\begin{pmatrix} \phi_1 \\ \phi_2 \\ \phi_3 \end{pmatrix} = \begin{pmatrix} \frac{1}{\sqrt{2}} & \frac{1}{\sqrt{6}} & \frac{1}{\sqrt{3}} \\ -\frac{1}{\sqrt{2}} & \frac{1}{\sqrt{6}} & \frac{1}{\sqrt{3}} \\ 0 & -\frac{2}{\sqrt{3}} & \frac{1}{\sqrt{3}} \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}.$$
 (2.1.6)

The most general renormalizable $S_3 \otimes U(1)$ scalar potential can be written as [32, 35–37]:

$$V = V_{2} + V_{4},$$

$$V_{2} = \mu_{1}^{2} \left(h_{1}^{\dagger} h_{1} + h_{2}^{\dagger} h_{2} \right) + \mu_{0}^{2} h_{S}^{\dagger} h_{S},$$

$$V_{4} = \lambda_{1} \left(h_{1}^{\dagger} h_{1} + h_{2}^{\dagger} h_{2} \right)^{2} + \lambda_{2} \left(h_{1}^{\dagger} h_{2} - h_{2}^{\dagger} h_{1} \right)^{2} + \lambda_{3} \left[\left(h_{1}^{\dagger} h_{1} - h_{2}^{\dagger} h_{2} \right)^{2} + \left(h_{1}^{\dagger} h_{2} + h_{2}^{\dagger} h_{1} \right)^{2} \right]$$

$$+ \lambda_{4} \left[\left(h_{S}^{\dagger} h_{1} \right) \left(h_{1}^{\dagger} h_{2} + h_{2}^{\dagger} h_{1} \right) + \left(h_{S}^{\dagger} h_{2} \right) \left(h_{1}^{\dagger} h_{1} - h_{2}^{\dagger} h_{2} \right) + \text{h.c.} \right]$$

$$+ \lambda_{5} \left[\left(h_{S}^{\dagger} h_{S} \right) \left(h_{1}^{\dagger} h_{1} + h_{2}^{\dagger} h_{2} \right) \right] + \lambda_{6} \left[\left(h_{1}^{\dagger} h_{S} \right) \left(h_{S}^{\dagger} h_{1} \right) + \left(h_{2}^{\dagger} h_{S} \right) \left(h_{S}^{\dagger} h_{2} \right) \right]$$

$$+ \lambda_{7} \left[\left(h_{S}^{\dagger} h_{1} \right)^{2} + \left(h_{S}^{\dagger} h_{2} \right)^{2} + \text{h.c.} \right] + \lambda_{8} \left(h_{S}^{\dagger} h_{S} \right)^{2}.$$

$$(2.1.7)$$

Couplings μ and λ are assumed to be real provided that one is interested in the case when the Charge-Parity (*CP*) symmetry is not broken explicitly. Another possible way of writing down the scalar potential was presented by Derman [33, 38], which is covered in Appendix A. We also present the S_3 scalar potential in matrix form in Appendix A.

2.2 Two Possible Choices: Real and Complex Vacua

The unique characteristic of different models are the vacuum configurations; different configurations result in different minimization conditions. Not all of the first derivatives of the potential result in a unique set. We are interested in defining different vacuum configurations and thus a way around is to consider derivatives with respect to VEVs, which are all independent. A more thorough guideline can be found in Ref. [34] in sections 3 to 5. If a vacuum configuration to be analysed is known and some of the VEVs vanish, the set of derivatives no longer spans the full set of minimization conditions. This is true if the VEVs are substituted before differentiating the scalar potential with respect to the fields. We take a look at all possible first derivatives of the potential with respect to different fields and VEVs. Derivatives are written down in Appendix B.

2.2.1 Real Vacua

First of all, we review possible real vacuum configurations. Those do not violate CP spontaneously. For convenience, one can verify minimization conditions by considering the following derivatives:

$$\frac{\partial V}{\partial w_1}\Big|_{\langle v \rangle}, \quad \frac{\partial V}{\partial w_2}\Big|_{\langle v \rangle}, \quad \frac{\partial V}{\partial w_S}\Big|_{\langle v \rangle}, \quad (2.2.1)$$

where $\langle v \rangle$ indicates that all of the fields are set to zero. The minimization conditions lead to the following relations¹:

$$\mu_1^2 = -(\lambda_1 + \lambda_3) \left(w_1^2 + w_2^2 \right) - \frac{1}{2} w_S \left[6\lambda_4 w_2 + (\lambda_5 + \lambda_6 + 2\lambda_7) w_S \right], \qquad (2.2.2a)$$

¹We consider that the minimization conditions are satisfied exactly and not in the limit.

$$\mu_1^2 = -(\lambda_1 + \lambda_3) \left(w_1^2 + w_2^2 \right) - \frac{1}{2} \frac{w_S}{w_2} \left[3\lambda_4 \left(w_1^2 - w_2^2 \right) + (\lambda_5 + \lambda_6 + 2\lambda_7) w_2 w_S \right], \qquad (2.2.2b)$$

$$\mu_0^2 = \frac{1}{2w_S} \left[\lambda_4 \left(-3w_1^2 w_2 + w_2^3 \right) - w_S \left(\left(\lambda_5 + \lambda_6 + 2\lambda_7 \right) \left(w_1^2 + w_2^2 \right) + 2\lambda_8 w_S^2 \right) \right], \tag{2.2.2c}$$

It should be noted that additional conditions should be taken into account as eq. (2.2.2a) and eq. (2.2.2b) are not consistent for all possible values. For self-consistency the following relation should be satisfied:

$$\lambda_4 w_S \left(w_1^2 - 3w_2^2 \right) = 0. \tag{2.2.3}$$

As a result, possible solutions are $\lambda_4 = 0$ or $w_1 = \pm \sqrt{3}w_2$, or $w_S = 0$. An interesting case is that one of the solutions involves $w_S = 0$. We take a closer look at what happens if we consider that one or several VEVs can acquire zero value.

If one of the VEVs acquires a zero value, the corresponding derivative vanishes automatically. For $w_1 = 0$, the self-consistency condition (2.2.3) is automatically satisfied as the μ_1^2 coupling is uniquely defined in this case. The choice of $w_2 = 0$ leads to additional minimization conditions in terms of eq. (2.2.3) as derivative with respect to $\tilde{\eta}_2$ does not vanish. In case of $w_S = 0$, the non-vanishing derivative is the one with respect to the field $\tilde{\eta}_S$ and results in:

$$\lambda_4 w_2 \left(3w_1^2 - w_2^2 \right) = 0, \tag{2.2.4}$$

although it might seem that $w_S = 0$ is a viable solution for eq. (2.2.3). As a result, vacuum configurations with $w_S = 0$ should be supplemented by either $\lambda_4 = 0$, or $w_2 = 0$, or $w_2 = \pm \sqrt{3}w_1$. The only non-trivial case when two of the VEVs acquire zero values is when $w_1 = 0$ and $w_S = 0$. In this case additional condition is given by:

$$\lambda_4 w_2^3 = 0. \tag{2.2.5}$$

We expand the scalar potential with respect to different conditions. The different vacuum configurations are presented in Table 2.1. We find that all of the real vacuum configurations match those of Ref. [34]. It is worth mentioning that there is a special case of the R-III vacuum configuration $\{w_1, 0, w_s\}$, which was not mentioned in the original paper.

2.2.2 Complex Vacua

We take a look at a possible case when the vacuum can acquire complex values. We work in the following notation:

$$\{w_1, w_2, w_S\} \to \{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, \hat{w}_S\},\tag{2.2.6}$$

where the hatted \hat{w}_i value indicates the absolute value and σ_i stands for a phase. Because of the $U(1)_Y$ gauge invariance of the scalar potential it is always possible to rotate one of the phases away. For convenience, we rotated the S_3 singlet phase. We work with the following set of independent derivatives:

$$\frac{\partial V}{\partial \hat{w}_1}\Big|_{\langle v \rangle}, \quad \frac{\partial V}{\partial \hat{w}_2}\Big|_{\langle v \rangle}, \quad \frac{\partial V}{\partial \hat{w}_S}\Big|_{\langle v \rangle}, \quad \frac{\partial V}{\partial \sigma_1}\Big|_{\langle v \rangle}, \quad \frac{\partial V}{\partial \sigma_2}\Big|_{\langle v \rangle}.$$
(2.2.7)

Full equations can be found in Appendix B.

We start by analysing the most general case, *i.e.*, of the form of eq. (2.2.6). It leads to several possible vacuum configurations. The self-consistency condition requires that μ_1^2 values should coincide. The μ_1^2 coupling can be defined by taking a look at the derivative of the potential with respect to either \hat{w}_1 or \hat{w}_2 . We can write down the self-consistency condition as:

$$-2 (\lambda_{2} + \lambda_{3}) \hat{w}_{2} (\hat{w}_{1}^{2} - \hat{w}_{2}^{2}) s_{(\sigma_{1} - \sigma_{2})}^{2} + \frac{1}{2} \lambda_{4} \hat{w}_{S} [c_{\sigma_{2}} (2\hat{w}_{1}^{2} - 7\hat{w}_{2}) + c_{(2\sigma_{1} - \sigma_{2})} (\hat{w}_{1}^{2} - 2\hat{w}_{2})] - \lambda_{7} \hat{w}_{2} \hat{w}_{S}^{2} (c_{2\sigma_{1}} - c_{2\sigma_{2}}) = 0,$$

$$(2.2.8)$$

Vacuum	$\{w_1,w_2,w_S\}$	Constraints		
R-0	$\{0, 0, 0\}$	None		
R-I-1	$\{0, 0, w_S\}$	$\mu_0^2 = -\lambda_8 w_S^2$		
R-I-2a	$\{w, 0, 0\}$	$\mu_1^2 = -\left(\lambda_1 + \lambda_3\right) w_1^2$		
R-I-2b	$\{w,\sqrt{3}w,0\}$	$\mu_1^2 = -\frac{4}{3} \left(\lambda_1 + \lambda_3 \right) w_2^2$		
R-I-2c	$\{w, -\sqrt{3}w, 0\}$	$\mu_1^2 = -\frac{4}{3} \left(\lambda_1 + \lambda_3 \right) w_2^2$		
B II 1a	$\begin{cases} 0 & w & w \\ \end{pmatrix}$	$\mu_0^2 = \frac{1}{2}\lambda_4 \frac{w_2^3}{w_S} - \frac{1}{2} \left(\lambda_5 + \lambda_6 + 2\lambda_7\right) w_2^2 - \lambda_8 w_S^2,$		
n-11-1a	$\{0, w, w_S\}$	$\mu_1^2 = -(\lambda_1 + \lambda_3) w_2^2 + \frac{3}{2} \lambda_4 w_2 w_S - \frac{1}{2} (\lambda_5 + \lambda_6 + 2\lambda_7) w_S^2$		
B-II-1b	$\{w = w/\sqrt{3} \ w_{q}\}$	$\mu_0^2 = -4\lambda_4 \frac{w_2^3}{w_S} - 2\left(\lambda_5 + \lambda_6 + 2\lambda_7\right) w_2^2 - \lambda_8 w_S^2,$		
		$\mu_1^2 = -4(\lambda_1 + \lambda_3)w_2^2 - 3\lambda_4w_2w_S - \frac{1}{2}(\lambda_5 + \lambda_6 + 2\lambda_7)w_S^2$		
B-II-1c	$\{w, w/\sqrt{3}, w_{g}\}$	$\mu_0^2 = -4\lambda_4 \frac{w_2^3}{w_S} - 2\left(\lambda_5 + \lambda_6 + 2\lambda_7\right) w_2^2 - \lambda_8 w_S^2,$		
		$\mu_1^2 = -4(\lambda_1 + \lambda_3)w_2^2 - 3\lambda_4w_2w_S - \frac{1}{2}(\lambda_5 + \lambda_6 + 2\lambda_7)w_S^2$		
B-II-2	$\{0, w, 0\}$	$\mu_1^2 = -\left(\lambda_1 + \lambda_3\right) w_2^2,$		
10 11 2	[0, 0, 0]	$\lambda_4 = 0$		
B-II-3	$\{w_1, w_2, 0\}$	$\mu_1^2 = -(\lambda_1 + \lambda_3) \left(w_1^2 + w_2^2 \right),$		
	$1w_1, w_2, 0f$	$\lambda_4 = 0$		
		$\mu_0^2 = -\frac{1}{2} \left(\lambda_5 + \lambda_6 + 2\lambda_7\right) \left(w_1^2 + w_2^2\right) - \lambda_8 w_S^2,$		
R-III	$\{w_1, w_2, w_S\}$	$\mu_1^2 = -(\lambda_1 + \lambda_3) \left(w_1^2 + w_2^2 \right) - \frac{1}{2} \left(\lambda_5 + \lambda_6 + 2\lambda_7 \right) w_S^2,$		
		$\lambda_4 = 0$		

Table 2.1: Possible real vacuum configurations. The classification is based on and adopts the notation of Ref. [34]. R stands for real vacuum configuration. The Roman numeral shows the total number of constraints. The last combination of numeral and letter is used to distinguish different configurations in the same category.

where we used the following symbols to denote trigonometric functions: $s_{\xi} \equiv \sin \xi$, $c_{\xi} \equiv \cos \xi$ and $t_{\xi} \equiv \tan \xi$.

Additional constraints arise from derivatives with respect to σ_1 and σ_2 due to the fact that those do not depend on the quadratic couplings μ_0^2 and μ_1^2 :

$$-(\lambda_2 + \lambda_3)\hat{w}_1^2\hat{w}_2^2\mathbf{s}_{(2\sigma_1 - 2\sigma_2)} - \lambda_4\hat{w}_1^2\hat{w}_2\hat{w}_S\mathbf{s}_{(2\sigma_1 - \sigma_2)} - \lambda_7\hat{w}_1^2\hat{w}_S^2\mathbf{s}_{2\sigma_1} = 0, \quad (2.2.9a)$$

$$(\lambda_2 + \lambda_3) \,\hat{w}_1^2 \hat{w}_2^2 \mathbf{s}_{(2\sigma_1 - 2\sigma_2)} + \frac{1}{2} \lambda_4 \hat{w}_2 \hat{w}_S \left[\hat{w}_1^2 \mathbf{s}_{(2\sigma_1 - \sigma_2)} - \left(2\hat{w}_1^2 - \hat{w}_2^2 \right) \mathbf{s}_{\sigma_2} \right] - \lambda_7 \hat{w}_2^2 \hat{w}_S^2 \mathbf{s}_{2\sigma_2} = 0. \quad (2.2.9b)$$

By taking a look at all of the possible cases in Table 2.2 we verify that our solutions coincide with the ones presented in Ref. [34], although some additional explanation is needed.

Vacuum	$\{w_1, w_2, w_S\}$	Constraints	
C-I-a $\{\hat{w}_1, \pm i\hat{w}_1, 0\}$		$\mu_1^2 = -2\left(\lambda_1 - \lambda_2\right)\hat{w}_1^2$	
		$\mu_0^2 = -\frac{1}{2} \left(\lambda_5 + \lambda_6 - 2\lambda_7\right) \hat{w}_2^2 - \lambda_8 \hat{w}_S^2,$	
C-III-a	$\{0, \hat{w}_2 e^{i\sigma_2}, \hat{w}_S\}$	$\mu_1^2 = -(\lambda_1 + \lambda_3)\hat{w}_2^2 - \frac{1}{2}\left(\lambda_5 + \lambda_6 - 2\lambda_7 - 8c_{\sigma_2}^2\lambda_7\right)\hat{w}_S^2,$	
		$\lambda_4 = rac{4 c_{\sigma_2} \hat{w}_S}{\hat{w}_2} \lambda_7$	

		$\mu_0^2 = -\frac{1}{2} \left(\lambda_5 + \lambda_6 - 2\lambda_7\right) \hat{w}_1^2 - \lambda_8 \hat{w}_S^2,$				
C-III-b	$\{\pm i\hat{w}_1, 0, \hat{w}_S\}$	$\mu_1^2 = -(\lambda_1 + \lambda_3)\hat{w}_1^2 - \frac{1}{2}(\lambda_5 + \lambda_6 - 2\lambda_7)\hat{w}_S^2,$				
		$\lambda_4 = 0$				
		$\mu_1^2 = -(\lambda_1 + \lambda_3) \left(\hat{w}_1^2 + \hat{w}_2^2 \right),$				
C-III-c	$\{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, 0\}$	$\lambda_2 + \lambda_3 = 0,$				
		$\lambda_4 = 0$				
		$\mu_0^2 = (\lambda_2 + \lambda_3) \frac{(\hat{w}_1^2 - \hat{w}_2^2)^2}{\hat{w}_S^2} - \frac{(\hat{w}_1^2 - \hat{w}_2^2)(\hat{w}_1^2 - 3\hat{w}_2^2)}{4\hat{w}_2\hat{w}_S} \lambda_4$				
C-III-d	$\{\pm i \hat{w}_1, \hat{w}_2, \hat{w}_S\}$	$-\frac{1}{2}\left(\lambda_5+\lambda_6\right)\left(\hat{w}_1^2+\hat{w}_2^2\right)-\lambda_8\hat{w}_S^2,$				
0 III u		$\mu_1^2 = -\left(\lambda_1 - \lambda_2\right) \left(\hat{w}_1^2 + \hat{w}_2^2\right) - \frac{\hat{w}_S(\hat{w}_1^2 - \hat{w}_2^2)}{4\hat{w}_2}\lambda_4 - \frac{1}{2}\left(\lambda_5 + \lambda_6\right)\hat{w}_S^2,$				
		$\lambda_7 = \frac{(\hat{w}_1^2 - \hat{w}_2^2)}{\hat{w}_S^2} \left(\lambda_2 + \lambda_3\right) - \frac{(\hat{w}_1^2 - 5\hat{w}_2^2)}{4\hat{w}_2\hat{w}_S}\lambda_4$				
		$\mu_0^2 = (\lambda_2 + \lambda_3) \frac{(\hat{w}_1^2 - \hat{w}_2^2)^2}{\hat{w}_S^2} + \frac{(\hat{w}_1^2 - \hat{w}_2^2)(\hat{w}_1^2 - 3\hat{w}_2^2)}{4\hat{w}_2\hat{w}_S} \lambda_4$				
CIII.o	$\{\pm i\hat{w}_1, -\hat{w}_2, \hat{w}_S\}$	$-\frac{1}{2}\left(\lambda_5+\lambda_6\right)\left(\hat{w}_1^2+\hat{w}_2^2\right)-\lambda_8\hat{w}_S^2,$				
0-111-6		$\mu_1^2 = -\left(\lambda_1 - \lambda_2\right) \left(\hat{w}_1^2 + \hat{w}_2^2\right) + \frac{\hat{w}_S(\hat{w}_1^2 - \hat{w}_2^2)}{4\hat{w}_2}\lambda_4 - \frac{1}{2}\left(\lambda_5 + \lambda_6\right)\hat{w}_S^2,$				
		$\lambda_7 = \frac{(\hat{w}_1^2 - \hat{w}_2^2)}{\hat{w}_S^2} \left(\lambda_2 + \lambda_3\right) + \frac{(\hat{w}_1^2 - 5\hat{w}_2^2)}{4\hat{w}_2\hat{w}_S}\lambda_4$				
		$\mu_0^2 = -\frac{1}{2} \left(\lambda_5 + \lambda_6 - 2\lambda_7\right) \left(\hat{w}_1^2 + \hat{w}_2^2\right) - \lambda_8 \hat{w}_S^2,$				
C-III-f	$\{\pm i\hat{w}_1, i\hat{w}_2, \hat{w}_S\}$	$\mu_1^2 = -(\lambda_1 + \lambda_3) \left(\hat{w}_1^2 + \hat{w}_2^2 \right) - \frac{1}{2} \left(\lambda_5 + \lambda_6 - 2\lambda_7 \right) \hat{w}_S^2,$				
		$\lambda_4 = 0$				
	$\{\pm i\hat{w}_1, -i\hat{w}_2, \hat{w}_S\}$	$\mu_0^2 = -\frac{1}{2} \left(\lambda_5 + \lambda_6 - 2\lambda_7\right) \left(\hat{w}_1^2 + \hat{w}_2^2\right) - \lambda_8 \hat{w}_S^2,$				
C-III-g		$\mu_1^2 = -(\lambda_1 + \lambda_3) \left(\hat{w}_1^2 + \hat{w}_2^2 \right) - \frac{1}{2} \left(\lambda_5 + \lambda_6 - 2\lambda_7 \right) \hat{w}_S^2,$				
		$\lambda_4 = 0$				
		$\mu_0^2 = -2 \left(\lambda_5 + \lambda_6 - 2\lambda_7\right) \hat{w}_2^2 - \lambda_8 \hat{w}_S^2,$				
C-III-h	$\{\sqrt{3}\hat{w}_2e^{i\sigma_2}, \pm\hat{w}_2e^{i\sigma_2}, \hat{w}_S\}$	$\mu_1^2 = -4 \left(\lambda_1 + \lambda_3\right) \hat{w}_2^2 - \frac{1}{2} \left(\lambda_5 + \lambda_6 - 2\lambda_7 - 8c_{\sigma_2}^2 \lambda_7\right) \hat{w}_S^2,$				
		$\lambda_4=\mprac{2\mathrm{c}_{\sigma_2}\hat{w}_S}{\hat{w}_2}\lambda_7$				
		$\mu_0^2 = \frac{16\left(1 - 3t_{\sigma_1}^2\right)^2}{\left(1 + 9t_{\sigma_1}^2\right)^2} \left(\lambda_2 + \lambda_3\right) \frac{\hat{w}_2^4}{\hat{w}_S^2} \pm \frac{6\left(1 - t_{\sigma_1}^2\right)\left(1 - 3t_{\sigma_1}^2\right)}{\left(1 + 9t_{\sigma_1}^2\right)^{3/2}} \lambda_4 \frac{\hat{w}_2^3}{\hat{w}_S}$				
	$\left\{\sqrt{\frac{3(1+t_{\sigma_1}^2)}{1+0t^2}}\hat{w}_2e^{i\sigma_1},\right.$	$-\frac{2(1+3t_{\sigma_1}^2)}{(1+9t^2)} (\lambda_5 + \lambda_6) \hat{w}_2^2 - \lambda_8 \hat{w}_S^2,$				
C-III-i	$ + \hat{w}_2 e^{-i \arctan(3t_{\sigma_1})} + \hat{w}_S $	$\mu_1^2 = -\frac{4\left(1+3t_{\sigma_1}^2\right)}{\left(\lambda_1 - \lambda_2\right)\hat{w}_2^2} \mp \frac{\left(1-3t_{\sigma_1}^2\right)}{\left(\lambda_1 - \lambda_2\right)\hat{w}_2^2} \lambda_4 \hat{w}_2 \hat{w}_S - \frac{1}{2}\left(\lambda_5 + \lambda_6\right)\hat{w}_S^2,$				
	2- , ~.5]	$\begin{pmatrix} (1+9t_{\sigma_1}^2) & (1-3t_{\sigma_1}^2) \\ & & (1-3t_{\sigma_$				
		$\lambda_{7}\frac{1}{\left(1+9t_{\sigma_{1}}^{2}\right)} \left(\lambda_{2} + \lambda_{3}\right) \frac{1}{\hat{w}_{S}^{2}} + \frac{1}{2\sqrt{1+9t_{\sigma_{1}}^{2}}} \lambda_{4} \frac{1}{\hat{w}_{S}}$				

	$\{\hat{w}_1 e^{i\sigma_1}, 0, \hat{w}_S\}$	$\mu_0^2 = -\frac{1}{2} \left(\lambda_5 + \lambda_6\right) \hat{w}_1^2 - \lambda_8 \hat{w}_S^2,$
C-IV-a		$\mu_1^2 = -(\lambda_1 + \lambda_3)\hat{w}_1^2 - \frac{1}{2}(\lambda_5 + \lambda_6)\hat{w}_S^2,$
		$\lambda_4 = 0,$
		$\lambda_7 = 0$
		$\mu_0^2 = (\lambda_2 + \lambda_3) \frac{(\hat{w}_1^2 - \hat{w}_2^2)^2}{\hat{w}_S^2} - \frac{1}{2} (\lambda_5 + \lambda_6) \left(\hat{w}_1^2 + \hat{w}_2^2\right) - \lambda_8 \hat{w}_S^2,$
CWb	[aû. +iaû. aû_]	$\mu_1^2 = -(\lambda_1 - \lambda_2) \left(\hat{w}_1^2 + \hat{w}_2^2 \right) - \frac{1}{2} \left(\lambda_5 + \lambda_6 \right) \hat{w}_S^2,$
0-10-0	$\{w_1, \pm i w_2, w_S\}$	$\lambda_4 = 0,$
		$\lambda_7=-rac{(\hat{w}_1^2-\hat{w}_2^2)}{\hat{w}_S^2}\left(\lambda_2+\lambda_3 ight)$
		$\mu_0^2 = 2c_{\sigma_2}^2 \left(1 + c_{\sigma_2}^2\right) \left(\lambda_2 + \lambda_3\right) \frac{\hat{w}_2^4}{\hat{w}_S^2} - \left(1 + c_{\sigma_2}^2\right) \left(\lambda_5 + \lambda_6\right) \hat{w}_2^2 - \lambda_8 \hat{w}_S^2,$
O W	$\{\sqrt{1+2\mathbf{c}_{\sigma_2}^2}\hat{w}_2,$	$\mu_1^2 = -\left[2\left(1 + c_{\sigma_2}^2\right)\lambda_1 - \left(2 + 3c_{\sigma_2}^2\right)\lambda_2 - c_{\sigma_2}^2\lambda_3\right]\hat{w}_2^2 - \frac{1}{2}\left(\lambda_5 + \lambda_6\right)\hat{w}_S^2,$
C-IV-c	$\hat{w}_2 e^{i\sigma_2}, \hat{w}_S \}$	$\lambda_4 = -2\mathbf{c}_{\sigma_2} \left(\lambda_2 + \lambda_3\right) \frac{\hat{w}_2}{\hat{w}_S},$
		$\lambda_7 = \mathrm{c}_{\sigma_2}^2 \left(\lambda_2 + \lambda_3 ight) rac{\hat{w}_2^2}{\hat{w}_S^2}$
		$\mu_0^2 = -\frac{1}{2} \left(\lambda_5 + \lambda_6\right) \left(\hat{w}_1^2 + \hat{w}_2^2\right) - \lambda_8 \hat{w}_S^2,$
a	$\{\hat{w}_1e^{i\sigma_1},\pm\hat{w}_2e^{i\sigma_1},\hat{w}_S\}$	$\mu_1^2 = -(\lambda_1 + \lambda_3) \left(\hat{w}_1^2 + \hat{w}_2^2 \right) - \frac{1}{2} \left(\lambda_5 + \lambda_6 \right) \hat{w}_S^2,$
C-IV-d		$\lambda_4 = 0,$
		$\lambda_7=0$
		$\mu_0^2 = \frac{s_{2(\sigma_1 - \sigma_2)}^2}{s_{2\sigma_1}^2} \left(\lambda_2 + \lambda_3\right) \frac{\hat{w}_2^4}{\hat{w}_s^2} - \frac{1}{2} \left(1 - \frac{s_{2\sigma_2}}{s_{2\sigma_1}}\right) \left(\lambda_5 + \lambda_6\right) \hat{w}_2^2 - \lambda_8 \hat{w}_S^2,$
C W -	$\{ \sqrt{-\frac{\mathbf{s}_{2\sigma_{2}}}{\mathbf{s}_{2\sigma_{1}}}} \hat{w}_{2} e^{i\sigma_{1}}, \\ \hat{w}_{2} e^{i\sigma_{2}}, \hat{w}_{S} \}$	$\mu_1^2 = -\left(1 - \frac{s_{2\sigma_2}}{s_{2\sigma_1}}\right) (\lambda_1 - \lambda_2) \hat{w}_2^2 - \frac{1}{2} \left(\lambda_5 + \lambda_6\right) \hat{w}_S^2,$
С-1v-е		$\lambda_4 = 0,$
		$\lambda_7 = -\frac{\mathrm{s}_{2(\sigma_1 - \sigma_2)}}{\mathrm{s}_{2\sigma_1}} \left(\lambda_2 + \lambda_3\right) \frac{\hat{w}_2^2}{\hat{w}_S^2}$
		$\mu_0^2 = -\frac{(\mathbf{c}_{(\sigma_1 - 2\sigma_2)} + 3\mathbf{c}_{\sigma_1})\mathbf{c}_{(\sigma_2 - \sigma_1)}}{2\mathbf{c}_{\sigma_*}^2}\lambda_4 \frac{\hat{w}_2^3}{\hat{w}_S}$
		$-\frac{\mathrm{c}_{(\sigma_1-2\sigma_2)}+3\mathrm{c}_{\sigma_1}}{2\mathrm{c}_{\sigma_1}}\left(\lambda_5+\lambda_6\right)\hat{w}_2^2-\lambda_8\hat{w}_S^2,$
C IV C	$\left\{\sqrt{2 + \frac{\mathbf{c}_{(\sigma_1 - 2\sigma_2)}}{\mathbf{c}_{\sigma_1}}}\hat{w}_2 e^{i\sigma_1},\right.$	$\mu_1^2 = -\frac{c_{(\sigma_1 - 2\sigma_2)} + 3c_{\sigma_1}}{c_{\sigma_1}} \left(\lambda_1 + \lambda_3\right) \hat{w}_2^2$
C-IV-I	$\hat{w}_2 e^{i\sigma_2}, \hat{w}_S$	$-\frac{3c_{2\sigma_{1}}+2c_{2(\sigma_{1}-\sigma_{2})}+c_{2\sigma_{2}}+4}{4c_{(\sigma_{1}-\sigma_{2})}c_{\sigma_{1}}}\lambda_{4}\hat{w}_{2}\hat{w}_{S}-\frac{1}{2}\left(\lambda_{5}+\lambda_{6}\right)\hat{w}_{S}^{2},$
		$\lambda_4 = -2\frac{\mathbf{c}_{\sigma_2 - \sigma_1}}{\mathbf{c}_{\sigma_1}} \left(\lambda_2 + \lambda_3\right) \frac{\hat{w}_2}{\hat{w}_S},$
		$\lambda_7 = rac{\mathrm{c}^2_{\sigma_2-\sigma_1}}{\mathrm{c}^2_{\sigma_1}} \left(\lambda_2+\lambda_3 ight) rac{\hat{w}_2^2}{\hat{w}_S^2}$
		$\mu_0^2 = -\frac{1}{2} \left(\lambda_5 + \lambda_6\right) \left(\hat{w}_1^2 + \hat{w}_2^2\right) - \lambda_8 \hat{w}_S^2,$
		$\mu_1^2 = -(\lambda_1 + \lambda_3) \left(\hat{w}_1^2 + \hat{w}_2^2 \right) - \frac{1}{2} \left(\lambda_5 + \lambda_6 \right) \hat{w}_S^2,$
C-V	$\{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, \hat{w}_S\}$	$\lambda_2 + \lambda_3 = 0,$
		$\lambda_4 = 0,$
		$\lambda_7 = 0$

Table 2.2: Possible complex vacuum configurations. The classification is based on and adopts the notation of Ref. [34]. C stands for complex vacuum configuration. The Roman numeral shows the total number of constraints. The last combination of numeral and letter is used to distinguish different configurations in the same category.

We note that not all of the possible vacuum configurations and constraints can be applied directly to minimize the scalar potential *a priori*. If we take into consideration vacuum configurations with additional constraints in terms of λ couplings, some caution is required. If one of the couplings depends explicitly on another, *e.g.*, in case of the C-III-a vacuum configuration one of the minimization constraints is:

$$\lambda_4 = \frac{4\mathbf{c}_{\sigma_2}\hat{w}_S}{\hat{w}_2}\lambda_7,\tag{2.2.10}$$

such constraints cannot be applied to the potential before differentiating it. If one changed the form of the scalar potential before differentiating it, that would potentially lead to additional terms. These additional terms potentially change the scalar potential structure. This is not the case when $\lambda_2 + \lambda_3 = 0$.

2.3 Identifying the Goldstone States

We proceed to a general check of hidden symmetries of vacuum configurations presented in Table 2.1 and Table 2.2. We try to uncover hidden continuous symmetries of the scalar potential to identify additional Goldstone bosons. These Goldstone bosons are distinct from the longitudinal polarization components of the W and Z. The Goldstone bosons which are "eaten" by the three gauge bosons are called the would-be Goldstone bosons. Therefore we identify additional massless states as those which do not coincide with the would-be Goldstone bosons.

One way to determine if there is at least one additional massless state is to consider the determinant of the mass-squared matrix after identifying the would-be Goldstone bosons. If the determinant results in zero, this indicates that some of the scalar states are massless. We are, however, interested in the total number of massless states, and, in principle, we are not interested in fields expressed in terms of the mass-eigenstates and therefore adopt a more straightforward scheme. We found that exact S_3 -symmetric models discussed in section 2.2 result in only neutral massless states. The additional charged massless scalars would result in a decent amount of issues, especially if additional charged massless scalars, not the would-be Goldstone bosons, coupled to photons.

The most general neutral mass-squared matrix $\mathcal{M}_{6\times 6}^0$ is of dim = 6. The determinant of $\mathcal{M}_{6\times 6}^0$ is a product of all eigenvalues:

$$\det\left(\mathcal{M}_{6\times 6}^{0}\right) = \prod_{i=1}^{6} m_{H_{i}}^{2}, \qquad (2.3.1)$$

where $m_{H_i}^2$ are the corresponding masses squared of the scalar fields. The determinant det $(\mathcal{M}_{6\times 6}^0)$ obviously results in zero as there is at least one would-be Goldstone boson present. The determinant can be used to find the eigenvalues of the mass-squared matrix by solving the characteristic equation:

$$\det\left(\mathcal{M}_{6\times 6}^{0} - \lambda \mathcal{I}_{6}\right) = 0, \qquad (2.3.2)$$

where \mathcal{I}_n is the identity matrix of dimension n. We are only interested in determining the total number of massless states and are not interested in explicit analytic expressions. The way around is to consider:

$$\det \left(\hat{\mathcal{M}}_{6\times 6}^{0} - \lambda \,\mathcal{I}_{6} \right) = \left(m_{H_{1}}^{2} - \lambda \right) \left(m_{H_{2}}^{2} - \lambda \right) \dots \left(m_{H_{6}}^{2} - \lambda \right)$$

= $\sum_{i=0}^{6} (-1)^{i} \lambda^{i} c_{i},$ (2.3.3)

where $\hat{\mathcal{M}}^{0}_{6\times 6}$ is a diagonalized mass-squared matrix and c_i are exponential Bell polynomials which satisfy the Cayley-Hamiltonian theorem [39, 40]:

$$c_{0} = \det\left(\hat{\mathcal{M}}_{6\times6}^{0}\right),$$

$$c_{1} = \frac{1}{120} \left[\operatorname{tr}\left(\hat{\mathcal{M}}_{6\times6}^{0}\right)^{5} - 10\operatorname{tr}\left(\hat{\mathcal{M}}_{6\times6}^{0}\right)^{3}\operatorname{tr}\left((\hat{\mathcal{M}}_{6\times6}^{0})^{2}\right) + 15\operatorname{tr}\left(\hat{\mathcal{M}}_{6\times6}^{0}\right)\operatorname{tr}\left((\hat{\mathcal{M}}_{6\times6}^{0})^{2}\right)^{2} \right]$$

$$(2.3.4a)$$

$$+ 20 \operatorname{tr} \left(\hat{\mathcal{M}}_{6\times 6}^{0} \right)^{2} \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{3} \right) - 20 \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{2} \right) \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{3} \right)$$

$$- 30 \operatorname{tr} \left(\hat{\mathcal{M}}_{6\times 6}^{0} \right) \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{4} \right) + 24 \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{5} \right) \right],$$

$$(2.3.4b)$$

$$c_{2} = \frac{1}{24} \left[\operatorname{tr} \left(\hat{\mathcal{M}}_{6\times 6}^{0} \right)^{4} - 6 \operatorname{tr} \left(\hat{\mathcal{M}}_{6\times 6}^{0} \right)^{2} \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{2} \right) + 3 \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{2} \right)^{2} + 8 \operatorname{tr} \left(\hat{\mathcal{M}}_{6\times 6}^{0} \right) \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{3} \right) - 6 \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{4} \right) \right],$$

$$(2.3.4c)$$

$$c_{3} = \frac{1}{6} \left[\operatorname{tr} \left(\hat{\mathcal{M}}_{6\times 6}^{0} \right)^{3} - 3 \operatorname{tr} \left(\hat{\mathcal{M}}_{6\times 6}^{0} \right) \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{2} \right) + 2 \operatorname{tr} \left((\hat{\mathcal{M}}_{6\times 6}^{0})^{3} \right) \right],$$
(2.3.4d)

$$c_4 = \frac{1}{2} \left[\operatorname{tr} \left(\hat{\mathcal{M}}^0_{6 \times 6} \right)^2 - \operatorname{tr} \left((\hat{\mathcal{M}}^0_{6 \times 6})^2 \right) \right], \qquad (2.3.4e)$$

$$c_5 = \operatorname{tr}\left(\tilde{\mathcal{M}}_{6\times 6}^0\right),\tag{2.3.4f}$$

$$c_6 = 1.$$
 (2.3.4g)

Of particular interest are coefficients $c_{1..5}$ as the highest order polynomial among these, which satisfies $c_j = 0$, indicates that there are exactly j additional massless states. The coefficient c_0 is zero due to the would-be Goldstone boson and c_6 is of no particular interest as it does not depend on the mass-squared parameters. Therefore coefficients $c_{1..5}$ are checked.

The mass eigenvalues $\hat{\mathcal{M}}_{6\times 6}^0$ are, in general, not known and therefore the following identity can be used:

$$\det\left(\mathcal{M}_{6\times 6}^{0}-\lambda \mathcal{I}_{6}\right)=\det\left(\hat{\mathcal{M}}_{6\times 6}^{0}-\lambda \mathcal{I}_{6}\right).$$
(2.3.5)

If a specific vacuum configuration results in at least two massless states, *i.e.*, $c_1 = 0$, the corresponding model is further analysed. Vacuum configurations, with at least $c_1 = 0$, are presented in Table 2.3.

The lowest non-vanishing $c_j \neq 0$ can be factorized to determine the number of mass-degenerate states. The power n, to which the $\mathcal{M}_{6\times 6}^0$ elements of the c_j polynomial are raised, indicates that there are in total n mass-degenerate states. We are not interested in such states and therefore the general result is only presented in Table 2.5 without further discussion.

An interesting observation is that there seems to be some sort of correlation between the number of constraints in terms of λ_i and additional massless states. Moreover every vacuum configuration with at least one minimization condition given by $\lambda_i = 0$ results in massless states. One of the possible explanations is to take a look at hidden symmetries of the potential after applying the minimization conditions in terms of λ_i and relate those to the Goldstone theorem [41–43].

Consider that the hidden symmetry, at most, results in a U(3) transformation U. Assume that the scalar potential transforms under the unitary transformation as $V \xrightarrow{U} V'$ and the SU(2)doublets transform as $h'_i = U_{ij}h_j^{-2}$. The scalar potential couplings μ_i^2 and λ_j are left intact after performing the aforementioned transformation U. In other words, the transformation U should leave the scalar potential invariant after changing back to the original basis $h'_i \to h_j$, so that V' - V = 0. This is not a property of the most general transformation U of the S_3 -symmetric potential V (2.1.7).

An obvious question is which of the SU(2) doublet combinations are invariant under U and if the most general form of the transformation matrix U can be simplified. The simplest way

²The S_3 singlet transforms as $h'_S = U_{3j}h_j$.

Vacuum	Vacuum configuration	$\begin{array}{l} \text{Additional} \\ m^0 = 0 \end{array}$	λ constraints
R-II-2	$\{0, w, 0\}$	1	$\lambda_4 = 0$
R-II-3	$\{\hat{w}_1,\hat{w}_2,0\}$	1	$\lambda_4 = 0$
R-III	$\{\hat{w}_1, \hat{w}_2, w_S\}$	1	$\lambda_4 = 0$
C-III-b	$\{\pm i\hat{w}_1, 0, \hat{w}_S\}$	1	$\lambda_4 = 0$
C-III-c	$\{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, 0\}$	2	$\lambda_2 + \lambda_3 = 0,$
			$\lambda_4 = 0$
C-III-f	$\{\pm i\hat{w}_1, i\hat{w}_2, \hat{w}_S\}$	1	$\lambda_4 = 0$
C-III-g	$\{\pm i\hat{w}_1, -i\hat{w}_2, \hat{w}_S\}$	1	$\lambda_4 = 0$
CWa	$\{\hat{w}_1 e^{i\sigma_1},0,\hat{w}_S\}$	2	$\lambda_4 = 0,$
0-1v-a			$\lambda_7 = 0$
C-IV-b	$\{\hat{w}_1, \pm i\hat{w}_2, \hat{w}_S\}$	1	$\lambda_4 = 0$
CWa	$\{\sqrt{1+2\mathbf{c}_{\sigma_2}^2}\hat{w}_2,$	1	$\lambda_4 = -2A \left(\lambda_2 + \lambda_3\right),$
0-10-0	$\hat{w}_2 e^{i\sigma_2}, \hat{w}_S \}$	I	$\lambda_7 = A^2 \left(\lambda_2 + \lambda_3 \right)$
CWd	$\int \hat{w}_{i} e^{i\sigma_{1}} + \hat{w}_{2} e^{i\sigma_{1}} \hat{w}_{3}$	9	$\lambda_4 = 0,$
0-1 v -u	$\begin{bmatrix} 1 & 1e & 1e & 1e \\ 1 & 1e & 1e & 1e \end{bmatrix}$	2	$\lambda_7 = 0$
CWo	$\{\sqrt{-\frac{\mathbf{s}_{2\sigma_2}}{\mathbf{s}_{2\sigma_1}}}\hat{w}_2e^{i\sigma_1},$	1	
0-11-6	$\hat{w}_2 e^{i\sigma_2}, \hat{w}_S$	1	$\lambda_4 = 0$
C IV C	$\left\{\sqrt{2 + \frac{\mathbf{c}_{(\sigma_1 - 2\sigma_2)}}{\mathbf{c}_{\sigma_1}}}\hat{w}_2 e^{i\sigma_1},\right.$	1	$\lambda_4 = -2A\left(\lambda_2 + \lambda_3\right),$
C-1V-1	$\hat{w}_2 e^{i\sigma_2}, \hat{w}_S \}$	L	$\lambda_7 = A^2 \left(\lambda_2 + \lambda_3 \right)$
		3	$\lambda_2 + \lambda_3 = 0,$
C-V	$\{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, \hat{w}_S\}$		$\lambda_4 = 0,$
			$\lambda_7 = 0$

Table 2.3: Vacuum configurations with additional neutral massless states $m^0 = 0$ not counting the would-be Goldstone bosons. The coefficient A for the C-IV-c and C-IV-f cases is expressed in eq. (2.3.17).

to determine the form of U is to consider transformation of the quadratic terms V_2 of the scalar potential:

$$\mu_0^2 h_S^{\dagger} h_S \xrightarrow{U} \mu_0^2 (h_S')^{\dagger} h_S' = \mu_0^2 h_S^{\dagger} h_S + \mu_0^2 \sum_{\substack{i,j=\{1,2,S\}\\(i,j)\neq(S,S)}} \mathcal{G}_{ij} h_i^{\dagger} h_j, \qquad (2.3.6a)$$

$$\mu_1^2 \left(h_1^{\dagger} h_1 + h_2^{\dagger} h_2 \right) \xrightarrow{U} \mu_1^2 \left({h_1'}^{\dagger} h_1' + {h_2'}^{\dagger} h_2' \right) = \mu_1^2 \left(h_1^{\dagger} h_1 + h_2^{\dagger} h_2 \right) + \mu_1^2 \sum_{\substack{i,j = \{1,2,S\} \\ (i,j) \neq (1,1) \\ (i,j) \neq (2,2)}} \mathcal{G}_{ij} h_i^{\dagger} h_j, \qquad (2.3.6b)$$

where \mathcal{G}_{ij} are coefficients which arise due to the U transformation, e.g., $\mathcal{G}_{21} = U_{12}^* U_{11} + U_{22}^* U_{21}$. The quadratic terms remain invariant under the U transformation provided that the off-diagonal coefficients \mathcal{G}_{ij} vanish, and is only true for quadratic terms as those depend on $h_i^{\dagger} h_i$. Consider transformation of the S_3 singlet h_S under U (2.3.6a). It follows that the unitary matrix U should not mix the S_3 doublet and singlet. If there is no mixing present, eq. (2.3.6b) results in $\mathcal{G}_{ij} = 0$. Thus, the U transformation, which is compatible with our primary assumption that it leaves the scalar potential invariant, is of the form:

$$\begin{pmatrix} h_1' \\ h_2' \\ h_S' \end{pmatrix} = \begin{pmatrix} U_{11} & U_{12} & 0 \\ U_{21} & U_{22} & 0 \\ 0 & 0 & U_{33} \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix},$$
(2.3.7)

and the coefficients U_{ij} are such that U is unitary. In principle, this corresponds to the $U(2) \otimes U(1)$ transformation. The non-invariant terms under the U transformation are the ones which are multiplied by the following quartic couplings: $\{\lambda_2, \lambda_3, \lambda_4, \lambda_7\}$. We note that although terms multiplied by λ_2 and λ_3 are not invariant under the U transformation, there is a direction $\lambda_2 + \lambda_3 = 0$ of the scalar potential which is invariant under the U(2) transformation. After substituting $\lambda_3 = -\lambda_2$, the resulting term in the primed basis is:

$$V_{23}' = -\lambda_2 \left[4 \left(h_1'^{\dagger} h_2' \right) \left(h_2'^{\dagger} h_1' \right) + \left(h_1'^{\dagger} h_1' - h_2'^{\dagger} h_2' \right)^2 \right].$$
(2.3.8)

It follows that a specific combination of the terms

$$\{\lambda_2 + \lambda_3, \lambda_4, \lambda_7\},\tag{2.3.9}$$

might result in a non identity transformation U (2.3.7). The scalar potential results in the following dependence between the λ_i coupling of eq. (2.3.9), written in terms of the SU(2) singlets:

$$(\lambda_2 + \lambda_3) \sim \{h_1^{\dagger}h_1, h_2^{\dagger}h_2, h_1^{\dagger}h_2\} + \text{h.c.},$$
 (2.3.10a)

$$\lambda_4 \sim \{h_1^{\dagger}h_1, h_2^{\dagger}h_2, h_1^{\dagger}h_1, h_2^{\dagger}h_2, h_1^{\dagger}h_2\} + \text{h.c.} , \qquad (2.3.10b)$$

$$\lambda_7 \sim \{h_S^{\dagger} h_1, h_S^{\dagger} h_2\} + \text{h.c.}$$
 (2.3.10c)

It can be seen that the SU(2) singlets multiplied by $(\lambda_2 + \lambda_3)$ and λ_7 are contained within the set (2.3.10b). Moreover, from Table 2.3 we see that there is a common feature of massless states associated with the $\lambda_4 = 0$ constraint; this is true with the only exception of vacuum configurations C-IV-c and C-IV-f. It turns out that if $\lambda_4 = 0$, the scalar potential is symmetric under SO(2):

$$U = \begin{pmatrix} c_{\theta_4} & s_{\theta_4} & 0\\ -s_{\theta_4} & c_{\theta_4} & 0\\ 0 & 0 & 1 \end{pmatrix}.$$
 (2.3.11)

In total, eight vacuum configurations fall into this category: {R-II-2, R-II-3, R-III, C-III-b, C-III-f, C-III-g, C-IV-b, C-IV-e}.

The constraints $\lambda_4 = 0$ and $\lambda_7 = 0$ result in an additional U(1) symmetry alongside the SO(2):

$$U = \begin{pmatrix} e^{i\theta_7} & 0 & 0\\ 0 & e^{i\theta_7} & 0\\ 0 & 0 & e^{i\theta_7} \end{pmatrix},$$
 (2.3.12)

with either $\theta'_7 = 0$ or $\theta_7 = 0$. Due to the U(1) invariance of the scalar potential, it makes little sense to consider both non-zero phases of the S_3 doublet and singlet. The case of $\lambda_4 = 0$ and $\lambda_7 = 0$ corresponds to vacuum configurations: {C-IV-a, C-IV-d}. The vacuum configuration C-IV-a is a special case of C-IV-d for $\hat{w}_2 = 0$. The choice of $\theta'_7 = 0$ leads to the fact that the σ_1 phase of the vacuum configuration C-IV-d { $\hat{w}_1 e^{i\sigma_1}, \pm \hat{w}_2 e^{i\sigma_1}, \hat{w}_S$ } can be rotated away by setting $\theta_7 = -\sigma_1$. Therefore both C-IV-a and C-IV-d become real.

Finally, when the minimization conditions $\lambda_4 = 0$, and $\lambda_2 + \lambda_3 = 0$, and $\lambda_7 = 0$ are applied, the scalar potential acquires an additional SU(2) symmetry:

$$U = \begin{pmatrix} e^{i\varphi_{1}}c_{\theta} & e^{i\varphi_{2}}s_{\theta} & 0\\ -e^{-i\varphi_{2}}s_{\theta} & e^{-i\varphi_{1}}c_{\theta} & 0\\ 0 & 0 & 1 \end{pmatrix}.$$
 (2.3.13)

The only vacuum configuration which falls under this category is C-V. It follows that due to the SU(2) transformation, the σ_i phases can be rotated away. As a result, C-V becomes real.

So far we have assumed that there is a unitary transformation U (2.3.7) that leaves the scalar potential invariant. This implies that the VEVs were considered to be independent as the whole SU(2) doublets were transformed. To be more precise, the Goldstone theorem examines a continuous symmetry which is spontaneously broken by the ground state. The aforementioned transformations still hold true as the minimization conditions in terms of λ_i are characteristics of a specific model. The only models with additional Goldstone bosons which were not considered are: {C-III-c, C-IV-c, C-IV-f }.

Before we proceed, we need to define which parameters contribute to the mass-squared matrices. The mass-squared matrix parameters arise from either bilinear terms or quartic terms of the scalar potential:

$$\mathcal{M}_{\xi_i^*\xi_j}^2 \propto V_2\left(\xi_i^*\xi_j\right),\tag{2.3.14a}$$

$$\mathcal{M}_{\xi_i^*\xi_i}^2 \propto V_4\left(\hat{w}_k \hat{w}_l \xi_i^* \xi_j\right),\tag{2.3.14b}$$

where ξ are the gauge fields of eq. (2.1.3). Of particular interest are the λ terms (2.3.14b) as the self-consistency conditions result in some of them set to zero, see eq. (2.2.3) for the real case and eq. (2.2.8) for the complex case. Consider the R-I-1 model. Vacuum configuration is given by: $\{0, 0, w_S\}$. From eq. (2.3.14) it follows that the mass-squared terms can be written as:

$$\mathcal{M}^2 = \mu_0^2 \mathcal{M}_0^2 + \sum_{i=5}^8 \lambda_i \mathcal{M}_i^2.$$
(2.3.15)

The R-I-1 does not result in any minimization conditions in terms of λ_i and therefore there are no additional Goldstone bosons. On the other hand, the R-II-2 vacuum configuration is given by $\{0, w, 0\}$, and the model is constrained by $\lambda_4 = 0$. The general mass-squared matrix is given by:

$$\mathcal{M}^2 = \mu_1^2 \mathcal{M}_0^2 + \sum_{i=1}^7 \lambda_i \mathcal{M}_i^2, \qquad (2.3.16)$$

and if not for the $\lambda_4 = 0$ constraint, there would have been an additional contribution in terms of the λ_4 coupling. However, the $\lambda_4 = 0$ constraint results in an additional SO(2) symmetry and this leads to the Goldstone boson.

For the C-III-c vacuum configuration we get a continuous SO(2) symmetry due to the minimization condition $\lambda_4 = 0$. This explains why there is one massless state. A closer inspection reveals that there are two massless states and not just one, see Table 2.3. Therefore another group of dimension dim = 1 should be broken for the Goldstone theorem to hold. The VEV of the S_3 singlet is given by $\langle h_S \rangle = 0$. This leads to the fact that although $\lambda_7 = 0$ is inconsistent with the C-III-c model, it is the VEV of the S_3 singlet which results in an additional U(1) symmetry, with U(1) given by eq. (2.3.12) with $\theta_7 = 0$. To distinguish this symmetry from the one which holds for the SU(2) doublets transformations, we denote it $U(1)_w$.

In section 3.2 we discuss a possible solution to promote massless scalars to massive ones with soft symmetry breaking. Models with soft symmetry breaking parameters, which are discussed in section 3.2, "eat" the minimization condition $\lambda_2 + \lambda_3 = 0$ and thus the corresponding massless states are promoted to massive ones.

Other interesting cases are the vacuum configurations C-IV-c and C-IV-f. The minimization conditions in terms of quartic couplings $\lambda_4 = -2A(\lambda_2 + \lambda_3)$ and $\lambda_7 = A^2(\lambda_2 + \lambda_3)$ are more involved but both of these minimization conditions depend on the sum of couplings $(\lambda_2 + \lambda_3)$ and a proportionality coefficient:

$$A = \frac{c_{\sigma_2 - \sigma_1} w_2}{c_{\sigma_1} \hat{w}_S}.$$
 (2.3.17)

The C-IV-c vacuum configuration is contained within the C-IV-f model in the limit $\sigma_1 = 0$. Therefore, we consider a single coefficient A for both cases. The minimization conditions can be expressed in a slightly different way:

$$\lambda_3 = -\lambda_2 - \frac{1}{2A}\lambda_4,$$

$$\lambda_7 = -\frac{A}{2}\lambda_4.$$
(2.3.18)

This way the V'_{23} of eq. (2.3.8) is visible, which is U(2) invariant. The terms which are not $U(2) \otimes U(1)$ invariant are multiplied by the λ_4 coupling. After analysing the expanded scalar potential in terms of the gauge fields and VEVs, it becomes obvious that the vacuum acquires an additional $U(1)_w$ symmetry. This is quite an involved statement but becomes more apparent when three facts are taken into consideration. First of all, both members of the S_3 doublets depend on a single \hat{w}_2 . Secondly, the minimization conditions (2.3.18) depend on $\frac{\hat{w}_2}{\hat{w}_S}$. Finally, there are two constraint in terms of the λ_i couplings which result in a specific direction of the scalar potential. A more convincing statement is that models C-III-d/e³, and C-III-h do not result in additional massless states as not all of the aforementioned conditions are satisfied. The minimization conditions for the C-III-d and C-III-e models can be expressed as $\lambda_3 = -\lambda_2 + \text{terms with } \lambda_4$ and λ_7 . Consider the C-III-d model. There is a direction:

$$(\lambda_2 + \lambda_3) \left(\hat{w}_1^2 - \hat{w}_2^2 \right) + \lambda_4 \hat{w}_2 \hat{w}_S = 0, \text{ not satisfied},$$
 (2.3.19)

which results in an additional massless state. Although there exists such direction, it is not fixed by the minimization conditions. The other case is the C-III-h model. This time, an additional massless state would be present if the following constraint would have been satisfied:

$$4(\lambda_2 + \lambda_3) c_{\sigma_2}^2 \hat{w}_2^2 - 4\lambda_7 c_{\sigma_2}^4 \hat{w}_S^2 = 0, \text{ not satisfied.}$$
(2.3.20)

Therefore this explains why C-III-d, and C-III-e, and C-III-h share some of the properties with C-IV-c and C-IV-f, but only the last two vacuum configurations result in additional massless states. On the other hand, the C-IV-e model resembles the C-IV-f model. However, the C-IV-e model is supplemented by $\lambda_4 = 0$ and results in an additional SO(2) symmetry.

The Goldstone theorem should be applicable to all of the discussed cases in this section. We take a look at which symmetries are broken and if the number of massless states is in agreement with the number of broken generators in Table 2.4.

2.4 Dark Matter Candidates

We are interested in a possible scalar DM candidate and thus take a look at which vacuum configurations might result in a plausible explanation. For the DM study we may⁴ impose an additional \mathbb{Z}_2 symmetry:

$$\begin{pmatrix} h_{\text{active}} \\ h_{\text{inert}} \end{pmatrix} = \text{diag}\left(1, -1\right) \begin{pmatrix} h_i \\ h_j \end{pmatrix},$$
 (2.4.1)

under which at least one of the SU(2) scalar doublets is even, and such doublets are called active, while the other doublets are odd, and those are inert doublets, and would accommodate the DM. The \mathbb{Z}_2 symmetry prevents couplings between the SM particles and a single inert particle at treelevel. As a consequence, the lightest \mathbb{Z}_2 -odd particle is stable and is a possible DM candidate.

All of the SU(2) scalar doublets in the S_3 scalar potential come in pairs except for the SU(2)doublets that are multiplied by λ_4 . This is the only coupling which breaks the \mathbb{Z}_2 symmetry for h_2 and h_S . Provided that the DM candidate resides in h_2 or h_S it is a must to impose the constraint $\lambda_4 = 0$, but if the inert SU(2) doublet corresponds to h_1 , then the $\lambda_4 = 0$ constraint is not necessary.

Not all of the vacuum configurations with zero VEV components result in a possible DM candidate. The \mathbb{Z}_2 can be broken, based on the $\lambda_4 = 0$ constraint. Also we take a look at

³These two vacuum configurations can be analysed together as those differ only by the sign of $\langle h_2 \rangle$.

⁴It should be noted that, in principle, the \mathbb{Z}_2 symmetry might be a smaller symmetry and thus a subgroup of a larger symmetry under which the scalar potential is invariant.

Vacuum	Broken symmetries	Number of broken generators	Additional $m^0 = 0$	λ constraints
C-IV-c, C-IV-f	$U(1)_w$	1	1	$\lambda_4 = -2A \left(\lambda_2 + \lambda_3\right),$ $\lambda_7 = A^2 \left(\lambda_2 + \lambda_3\right)$
R-II-2, R-II-3, R-III,				
C-III-b, C-III-f, C-III-g,	SO(2)	1	1	$\lambda_4 = 0$
C-IV-b, C-IV-e				
C-III-c	$SO(2) \otimes U(1)_w$	1+1	2	$\lambda_4 = 0,$
				$\lambda_2 + \lambda_3 = 0$
C-IV-a C-IV-d	$SO(2) \otimes U(1)$	1+1	2	$\lambda_4 = 0,$
		I I	-	$\lambda_7 = 0$
	SU(2)	3	3	$\lambda_4 = 0,$
C-V				$\lambda_2 + \lambda_3 = 0,$
				$\lambda_7 = 0$

Table 2.4: Comparison of the number of broken generators with the number of additional massless states. The $U(1)_w$ group indicates that additional symmetry is reached through the vacuum. The other cases result in a symmetry of the SU(2) doublets.

the mass-squared matrices to determine if there is mixing present between the inert and active SU(2) doublets. This is a trivial task and is based on determining if the mass-squared matrix is block-diagonal in the basis of the fields which correspond to the active-inert SU(2) doublets. For simplicity, we write down all of the possible vacuum configurations and specify their properties. Results are presented in Table 2.5. It should be noted that only a basic check was performed and thus the mentioned DM candidates are possible but may not result in a viable model.

The only two vacuum configurations with a single inert doublet h_1 are R-II-1a and C-III-a. An interesting observation is that although vacua R-I-2b, and R-I-2c, and C-I-a involve zero VEVs, there is mixing present in the mass-squared matrices and therefore these models fail to describe DM. Cases R-II-2, and C-III-b, and C-IV-a result in one neutral inert massless scalar state. Another interesting observation is that if the h_2 or $h_S SU(2)$ doublets are the DM candidates, this results in a requirement that $\lambda_4 = 0$, and therefore at least one massless scalar state is present provided that the scalar potential is exactly S_3 -symmetric.

Vacuum	Vacuum	Additional	Mass-degenerate	Symmetries	Possible DM
Vacuum	Configuration	$m^0 = 0$	states		candidate
R-I-1	$\{0, 0, w_S\}$		$1H^{\pm}$ and $2H^0$	\mathbb{Z}_2	
R-I-2a	$\{w,0,0\}$			\mathbb{Z}_2	
R-I-2b	$\{w,\sqrt{3}w,0\}$				
R-I-2c	$\{w,-\sqrt{3}w,0\}$				
R-II-1a	$\{0, w, w_S\}$				h_1
R-II-1b	$\{w, -w/\sqrt{3}, w_S\}$				
R-II-1c	$\{w, w/\sqrt{3}, w_S\}$				
R-II-2	$\{0, w, 0\}$	1		SO(2)	h_1, h_S

R-II-3	$\{w_1, w_2, 0\}$	1		SO(2)	h_S	
R-III	$\{w_1, w_2, w_S\}$	1		SO(2)		
C-I-a	$\{\hat{w}_1, \pm i\hat{w}_1, 0\}$		$2H^0$			
C-III-a	$\{0, \hat{w}_2 e^{i\sigma_2}, \hat{w}_S\}$				h_1	
C-III-b	$\{\pm i\hat{w}_1, 0, \hat{w}_S\}$	1		SO(2)	h_2	
C-III-c	$\{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, 0\}$	2		$SO(2) \otimes U(1)_w$	h_S	
C-III-d	$\{\pm i \hat{w}_1, \hat{w}_2, \hat{w}_S\}$					
C-III-e	$\{\pm i \hat{w}_1, -\hat{w}_2, \hat{w}_S\}$					
C-III-f	$\{\pm i\hat{w}_1,i\hat{w}_2,\hat{w}_S\}$	1		SO(2)		
C-III-g	$\{\pm i\hat{w}_1, -i\hat{w}_2, \hat{w}_S\}$	1		SO(2)		
C-III-h	$\{\sqrt{3}\hat{w}_2e^{i\sigma_2}, \pm\hat{w}_2e^{i\sigma_2}, \hat{w}_S\}$					
C-III-i	$\{\sqrt{\frac{3(1+\mathbf{t}_{\sigma_{1}}^{2})}{1+9\mathbf{t}_{\sigma_{1}}^{2}}}\hat{w}_{2}e^{i\sigma_{1}},$					
	$\pm \hat{w}_2 e^{-i \arctan(3 t_{\sigma_1})}, \hat{w}_S \}$					
C-IV-a	$\{\hat{w}_1 e^{i\sigma_1},0,\hat{w}_S\}$	2		$SO(2)\otimes U(1)$	h_2	
C-IV-b	$\{\hat{w}_1, \pm i\hat{w}_2, \hat{w}_S\}$	1		SO(2)		
C-IV-c	$\{\sqrt{1+2c_{\sigma_2}^2}\hat{w}_2,$	1	1		U(1)	
	$\hat{w}_2 e^{i\sigma_2}, \hat{w}_S \}$	1				
C-IV-d	$\{\hat{w}_1 e^{i\sigma_1}, \pm \hat{w}_2 e^{i\sigma_1}, \hat{w}_S\}$	2		$SO(2)\otimes U(1)$		
C-IV-e	$\{\sqrt{-\frac{\mathbf{s}_{2\sigma_2}}{\mathbf{s}_{2\sigma_1}}}\hat{w}_2e^{i\sigma_1},$	1		SO(2)		
	$\hat{w}_2 e^{i\sigma_2},\hat{w}_S\}$	Ĩ		50(2)		
C-IV-f	$\{\sqrt{2+\frac{\mathbf{c}_{(\sigma_1-2\sigma_2)}}{\mathbf{c}_{\sigma_1}}}\hat{w}_2e^{i\sigma_1},$	1		U(1)		
U-1 V-1	$\hat{w}_2 e^{i\sigma_2}, \hat{w}_S \}$	1		U (1)w		
C-V	$\{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, \hat{w}_S\}$	3		SU(2)		

Table 2.5: Properties of different vacuum configurations. In the third column, the number of additional massless states is presented. In the fourth column, the number of mass-degenerate states is shown, *e.g.*, $2H^0$ indicates that there are in total two massive mass-degenerate pairs, values of which are not the same, *i.e.*, $m_{H_i} = m_{H_j} \neq 0$, and $m_{H_k} = m_{H_i} \neq 0$, and $m_{H_i} \neq m_{H_k}$. Same notation applies to the charged mass-degenerate states denoted by $1H^{\pm}$. In the fifth column, additional symmetries of the potential, after applying the minimization conditions in terms of λ_i , are presented, and \mathbb{Z}_2 indicates that the \mathbb{Z}_2 symmetry is broken. In the last column, inert SU(2) doublets are written down. These doublets do not violate the \mathbb{Z}_2 symmetry and the mass-squared matrices do not mix with active SU(2) doublets.

2.5 The Higgs Basis Transformation

Physical quantities are basis invariant and therefore a convenient choice would be to work in the so-called Higgs basis. In case of the 2HDM, the Higgs basis [44, 45] is defined as a basis in which

one of the VEVs has a zero value. A more general definition of the Higgs basis, applicable to the multi-Higgs-doublet model, is that it is a basis in which only one VEV acquires a non-zero value:

$$\langle H_1 \rangle = \frac{v}{\sqrt{2}}, \quad \langle H_i \rangle = 0,$$
 (2.5.1)

where $\langle H_i \rangle$ indicates the vacuum of the specific SU(2) doublet, the H_i SU(2) doublets indicate that those are in the Higgs basis, and the VEV of the H_1 doublet is real. Another property of the Higgs basis is that the would-be Goldstone bosons are isolated in the H_1 doublet.

Assume that the Higgs basis rotation from the generic basis h_i to the Higgs basis H_i is given by the following transformation:

$$H_i = \mathcal{R}_{ij} h_j, \tag{2.5.2}$$

or equivalently:

$$h_i = \mathcal{R}^*_{ji} H_j, \tag{2.5.3}$$

where the rotation matrix \mathcal{R} is unitary. In the Higgs basis the S_3 symmetry is not explicit anymore and therefore we define the SU(2) doublets as: $\{H_1, H_2, H_3\}$.

The scalar potential under eq. (2.5.3) transforms as⁵:

$$\left(h_{\bar{i}}^{\dagger}h_{\bar{j}}\right) = \mathcal{R}_{i\bar{i}}\mathcal{R}_{j\bar{j}}^{*}\left(H_{i}^{\dagger}H_{j}\right), \qquad (2.5.4a)$$

$$\left(h_{\bar{i}}^{\dagger}h_{\bar{j}}\right)\left(h_{\bar{k}}^{\dagger}h_{\bar{l}}\right) = \frac{1}{2}c_{ijkl}\left[\mathcal{R}_{i\bar{i}}\mathcal{R}_{j\bar{j}}^{*}\mathcal{R}_{k\bar{k}}\mathcal{R}_{l\bar{l}}^{*} + \mathcal{R}_{k\bar{i}}\mathcal{R}_{l\bar{j}}^{*}\mathcal{R}_{i\bar{k}}\mathcal{R}_{j\bar{l}}^{*}\right]\left(H_{i}^{\dagger}H_{j}\right)\left(H_{k}^{\dagger}H_{l}\right), \qquad (2.5.4b)$$

where additional terms of the quadratic transformation arise due to

$$\left(H_{i}^{\dagger}H_{j}\right)\left(H_{k}^{\dagger}H_{l}\right) = \left(H_{k}^{\dagger}H_{l}\right)\left(H_{i}^{\dagger}H_{j}\right), \qquad (2.5.5)$$

and c_{ijkl} is a symmetry factor:

$$c_{ijkl} = \begin{cases} 1, & \text{if } i = k \text{ and } j = l \\ 2, & \text{otherwise} \end{cases}$$
(2.5.6)

The scalar potential in the Higgs basis in the SU(2)-covariant form is given by [13, 46]:

$$V^{\rm HB} = Y_{ij} \left(H_i^{\dagger} H_j \right) + Z_{ijkl} \left(H_i^{\dagger} H_j \right) \left(H_k^{\dagger} H_l \right), \qquad (2.5.7)$$

where due to hermiticity the following relations hold:

$$Y_{ij} = Y_{ji}^*,$$
 (2.5.8a)

$$Z_{ijkl} = Z_{jilk}^*, \qquad Z_{ijkl} = Z_{klij}. \tag{2.5.8b}$$

The quadratic coupling in the Higgs basis are^6 :

$$Y_{ij} = \frac{1}{2}\nu^{2} \left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{j\bar{2}}^{*} + \text{h.c.} \right) + \mu_{0}^{2}\mathcal{R}_{i\bar{3}}\mathcal{R}_{j\bar{3}}^{*} + \mu_{1}^{2} \left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{j\bar{1}}^{*} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{j\bar{2}}^{*} \right), + \mu_{2}^{2} \left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{j\bar{1}}^{*} - \mathcal{R}_{i\bar{2}}\mathcal{R}_{j\bar{2}}^{*} \right) + \frac{1}{2}\mu_{3}^{2} \left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{j\bar{3}}^{*} + \text{h.c.} \right) + \frac{1}{2}\mu_{4}^{2} \left(\mathcal{R}_{i\bar{2}}\mathcal{R}_{j\bar{3}}^{*} + \text{h.c.} \right),$$
(2.5.9)

 $^{{}^{5}}$ We are not using the summation convention. The barred indices are not contracted with the un-barred.

 $^{^{6}}$ For completeness, we assume that soft symmetry breaking terms are added, see section 3.2.

and the quartic couplings are:

$$\begin{split} Z_{ijkl} = &\lambda_{1}c_{ijkl} \left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{j\bar{1}}^{*} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{j\bar{2}}^{*} \right) \left(\mathcal{R}_{k\bar{1}}\mathcal{R}_{l\bar{1}}^{*} + \mathcal{R}_{k\bar{2}}\mathcal{R}_{l\bar{2}}^{*} \right) \\ &+ \lambda_{2}c_{ijkl} \left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{j\bar{2}}^{*} - \mathcal{R}_{i\bar{2}}\mathcal{R}_{j\bar{1}}^{*} \right) \left(\mathcal{R}_{k\bar{1}}\mathcal{R}_{l\bar{2}}^{*} - \mathcal{R}_{k\bar{2}}\mathcal{R}_{l\bar{1}}^{*} \right) \\ &+ \lambda_{3}c_{ijkl} \left[\mathcal{R}_{j\bar{1}}^{*} \left(\left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{1}} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{2}} \right) \mathcal{R}_{l\bar{1}}^{*} - \left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{1}} - \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{1}} \right) \mathcal{R}_{l\bar{2}}^{*} \right) \\ &+ \mathcal{R}_{j\bar{2}}^{*} \left(\left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{1}} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{2}} \right) \mathcal{R}_{l\bar{2}}^{*} - \left(\mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{1}} - \mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{2}} \right) \mathcal{R}_{l\bar{1}}^{*} \right) \right] \\ &+ \lambda_{4}c_{ijkl} \left[\mathcal{R}_{j\bar{1}}^{*} \left(\left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{2}} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{3}} \right) \mathcal{R}_{l\bar{1}}^{*} + \left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{1}} + \mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{2}} \right) \mathcal{R}_{l\bar{3}}^{*} \right) \\ &+ \mathcal{R}_{j\bar{2}}^{*} \left(\left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{2}} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{3}} \right) \mathcal{R}_{l\bar{1}}^{*} - \left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{2}} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{3}} \right) \mathcal{R}_{l\bar{2}}^{*} \right) \\ &+ \mathcal{R}_{j\bar{2}}^{*} \left(\left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{1}} + \mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{3}} \right) \mathcal{R}_{l\bar{1}}^{*} - \left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{2}} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{3}} \right) \mathcal{R}_{l\bar{2}}^{*} \right) \\ &+ \mathcal{R}_{j\bar{2}}^{*} \left(\left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{1}} + \mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{3}} \right) \mathcal{R}_{l\bar{1}}^{*} - \left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{2}} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{3}} \right) \mathcal{R}_{l\bar{2}}^{*} \right) \right] \\ &+ \mathcal{R}_{j\bar{3}}^{*} \left(\left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{1}} + \mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{3}} \right) \mathcal{R}_{l\bar{1}}^{*} - \left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{k\bar{2}} + \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{2}} \right) \mathcal{R}_{l\bar{2}}^{*} \right) \right] \\ &+ \mathcal{R}_{j\bar{3}}^{*} \left(\left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{j\bar{3}}^{*} \left(\mathcal{R}_{k\bar{1}}\mathcal{R}_{l\bar{1}} + \mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{3}} \right) \mathcal{R}_{l\bar{1}}^{*} + \left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{k\bar{1}} - \mathcal{R}_{i\bar{2}}\mathcal{R}_{k\bar{2}} \right) \right] \\ &+ \mathcal{R}_{j\bar{3}}^{*} \left(\left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{l\bar{3}}^{*} \left(\mathcal{R}_{k\bar{1}}\mathcal{R}_{l\bar{1}}^{*} + \mathcal{R}_{k\bar{2}}\mathcal{R}_{l\bar{2}}^{*} \right) \right) \right) \\ &+ \mathcal{R}_{j\bar{3}}^{*} \left(\mathcal{R}_{i\bar{3}}\mathcal{R}_{l\bar{3}}^{*} \left(\mathcal{R}_{i\bar{1}}\mathcal{R}_{l\bar{1}}^{*} + \mathcal{R}_{k\bar{2}}\mathcal{R}_{$$

Consider the most general vacuum configuration:

$$\{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, \hat{w}_S\}.$$
 (2.5.11)

Such vacuum configuration is first rotated into an intermediate basis, where VEVs are expressed only in terms of the absolute value \hat{w}_i :

$$\begin{pmatrix} h_1' \\ h_2' \\ h_S' \end{pmatrix} = \operatorname{diag} \left(e^{-i\sigma_1}, e^{-i\sigma_2}, 1 \right) \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}.$$
 (2.5.12)

The next step is to rotate the modulus of VEVs. The most trivial approach would be to consider the Euler rotation matrices. In total, three components \hat{w}_i need to be rotated into a single v(2.1.4). Such rotation can be performed in terms of just two angles. This indicates that at least one of the \mathcal{R} components of eq. (2.5.2) will become zero. Based on a choice of the Euler rotation matrix, appropriate changes of the couplings given by eqs. (2.5.9, 2.5.10) are expected as there is a migrating zero. It follows that the Higgs basis is not unique and results in a spectrum of Higgs bases. Moreover, the Higgs basis is not well-defined as there is a freedom to redefine the SU(2)doublets with vanishing VEVs, see Refs. [12, 13, 47]. In our case it is possible to rotate the H_2 and H_3 doublets by an additional U(2) transformation. Such transformation could potentially simplify some of the couplings.

As stressed, the Higgs basis is not uniquely defined. One of the possibilities is to consider that we first rotate $\langle h_S \rangle$ into $\langle h_2 \rangle$ so it becomes $\langle h'_2 \rangle$ and then $\langle h'_2 \rangle$ into $\langle h_1 \rangle$. The Higgs basis rotation in this case is given by:

$$\begin{pmatrix}
H_{1} \\
H_{2} \\
H_{3}
\end{pmatrix} = \begin{pmatrix}
c_{\beta_{1}} & s_{\beta_{1}} & 0 \\
-s_{\beta_{1}} & c_{\beta_{1}} & 0 \\
0 & 0 & 1
\end{pmatrix} \begin{pmatrix}
1 & 0 & 0 \\
0 & c_{\beta_{2}} & s_{\beta_{2}} \\
0 & -s_{\beta_{2}} & c_{\beta_{2}}
\end{pmatrix} \begin{pmatrix}
e^{-i\sigma_{1}} & 0 & 0 \\
0 & e^{-i\sigma_{2}} & 0 \\
0 & 0 & 1
\end{pmatrix} \begin{pmatrix}
h_{1} \\
h_{2} \\
h_{S}
\end{pmatrix}$$

$$= \begin{pmatrix}
e^{-i\sigma_{1}}c_{\beta_{1}} & e^{-i\sigma_{2}}s_{\beta_{1}}c_{\beta_{2}} & s_{\beta_{1}}s_{\beta_{2}} \\
-e^{-i\sigma_{1}}s_{\beta_{1}} & e^{-i\sigma_{2}}c_{\beta_{1}}c_{\beta_{2}} & c_{\beta_{1}}s_{\beta_{2}} \\
0 & -e^{-i\sigma_{2}}s_{\beta_{2}} & c_{\beta_{2}}
\end{pmatrix} \begin{pmatrix}
h_{1} \\
h_{2} \\
h_{S}
\end{pmatrix},$$
(2.5.13)

The rotation of the SU(2) doublets from the generic basis (2.1.3, 2.1.7) to the Higgs basis is therefore:

$$h_1 = e^{i\sigma_1} \left(H_1 c_{\beta_1} - H_2 s_{\beta_1} \right), \tag{2.5.14a}$$

$$h_2 = e^{i\sigma_2} \left(H_1 c_{\beta_2} s_{\beta_1} + H_2 c_{\beta_1} c_{\beta_2} - H_3 s_{\beta_2} \right), \qquad (2.5.14b)$$

$$h_S = (H_1 s_{\beta_1} + H_2 c_{\beta_1}) s_{\beta_2} + H_3 c_{\beta_2}.$$
(2.5.14c)

The choice of the Euler rotation matrix results in $\mathcal{R}_{31} = 0$ and therefore h_1 does not contribute to H_3 .

The Higgs basis rotation depends on whether one of the VEVs is zero. When $\hat{w}_2 \neq 0$, the β_i angles are given by:

$$\mathbf{c}_{\beta_1} = \frac{\hat{w}_1}{v},\tag{2.5.15a}$$

$$t_{\beta_2} = \frac{\hat{w}_S}{\hat{w}_2},$$
 (2.5.15b)

and if $\hat{w}_2 = 0$, *i.e.*, {R-I-1, R-I-2a, C-III-b, C-IV-a}, we get that:

$$\mathbf{t}_{\beta_1} = \frac{\hat{w}_S}{\hat{w}_1},\tag{2.5.16a}$$

$$\beta_2 = \frac{\pi}{2}.$$
 (2.5.16b)

If both \hat{w}_1 and \hat{w}_2 vanish, *i.e.*, the R-I-1 model case, then the β_i angles are fixed:

$$\beta_1 = \beta_2 = \frac{\pi}{2},\tag{2.5.17}$$

so that such rotation results in a translation of the h_S doublet into h_1 .

Relations between the Higgs basis and the generic basis are presented in Appendix C.1. An interesting observation is that not all of the Z couplings depend on the λ_2 coupling. This is due to the basis transformation of eq. (2.5.14) $\mathcal{R}_{31} = 0$. The Z_{ijkl} couplings, which do not depend on the λ_2 coupling are: { Z_{1133} , Z_{2233} , Z_{3333} , Z_{1233} , Z_{1333} , Z_{2333} }. The latter three couplings are complex. Another interesting consequence of the choice of transformation (2.5.14) is that the couplings Z_{1233} and Z_{1332} become real.

For completeness, we take a look at a more general transformation, *i.e.*, when H_2 and H_3 doublets are transformed by U(2). The most general U(2) transformation is as follows:

$$U = e^{\frac{i\phi}{2}} \begin{pmatrix} e^{i\phi_1} \mathbf{c}_{\theta} & e^{i\phi_2} \mathbf{s}_{\theta} \\ -e^{-i\phi_2} \mathbf{s}_{\theta} & e^{-i\phi_1} \mathbf{c}_{\theta} \end{pmatrix}.$$
 (2.5.18)

Promoting this to the Higgs basis transformation results in:

$$\begin{pmatrix} H_1 \\ H_2 \\ H_3 \end{pmatrix} = \begin{pmatrix} \mathcal{U}_{11} & \mathcal{U}_{12} & \mathcal{U}_{13} \\ \mathcal{U}_{21} & \mathcal{U}_{22} & \mathcal{U}_{23} \\ \mathcal{U}_{31} & \mathcal{U}_{32} & \mathcal{U}_{33} \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix},$$
(2.5.19)

where

$$\mathcal{U}_{11} = e^{-i\sigma_1} c_{\beta_1}, \tag{2.5.20a}$$

$$\mathcal{U}_{12} = e^{-i\sigma_2} c_{\beta_2} s_{\beta_1}, \tag{2.5.20b}$$

$$\mathcal{U}_{13} = s_{\beta_1} s_{\beta_2}, \tag{2.5.20c}$$

$$\mathcal{U}_{21} = -e^{\frac{1}{2}i(-2\sigma_1 + \phi + 2\phi_1)} c_\theta s_{\beta_1}, \qquad (2.5.20d)$$

$$\mathcal{U}_{22} = e^{\frac{1}{2}i(\phi - 2\sigma_2)} \left(e^{i\phi_1} c_{\beta_1} c_{\beta_2} c_{\theta} - e^{i\phi_2} s_{\beta_2} s_{\theta} \right),$$
(2.5.20e)

$$\mathcal{U}_{23} = e^{\frac{i\phi}{2}} \left(e^{i\phi_2} c_{\beta_2} s_\theta + e^{i\phi_1} c_{\beta_1} c_\theta s_{\beta_2} \right),$$
(2.5.20f)

$$\mathcal{U}_{31} = s_{\beta_1} s_{\theta} e^{\frac{1}{2}i(\phi - 2(\sigma_1 + \phi_2))}, \tag{2.5.20g}$$

$$\mathcal{U}_{32} = -e^{\frac{1}{2}i(\phi - 2(\sigma_2 + \phi_1 + \phi_2))} \left(e^{i\phi_1} c_{\beta_1} c_{\beta_2} s_\theta + e^{i\phi_2} c_\theta s_{\beta_2} \right), \qquad (2.5.20h)$$

$$\mathcal{U}_{33} = e^{\frac{i\phi}{2}} \left(e^{-i\phi_1} c_{\beta_2} c_{\theta} - e^{-i\phi_2} c_{\beta_1} s_{\beta_2} s_{\theta} \right).$$
(2.5.20i)

The resulting rotation from the generic basis to the Higgs basis is given by eq. (2.5.3) with $\mathcal{R}_{ji} \to \mathcal{U}_{ji}$. From the definition of the \mathcal{U}_{ij} components (2.5.20) it follows that in this case all of the generic doublets h_i transform into all of the Higgs doublets H_j .

To emphasize the migrating zero feature of the Higgs basis transformation, when an additional U(2) transformation of the H_2 and H_3 is not regarded, consider that $\langle h_2 \rangle$ is rotated into $\langle h_1 \rangle$ and $\langle h_S \rangle$ into $\langle h'_1 \rangle$. In this case, the Higgs basis transformation is:

$$\begin{pmatrix} H_1 \\ H_2 \\ H_3 \end{pmatrix} = \begin{pmatrix} c_{\beta_2} & 0 & s_{\beta_2} \\ 0 & 1 & 0 \\ -s_{\beta_2} & 0 & c_{\beta_2} \end{pmatrix} \begin{pmatrix} c_{\beta_1} & s_{\beta_1} & 0 \\ -s_{\beta_1} & c_{\beta_1} & 0 \\ 0 & 0 & 1 \end{pmatrix} \begin{pmatrix} e^{-i\sigma_1} & 0 & 0 \\ 0 & e^{-i\sigma_2} & 0 \\ 0 & 0 & 1 \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}$$

$$= \begin{pmatrix} e^{-i\sigma_1}c_{\beta_1}c_{\beta_2} & e^{-i\sigma_2}s_{\beta_1}c_{\beta_2} & s_{\beta_2} \\ -e^{-i\sigma_1}s_{\beta_1} & e^{-i\sigma_2}c_{\beta_1} & 0 \\ -e^{-i\sigma_1}c_{\beta_1}s_{\beta_2} & -e^{-i\sigma_2}s_{\beta_1}s_{\beta_2} & c_{\beta_2} \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix},$$

$$(2.5.21)$$

and the transformation from the generic basis to the Higgs basis is given by:

$$h_1 = -e^{i\sigma_1} \left(-H_1 c_{\beta_1} c_{\beta_2} + H_2 s_{\beta_1} + H_3 c_{\beta_1} s_{\beta_2} \right), \qquad (2.5.22a)$$

$$h_2 = e^{i\sigma_2} \left(H_1 \mathbf{s}_{\beta_1} \mathbf{c}_{\beta_2} + H_2 \mathbf{c}_{\beta_1} - H_3 \mathbf{s}_{\beta_1} \mathbf{s}_{\beta_2} \right),$$
(2.5.22b)

$$h_S = H_1 \mathbf{s}_{\beta_2} + H_3 \mathbf{c}_{\beta_2}. \tag{2.5.22c}$$

The zero element is \mathcal{R}_{23} and thus h_S does not depend on H_2 . This transformation is equivalent to the Higgs basis transformation suggested in Ref. [48]:

$$\begin{pmatrix} H_1 \\ H_2 \\ H_3 \end{pmatrix} = \begin{pmatrix} \frac{\hat{w}_1}{N_1} & \frac{\hat{w}_2}{N_1} & \frac{\hat{w}_S}{N_1} \\ \frac{\hat{w}_2}{N_2} & -\frac{\hat{w}_1}{N_2} & 0 \\ \frac{\hat{w}_1}{N_3} & \frac{\hat{w}_2}{N_3} & \frac{X}{N_3} \end{pmatrix} \begin{pmatrix} e^{-i\sigma_1} & 0 & 0 \\ 0 & e^{-i\sigma_2} & 0 \\ 0 & 0 & 1 \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix},$$
(2.5.23)

where

$$N_1^2 = \hat{w}_1^2 + \hat{w}_2^2 + \hat{w}_S^2 \equiv v^2, \qquad (2.5.24a)$$

$$N_2^2 = \hat{w}_1^2 + \hat{w}_2^2 \equiv w^2, \qquad (2.5.24b)$$

$$N_3^2 = \hat{w}_1^2 + \hat{w}_2^2 + X^2, \qquad (2.5.24c)$$

with

$$X = -\frac{w^2}{\hat{w}_S}.$$
 (2.5.25)

Inverting eq. (2.5.23) results in transformation from the generic basis to the Higgs basis:

$$h_1 = \frac{e^{i\sigma_1}}{N_1 N_2 N_3} (N_2 N_3 \hat{w}_1 H_1 + N_1 N_3 \hat{w}_2 H_2 + N_1 N_2 \hat{w}_1 H_3), \qquad (2.5.26a)$$

$$h_2 = \frac{e^{i\sigma_2}}{N_1 N_2 N_3} (N_2 N_3 \hat{w}_2 H_1 - N_1 N_3 \hat{w}_1 H_2 + N_1 N_2 \hat{w}_2 H_3), \qquad (2.5.26b)$$

$$h_S = \frac{1}{N_1 N_3 \hat{w}_S} (N_3 \hat{w}_S^2 H_1 - N_1 w^2 H_3).$$
(2.5.26c)

Relations between the Higgs basis and the generic basis are presented in Appendix C.2.

2.5.1 Choice of the Higgs Basis

For completeness, we discuss general properties of a specific Higgs basis choice. Consider the following Euler rotation matrices:

$$\mathcal{R}_x = \begin{pmatrix} 1 & 0 & 0\\ 0 & c_{\beta_x} & s_{\beta_x}\\ 0 & -s_{\beta_x} & c_{\beta_x} \end{pmatrix}, \qquad (2.5.27a)$$

$$\mathcal{R}_{y} = \begin{pmatrix} c_{\beta_{y}} & 0 & s_{\beta_{y}} \\ 0 & 1 & 0 \\ -s_{\beta_{y}} & 0 & c_{\beta_{y}} \end{pmatrix}, \qquad (2.5.27b)$$

$$\mathcal{R}_{z} = \begin{pmatrix} c_{\beta_{z}} & s_{\beta_{z}} & 0\\ -s_{\beta_{z}} & c_{\beta_{z}} & 0\\ 0 & 0 & 1 \end{pmatrix}.$$
 (2.5.27c)

The Higgs basis transformation is thus:

$$\begin{pmatrix} H_1 \\ H_2 \\ H_3 \end{pmatrix} = \mathcal{R}_{\beta_i} \mathcal{R}_{\beta_j} \begin{pmatrix} e^{-i\sigma_1} & 0 & 0 \\ 0 & e^{-i\sigma_2} & 0 \\ 0 & 0 & 1 \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}.$$
 (2.5.28)

We consider different combinations of $\mathcal{R}_{\beta_i}\mathcal{R}_{\beta_j}$ and in what specific direction of the scalar potential those result. Note that both H_2 and H_3 can further be mixed by a U(2) transformation, but we do not consider this additional transformation.

 $\mathcal{R}_{\beta_z}\mathcal{R}_{\beta_x}$ results in a rotation element $\mathcal{R}_{31} = 0$. This is the case of eq. (2.5.13). The Z_{ijkl} couplings then satisfy the following relations:

$$\operatorname{Im}(Z_{1112}) = -\operatorname{Im}(Z_{1222}), \qquad (2.5.29a)$$

$$Im(Z_{1123}) = -Im(Z_{1231}), \qquad (2.5.29b)$$

$$Im(Z_{1223}) = Im(Z_{1322}), \qquad (2.5.29c)$$

$$\operatorname{Im}(Z_{1233}) = \operatorname{Im}(Z_{1332}) = 0. \tag{2.5.29d}$$

 $\mathcal{R}_{\beta_x}\mathcal{R}_{\beta_z}$ results in a rotation element $\mathcal{R}_{13} = 0$. The Z_{ijkl} couplings then satisfy the following relations:

$$Z_{1233} = Z_{1332}, \tag{2.5.30a}$$

$$Z_{1223} = Z_{1322}, \tag{2.5.30b}$$

$$\operatorname{Im}(Z_{1123}) = -\operatorname{Im}(Z_{1231}). \tag{2.5.30c}$$

 $\mathcal{R}_{\beta_y}\mathcal{R}_{\beta_z}$ results in a rotation element $\mathcal{R}_{23} = 0$. This is the case of eq. (2.5.21). The Z_{ijkl} couplings then satisfy the following relations:

$$Z_{1123} = Z_{1231}^*, \tag{2.5.31a}$$

$$Z_{1233} = Z_{1332}, \tag{2.5.31b}$$

$$\operatorname{Im}(Z_{1223}) = -\operatorname{Im}(Z_{1322}). \qquad (2.5.31c)$$

 $\mathcal{R}_{\beta_z}\mathcal{R}_{\beta_y}$ results in a rotation element $\mathcal{R}_{32} = 0$. The Z_{ijkl} couplings then satisfy the following relations:

$$Im(Z_{1233}) = Im(Z_{1332}), \qquad (2.5.32a)$$

$$\operatorname{Im}(Z_{1123}) = -\operatorname{Im}(Z_{1231}), \qquad (2.5.32b)$$

$$\operatorname{Im}(Z_{1223}) = \operatorname{Im}(Z_{1322}). \tag{2.5.32c}$$

 $\mathcal{R}_{\beta_y}\mathcal{R}_{\beta_x}$ results in a rotation element $\mathcal{R}_{21} = 0$. The Z_{ijkl} couplings then satisfy the following relations:

$$\operatorname{Im}(Z_{1223}) = \operatorname{Im}(Z_{1322}) = 0, \qquad (2.5.33a)$$

$$Im(Z_{1233}) = Im(Z_{1332}), \qquad (2.5.33b)$$

$$\operatorname{Im}(Z_{1123}) = -\operatorname{Im}(Z_{1231}), \qquad (2.5.33c)$$

$$\operatorname{Im}(Z_{1113}) = -\operatorname{Im}(Z_{1333}). \tag{2.5.33d}$$

 $\mathcal{R}_{\beta_x}\mathcal{R}_{\beta_y}$ results in a rotation element $\mathcal{R}_{12} = 0$. The Z_{ijkl} couplings then satisfy the following relations:

$$Im(Z_{1123}) = -Im(Z_{1231}), \qquad (2.5.34a)$$

$$Im(Z_{1233}) = Im(Z_{1332}), \qquad (2.5.34b)$$

 $Im(Z_{1223}) = Im(Z_{1322}).$ (2.5.34c)

The total number of free parameters can be counted:

$$Y^{\mathbb{R}} + 2Y^{\mathbb{C}} + Z^{\mathbb{R}} + 2Z^{\mathbb{C}} = 3 + 2 \times 3 + 9 + 2 \times 18 = 54,$$
(2.5.35)

where $Y^{\mathbb{R}}$, $Z^{\mathbb{R}}$ indicate real couplings Y_{ii} , and Z_{iiii} , and Z_{ijji} , and $Y^{\mathbb{C}}$, $Z^{\mathbb{C}}$ stand for complex couplings. Assume that the Higgs basis transformation is given by eq. (2.5.13). Constraints (2.5.29) result in:

$$Y^{\mathbb{R}} + Y^{\mathbb{C}} + Z^{\mathbb{R}} + Z^{\mathbb{C}} = 3 + 3 + 11 + 16 = 33, \qquad (2.5.36)$$

counting the complex couplings as a single free parameter, e.g., Z_{1113} and Z_{1113}^* . This coincides with the number of free parameters (33) for the general renormalizable C-invariant potential in Ref. [30]:

$$\mathcal{N}_{\mathcal{C}} = \frac{1}{4} N \left(N^3 + 5N + 2 \right), \qquad (2.5.37)$$

where $\mathcal{N}_{\mathcal{C}}$ is the total number of free parameters and N indicates how many SU(2) doublets are considered.

The number of free parameters in the Higgs basis by far surpasses the number of couplings in the generic basis. The exact S_3 -symmetric scalar potential results in 2 quadratic and 8 quartic couplings, assuming that those are real. The soft breaking terms add another 4 quadratic couplings. All in all, there are (10+4) real parameters needed to specify the scalar potential. Some of the couplings in the generic basis could be promoted to complex ones: { ν^2 , μ_3^2 , μ_4^2 , λ_4 , λ_7 }. At most, this results in 12 parameters for the exact S_3 potential with complex couplings and 19 with softly broken ones. In both cases it is nowhere near the 54 free parameters. Therefore, a great amount of parameters in the Higgs basis are interdependent.

2.5.2 The Mass-Squared Matrices

We consider that the SU(2) doublets in the Higgs basis are given by:

$$H_1 = \begin{pmatrix} G^+ \\ \frac{1}{\sqrt{2}} \left(v + \tilde{\eta}_1 + iG^0 \right) \end{pmatrix}, \qquad (2.5.38a)$$

$$H_{2} = \begin{pmatrix} h_{2}^{+} \\ \frac{1}{\sqrt{2}} (\tilde{\eta}_{2} + i\tilde{\chi}_{2}) \end{pmatrix}, \qquad (2.5.38b)$$

$$H_{3} = \begin{pmatrix} h_{3}^{+} \\ \frac{1}{\sqrt{2}} \left(\tilde{\eta}_{3} + i \tilde{\chi}_{3} \right) \end{pmatrix}.$$
 (2.5.38c)

The minimization conditions in the Higgs basis are drastically simplified due to a single VEV:

$$Y_{11} = -v^2 Z_{1111}, (2.5.39a)$$

$$Y_{12} = -\frac{1}{2}v^2 Z_{1112}, \tag{2.5.39b}$$

$$Y_{12}^* = -\frac{1}{2}v^2 Z_{1112}^*, \qquad (2.5.39c)$$

$$Y_{13} = -\frac{1}{2}v^2 Z_{1113}, \qquad (2.5.39d)$$

$$Y_{13}^* = -\frac{1}{2}v^2 Z_{1113}^*.$$
 (2.5.39e)

The charged mass-squared matrix without the would-be Goldstone boson is:

$$\mathcal{M}_{\text{Charged}}^{2} = \begin{pmatrix} Y_{22} + \frac{1}{2}v^{2}Z_{1122} & Y_{23}^{*} + \frac{1}{2}v^{2}Z_{1123}^{*} \\ Y_{23} + \frac{1}{2}v^{2}Z_{1123} & Y_{33} + \frac{1}{2}v^{2}Z_{1133} \end{pmatrix}, \qquad (2.5.40)$$

with eigenvalues:

$$m_{H_1^{\pm}}^2 = \frac{1}{4} \left[2Y_{22} + 2Y_{33} + v^2 \left(Z_{1122} + Z_{1133} \right) - \Delta \right], \qquad (2.5.41a)$$

$$m_{H_2^{\pm}}^2 = \frac{1}{4} \left[2Y_{22} + 2Y_{33} + v^2 \left(Z_{1122} + Z_{1133} \right) + \Delta \right], \qquad (2.5.41b)$$

where

$$\Delta^{2} = 4Y_{22}^{2} + 4Y_{33}^{2} + 8Y_{23} \left(2Y_{23}^{*} + v^{2}Z_{1123}^{*}\right) - 4Y_{22} \left[2Y_{33} - v^{2} \left(Z_{1122} - Z_{1133}\right)\right] + v^{2} \left[8Y_{23}^{*}Z_{1123} + v^{2} \left(4|Z_{1123}|^{2} + \left(Z_{1122} - Z_{1133}\right)^{2}\right) - 4Y_{33} \left(Z_{1122} - Z_{1133}\right)\right].$$
(2.5.42)

The neutral mass-squared matrix without the would-be Goldstone boson in the basis

$$\{\tilde{\chi}_2, \, \tilde{\chi}_3, \, \tilde{\eta}_1, \, \tilde{\eta}_2, \, \tilde{\eta}_3\}$$
 (2.5.43)

is as follows:

$$\mathcal{M}_{\text{Neutral}}^{2} = \begin{pmatrix} (\mathcal{M}^{2})_{11} & (\mathcal{M}^{2})_{12} & (\mathcal{M}^{2})_{13} & 0 & (\mathcal{M}^{2})_{15} \\ (\mathcal{M}^{2})_{12} & (\mathcal{M}^{2})_{22} & (\mathcal{M}^{2})_{23} & (\mathcal{M}^{2})_{24} & (\mathcal{M}^{2})_{25} \\ (\mathcal{M}^{2})_{13} & (\mathcal{M}^{2})_{23} & (\mathcal{M}^{2})_{33} & (\mathcal{M}^{2})_{34} & (\mathcal{M}^{2})_{35} \\ 0 & (\mathcal{M}^{2})_{24} & (\mathcal{M}^{2})_{34} & (\mathcal{M}^{2})_{44} & (\mathcal{M}^{2})_{45} \\ (\mathcal{M}^{2})_{15} & (\mathcal{M}^{2})_{25} & (\mathcal{M}^{2})_{35} & (\mathcal{M}^{2})_{45} & (\mathcal{M}^{2})_{55} \end{pmatrix},$$
(2.5.44)

where

$$\left(\mathcal{M}^{2}\right)_{11} = Y_{22} + \frac{1}{2}v^{2}\left(Z_{1122} - 2Z_{1212} + Z_{1221}\right), \qquad (2.5.45a)$$

$$\left(\mathcal{M}^{2}\right)_{12} = \operatorname{Re}\left(Y_{23}\right) + \frac{1}{2}v^{2}\left[Z_{1231} + \operatorname{Re}\left(Z_{1123}\right) - \operatorname{Re}\left(Z_{1213}\right)\right], \qquad (2.5.45b)$$

$$\left(\mathcal{M}^2\right)_{13} = -v^2 \operatorname{Im}\left(Z_{1112}\right), \qquad (2.5.45c)$$

$$\left(\mathcal{M}^2\right)_{15} = \operatorname{Im}\left(Y_{23}\right) + \frac{1}{2}v^2 \operatorname{Im}\left(Z_{1123} - Z_{1213}\right), \qquad (2.5.45d)$$

$$\left(\mathcal{M}^2\right)_{22} = Y_{33} + \frac{1}{2}v^2 \left[Z_{1133} + Z_{1331} - 2\operatorname{Re}\left(Z_{1313}\right)\right], \qquad (2.5.45e)$$

$$\left(\mathcal{M}^2\right)_{23} = -v^2 \operatorname{Im}\left(Z_{1113}\right),\tag{2.5.45f}$$

$$\left(\mathcal{M}^2\right)_{24} = -\operatorname{Im}\left(Y_{23}\right) - \frac{1}{2}v^2\operatorname{Im}\left(Z_{1123} + Z_{1213}\right), \qquad (2.5.45g)$$

$$\left(\mathcal{M}^2\right)_{25} = -v^2 \operatorname{Im}\left(Z_{1313}\right), \qquad (2.5.45h)$$

$$\left(\mathcal{M}^2\right)_{33} = 2v^2 Z_{1111},\tag{2.5.45i}$$

$$(\mathcal{M}^2)_{34} = v^2 \operatorname{Re}(Z_{1112}),$$
 (2.5.45j)

$$(\mathcal{M}^2)_{35} = v^2 \operatorname{Re}(Z_{1113}),$$
 (2.5.45k)

$$\left(\mathcal{M}^2\right)_{44} = Y_{22} + \frac{1}{2}v^2 \left(Z_{1122} + 2Z_{1212} + Z_{1221}\right), \qquad (2.5.451)$$

$$\left(\mathcal{M}^2\right)_{45} = \operatorname{Re}\left(Y_{23}\right) + \frac{1}{2}v^2\left[Z_{1231} + \operatorname{Re}\left(Z_{1123}\right) + \operatorname{Re}\left(Z_{1213}\right)\right], \qquad (2.5.45\mathrm{m})$$

$$\left(\mathcal{M}^2\right)_{55} = Y_{33} + \frac{1}{2}v^2 \left[Z_{1133} + Z_{1331} + 2\operatorname{Re}\left(Z_{1313}\right)\right].$$
(2.5.45n)

It follows that the neutral mass-squared matrix is block-diagonal in the real basis:

$$\mathcal{M}_{\text{Neutral}}^2 = \text{diag}\left(\mathcal{M}_{\tilde{\eta}}^2, \, \mathcal{M}_{\tilde{\chi}}^2\right).$$
(2.5.46)

The neutral mass-squared matrix for the $\tilde{\eta}$ sector is:

$$\mathcal{M}_{\tilde{\eta}}^{2} = \begin{pmatrix} 2v^{2}Z_{1111} & v^{2}Z_{1112} & v^{2}Z_{1113} \\ v^{2}Z_{1112} & Y_{22} + \frac{1}{2}v^{2}\left(Z_{1122} + 2Z_{1212} + Z_{1221}\right) & Y_{23} + \frac{1}{2}v^{2}\left(Z_{1123} + Z_{1213} + Z_{1231}\right) \\ v^{2}Z_{1113} & Y_{23} + \frac{1}{2}v^{2}\left(Z_{1123} + Z_{1213} + Z_{1231}\right) & Y_{33} + \frac{1}{2}v^{2}\left(Z_{1133} + 2Z_{1313} + Z_{1331}\right) \end{pmatrix}.$$

$$(2.5.47)$$

The neutral mass-squared matrix for the $\tilde{\chi}$ sector is:

$$\mathcal{M}_{\tilde{\chi}}^{2} = \begin{pmatrix} Y_{22} + \frac{1}{2}v^{2} \left(Z_{1122} - 2Z_{1212} + Z_{1221} \right) & Y_{23} + \frac{1}{2}v^{2} \left(Z_{1123} - Z_{1213} + Z_{1231} \right) \\ Y_{23} + \frac{1}{2}v^{2} \left(Z_{1123} - Z_{1213} + Z_{1231} \right) & Y_{33} + \frac{1}{2}v^{2} \left(Z_{1133} - Z_{1313} + Z_{1331} \right) \end{pmatrix}, \quad (2.5.48)$$

In both of these cases the mass-squared parameters are too involved to be analytically expressed.

By inspecting the mass-squared matrices (2.5.40, 2.5.44) an interesting conclusion can be drawn: although the scalar potential can be expressed in terms of 33 parameters, the mass-squared matrices depend only on 12 Z_{ijkl} and 3 Y_{ij} couplings. The VEV of only the first doublet is non-zero (2.5.1)and thus the mass-squared parameters depend on the following combinations:

$$Z_{111i}, Z_{11ij}, Z_{1i1j}, Z_{1ij1}, Y_{ij}, \text{ for } \{i, j\} = \overline{1, 3}.$$
 (2.5.49)

Chapter 3 The C-III-c Model

In this chapter we consider the C-III-c vacuum configuration, given by:

$$\left\{ \hat{w}_1 e^{i\sigma_1}, \, \hat{w}_2 e^{i\sigma_2}, \, 0 \right\}. \tag{3.0.1}$$

First of all, we analyse the mass-squared matrices of the exact S_3 symmetry and find that this case is unappealing due to two massless states as presented in Table 2.5. We apply the concept of soft symmetry breaking to promote massless scalars to massive ones. Of particular interest are the possible DM models and thus only properties of such models are taken into account: the mass-squared matrices and if the models are CP violating.

3.1 Model With Exact S_3 Symmetry

We consider that the gauge-eigenstates are given by eq. (2.1.3). The gauge-eigenstates are not uniquely defined and the SU(2) doublets can be expressed in other ways.

The exact C-III-c vacuum configuration is given by eq. (3.0.1) alongside with the constraints:

$$\mu_1^2 = -(\lambda_1 - \lambda_2) v^2, \qquad (3.1.1a)$$

$$\lambda_2 + \lambda_3 = 0, \tag{3.1.1b}$$

$$\lambda_4 = 0, \tag{3.1.1c}$$

where we made use of eq. (2.1.4). From now on we will be using $\hat{w}_1^2 + \hat{w}_2^2 = v^2$ to simplify the look of the VEVs whenever it is possible.

3.1.1 Freedom of the Basis Redefinition

The Higgs basis rotation for the C-III-c vacuum configuration is given by:

$$\begin{pmatrix} H_1 \\ H_2 \\ H_3 \end{pmatrix} = \frac{1}{v} \begin{pmatrix} \hat{w}_1 & \hat{w}_2 & 0 \\ -\hat{w}_2 & \hat{w}_1 & 0 \\ 0 & 0 & v \end{pmatrix} \begin{pmatrix} e^{-i\sigma_1} & 0 & 0 \\ 0 & e^{-i\sigma_2} & 0 \\ 0 & 0 & 1 \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}$$

$$= \frac{1}{v} \begin{pmatrix} e^{-i\sigma_1}\hat{w}_1 & e^{-i\sigma_2}\hat{w}_2 & 0 \\ -e^{-i\sigma_1}\hat{w}_2 & e^{-i\sigma_2}\hat{w}_1 & 0 \\ 0 & 0 & v \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}$$

$$= \begin{pmatrix} \mathcal{R}_{\text{HB}-2} & 0 \\ 0 & 1 \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix} \equiv \mathcal{R}_{\text{HB}} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}.$$

$$(3.1.2)$$

Let us discuss what was done here. First of all, the phases of the VEVs were rotated away so that VEVs depend only on the absolute value \hat{w}_i ; note that when going into the Higgs basis, the full

SU(2) doublets are rotated and not only the VEVs. Next, as in the Higgs basis only one VEV acquires a non-zero value, we performed a rotation of the absolute values \hat{w}_i so that $\langle H_1 \rangle = v$ as given by eq. (2.5.1). The rotation of the absolute values can be expressed in terms of the well-known t_β parameter:

$$\mathcal{R}_{\beta} = \begin{pmatrix} c_{\beta} & s_{\beta} & 0\\ -s_{\beta} & c_{\beta} & 0\\ 0 & 0 & 1 \end{pmatrix} = \operatorname{diag}(\mathcal{R}_{\beta-2}, 1), \qquad (3.1.3)$$

where the rotation angle β is determined by:

$$\mathbf{t}_{\beta} = \frac{\hat{w}_2}{\hat{w}_1} = \frac{\mathbf{s}_{\beta} \, v}{\mathbf{c}_{\beta} \, v}.\tag{3.1.4}$$

There is a freedom to rotate the VEVs (3.0.1) of the C-III-c model. We can parametrize the fields, for instance, in the following way:

$$h_{i} = e^{i(\sigma_{i} - \sigma_{i})} \begin{pmatrix} h_{i}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}_{i} e^{i\sigma_{i}} + \tilde{\eta}_{i} + i\tilde{\chi}_{i} \right) \end{pmatrix}$$
$$= e^{i\sigma_{i}} \begin{pmatrix} e^{-i\sigma_{i}} h_{i}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}_{i} + e^{-i\sigma_{i}} \tilde{\eta}_{i} + ie^{-i\sigma_{i}} \tilde{\chi}_{i} \right) \end{pmatrix}$$
$$\equiv e^{i\sigma_{i}} \begin{pmatrix} h_{i}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}_{i} + \tilde{\eta}_{i} + i\tilde{\chi}_{i} \right) \end{pmatrix}, \qquad (3.1.5)$$

where in the last equality we abused the mathematical notation not to introduce additional abbreviations for the fields. We, however, will make a note and relate both cases so that the fields are expressed in terms of eq. (2.1.3) if a non-trivial re-definition of the fields is performed. After changing the basis, the SU(2) doublets are no longer consistent with eq. (2.1.3).

Due to the U(1) invariance of the scalar potential, it is possible to rotate all of the doublets by the same phase:

$$\begin{pmatrix} h_1' \\ h_2' \\ h_S' \end{pmatrix} = e^{-i\sigma_2} \mathcal{I}_3 \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix},$$
(3.1.6)

so that VEVs are now:

$$\left\{\hat{w}_1 e^{i(\sigma_1 - \sigma_2)}, \, \hat{w}_2, \, 0\right\} \equiv \left\{\hat{w}_1 e^{i\sigma}, \, \hat{w}_2, \, 0\right\}.$$
(3.1.7)

In this case, the Higgs basis rotation \mathcal{R}_{HB} is given by:

$$\begin{pmatrix} H_1 \\ H_2 \\ H_3 \end{pmatrix} = \begin{pmatrix} c_\beta & s_\beta & 0 \\ -s_\beta & c_\beta & 0 \\ 0 & 0 & 1 \end{pmatrix} \begin{pmatrix} e^{-i\sigma} & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{pmatrix} \begin{pmatrix} h'_1 \\ h'_2 \\ h'_S \end{pmatrix}.$$
 (3.1.8)

3.1.2 The Mass-Squared Matrices

Of particular interest are mass-eigenstates and therefore we need to find rotation matrices \mathcal{R} of gauge-eigenstates, which would result in diagonal mass-squared matrices:

$$\hat{\mathcal{M}}^2 = \mathcal{R}\mathcal{M}^2\mathcal{R}^{\dagger}, \qquad (3.1.9)$$

where the hatted mass-squared matrix $\hat{\mathcal{M}}^2$ indicates that it is a diagonal matrix.

Before we evaluate the mass-squared matrices we want to show that those are block diagonal for the C-III-c model and thus there is no mixing between the S_3 doublet and S_3 singlet parts as we are interested in the inert S_3 singlet. The only couplings, apart from λ_4 as in C-III-c it is set
to zero, which could result in a mixing of the S_3 doublet and singlet are λ_5 , or λ_6 , or λ_7 . In terms of the doublets h_i , the dependence is either $\sim h_1^2 h_S^2$ or $\sim h_2^2 h_S^2$. If we take into consideration the vacuum configuration, especially that $\hat{w}_S = 0$, it becomes obvious that there is no mixing: the only case when the terms are non-zero is when $h_i^2 \to \hat{w}_i^2$, for $i = \overline{1,2}$, and therefore from eq. (2.3.14) it follows that the S_3 singlet fields acquire mass terms $\mathcal{M}^2_{(\xi_S^*)_i(\xi_S)_i}$ from:

$$V_2((\xi_k^S)^* \xi_l^S)$$
 and $V_4(\hat{w}_i \hat{w}_j (\xi_k^S)^* \xi_l^S)$, for $i = \overline{1, 2}$, (3.1.10)

where $\xi^S = \{h_S^{\pm}, \tilde{\eta}_S, \tilde{\chi}_S\}$. Thus we can treat the S_3 singlet mass terms separately. On the other hand, if we did not have the $\lambda_4 = 0$ constraint, this would obviously lead to mixing. This is governed by the fact that there would be terms proportional to $h_1^2 h_2 h_S$ and $h_2^3 h_S$, *i.e.*, even in the limit of $\hat{w}_S = 0$, mixing would arise between the S_3 doublet and singlet gauge fields. Since $\lambda_4 = 0$ we get no mixing.

We consider that the SU(2) doublets are parameterized by (3.1.6) and the overall phase σ is extracted afterwards as in eq. (3.1.5):

$$h_1 \to h_1 e^{i\sigma}, \quad h_2 \to h_2, \quad h_S \to h_S,$$

$$(3.1.11)$$

and the VEVs are given by (3.1.7). Therefore, only a single overall phase $\sigma = \sigma_1 - \sigma_2$ is present. All further calculations are performed assuming that the SU(2) doublets in terms of the gauge fields are given by:

$$h_{1} = e^{i(\sigma_{1} - \sigma_{2})} \begin{pmatrix} e^{-i\sigma_{1}}h_{1}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}_{1} + e^{-i\sigma_{1}}\tilde{\eta}_{1} + ie^{-i\sigma_{1}}\tilde{\chi}_{1}\right) \end{pmatrix} \equiv e^{i\sigma} \begin{pmatrix} h_{1}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}_{1} + \tilde{\eta}_{1} + i\tilde{\chi}_{1}\right) \end{pmatrix},$$

$$h_{2} = e^{i(\sigma_{2} - \sigma_{2})} \begin{pmatrix} e^{-i\sigma_{2}}h_{2}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}_{2} + e^{-i\sigma_{2}}\tilde{\eta}_{2} + ie^{-i\sigma_{2}}\tilde{\chi}_{2}\right) \end{pmatrix} \equiv \begin{pmatrix} h_{2}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}_{2} + \tilde{\eta}_{2} + i\tilde{\chi}_{2}\right) \end{pmatrix},$$

$$h_{S} = e^{i(\sigma_{2} - \sigma_{2})} \begin{pmatrix} e^{-i\sigma_{2}}h_{S}^{+} \\ \frac{1}{\sqrt{2}} \left(e^{-i\sigma_{2}}\tilde{\eta}_{S} + ie^{-i\sigma_{2}}\tilde{\chi}_{S}\right) \end{pmatrix} \equiv \begin{pmatrix} h_{S}^{+} \\ \frac{1}{\sqrt{2}} \left(\tilde{\eta}_{S} + i\tilde{\chi}_{S}\right) \end{pmatrix}$$
(3.1.12)

After performing the rotation \mathcal{R}_{β} of eq. (3.1.3), the charged mass-squared matrix is already block-diagonal:

$$\hat{\mathcal{M}}_{\text{Charged}}^2 = \mathcal{R}_\beta \ \mathcal{M}_{\text{Charged}}^2 \ \mathcal{R}_\beta^{-1}.$$
(3.1.13)

We get the following masses of the charged scalars:

$$m_{H^{\pm}}^2 = 2\lambda_2 v^2,$$
 (3.1.14a)

$$m_{S^{\pm}}^2 = \mu_0^2 + \frac{1}{2}\lambda_5 v^2.$$
 (3.1.14b)

In order to get the charged physical fields and to identify the would-be Goldstone boson, one needs to apply the rotation matrix to the gauge fields. In terms of the charged sector, the masseigenstates are given by:

$$\mathcal{R}_{\beta} \begin{pmatrix} h_{1}^{+} \\ h_{2}^{+} \\ h_{S}^{+} \end{pmatrix} = \begin{pmatrix} G^{+} \\ H^{+} \\ S^{+} \end{pmatrix}, \quad \begin{pmatrix} G^{-} \\ H^{-} \\ S^{-} \end{pmatrix} = \begin{pmatrix} (G^{+})^{\dagger} \\ (H^{+})^{\dagger} \\ (S^{+})^{\dagger} \end{pmatrix}.$$
(3.1.15)

We write down explicitly the charged fields:

$$G^{\pm} = c_{\beta}h_{1}^{\pm} + s_{\beta}h_{2}^{\pm}, \qquad (3.1.16a)$$

$$H^{\pm} = -\mathbf{s}_{\beta}h_1^{\pm} + \mathbf{c}_{\beta}h_2^{\pm}, \qquad (3.1.16b)$$

$$S^{\pm} = h_S^{\pm}.$$
 (3.1.16c)

Next, we consider the neutral scalar sector. As discussed before, there is no mixing between the S_3 doublet and singlet, *i.e.*, in the basis $(h_1, h_2) - h_s$. Consider the gauge fields of the S_3 doublet. The \mathcal{R}_β rotation results in the following fields:

$$\begin{pmatrix} H_1' \\ H_2' \\ G^0 \\ H_3' \end{pmatrix} = \mathcal{R}_{\beta-4} \begin{pmatrix} \tilde{\eta}_1 \\ \tilde{\eta}_2 \\ \tilde{\chi}_1 \\ \tilde{\chi}_2 \end{pmatrix}, \qquad (3.1.17)$$

where

$$\mathcal{R}_{\beta-4} = \mathcal{I}_2 \otimes \mathcal{R}_{\beta-2}. \tag{3.1.18}$$

We identify the neutral would-be Goldstone boson as $G^0 = c_\beta \tilde{\chi}_1 + s_\beta \tilde{\chi}_2$, governed by the same logic as when identifying the charged would-be Goldstone boson G^{\pm} in eq. (3.1.15).

After identifying the state corresponding to the would-be Goldstone boson, we take a look at the neutral sector mass-squared matrix:

$$\mathcal{M}_{\text{Neutral}-12}^{2} = \frac{\partial^{2} V}{\partial \zeta_{i}^{12} \partial \zeta_{j}^{12}} \Big|_{\langle v \rangle} = \begin{pmatrix} (\mathcal{M}_{a}^{2})_{11} & 0 & 0\\ 0 & 0 & 0\\ 0 & 0 & 0 \end{pmatrix}, \qquad (3.1.19)$$

where

$$\zeta_i^{12} = \{H_1', \ H_2', \ H_3'\}. \tag{3.1.20}$$

An interesting feature of the C-III-c vacuum configuration can be seen. It is obvious that there are two massless scalars as there is only one non-zero element present in the matrix $\mathcal{M}^2_{\text{Neutral-12}}$. This was explained in section 2.3. Due to the structure of $\mathcal{M}^2_{\text{Neutral-12}}$, as it is already diagonal, we identify the physical states:

$$H_1 = c_\beta \tilde{\eta}_1 + s_\beta \tilde{\eta}_2, \qquad (3.1.21a)$$

$$H_2 = -s_\beta \tilde{\eta}_1 + c_\beta \tilde{\eta}_2, \qquad (3.1.21b)$$

$$H_3 = -\mathbf{s}_\beta \tilde{\chi}_1 + \mathbf{c}_\beta \tilde{\chi}_2. \tag{3.1.21c}$$

Out of all these states only one is massive:

$$m_{H_1}^2 = 2 \left(\lambda_1 - \lambda_2\right) v^2, \tag{3.1.22a}$$

$$m_{H_2}^2 = 0,$$
 (3.1.22b)

$$m_{H_3}^2 = 0. (3.1.22c)$$

In order to get massive scalar states H_2 and H_3 one needs to softly break the S_3 -symmetric potential. This case is taken into consideration in section 3.2.

The only part we are left with is to determine the mass-squared matrix of the S_3 singlet:

$$\mathcal{M}_{\text{Neutral-S}}^2 = \frac{\partial^2 V}{\partial \zeta_i^S \partial \zeta_j^S} \Big|_{\langle v \rangle} = \begin{pmatrix} (\mathcal{M}_b^2)_{11} & (\mathcal{M}_b^2)_{12} \\ (\mathcal{M}_b^2)_{12} & (\mathcal{M}_b^2)_{22} \end{pmatrix}, \qquad (3.1.23)$$

where

$$\zeta_i^S = \{ \tilde{\eta}_S, \ \tilde{\chi}_S \}. \tag{3.1.24}$$

The elements of the mass-squared matrix are:

$$(\mathcal{M}_b^2)_{11} = \mu_0^2 + \frac{1}{2} \left(\lambda_5 + \lambda_6\right) v^2 + \lambda_7 \left(c_{2\sigma} \hat{w}_1^2 + \hat{w}_2^2\right), \qquad (3.1.25a)$$

$$(\mathcal{M}_b^2)_{12} = \lambda_7 \mathbf{s}_{2\sigma} \hat{w}_1^2,$$
 (3.1.25b)

$$(\mathcal{M}_b^2)_{22} = \mu_0^2 + \frac{1}{2} \left(\lambda_5 + \lambda_6\right) v^2 - \lambda_7 \left(c_{2\sigma} \hat{w}_1^2 + \hat{w}_2^2\right).$$
(3.1.25c)

The mass-squared matrix $\mathcal{M}^2_{Neutral-S}$ is diagonalizable by performing the following rotation:

$$\hat{\mathcal{M}}_{\text{Neutral}-S}^2 = \mathcal{R}_{\gamma} \mathcal{M}_{\text{Neutral}-S}^2 \mathcal{R}_{\gamma}^{-1}, \qquad (3.1.26)$$

where the rotation matrix is:

$$\mathcal{R}_{\gamma} = \begin{pmatrix} c_{\gamma} & s_{\gamma} \\ -s_{\gamma} & c_{\gamma} \end{pmatrix}.$$
 (3.1.27)

In this case the rotation angle γ is not trivial anymore. Calculations yield that the rotation angle γ is determined by¹:

$$t_{\gamma} = \frac{s_{2\sigma}\hat{w}_1^2}{c_{2\sigma}\hat{w}_1^2 + \hat{w}_2^2 + \sqrt{\hat{w}_1^4 + 2c_{2\sigma}\hat{w}_1^2\hat{w}_2^2 + \hat{w}_2^4}}.$$
(3.1.28)

We get the following masses of the S_3 singlet scalars:

$$m_{S_1}^2 = \mu_0^2 + \frac{1}{2} \left(\lambda_5 + \lambda_6\right) v^2 - \lambda_7 \sqrt{\hat{w}_1^4 + 2c_{2\sigma}\hat{w}_1^2\hat{w}_2^2 + \hat{w}_2^4}, \qquad (3.1.29a)$$

$$m_{S_2}^2 = \mu_0^2 + \frac{1}{2} \left(\lambda_5 + \lambda_6\right) v^2 + \lambda_7 \sqrt{\hat{w}_1^4 + 2c_{2\sigma}\hat{w}_1^2\hat{w}_2^2 + \hat{w}_2^4}.$$
 (3.1.29b)

The corresponding neutral fields are:

$$S_1 = c_{\gamma} \tilde{\eta}_S + s_{\gamma} \tilde{\chi}_S, \qquad (3.1.30a)$$

$$S_2 = -\operatorname{s}_{\gamma} \tilde{\eta}_S + \operatorname{c}_{\gamma} \tilde{\chi}_S. \tag{3.1.30b}$$

The SU(2) doublets in terms of the mass-eigenstates in the basis of eq. (3.1.12) are:

$$h_{1} = e^{i\sigma} \begin{pmatrix} c_{\beta}G^{+} - s_{\beta}H^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}_{1} + c_{\beta}H_{1} - s_{\beta}H_{2} + i\left(c_{\beta}G^{0} - s_{\beta}H_{3}\right) \right) \end{pmatrix}, \qquad (3.1.31a)$$

$$h_{2} = \begin{pmatrix} s_{\beta}G^{+} + c_{\beta}H^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}_{2} + s_{\beta}H_{1} + c_{\beta}H_{2} + i\left(s_{\beta}G^{0} + c_{\beta}H_{3}\right) \right) \end{pmatrix}, \qquad (3.1.31b)$$

$$h_S = \begin{pmatrix} S^+ \\ \frac{1}{\sqrt{2}} \left(e^{i\gamma} \left(S_1 + iS_2 \right) \right) \end{pmatrix}.$$
(3.1.31c)

The SU(2) doublets, denoted as H_i , in the Higgs basis are trivially simplified to:

$$H_1 = \begin{pmatrix} G^+ \\ \frac{1}{\sqrt{2}} \left(v + \varphi_1 + iG^0 \right) \end{pmatrix}, \qquad (3.1.32a)$$

$$H_2 = \begin{pmatrix} H^+ \\ \frac{1}{\sqrt{2}} \left(\varphi_2 + i\varphi_3\right) \end{pmatrix}, \qquad (3.1.32b)$$

$$H_3 = \begin{pmatrix} S^+ \\ \frac{1}{\sqrt{2}} \left(e^{i\gamma} \left(S_1 + iS_2 \right) \right) \end{pmatrix}, \qquad (3.1.32c)$$

where we used φ_i for the neutral fields to distinguish those from the SU(2) doublets as it is common to use the H_i notation for the SU(2) doublets in the Higgs basis. The neutral fields φ_i are the same as the H_i fields of eq. (3.1.21) discussed in this section.

 $^{^1\}mathrm{It}$ should be noted that this solution is not unique.

3.1.3 The Mass-Squared Matrices in Different Bases

So far we have considered that the C-III-c vacuum is: $\{\hat{w}_1 e^{i\sigma}, \hat{w}_2, 0\}$. For completeness, we would like to briefly mention the other possible choices and what they result in.

The generic vacuum configuration $\{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, 0\}$ results in a hermitian charged masssquared matrix:

$$\mathcal{M}_{\text{Charged}}^{2} = \begin{pmatrix} \mathcal{M}_{11} & \mathcal{M}_{12} & 0 \\ \mathcal{M}_{12}^{*} & \mathcal{M}_{22} & 0 \\ 0 & 0 & \mathcal{M}_{33} \end{pmatrix}, \qquad (3.1.33)$$

and a symmetric neutral mass-squared matrices:

$$\mathcal{M}_{\text{Neutral}}^{2} = \begin{pmatrix} \mathcal{M}_{11} & \mathcal{M}_{12} & \mathcal{M}_{13} & \mathcal{M}_{14} & 0 & 0 \\ \mathcal{M}_{12} & \mathcal{M}_{22} & \mathcal{M}_{23} & \mathcal{M}_{24} & 0 & 0 \\ \mathcal{M}_{13} & \mathcal{M}_{23} & \mathcal{M}_{33} & \mathcal{M}_{34} & 0 & 0 \\ \mathcal{M}_{14} & \mathcal{M}_{24} & \mathcal{M}_{34} & \mathcal{M}_{44} & 0 & 0 \\ 0 & 0 & 0 & \mathcal{M}_{55} & \mathcal{M}_{56} \\ 0 & 0 & 0 & \mathcal{M}_{56} & \mathcal{M}_{66} \end{pmatrix},$$
(3.1.34)

in the basis:

$$\{\tilde{\eta}_1, \tilde{\eta}_2, \tilde{\chi}_1, \tilde{\chi}_2, \tilde{\eta}_S, \tilde{\chi}_S\}.$$
 (3.1.35)

The following $\mathcal{M}^2_{\text{Neutral}}$ of this section are evaluated in this basis. Although this is the most trivial choice of the SU(2) doublets, it results in a fact that the charged mass-squared matrix is diagonalizable by a complex matrix while the neutral states are potentially combinations of the gauge-eigenstates $\tilde{\eta}_i - \tilde{\chi}_i$.

If the σ_i phases are extracted from the SU(2) doublets $h_i \to e^{i\sigma_i}h_i$ as in eq. (3.1.5), this results in:

The charged mass-eigenstates no longer depend on a phase and the neutral mass-eigenstates do not mix the $\tilde{\eta}_i$ and $\tilde{\chi}_i$ fields.

In Ref. [48] it was shown that the C-III-c vacuum configuration can be written in a basis in which VEVs are given by:

$$\{ae^{i\delta}, ae^{-i\delta}, 0\}.$$
 (3.1.37)

This specific choice of the basis results in $t_{\beta} = 1$ of eq. (3.1.4). It can be proven that eq. (3.1.37) is a special basis of the C-III-c model by considering additional symmetries of section 2.3. The $\lambda_4 = 0$ constraint results in a continuous SO(2) symmetry. This fact can be used to rotate the active doublets:

$$\begin{pmatrix} c_{\theta} & s_{\theta} \\ -s_{\theta} & c_{\theta} \end{pmatrix} \begin{pmatrix} e^{-i\sigma_2} & 0 \\ 0 & e^{-i\sigma_2} \end{pmatrix} \begin{pmatrix} \hat{w}_1 e^{i\sigma_1} \\ \hat{w}_2 e^{i\sigma_2} \end{pmatrix} = \begin{pmatrix} \hat{w}_1 c_{\theta} c_{\sigma} + \hat{w}_2 s_{\theta} + i\hat{w}_1 c_{\theta} s_{\sigma} \\ -\hat{w}_1 c_{\sigma} s_{\theta} + \hat{w}_2 c_{\theta} - i\hat{w}_1 s_{\theta} s_{\sigma} \end{pmatrix} \equiv \begin{pmatrix} \tilde{w}_1 \\ \tilde{w}_2 \end{pmatrix}.$$
(3.1.38)

The complex VEVs \tilde{w}_i can be written as:

$$\tilde{w}_i' = |\tilde{w}_i|e^{i\arg(\tilde{w}_i)}.\tag{3.1.39}$$

The modulus or the phases² of the complex values \tilde{w}'_i can be related as there is a free θ parameter. Provided that the absolute values are equal

$$\mathbf{t}_{2\theta} = \frac{\hat{w}_2^2 - \hat{w}_1^2}{2\hat{w}_1\hat{w}_2\mathbf{c}_{\sigma}},\tag{3.1.40}$$

this results in the vacuum configuration: $\{ae^{i\delta_1}, ae^{i\delta_2}, 0\}$, with

$$a = \sqrt{\hat{w}_1^2 c_{\theta}^2 s_{\sigma}^2 + (\hat{w}_1 c_{\theta} c_{\sigma} + \hat{w}_2 s_{\theta})^2}.$$
(3.1.41)

Due to the U(1) invariance of the potential, an overall rotation by $e^{-\frac{i}{2}(\delta_1+\delta_2)}$ is possible. This results in the vacuum configuration: $\{ae^{i\delta}, ae^{-i\delta}, 0\}$. The δ phases can be extracted as follows: $h_1 \to e^{i\delta}h_1$ and $h_2 \to e^{-i\delta}h_2$. In this case, the mass-squared matrices are:

Benefits of this basis are a fixed $\beta = \pi/4$ parameter as well as the diagonalized S_3 singlet states.

It can be seen that in different bases the mass-eigenstates can be expressed differently. The mass squared parameters are physical quantities and therefore are basis independent. Although there is no preferred basis, it should be noted that based on what one is interested in, a re-definition of the SU(2) doublets is valid as long as it is unitary and results in a special direction of the mass-eigenstates.

3.2 Soft Symmetry Breaking

We next consider the possibility to promote massless states of the C-III-c model to massive ones. One of the possible solutions is the soft symmetry breaking concept. The idea behind this method is to introduce additional bilinear terms V'_2 and therefore softly break the S_3 symmetry:

$$\tilde{V} = V_2 + V_2' + V_4, \tag{3.2.1}$$

The most general form of these terms is [49]:

$$V_{2}' = \frac{1}{2}\nu^{2} \left(h_{1}^{\dagger}h_{2} + \text{h.c.}\right) + \mu_{2}^{2} \left(h_{1}^{\dagger}h_{1} - h_{2}^{\dagger}h_{2}\right) + \frac{1}{2}\mu_{3}^{2} \left(h_{S}^{\dagger}h_{1} + \text{h.c.}\right) + \frac{1}{2}\mu_{4}^{2} \left(h_{S}^{\dagger}h_{2} + \text{h.c.}\right). \quad (3.2.2)$$

Due to the fact that there are new terms present in the scalar potential, this leads to altered minimization conditions. We assume that the C-III-c model VEVs are consistent with eq. (3.0.1): $\{\hat{w}_1e^{i\sigma_1}, \hat{w}_2e^{i\sigma_2}, 0\}$ if not specified otherwise. The SU(2) doublets are expanded in terms of VEVs

²There is no meaningful solutions when trying to solve for θ in terms of $\arg(\tilde{w}_1) = \arg(\tilde{w}_2)$.

before differentiating the potential and regardless can be fixed if those are a subject of the minimization conditions. To distinguish the softly broken models from the exact S_3 -symmetric C-III-c model we denote these models as C-III-c-X, where X is a string of the softly broken parameters.

The most general model is C-III-c- $\nu^2 - \mu_2^2 - \mu_3^2 - \mu_4^2$. This model results in the following constraints after minimizing the scalar potential:

$$\nu^{2} = -4 \left(\lambda_{2} + \lambda_{3}\right) c_{\sigma_{1} - \sigma_{2}} \hat{w}_{1} \hat{w}_{2}, \qquad (3.2.3a)$$

$$\mu_1^2 = -(\lambda_1 - \lambda_2) v^2, \qquad (3.2.3b)$$

$$\mu_2^2 = -(\lambda_2 + \lambda_3) \left(\hat{w}_1^2 - \hat{w}_2^2 \right), \qquad (3.2.3c)$$

$$\mu_3^2 = -2\lambda_4 c_{2(\sigma_1 - \sigma_2)} \hat{w}_1 \hat{w}_2, \qquad (3.2.3d)$$

$$\mu_4^2 = -\lambda_4 \left(\hat{w}_1^2 - \hat{w}_2^2 \right). \tag{3.2.3e}$$

In this case, we no longer get the $\lambda_4 = 0$ constraint and thus there arises additional mixing between the S_3 doublet and singlet. It is clear that the soft breaking terms μ_3^2 and μ_4^2 are the ones responsible for the non-zero coupling value of λ_4 . To be more precise, the aforementioned soft breaking terms are proportional to $h_S^{\dagger}h_i$ and are not consistent with the constraint $\lambda_4 = 0$. From another point of view, it should be noted that both μ_3^2 and μ_4^2 couplings break the \mathbb{Z}_2 symmetry.

In principle, one could solve for minimization conditions not in terms of μ_3^2 and μ_4^2 , but in terms of λ_4 :

$$\lambda_4 = \frac{\mu_3^2 \mathbf{c}_{\sigma_1} \hat{w}_1 + \mu_4^2 \mathbf{c}_{\sigma_2} \hat{w}_2}{-\left(\mathbf{c}_{2\sigma_1 - \sigma_2} + 2\mathbf{c}_{\sigma_2}\right) \hat{w}_1^2 \hat{w}_2 + \mathbf{c}_{\sigma_2} \hat{w}_2^3}.$$
(3.2.4)

Assume that μ_4^2 depends on μ_3^2 so that:

$$\mu_4^2 = -\mu_3^2 \frac{\mathbf{c}_{\sigma_1} \hat{w}_1}{\mathbf{c}_{\sigma_2} \hat{w}_2}.$$
(3.2.5)

Although it might seem that such dependence leads to the constraint $\lambda_4 = 0$, this is not entirely correct. The λ_4 minimization condition of eq. (3.2.4) is not sufficient to minimize the scalar potential as before two equations (3.2.3d, 3.2.3e) were solved and not a single one. The other derivative results in:

$$\mu_3^2 \frac{s_{\sigma_1 - \sigma_2}}{c_{\sigma_2}} \hat{w}_1 = 0. \tag{3.2.6}$$

A trivial $\mu_3^2 = 0$ is consistent with $\lambda_4 = 0$, but in this case both μ_3^2 and μ_4^2 vanish.

Another possible solution would be to consider $\sigma_2 = \sigma_1$. The derivatives with respect to σ_i vanish and therefore the total number of independent minimization conditions is reduced to three:

$$\mu_1^2 = -\frac{\nu^2 + 4\left(\lambda_1 + \lambda_3\right)\hat{w}_1\hat{w}_2}{4\hat{w}_1\hat{w}_2}v,\tag{3.2.7a}$$

$$\mu_2^2 = \frac{1}{4}\nu^2 \frac{\hat{w}_1^2 - \hat{w}_2^2}{\hat{w}_1 \hat{w}_2},\tag{3.2.7b}$$

$$\mu_3^2 = -\frac{\hat{w}_2}{\hat{w}_1} \left[\mu_4^2 + \lambda_4 \left(3\hat{w}_1^2 - \hat{w}_2^2 \right) \right].$$
(3.2.7c)

This case does not result in the $\lambda_4 = 0$ constraint.

For completeness, we once again consider the minimization condition in terms of eq. (3.2.4), but this time relax the (3.2.5) condition as it was artificially introduced. The scalar potential is minimized provided that an additional constraint is satisfied:

$$\mu_3^2 \left(\hat{w}_1^2 - \hat{w}_2^2 \right) - 2\mu_4^2 \mathbf{c}_{\sigma_1 - \sigma_2} \hat{w}_1 \hat{w}_2 = 0.$$
(3.2.8)

Assume that we express μ_4^2 in terms of μ_3^2 . Full minimization of the potential results in the $\mu_3^2 \sim \lambda_4$ dependence. Provided that $\lambda_4 = 0$, both μ_3^2 and μ_4^2 vanish.

We are interested in a possible DM candidate and therefore the soft breaking terms of interest are reduced:

$$V_2' = \frac{1}{2}\nu^2 \left(h_1^{\dagger}h_2 + h_2^{\dagger}h_1\right) + \mu_2^2 \left(h_1^{\dagger}h_1 - h_2^{\dagger}h_2\right).$$
(3.2.9)

The quadratic part of the potential with respect to VEVs is therefore:

$$V_2'|_{\langle v \rangle} = \frac{1}{2} \left[\nu^2 c_{\sigma_1 - \sigma_2} \hat{w}_1 \hat{w}_2 + \mu_2^2 \left(\hat{w}_1^2 - \hat{w}_2^2 \right) \right].$$
(3.2.10)

For simplicity, we do not identify the fields in terms of the mass-eigenstates and therefore the most general mass-squared matrices are considered, *i.e.*, without defining the Goldstone bosons. All of the calculations are performed in the basis of eq. (2.1.3) unless specified otherwise.

We show that models with soft breaking parameters, which are consistent with the $\lambda_4 = 0$ constraint, result in the same eigenvalues for the charged sector and the S_3 singlet as for the exact C-III-c model by taking appropriate substitutions of the vacuum configuration if needed³. The only significant change remains for the S_3 doublet neutral eigenvalues. After all, the main idea behind why we consider soft symmetry breaking is to give masses to otherwise massless states.

3.2.1 The C-III-c- ν^2 Model

In the C-III-c- ν^2 model, the additional bilinear term is:

$$V_2' = \frac{1}{2}\nu^2 \left(h_1^{\dagger}h_2 + h_2^{\dagger}h_1\right).$$
(3.2.11)

In principle, we do not assume that the numerical value of the term ν^2 is bounded by the μ_0^2 and μ_1^2 terms.

Due to the soft symmetry breaking term ν^2 derivatives get modified as follows:

$$\frac{\partial \tilde{V}}{\partial \hat{w}_1}\Big|_{\langle v \rangle} = \frac{\partial V}{\partial \hat{w}_1}\Big|_{\langle v \rangle} + \frac{1}{2}\nu^2 \hat{w}_2 \mathbf{c}_{(\sigma_1 - \sigma_2)}, \qquad (3.2.12a)$$

$$\frac{\partial \tilde{V}}{\partial \hat{w}_2}\Big|_{\langle v \rangle} = \frac{\partial V}{\partial \hat{w}_2}\Big|_{\langle v \rangle} + \frac{1}{2}\nu^2 \hat{w}_1 c_{(\sigma_1 - \sigma_2)}, \qquad (3.2.12b)$$

$$\frac{\partial \tilde{V}}{\partial \sigma_1}\Big|_{\langle v \rangle} = \frac{\partial V}{\partial \sigma_1}\Big|_{\langle v \rangle} - \frac{1}{2}\nu^2 \hat{w}_1 \hat{w}_{2} \mathbf{s}_{(\sigma_1 - \sigma_2)}, \qquad (3.2.12c)$$

$$\frac{\partial \tilde{V}}{\partial \sigma_2}\Big|_{\langle v \rangle} = \frac{\partial V}{\partial \sigma_2}\Big|_{\langle v \rangle} + \frac{1}{2}\nu^2 \hat{w}_1 \hat{w}_{2} \mathbf{s}_{(\sigma_1 - \sigma_2)}.$$
(3.2.12d)

The soft breaking term ν^2 can be defined by taking a look at the derivatives with respect to phases eq. (3.2.12c) and eq. (3.2.12d):

$$\nu^2 = -4 \left(\lambda_2 + \lambda_3\right) \hat{w}_1 \hat{w}_2 c_{(\sigma_1 - \sigma_2)}.$$
(3.2.13)

In case of the exact S_3 -symmetric potential we found that one of the minimization condition (3.1.1b) required $\lambda_2 + \lambda_3 = 0$. The soft symmetry breaking term ν^2 "eats" the $\lambda_2 + \lambda_3 = 0$ constraint and this results in a massive state. The constraint $\lambda_4 = 0$ survives as there are no additional terms coming from the derivative with respect to \hat{w}_S (3.2.10). A direct inspection of the derivatives eq. (3.2.12a) and eq. (3.2.12b) leads to the constraint $\hat{w}_1 = \hat{w}_2$.

Taking everything into consideration we end up with the following vacuum configuration:

$$\left\{\hat{w}_1 e^{i\sigma_1}, \, \hat{w}_1 e^{i\sigma_2}, \, 0\right\},\tag{3.2.14}$$

³We consider two models, *i.e.*, C-III-c- ν^2 and C-III-c- μ_2^2 , which after solving for the minimization conditions constrain the vacuum configuration. In the C-III-c- ν^2 model absolute values \hat{w}_i are further limited while in the C-III-c- μ_2^2 model the overall phases σ_i are fixed.

with constraints:

$$\lambda_4 = 0, \tag{3.2.15a}$$

$$\mu_1^2 = -(\lambda_1 - \lambda_2) v^2, \qquad (3.2.15b)$$

$$\nu^{2} = -2 \left(\lambda_{2} + \lambda_{3}\right) c_{(\sigma_{1} - \sigma_{2})} v^{2}, \qquad (3.2.15c)$$

where

$$v^2 = 2\hat{w}_1^2. \tag{3.2.16}$$

This model is analysed more thoroughly in chapter 4.

3.2.2 The C-III-c- μ_2^2 Model

Another possible soft breaking term is μ_2^2 :

$$V_2' = \mu_2^2 \left(h_1^{\dagger} h_1 - h_2^{\dagger} h_2 \right).$$
(3.2.17)

In this case derivatives get modified:

$$\frac{\partial \tilde{V}}{\partial \hat{w}_1}\Big|_{\langle v \rangle} = \frac{\partial V}{\partial \hat{w}_1}\Big|_{\langle v \rangle} + \mu_2^2 \hat{w}_1, \qquad (3.2.18a)$$

$$\frac{\partial \tilde{V}}{\partial \hat{w}_2}\Big|_{\langle v \rangle} = \frac{\partial V}{\partial \hat{w}_2}\Big|_{\langle v \rangle} - \mu_2^2 \hat{w}_2.$$
(3.2.18b)

This results in the following minimization conditions:

$$\lambda_4 = 0, \tag{3.2.19a}$$

$$\mu_1^2 = -\frac{1}{2} \left(2\lambda_1 - \lambda_2 + \lambda_3 + (\lambda_2 + \lambda_3) c_{2(\sigma_1 - \sigma_2)} \right) v^2, \qquad (3.2.19b)$$

$$\mu_2^2 = -(\lambda_2 + \lambda_3) s_{\sigma_1 - \sigma_2}^2 \left(\hat{w}_1^2 - \hat{w}_2^2 \right)$$
(3.2.19c)

alongside with

$$(\lambda_2 + \lambda_3) \,\mathbf{s}_{2(\sigma_1 - \sigma_2)} \hat{w}_1 \hat{w}_2 = 0. \tag{3.2.20}$$

Provided that $\mu_2^2 \neq 0$, either phases should be constrained or one of the VEVs should vanish. This leads to several possibilities.

As a first case, we consider the limit $\sigma_1 = \frac{1}{4}\pi$ and $\sigma_2 = -\sigma_1$, the minimization conditions are reduced to:

$$\lambda_4 = 0, \tag{3.2.21a}$$

$$\mu_1^2 = -(\lambda_1 - \lambda_2) v^2, \qquad (3.2.21b)$$

$$\mu_2^2 = -(\lambda_2 + \lambda_3) \left(\hat{w}_1^2 - \hat{w}_2^2 \right).$$
(3.2.21c)

The charged mass-squared matrix is given by:

$$\mathcal{M}_{\text{Charged}}^{2} = \begin{pmatrix} 2\lambda_{2}\hat{w}_{2}^{2} & 2i\lambda_{2}\hat{w}_{1}\hat{w}_{2} & 0\\ -2i\lambda_{2}\hat{w}_{1}\hat{w}_{2} & 2\lambda_{2}\hat{w}_{1}^{2} & 0\\ 0 & 0 & \mu_{0}^{2} + \frac{1}{2}\lambda_{5}v^{2} \end{pmatrix}, \qquad (3.2.22)$$

with eigenstates:

$$m_{H^{\pm}}^2 = 2\lambda_2 v^2,$$
 (3.2.23a)

$$m_{S^{\pm}}^2 = \mu_0^2 + \frac{1}{2}\lambda_5 v^2.$$
 (3.2.23b)

The charged scalar masses are identical to the ones of the exact C-III-c model (3.1.14). Note that the mass-squared matrix $\mathcal{M}^2_{\text{Charged}}$ is diagonalizable by a complex rotation matrix \mathcal{R} .

The S_3 doublet neutral-mass squared matrix in the basis

$$\{\tilde{\eta}_1, \tilde{\eta}_2, \tilde{\chi}_1, \tilde{\chi}_2\}$$
 (3.2.24)

is of the following form:

$$\mathcal{M}_{\text{Neutral}-12}^{2} = \begin{pmatrix} (\mathcal{M}_{a}^{2})_{11} & (\mathcal{M}_{a}^{2})_{12} & (\mathcal{M}_{a}^{2})_{13} & (\mathcal{M}_{a}^{2})_{14} \\ (\mathcal{M}_{a}^{2})_{12} & (\mathcal{M}_{a}^{2})_{22} & (\mathcal{M}_{a}^{2})_{23} & (\mathcal{M}_{a}^{2})_{24} \\ (\mathcal{M}_{a}^{2})_{13} & (\mathcal{M}_{a}^{2})_{23} & (\mathcal{M}_{a}^{2})_{11} & (\mathcal{M}_{a}^{2})_{34} \\ (\mathcal{M}_{a}^{2})_{14} & (\mathcal{M}_{a}^{2})_{24} & (\mathcal{M}_{a}^{2})_{34} & (\mathcal{M}_{a}^{2})_{22} \end{pmatrix}, \quad (3.2.25)$$

where

$$\left(\mathcal{M}_{a}^{2}\right)_{11} = (\lambda_{1} + \lambda_{3})\,\hat{w}_{1}^{2} + (\lambda_{2} + \lambda_{3})\,\hat{w}_{2}^{2},\tag{3.2.26a}$$

$$(\mathcal{M}_{a}^{2})_{12} = (\lambda_{1} - \lambda_{2}) w_{1} w_{2}, \qquad (3.2.26b)$$

$$(\mathcal{M}_{a}^{2})_{12} = (\lambda_{1} + \lambda_{2}) \hat{w}^{2} - (\lambda_{2} + \lambda_{2}) \hat{w}^{2} \qquad (3.2.26c)$$

$$(\mathcal{M}_a)_{13} = (\lambda_1 + \lambda_3) w_1 - (\lambda_2 + \lambda_3) w_2, \qquad (3.2.20c)$$
$$(\mathcal{M}^2) = (\lambda_1 + \lambda_3) w_1 - (\lambda_2 + \lambda_3) w_2, \qquad (3.2.20c)$$

$$(\mathcal{M}_a)_{14} = -(\lambda_1 - 3\lambda_2 - 2\lambda_3)w_1w_2, \qquad (3.2.200)$$
$$(\mathcal{M}^2)_{14} = -(\lambda_2 + \lambda_3)w_2^2 + (\lambda_2 + \lambda_3)w_2^2 \qquad (3.2.266)$$

$$(\mathcal{M}_a)_{22} = (\lambda_2 + \lambda_3) w_1 + (\lambda_1 + \lambda_3) w_2, \qquad (3.2.206)$$
$$(\mathcal{M}_a^2)_{22} = (\lambda_2 - \lambda_3) w_1 + (\lambda_1 + \lambda_3) w_2, \qquad (3.2.266)$$

$$(\mathcal{M}_{a}^{2})_{23} = (\lambda_{1} - 3\lambda_{2} - 2\lambda_{3}) w_{1}w_{2}, \qquad (3.2.261)$$

$$\left(\mathcal{M}_{a}^{2}\right)_{24} = (\lambda_{2} + \lambda_{3})\,\hat{w}_{1}^{2} - (\lambda_{1} + \lambda_{3})\,\hat{w}_{2}^{2},\tag{3.2.26g}$$

$$\left(\mathcal{M}_{a}^{2}\right)_{34} = -\left(\lambda_{1} - \lambda_{2}\right)\hat{w}_{1}\hat{w}_{2}.$$
 (3.2.26h)

The neutral scalars of the S_3 doublet acquire the following masses:

$$m_{H_1}^2 = 2v^2 \left(\lambda_2 + \lambda_3\right),$$
 (3.2.27a)

$$m_{H_2}^2 = v^2 \left(\lambda_1 + \lambda_3\right) - \Delta,$$
 (3.2.27b)

$$m_{H_3}^2 = v^2 \left(\lambda_1 + \lambda_3\right) + \Delta,$$
 (3.2.27c)

where

$$\Delta^{2} = v^{4} \left(\lambda_{1} + \lambda_{3}\right)^{2} - 16 \left(\lambda_{1} - \lambda_{2}\right) \left(\lambda_{2} + \lambda_{3}\right) \hat{w}_{1}^{2} \hat{w}_{2}^{2}.$$
(3.2.28)

The mass-eigenstates are a combination of all gauge-eigenstates, which potentially leads to CP-indefinite states.

The mass-squared matrix of the S_3 singlet is:

$$\mathcal{M}_{\text{Neutral-S}}^{2} = \begin{pmatrix} \mu_{0}^{2} + \frac{1}{2} \left(\lambda_{5} + \lambda_{6}\right) v^{2} & \lambda_{7} \left(\hat{w}_{1}^{2} - \hat{w}_{2}^{2}\right), \\ \lambda_{7} \left(\hat{w}_{1}^{2} - \hat{w}_{2}^{2}\right) & \mu_{0}^{2} + \frac{1}{2} \left(\lambda_{5} + \lambda_{6}\right) v^{2} \end{pmatrix},$$
(3.2.29)

with eigenvalues:

$$m_{S_1}^2 = \mu_0^2 + \frac{1}{2} \left(\lambda_5 + \lambda_6\right) v^2 - \lambda_7 \left(\hat{w}_1^2 - \hat{w}_2^2\right), \qquad (3.2.30a)$$

$$m_{S_2}^2 = \mu_0^2 + \frac{1}{2} \left(\lambda_5 + \lambda_6\right) v^2 + \lambda_7 \left(\hat{w}_1^2 - \hat{w}_2^2\right).$$
(3.2.30b)

This coincides with masses of the exact C-III-c model (3.1.29) by substituting values of the σ_i phases.

Another possibility to satisfy eq. (3.2.20) is to consider that either \hat{w}_1 or \hat{w}_2 vanishes. If we substitute $\hat{w}_i = 0$ directly into the vacuum configuration, this no longer results in the C-III-c model. To be more precise, such change results in a special case of either R-I-1 or R-II-2. Both of these cases are of no particular interest. Therefore, we conclude that the C-III-c- μ_2^2 vacuum configuration is given by:

$$\{\hat{w}_1 e^{i\frac{\pi}{4}}, \, \hat{w}_2 e^{-i\frac{\pi}{4}}, \, 0\}. \tag{3.2.31}$$

3.2.3 The C-III-c- ν^2 - μ_2^2 Model

The aforementioned soft symmetry breaking terms ν^2 and μ_2^2 can be combined together resulting in the following additional bilinear terms:

$$V_2' = \frac{1}{2}\nu^2 \left(h_1^{\dagger}h_2 + h_2^{\dagger}h_1 \right) + \mu_2^2 \left(h_1^{\dagger}h_1 - h_2^{\dagger}h_2 \right).$$
(3.2.32)

In this case constraints are:

$$\lambda_4 = 0, \tag{3.2.33a}$$

$$\mu_1^2 = -(\lambda_1 - \lambda_2) v^2, \qquad (3.2.33b)$$

$$\mu_2^2 = -(\lambda_2 + \lambda_3) \left(\hat{w}_1^2 - \hat{w}_2^2 \right), \qquad (3.2.33c)$$

$$\nu^{2} = -4 \left(\lambda_{2} + \lambda_{3}\right) c_{\sigma_{1} - \sigma_{2}} \hat{w}_{1} \hat{w}_{2}. \qquad (3.2.33d)$$

The charged mass-squared matrix is given by:

$$\mathcal{M}_{\text{Charged}}^{2} = \begin{pmatrix} 2\lambda_{2}\hat{w}_{2}^{2} & -2e^{-i(\sigma_{1}-\sigma_{2})}\lambda_{2}\hat{w}_{1}\hat{w}_{2} & 0\\ -2e^{i(\sigma_{1}-\sigma_{2})}\lambda_{2}\hat{w}_{1}\hat{w}_{2} & 2\lambda_{2}\hat{w}_{1}^{2} & 0\\ 0 & 0 & \mu_{0}^{2} + \frac{1}{2}\lambda_{5}v^{2} \end{pmatrix}, \qquad (3.2.34)$$

with eigenvalues:

$$m_{H^{\pm}}^2 = 2v^2\lambda_2,$$
 (3.2.35a)

$$m_{S^{\pm}}^2 = \mu_0^2 + \frac{1}{2}v^2\lambda_5.$$
 (3.2.35b)

The charged scalar masses are identical to the ones of the exact C-III-c model (3.1.14).

The neutral-mass squared matrix of the S_3 doublet in the basis (3.2.24) is of the following form:

$$\mathcal{M}_{\text{Neutral}-12}^{2} = \begin{pmatrix} (\mathcal{M}_{a}^{2})_{11} & (\mathcal{M}_{a}^{2})_{12} & (\mathcal{M}_{a}^{2})_{13} & (\mathcal{M}_{a}^{2})_{14} \\ (\mathcal{M}_{a}^{2})_{12} & (\mathcal{M}_{a}^{2})_{22} & (\mathcal{M}_{a}^{2})_{23} & (\mathcal{M}_{a}^{2})_{24} \\ (\mathcal{M}_{a}^{2})_{13} & (\mathcal{M}_{a}^{2})_{23} & (\mathcal{M}_{a}^{2})_{33} & (\mathcal{M}_{a}^{2})_{34} \\ (\mathcal{M}_{a}^{2})_{14} & (\mathcal{M}_{a}^{2})_{24} & (\mathcal{M}_{a}^{2})_{34} & (\mathcal{M}_{a}^{2})_{44} \end{pmatrix}, \quad (3.2.36)$$

where

$$\left(\mathcal{M}_{a}^{2}\right)_{11} = 2\left(\lambda_{1} + \lambda_{3}\right)\hat{w}_{1}^{2}c_{\sigma_{1}}^{2} + 2\left(\lambda_{2} + \lambda_{3}\right)\hat{w}_{2}^{2}c_{\sigma_{2}}^{2}, \qquad (3.2.37a)$$

$$\left(\mathcal{M}_{a}^{2}\right)_{12} = 2\left(\lambda_{1} - \lambda_{2}\right)\hat{w}_{1}\hat{w}_{2}c_{\sigma_{1}}c_{\sigma_{2}},\tag{3.2.37b}$$

$$\left(\mathcal{M}_{a}^{2}\right)_{13} = (\lambda_{1} + \lambda_{3})\,\hat{w}_{1}^{2}s_{2\sigma_{1}} + (\lambda_{2} + \lambda_{3})\,\hat{w}_{2}^{2}s_{2\sigma_{2}},\tag{3.2.37c}$$

$$\left(\mathcal{M}_{a}^{2}\right)_{14} = 2\hat{w}_{1}\hat{w}_{2}\left[\left(\lambda_{1} - 2\lambda_{2} - \lambda_{3}\right)c_{\sigma_{1}}s_{\sigma_{2}} + \left(\lambda_{2} + \lambda_{3}\right)c_{\sigma_{2}}s_{\sigma_{1}}\right], \qquad (3.2.37d)$$

$$\left(\mathcal{M}_{a}^{2}\right)_{22} = 2\left(\lambda_{2} + \lambda_{3}\right)\hat{w}_{1}^{2}c_{\sigma_{1}}^{2} + 2\left(\lambda_{1} + \lambda_{3}\right)\hat{w}_{2}^{2}c_{\sigma_{2}}^{2}, \qquad (3.2.37e)$$

$$\left(\mathcal{M}_{a}^{2}\right)_{23} = 2\hat{w}_{1}\hat{w}_{2}\left[\left(\lambda_{1} - 2\lambda_{2} - \lambda_{3}\right)c_{\sigma_{2}}s_{\sigma_{1}} + \left(\lambda_{2} + \lambda_{3}\right)c_{\sigma_{1}}s_{\sigma_{2}}\right], \qquad (3.2.37f)$$

$$(\mathcal{M}_{a})_{24} = (\lambda_{2} + \lambda_{3}) w_{1} s_{2\sigma_{1}} + (\lambda_{1} + \lambda_{3}) w_{2} s_{2\sigma_{2}}, \qquad (3.2.57g)$$

$$(\mathcal{M}_{a}^{2})_{33} = 2 (\lambda_{1} + \lambda_{3}) \hat{w}_{1}^{2} s_{\sigma_{1}}^{2} + 2 (\lambda_{2} + \lambda_{3}) \hat{w}_{2}^{2} s_{\sigma_{2}}^{2},$$

$$(3.2.37h)$$

$$\left(\mathcal{M}_{a}^{2}\right)_{34} = 2\left(\lambda_{1} - \lambda_{2}\right)\hat{w}_{1}\hat{w}_{2}s_{\sigma_{1}}s_{\sigma_{2}},\tag{3.2.37i}$$

$$\left(\mathcal{M}_{a}^{2}\right)_{44} = 2\left(\lambda_{2} + \lambda_{3}\right)\hat{w}_{1}^{2}s_{\sigma_{1}}^{2} + 2\left(\lambda_{1} + \lambda_{3}\right)\hat{w}_{2}^{2}s_{\sigma_{2}}^{2}.$$
(3.2.37j)

This results in the following mass-squared terms:

$$m_{H_1}^2 = 2v^2 \left(\lambda_2 + \lambda_3\right),$$
 (3.2.38a)

$$m_{H_2}^2 = v^2 \left(\lambda_1 + \lambda_3\right) - \Delta,$$
 (3.2.38b)

$$m_{H_3}^2 = v^2 (\lambda_1 + \lambda_3) + \Delta$$
 (3.2.38c)

where

$$\Delta^{2} = v^{4} \left(\lambda_{1} + \lambda_{3}\right)^{2} - 16 \left(\lambda_{1} - \lambda_{2}\right) \left(\lambda_{2} + \lambda_{3}\right) \mathbf{s}_{\sigma_{1} - \sigma_{2}}^{2} \hat{w}_{1}^{2} \hat{w}_{2}^{2}.$$
(3.2.39)

The only difference from the C-III-c- μ_2^2 model is how the Δ parameter is defined. The Δ parameter for the C-III-c- μ_2^2 model was defined in (3.2.28). In the C-III-c- ν^2 - μ_2^2 model the σ_i phases are no longer fixed and thus this results in additional dependence of the Δ parameter on $s_{\sigma_1-\sigma_2}^2$.

The S_3 singlet mass-squared matrix is:

$$\mathcal{M}_{\text{Neutral}-S}^{2} = \begin{pmatrix} (\mathcal{M}_{b}^{2})_{11} & (\mathcal{M}_{b}^{2})_{12} \\ (\mathcal{M}_{b}^{2})_{12} & (\mathcal{M}_{b}^{2})_{22} \end{pmatrix}, \qquad (3.2.40)$$

where

$$\left(\mathcal{M}_{b}^{2}\right)_{11} = \mu_{0}^{2} + \frac{1}{2}v^{2}\left(\lambda_{5} + \lambda_{6}\right) + \lambda_{7}\left(c_{2\sigma_{1}}\hat{w}_{1}^{2} + c_{2\sigma_{2}}\hat{w}_{2}^{2}\right), \qquad (3.2.41a)$$

$$\left(\mathcal{M}_{b}^{2}\right)_{12} = \lambda_{7} \left(s_{2\sigma_{1}} \hat{w}_{1}^{2} + s_{2\sigma_{2}} \hat{w}_{2}^{2}\right), \qquad (3.2.41b)$$

$$\left(\mathcal{M}_{b}^{2}\right)_{22} = \mu_{0}^{2} + \frac{1}{2}v^{2}\left(\lambda_{5} + \lambda_{6}\right) - \lambda_{7}\left(c_{2\sigma_{1}}\hat{w}_{1}^{2} + c_{2\sigma_{2}}\hat{w}_{2}^{2}\right).$$
(3.2.41c)

The mass-eigenstates of the S_3 singlet are:

$$m_{S_1}^2 = \mu_0^2 + \frac{1}{2}v^2 \left(\lambda_5 + \lambda_6\right) - \lambda_7 \sqrt{\hat{w}_1^4 + 2c_{2(\sigma_1 - \sigma_2)}\hat{w}_1^2 \hat{w}_2^2 + \hat{w}_2^4}, \qquad (3.2.42a)$$

$$m_{S_2}^2 = \mu_0^2 + \frac{1}{2}v^2 \left(\lambda_5 + \lambda_6\right) + \lambda_7 \sqrt{\hat{w}_1^4 + 2c_{2(\sigma_1 - \sigma_2)}\hat{w}_1^2 \hat{w}_2^2 + \hat{w}_2^4}, \qquad (3.2.42b)$$

which coincide with the terms for the exact C-III-c model (3.1.29).

3.3 *CP* Violation in the Scalar Sector of the C-III-c Models

Our primarily assumption was that the couplings of the scalar potential are real. Although the scalar potential does not violate CP explicitly, CP can still be violated spontaneously. The concept of spontaneous CP violation makes sense only if the scalar Lagrangian conserves CP explicitly. Spontaneous CP violation in the context of the 2HDM was proposed in Ref. [50]. A method to check if there is spontaneous CP violation in the Multi-Higgs-Doublet Model (NHDM) was presented in Ref. [51]. The main idea behind the method is to check if there exists a symmetry U that leaves the scalar Lagrangian invariant, and the vacuum satisfies:

$$U: U_{ij} \langle h_j \rangle^* = \langle h_i \rangle. \tag{3.3.1}$$

If this is true, then there is no spontaneous CP violation. This relation follows from the general CP transformation of the SU(2) doublets:

$$h_i \xrightarrow{CP} U'_{ij} h_j^*. \tag{3.3.2}$$

If the aforementioned mapping is possible, the CP is explicitly conserved. In general, the transformation matrix U'_{ij} , in terms of the 3HDM it is $U_{ij} \in U(3)$, is not a symmetry of the scalar Lagrangian. Provided that this symmetry leaves the scalar Lagrangian intact, eq. (3.3.1) is automatically satisfied. Nevertheless, CP violation can still be achieved through the Yukawa Lagrangian.

In the context of the C-III-c vacuum configuration, only the S_3 doublet should be checked. In Ref. [34] it was shown that the C-III-c vacuum configuration does not result in spontaneous CPviolation. In Ref. [48] it was proposed that CP violation is easier to analyse in the Higgs basis⁴. We,

⁴After the Higgs basis transformation only a single real VEV is left. The complex couplings in the Higgs basis result in CP violation provided those cannot be rotated into the real ones.

governed by this assumption, however, use a slightly different method and check if the couplings in the Higgs basis are real. Provided that such reverse engineering results in a possible direction in the Higgs basis, the SU(2) doublets in the generic basis can be re-constructed and checked if the result is in agreement with Refs. [34, 48]. The Higgs basis transformations are quite involved. The general approach can be found in section 2.5.

We consider the C-III-c and C-III-c-X (3.2.2) models consistent with the $\lambda_4 = 0$ constraint. The generic SU(2) doublets (2.1.3) are rotated into the Higgs basis so that CP is explicitly conserved in the generic basis. The Higgs basis transformation is given by eq. (3.1.2). The Euler rotation matrices for the C-III-c model in the basis of eq. (2.5.14) are:

$$\mathbf{t}_{\beta_1} = \frac{\hat{w}_2}{\hat{w}_1},\tag{3.3.3a}$$

$$\beta_2 = 0.$$
 (3.3.3b)

The coefficients of the potential are:

$$Y_{11} = Y_{22} = \mu_1^2, \tag{3.3.4a}$$

$$Y_{33} = \mu_0^2, \tag{3.3.4b}$$

and

$$Z_{1111} = Z_{2222} = \lambda_1 - \lambda_2, \tag{3.3.5a}$$

$$Z_{1122} = 2(\lambda_1 + \lambda_2),$$
 (3.3.5b)

$$Z_{1133} = Z_{2233} = \lambda_5, \tag{3.3.5c}$$

$$Z_{1221} = -4\lambda_2,$$
 (3.3.5d)

$$Z_{1313} = \lambda_7 \left(e^{-2i\sigma_1} c_{\beta_1}^2 + e^{-2i\sigma_2} s_{\beta_1}^2 \right), \qquad (3.3.5e)$$

$$Z_{1323} = \lambda_7 S_{2\beta_1} \left(-e^{-\lambda_0 T} + e^{-\lambda_0 T} \right), \qquad (3.3.51)$$

$$Z_{1331} = Z_{2332} = \lambda_6, \tag{3.3.5g}$$

$$Z_{2323} = \lambda_7 \left(e^{-2i\sigma_1} \mathbf{s}_{\beta_1}^2 + e^{-2i\sigma_2} \mathbf{c}_{\beta_1}^2 \right), \qquad (3.3.5h)$$

$$Z_{3333} = \lambda_8. (3.3.5i)$$

At first, it might seem that the scalar potential in the Higgs basis is CP violating. In the Higgs basis there is a freedom to rotate the SU(2) doublets with zero VEVs by a U(2) transformation, see eq. (2.5.19). As a consequence, there is a direction in the Higgs basis where all of the couplings become real. The choice of the Higgs basis transformation

$$\begin{pmatrix} H_1 \\ H_2 \\ H_3 \end{pmatrix} = \frac{1}{\sqrt{2}} \begin{pmatrix} e^{-i\sigma} & e^{i\sigma} & 0 \\ -ie^{-i\sigma} & ie^{i\sigma} & 0 \\ 0 & 0 & \sqrt{2}i \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}, \qquad (3.3.6)$$

alongside the vacuum configuration $\{\hat{w}e^{i\sigma}, \hat{w}e^{-i\sigma}, 0\}$ results in:

/ \

$$Z_{1313} = -Z_{2323} = -c_{2\sigma}\lambda_7, \tag{3.3.7a}$$

$$Z_{1323} = -2s_{2\sigma}\lambda_7,$$
 (3.3.7b)

and the other couplings are the same as in eqs. (3.3.4, 3.3.5). Therefore, the vacuum configuration of eq. (3.1.37) is *CP* conserving. We verify the claim of Refs. [34, 48] that the exact C-III-c vacuum configuration does not result in spontaneous *CP* violation⁵. It should be noted that the aforementioned Higgs basis rotation is not uniquely defined, *e.g.*, transformation

$$\begin{pmatrix} H_1 \\ H_2 \\ H_3 \end{pmatrix} = \begin{pmatrix} \frac{1}{\sqrt{2}}e^{-i\sigma_1} & \frac{1}{\sqrt{2}}e^{i\sigma} & 0\\ -\frac{i}{2}e^{-i\sigma} & \frac{i}{2}e^{i\sigma} & \frac{i}{\sqrt{2}}\\ \frac{i}{2}e^{-i\sigma} & -\frac{i}{2}e^{i\sigma} & \frac{i}{\sqrt{2}} \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}$$
(3.3.8)

⁵It was shown that this specific vacuum configuration does not lead to explicit CP violation and there exists a unitary matrix which satisfies eq. (3.3.1).

with VEVs { $\hat{w}e^{i\sigma}$, $\hat{w}e^{-i\sigma}$, 0} results in real couplings, but this time there are no vanishing couplings $Z_{ijkl} \neq 0$.

An interesting consequence is that although the exact C-III-c vacuum configuration is CP conserving, the soft symmetry breaking terms may result in CP violation. The corresponding case in the 2HDM was studied in Ref. [52]. When soft symmetry breaking terms are added, for the Higgs basis transformation (3.1.2), the quadratic terms result in:

$$Y_{11} = -\frac{1}{2}\nu^2 \mathbf{s}_{2\beta_1} \mathbf{c}_{\sigma_1 - \sigma_2} + \mu_1^2 + \mathbf{c}_{2\beta_1} \mu_2^2, \qquad (3.3.9a)$$

$$Y_{12} = \frac{1}{4}\nu^2 \left(c_{2\beta_1 + \sigma_1 - \sigma_2} + c_{2\beta_1 - \sigma_1 + \sigma_2} - 2is_{\sigma_1 - \sigma_2} \right) + s_{2\beta_1}\mu_2^2,$$
(3.3.9b)

$$Y_{13} = \frac{1}{2} e^{-i\sigma_1} c_{\beta_1} \mu_3^2 - \frac{1}{2} e^{-i\sigma_2} s_{\beta_1} \mu_4^2, \qquad (3.3.9c)$$

$$Y_{22} = \frac{1}{2}\nu^2 \mathbf{s}_{2\beta_1} \mathbf{c}_{\sigma_1 - \sigma_2} + \mu_1^2 - \mathbf{c}_{2\beta_1} \mu_2^2, \qquad (3.3.9d)$$

$$Y_{23} = \frac{1}{2}e^{-i\sigma_1}\mathbf{s}_{\beta_1}\mu_3^2 + \frac{1}{2}e^{-i\sigma_2}\mathbf{c}_{\beta_1}\mu_4^2, \qquad (3.3.9e)$$

$$Y_{33} = \mu_0^2. \tag{3.3.9f}$$

The Higgs basis transformation of eq. (3.3.6) results in complex quadratic couplings Y_{ij} :

$$Y_{12} = -\frac{1}{2}\nu^2 s_{2\sigma} + i\mu_2^2, \qquad (3.3.10a)$$

$$Y_{13} = -\frac{i}{2\sqrt{2}} \left[e^{-i\sigma} \mu_3^2 + e^{i\sigma} \mu_4^2 \right], \qquad (3.3.10b)$$

$$Y_{23} = \frac{1}{2\sqrt{2}} \left[-e^{-i\sigma} \mu_3^2 + e^{i\sigma} \mu_4^2 \right].$$
 (3.3.10c)

Of particular interest is only the Y_{12} coupling as the μ_3^2 and μ_4^2 couplings are inconsistent with $\lambda_4 = 0$. From eq. (3.3.10a) it follows that the C-III-c- ν^2 vacuum configuration does not violate CP spontaneously. The C-III-c- μ_2^2 vacuum configuration $\{\hat{w}e^{i\frac{\pi}{4}}, \hat{w}e^{-i\frac{\pi}{4}}, 0\}$ can be rotated into: $\{\hat{w}e^{i\delta}, \hat{w}, 0\}$, see eq. (3.1.38). Provided that the overall phase value is fixed at $\sigma = \pi/2$, the couplings in the Higgs basis become real. The C-III-c- ν^2 - μ_2^2 model does not result in a real basis due to the dependence $Y \sim \mu_2^2 + i\nu^2 s_{\sigma}$. In terms of the generic vacuum configuration (3.0.1), the C-III-c- ν^2 - μ_2^2 model results in real couplings in the Higgs basis provided that the vacuum configuration is given by $\{\hat{w}_1e^{i\sigma}, \hat{w}_2e^{i\sigma}, 0\}$. This configuration becomes real and is inconsistent with the C-III-c model.

Chapter 4 The C-III-c- ν^2 Model

In this chapter we revisit the C-III-c- ν^2 model presented in section 3.2.1. This model is of particular interest as it provides the simplest extension of the S_3 scalar potential and results in massive scalars as desired. Moreover, it does not result in spontaneous CP violation as claimed in section 3.3. This model is analysed by taking a look at tree-level couplings involving scalars. Theoretical constraints of the C-III-c- ν^2 model are also considered.

4.1 The Mass-Squared Matrices

If we expanded the scalar potential in terms of the generic SU(2) doublet of eq. (2.1.3), the charged mass-squared matrix would be hermitian and the neutral mass-squared matrices would depend on trigonometric functions of both σ_1 and σ_2 . In both cases, eigenvalues of the mass-squared matrices depend on a single combination of angles $\sigma_1 - \sigma_2$. The minimization condition in terms ν^2 is also a function of a single trigonometric function $c_{\sigma_1-\sigma_2}$, see eq. (3.2.15c). It is possible to rotate one of the phases away. In this case the vacuum configuration of eq. (3.2.14) changes to the following form:

$$\{\hat{w}e^{i\sigma}, \hat{w}, 0\},$$
 (4.1.1)

where $\hat{w} \equiv \hat{w}_1$ and $\sigma \equiv \sigma_1 - \sigma_2$. The constraint ν^2 of eq. (3.2.15c) gets modified:

$$\nu^2 = -2\left(\lambda_2 + \lambda_3\right) c_\sigma v^2. \tag{4.1.2}$$

We expand the scalar potential in terms of the following SU(2) doublets:

$$h_{1} = e^{-i\sigma_{2}} \begin{pmatrix} h_{1}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}e^{i\sigma_{1}} + \tilde{\eta}_{1} + i\tilde{\chi}_{1} \right) \end{pmatrix} \equiv e^{i\sigma} \begin{pmatrix} h_{1}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{\omega} + \tilde{\eta}_{1} + i\tilde{\chi}_{1} \right) \end{pmatrix},$$

$$h_{2} = e^{-i\sigma_{2}} \begin{pmatrix} h_{2}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{w}e^{i\sigma_{2}} + \tilde{\eta}_{2} + i\tilde{\chi}_{2} \right) \end{pmatrix} \equiv \begin{pmatrix} h_{2}^{+} \\ \frac{1}{\sqrt{2}} \left(\hat{\omega} + \tilde{\eta}_{2} + i\tilde{\chi}_{2} \right) \end{pmatrix},$$

$$h_{S} = e^{-i\sigma_{2}} \begin{pmatrix} h_{S}^{+} \\ \frac{1}{\sqrt{2}} \left(\tilde{\eta}_{S} + i\tilde{\chi}_{S} \right) \end{pmatrix} \equiv \begin{pmatrix} h_{S}^{+} \\ \frac{1}{\sqrt{2}} \left(\tilde{\eta}_{S} + i\tilde{\chi}_{S} \right) \end{pmatrix},$$
(4.1.3)

which is equivalent to the transformation (3.1.12). This time, however, the SU(2) doublets were simultaneously rotated by $e^{-i\sigma_2}$ due to the U(1) invariance (3.1.6) without extracting the phases as in eq. (3.1.5).

The modulus of the VEVs is fixed, $\hat{w}_1 = \hat{w}_2$, the rotation angle is also defined, $\beta = \pi/4$. Therefore the rotation matrix \mathcal{R}_{β} in this specific model is given by:

$$\mathcal{R}_{\beta} = \frac{1}{\sqrt{2}} \begin{pmatrix} 1 & 1 & 0 \\ -1 & 1 & 0 \\ 0 & 0 & \sqrt{2} \end{pmatrix}.$$
 (4.1.4)

Taking everything into consideration we get the following Higgs basis rotation:

$$\mathcal{R}_{\rm HB} = \frac{1}{\sqrt{2}} \begin{pmatrix} e^{-i\sigma} & 1 & 0\\ -e^{-i\sigma} & 1 & 0\\ 0 & 0 & \sqrt{2} \end{pmatrix}.$$
 (4.1.5)

After we determined the modified vacuum configuration and constraints of the C-III-c- ν^2 vacuum, we are ready to consider the mass-squared matrices. We start by investigating the charged scalar sector. It is diagonalizable by performing the rotation \mathcal{R}_{β} . The charged would-be Goldstone boson and the charged physical scalar fields are:

$$G^{\pm} = \frac{1}{\sqrt{2}} \left(h_1^{\pm} + h_2^{\pm} \right), \qquad (4.1.6a)$$

$$H^{\pm} = \frac{1}{\sqrt{2}} \left(-h_1^{\pm} + h_2^{\pm} \right), \qquad (4.1.6b)$$

$$S^{\pm} = h_S^{\pm}.$$
 (4.1.6c)

The masses of the charged scalar bosons are:

$$m_{H^{\pm}}^2 = 2\lambda_2 v^2, \qquad (4.1.7a)$$

$$m_{S^{\pm}}^2 = \mu_0^2 + \frac{1}{2}\lambda_5 v^2.$$
 (4.1.7b)

We note that the masses $m_{H^{\pm}}^2$ and $m_{S^{\pm}}^2$ are the same as those of eq. (3.1.14) for the exact C-III-c model.

Next, we consider the neutral scalar sector. After performing the \mathcal{R}_{β} rotation one can identify the would-be Goldstone boson:

$$G^{0} = \frac{1}{\sqrt{2}} \left(\tilde{\chi}_{1} + \tilde{\chi}_{2} \right).$$
(4.1.8)

Afterwards we take a look at the neutral sector mass-squared matrix:

$$\mathcal{M}_{\text{Neutral}-12}^{2} = \frac{\partial^{2} \tilde{V}}{\partial \zeta_{i}^{12} \partial \zeta_{j}^{12}} \Big|_{\langle v \rangle} = \begin{pmatrix} (\mathcal{M}_{a}^{2})_{11} & 0 & (\mathcal{M}_{a}^{2})_{13} \\ 0 & (\mathcal{M}_{a}^{2})_{22} & 0 \\ (\mathcal{M}_{a}^{2})_{13} & 0 & (\mathcal{M}_{a}^{2})_{33} \end{pmatrix},$$
(4.1.9)

where ζ_i^{12} is identical to the one of eq. (3.1.20). The non-zero elements of the mass-squared matrix $\mathcal{M}^2_{\text{Neutral}-12}$ are:

$$(\mathcal{M}_a^2)_{11} = 2\left[\lambda_1 - \lambda_2 \mathbf{s}_{\sigma}^2 + \lambda_3 \mathbf{c}_{\sigma}^2\right] v^2, \qquad (4.1.10a)$$

$$(\mathcal{M}_{a}^{2})_{13} = (\lambda_{2} + \lambda_{3}) \operatorname{s}_{2\sigma} v^{2},$$
 (4.1.10b)

$$(\mathcal{M}_a^2)_{22} = 2\,(\lambda_2 + \lambda_3)\,v^2,\tag{4.1.10c}$$

$$(\mathcal{M}_a^2)_{33} = 2(\lambda_2 + \lambda_3) s_{\sigma}^2 v^2.$$
(4.1.10d)

Note that elements $(\mathcal{M}_a^2)_{13}$, and $(\mathcal{M}_a^2)_{22}$, and $(\mathcal{M}_a^2)_{33}$ vanish in the limit of $\lambda_2 + \lambda_3 = 0$. An additional rotation in terms of an angle α is needed for the complete diagonalization procedure:

$$\mathcal{R}_{\alpha} = \begin{pmatrix} c_{\alpha} & 0 & s_{\alpha} \\ 0 & 1 & 0 \\ -s_{\alpha} & 0 & c_{\alpha} \end{pmatrix}.$$
(4.1.11)

After performing the rotation in terms of α we get the following eigenvalues:

$$m_{H_1}^2 = 2 \left[\lambda_1 c_{\alpha}^2 - \lambda_2 \left(c_{\alpha}^2 - c_{\alpha-\sigma}^2 \right) + \lambda_3 c_{\alpha-\sigma}^2 \right] v^2, \qquad (4.1.12a)$$

$$m_{H_2}^2 = 2(\lambda_2 + \lambda_3)v^2,$$
 (4.1.12b)

$$m_{H_3}^2 = 2 \left[\lambda_1 \mathbf{s}_{\alpha}^2 + \lambda_2 \left(\mathbf{c}_{\alpha}^2 - \mathbf{c}_{\alpha-\sigma}^2 \right) + \lambda_3 \mathbf{s}_{\alpha-\sigma}^2 \right] v^2.$$
(4.1.12c)

These eigenvalues can further be simplified by identifying the rotation angle α :

$$t_{2\alpha} = \frac{(\lambda_2 + \lambda_3) s_{2\sigma}}{\lambda_1 - \lambda_2 + (\lambda_2 + \lambda_3) c_{2\sigma}} , \qquad (4.1.13)$$

and thus

$$m_{H_1}^2 = (\lambda_1 + \lambda_3 - \Delta) v^2,$$
 (4.1.14a)

$$m_{H_2}^2 = 2(\lambda_2 + \lambda_3)v^2,$$
 (4.1.14b)

$$m_{H_3}^2 = (\lambda_1 + \lambda_3 + \Delta) v^2,$$
 (4.1.14c)

where for simplification we introduced an additional abbreviation

$$\Delta^{2} = (\lambda_{1} - \lambda_{2})^{2} + (\lambda_{2} + \lambda_{3})^{2} + 2(\lambda_{1} - \lambda_{2})(\lambda_{2} + \lambda_{3})c_{2\sigma}.$$
(4.1.15)

It follows that we can fix $0 \le \sigma \le \pi/2$ and $0 \le \alpha \le \pi/2$.

From dependence of masses on λ couplings it follows that masses are strictly ordered as $m_{H_1} < m_{H_2} < m_{H_3}$. Therefore m_{H_1} is the lightest scalar and we associate it with the SM-like Higgs boson.

The neutral S_3 doublet mass-eigenstates are:

$$H_1 = \frac{1}{\sqrt{2}} \left[c_\alpha \left(\tilde{\eta}_1 + \tilde{\eta}_2 \right) + s_\alpha \left(-\tilde{\chi}_1 + \tilde{\chi}_2 \right) \right], \qquad (4.1.16a)$$

$$H_2 = \frac{1}{\sqrt{2}} \left(-\tilde{\eta}_1 + \tilde{\eta}_2 \right), \tag{4.1.16b}$$

$$H_3 = \frac{1}{\sqrt{2}} \left[-\mathbf{s}_\alpha \left(\tilde{\eta}_1 + \tilde{\eta}_2 \right) + \mathbf{c}_\alpha \left(-\tilde{\chi}_1 + \tilde{\chi}_2 \right) \right].$$
(4.1.16c)

Based on couplings ZH_iH_j in section 4.2, we note that both H_1 and H_3 are *CP*-even states and H_2 is the *CP*-odd state.

Relation between the gauge-eigenstates and the mass-eigenstates is given by the following rotation matrix:

$$\begin{pmatrix} H_1 \\ H_2 \\ G^0 \\ H_3 \end{pmatrix} = \begin{pmatrix} \frac{c_{\alpha}}{\sqrt{2}} & \frac{c_{\alpha}}{\sqrt{2}} & -\frac{s_{\alpha}}{\sqrt{2}} & \frac{s_{\alpha}}{\sqrt{2}} \\ -\frac{1}{\sqrt{2}} & \frac{1}{\sqrt{2}} & 0 & 0 \\ 0 & 0 & \frac{1}{\sqrt{2}} & \frac{1}{\sqrt{2}} \\ -\frac{s_{\alpha}}{\sqrt{2}} & -\frac{s_{\alpha}}{\sqrt{2}} & -\frac{c_{\alpha}}{\sqrt{2}} & \frac{c_{\alpha}}{\sqrt{2}} \end{pmatrix} \begin{pmatrix} \tilde{\eta}_1 \\ \tilde{\eta}_2 \\ \tilde{\chi}_1 \\ \tilde{\chi}_2 \end{pmatrix}.$$
(4.1.17)

We finally consider the S_3 singlet. The mass-squared matrix is given by:

$$\mathcal{M}_{\text{Neutral-S}}^{2} = \begin{pmatrix} \mu_{0}^{2} + \frac{1}{2} \left(\lambda_{5} + \lambda_{6}\right) v^{2} + \lambda_{7} c_{\sigma}^{2} v^{2} & \frac{1}{2} \lambda_{7} s_{2\sigma} v^{2} \\ \frac{1}{2} \lambda_{7} s_{2\sigma} v^{2} & \mu_{0}^{2} + \frac{1}{2} \left(\lambda_{5} + \lambda_{6}\right) v^{2} - \lambda_{7} c_{\sigma}^{2} v^{2} \end{pmatrix}.$$
(4.1.18)

It is possible to diagonalize the $\mathcal{M}^2_{Neutral-S}$ mass-squared matrix by the following rotation matrix:

$$\mathcal{R}_{\gamma} = \begin{pmatrix} c_{\gamma} & s_{\gamma} \\ -s_{\gamma} & c_{\gamma} \end{pmatrix}.$$
(4.1.19)

The off-diagonal elements are zero provided that:

$$s_{(2\gamma-\sigma)} = 0$$
, or $s_{2\sigma-2\gamma} = s_{2\gamma}$. (4.1.20)

Therefore we take the value of $\gamma = \sigma/2$.

We find the following neutral scalar masses of the S_3 singlet:

$$m_{S_1}^2 = \mu_0^2 + \frac{1}{2} \left(\lambda_5 + \lambda_6\right) v^2 - \lambda_7 c_\sigma v^2, \qquad (4.1.21a)$$

$$m_{S_2}^2 = \mu_0^2 + \frac{1}{2} \left(\lambda_5 + \lambda_6\right) v^2 + \lambda_7 c_\sigma v^2, \qquad (4.1.21b)$$

and as expected those coincide with the exact C-III-c model (3.1.29) by making appropriate changes.

The S_3 singlet neutral physical fields are:

$$S_1 = c_\gamma \,\tilde{\eta}_S + s_\gamma \,\tilde{\chi}_S, \qquad (4.1.22a)$$

$$S_2 = -\mathbf{s}_\gamma \,\tilde{\eta}_S + \mathbf{c}_\gamma \,\tilde{\chi}_S. \tag{4.1.22b}$$

Although it it known that both states S_1 and S_2 are of opposite CP parities, due to the interaction ZS_1S_2 , as presented in section 4.2, it is impossible to assign which of them is CP-even and which is CP-odd.

Taking everything into consideration, the SU(2) doublets in terms of the mass-eigenstates are given by:

$$h_1 = e^{i\sigma} \frac{1}{\sqrt{2}} \begin{pmatrix} G^+ - H^+ \\ \frac{1}{\sqrt{2}} \left(v + H_1 e^{-i\alpha} - H_2 - H_3 i e^{-i\alpha} + i G^0 \right) \end{pmatrix},$$
(4.1.23a)

$$h_2 = \frac{1}{\sqrt{2}} \begin{pmatrix} G^+ + H^+ \\ \frac{1}{\sqrt{2}} \left(v + H_1 e^{i\alpha} + H_2 + H_3 i e^{i\alpha} + i G^0 \right) \end{pmatrix},$$
(4.1.23b)

$$h_S = \begin{pmatrix} S^+ \\ \frac{1}{\sqrt{2}} \left(S_1 e^{i\gamma} + S_2 i e^{i\gamma} \right) \end{pmatrix}.$$
(4.1.23c)

In the Higgs basis, the SU(2) doublets in terms of the mass-eigenstates are as follows:

$$H_1 = \begin{pmatrix} G^+ \\ \frac{1}{\sqrt{2}} \left(v + \varphi_1 \mathbf{c}_\alpha - \varphi_3 \mathbf{s}_\alpha + iG^0 \right) \end{pmatrix}, \qquad (4.1.24a)$$

$$H_2 = \begin{pmatrix} H^+ \\ \frac{1}{\sqrt{2}} \left(i \left(\varphi_1 \mathbf{s}_\alpha + \varphi_3 \mathbf{c}_\alpha \right) + \varphi_2 \right) \end{pmatrix}, \qquad (4.1.24b)$$

$$H_3 = \begin{pmatrix} S^+ \\ \frac{1}{\sqrt{2}} \left(S_1 e^{i\gamma} + S_2 i e^{i\gamma} \right) \end{pmatrix}, \qquad (4.1.24c)$$

where $\varphi_i \equiv H_i$ of eq. (4.1.17).

4.1.1 Quartic Couplings in Terms of Masses

We invert equations eq. (4.1.7a), eq. (4.1.14) and eq. (4.1.21) containing the mass-squared parameters and get equations for couplings as functions of masses. The most straightforward way is to determine relations for λ_2 and λ_5 . This can be done by taking a look at $m_{H^{\pm}}^2$ and $m_{S^{\pm}}^2$:

$$\lambda_2 = \frac{1}{2v^2} m_{H^{\pm}}^2, \tag{4.1.25a}$$

$$\lambda_5 = \frac{2}{v^2} \left(m_{S^{\pm}}^2 - \mu_0^2 \right). \tag{4.1.25b}$$

Afterwards, one can determine relation for the coupling λ_3 from $m_{H_2}^2$:

$$\lambda_3 = \frac{1}{2v^2} \left(m_{H_2}^2 - m_{H^{\pm}}^2 \right) \tag{4.1.26}$$

Next, we take a look at the two combinations $m_{H_1}^2 \pm m_{H_3}^2$ and find that:

$$\lambda_1 = \frac{1}{2v^2} \left(m_{H_1}^2 - m_{H_2}^2 + m_{H_3}^2 + m_{H^{\pm}}^2 \right), \qquad (4.1.27a)$$

$$\Delta = \frac{1}{2v^2} \left(-m_{H_1}^2 + m_{H_3}^2 \right). \tag{4.1.27b}$$

From the Δ parameter we can determine the value of σ :

$$c_{2\sigma} = 1 - \frac{2m_{H_1}^2 m_{H_3}^2}{m_{H_2}^2 \left(m_{H_1}^2 - m_{H_2}^2 + m_{H_3}^2\right)}.$$
(4.1.28)

From here, an interesting conclusion can be drawn. Although σ is an overall phase between the two S_3 doublets, it turns out that it can be expressed in terms of physical quantities and thus an arbitrary phase is promoted to a physical parameter.

From the combination $m_{S_1}^2 \pm m_{S_2}^2$ it follows that:

$$\lambda_{6} = \frac{1}{v^{2}} \left(-2m_{S^{\pm}}^{2} + m_{S_{1}}^{2} + m_{S_{2}}^{2} \right),$$

$$\lambda_{7} = \frac{1}{2c_{\sigma}v^{2}} \left(-m_{S_{1}}^{2} + m_{S_{2}}^{2} \right).$$
(4.1.29)

The only undetermined couplings are μ_0^2 and λ_8 , which are left as free parameters of the model.

For convenience, we collect all of the relations together. In case of the C-III-c- ν^2 model we get the following relations:

$$\mu_1^2 = -\frac{1}{2} \left(m_{H_1}^2 - m_{H_2}^2 + m_{H_3}^2 \right), \tag{4.1.30a}$$

$$\nu^{2} = -m_{H_{2}}^{2} \sqrt{1 - \frac{m_{H_{1}}^{2} m_{H_{3}}^{2}}{m_{H_{2}}^{2} \left(m_{H_{1}}^{2} - m_{H_{2}}^{2} + m_{H_{3}}^{2}\right)}},$$
(4.1.30b)

$$\lambda_1 = \frac{1}{2v^2} \left(m_{H_1}^2 - m_{H_2}^2 + m_{H_3}^2 + m_{H^{\pm}}^2 \right), \qquad (4.1.30c)$$

$$\lambda_2 = \frac{1}{2v^2} m_{H^{\pm}}^2, \tag{4.1.30d}$$

$$\lambda_3 = \frac{1}{2v^2} \left(m_{H_2}^2 - m_{H^{\pm}}^2 \right), \tag{4.1.30e}$$

$$\lambda_4 = 0, \tag{4.1.30f}$$

$$\lambda_5 = \frac{2}{v_s^2} \left(m_{S^{\pm}}^2 - \mu_0^2 \right), \tag{4.1.30g}$$

$$\lambda_6 = \frac{1}{v^2} \left(-2m_{S^{\pm}}^2 + m_{S_1}^2 + m_{S_2}^2 \right), \qquad (4.1.30h)$$

$$\lambda_7 = \frac{1}{2c_\sigma v^2} \left(-m_{S_1}^2 + m_{S_2}^2 \right), \qquad (4.1.30i)$$

$$c_{2\sigma} = 1 - \frac{2m_{H_1}^2 m_{H_3}^2}{m_{H_2}^2 \left(m_{H_1}^2 - m_{H_2}^2 + m_{H_3}^2\right)}.$$
(4.1.30j)

4.2 Scalar-Gauge Boson Interactions

The kinetic part of the Lagrangian is given by:

$$\mathcal{L}_{K} = \sum_{i=1,2} \left(D^{\mu} h_{i} \right)^{\dagger} \left(D_{\mu} h_{i} \right) + \left(D^{\mu} h_{S} \right)^{\dagger} \left(D_{\mu} h_{S} \right), \qquad (4.2.1)$$

where D_{μ} is the covariant derivative:

$$D_{\mu} = \partial_{\mu} + ig\tau_i W^i_{\mu} + ig' Y B_{\mu}, \qquad (4.2.2)$$

where g' and g are the U(1) and SU(2) coupling constants, W^i_{μ} $(i = \overline{1,3})$ and B_{μ} are gauge fields, τ_i are the generators of the SU(2) group and Y is the hypercharge. The fields W^i_{μ} and B_{μ} are in the gauge basis. These fields are to be diagonalized to correspond to the physical observable states W^{\pm}_{μ} , Z_{μ} and A_{μ} .

The interaction Lagrangian of the gauge bosons with scalars can be expressed as:

$$\mathcal{L}_K = \mathcal{L}_{V-\text{mass}} + \mathcal{L}_{V-S}, \tag{4.2.3}$$

where the $\mathcal{L}_{V-\text{mass}}$ part is responsible for generation of the mass terms of the gauge bosons and \mathcal{L}_{V-S} is responsible for interactions of the scalar and gauge bosons fields.

Expansion of the covariant derivative with respect to the SU(2) generators yields:

$$D_{\mu} = \partial_{\mu} + ig \frac{1}{2} \begin{pmatrix} W_{\mu}^{3} & W_{\mu}^{1} - iW_{\mu}^{2} \\ W_{\mu}^{1} + iW_{\mu}^{2} & -W_{\mu}^{3} \end{pmatrix} + ig' Y \begin{pmatrix} B_{\mu} & 0 \\ 0 & B_{\mu} \end{pmatrix}$$

$$= \partial_{\mu} + \frac{i}{2} \begin{pmatrix} gW_{\mu}^{3} + g' \, 2YB_{\mu} & g\left(W_{\mu}^{1} - iW_{\mu}^{2}\right) \\ g\left(W_{\mu}^{1} + iW_{\mu}^{2}\right) & -gW_{\mu}^{3} + g' \, 2YB_{\mu} \end{pmatrix}.$$
(4.2.4)

We define the physical fields in the following way:

$$W^{\pm}_{\mu} = \frac{1}{\sqrt{2}} \left(W^{1}_{\mu} \mp i W^{2}_{\mu} \right), \qquad (4.2.5a)$$

$$Z_{\mu} = \frac{1}{\sqrt{g^2 + g'^2}} \left(g W_{\mu}^3 - g' B_{\mu} \right), \qquad (4.2.5b)$$

$$A_{\mu} = \frac{1}{\sqrt{g^2 + g'^2}} \left(g' W_{\mu}^3 + g B_{\mu} \right), \qquad (4.2.5c)$$

where we assigned the value $Y = \frac{1}{2}$ to the hypercharge of the scalar doublets. The specific choice of the hypercharge decouples Z_{μ} from A_{μ} . This is done as we are interested in massless photon states.

We extract masses of the gauge bosons by evaluating interactions at vacuum:

$$\mathcal{L}_{V-\text{mass}} = \sum_{i=1,2} \frac{1}{8} w_i^2 \left[g^2 (W_{\mu}^+)^2 + g^2 (W_{\mu}^-)^2 + \left(g^2 + g'^2 \right) Z_{\mu}^2 + 0 \times A_{\mu}^2 \right]$$

$$\equiv m_W^2 W_{\mu}^{\dagger} W^{\mu} + \frac{1}{2} m_Z^2 Z_{\mu} Z^{\mu}.$$
(4.2.6)

The mass terms of the gauge bosons can be evaluated using the above mass Lagrangian:

$$m_W = \frac{g}{2}v, \tag{4.2.7a}$$

$$m_Z = \frac{\sqrt{g^2 + g'^2}}{2}v,$$
 (4.2.7b)

$$m_A = 0.$$
 (4.2.7c)

We list some of the useful relations:

$$e = g \operatorname{s}_w = g' \operatorname{c}_w, \tag{4.2.8a}$$

$$c_w = \frac{g}{\sqrt{g^2 + g'^2}},$$
 (4.2.8b)

$$s_w = \frac{g'}{\sqrt{g^2 + g'^2}},$$
 (4.2.8c)

$$\mathbf{t}_w = \frac{g'}{g},\tag{4.2.8d}$$

$$\frac{m_W}{m_Z} = c_w. \tag{4.2.8e}$$

It is straightforward to rewrite the covariant derivative D_{μ} in terms of the physical gauge bosons fields. The only part which requires some calculation is the upper-left component of eq. (4.2.4):

$$\frac{i}{2} \left(g W_{\mu}^{3} + g' \, 2Y B_{\mu} \right) = \frac{ig}{2c_{w}} \left[s_{w} B_{\mu} + c_{w} W_{\mu}^{3} \right],
= \frac{ig}{2c_{w}} \left[s_{w} \left(-s_{w} Z_{\mu} + c_{w} A_{\mu} \right) + c_{w} \left(c_{w} Z_{\mu} + s_{w} A_{\mu} \right) \right],
= \frac{ig}{2c_{w}} \left[\left(c_{w}^{2} - s_{w}^{2} \right) Z_{\mu} + 2s_{w} c_{w} A_{\mu} \right],
= \frac{ig}{2} \frac{c_{2w}}{c_{w}} Z_{\mu} + ie A_{\mu}$$
(4.2.9)

Acting with the covariant derivative on the SU(2) scalar doublet yields:

$$D_{\mu} \begin{pmatrix} h_{\alpha}^{+} \\ h_{\alpha}^{0} \end{pmatrix} = \begin{pmatrix} \left[\partial_{\mu} + \frac{ig}{2} \frac{c_{2w}}{c_{w}} Z_{\mu} + ieA_{\mu} \right] h_{\alpha}^{+} + \frac{ig}{\sqrt{2}} W_{\mu}^{+} h_{\alpha}^{0} \\ \frac{ig}{\sqrt{2}} W_{\mu}^{-} h_{\alpha}^{+} + \left[\partial_{\mu} - \frac{ig}{2c_{w}} Z_{\mu} \right] h_{\alpha}^{0} \end{pmatrix}, \qquad \alpha = \{1, 2, S\}.$$
(4.2.10)

Before substituting eq. (4.2.10) into eq. (4.2.1), for convenience the \mathcal{L}_{V-S} of eq. (4.2.3) can be split into several parts based on interaction properties:

$$\mathcal{L}_{V-S} = \mathcal{L}_{VHH} + \mathcal{L}_{VVH} + \mathcal{L}_{VVHH}.$$
(4.2.11)

For the general result we present the interaction Lagrangian \mathcal{L}_{V-S} in the gauge basis of the scalars:

$$h_{\alpha} = \begin{pmatrix} h_{\alpha}^{+} \\ \frac{1}{\sqrt{2}} h_{\alpha}^{0} \end{pmatrix} = \begin{pmatrix} h_{\alpha}^{+} \\ \frac{1}{\sqrt{2}} (w_{\alpha} + \tilde{\eta}_{\alpha} + \tilde{\chi}_{\alpha}) \end{pmatrix}.$$
 (4.2.12)

The resulting interaction Lagrangian parts are:

$$\mathcal{L}_{VVH} = \left[\frac{g}{2c_w}m_Z Z_\mu Z^\mu + gm_W W^+_\mu W^{\mu-}\right]\tilde{\eta}_\alpha + \left\{ \left[em_W A^\mu W^+_\mu - gm_Z s^2_w Z^\mu W^+_\mu\right]h^-_\alpha + \text{h.c.} \right\},$$
(4.2.13a)

$$\mathcal{L}_{VHH} = -\frac{ig}{4c_w} Z^{\mu} (h^0_{\alpha})^* \overleftrightarrow{\partial_{\mu}} h^0_{\alpha} - \frac{g}{2} \left\{ iW^+_{\mu} h^-_{\alpha} \overleftrightarrow{\partial^{\mu}} h^0_{\alpha} \right\} + \left[ieA^{\mu} + \frac{ig}{2} \frac{c_{2w}}{c_w} Z^{\mu} \right] h^+_{\alpha} \overleftrightarrow{\partial_{\mu}} h^-_{\alpha}, \quad (4.2.13b)$$

$$\mathcal{L}_{VVHH} = \left[\frac{g^2}{8c_w^2} Z_\mu Z^\mu + \frac{g^2}{4} W^+_\mu W^{\mu-} \right] (h^0_\alpha)^* h^0_\alpha + \left\{ \left[\frac{eg}{2} A^\mu W^+_\mu - \frac{g^2}{2} \frac{s^2_w}{c_w} Z^\mu W^+_\mu \right] h^-_\alpha h^+_\alpha + \text{h.c.} \right\} \\ + \left[e^2 A_\mu A^\mu + eg \frac{c_{2w}}{c_w} A_\mu Z^\mu + \frac{g^2}{4} \frac{c^2_{2w}}{c^2_w} Z_\mu Z^\mu + \frac{g^2}{2} W^-_\mu W^{\mu+} \right] h^-_\alpha h^+_\alpha.$$

$$(4.2.13c)$$

In order to extract interactions of physical states from the kinetic Lagrangian one needs to work with the mass-eigenstates of eq. (4.1.23). The resulting parts of the interaction Lagrangian \mathcal{L}_{V-S} are:

$$\mathcal{L}_{VVH} = \left[\frac{g}{2c_w}m_Z Z_\mu Z^\mu + gm_W W^+_\mu W^{\mu-}\right] (c_\alpha H_1 - s_\alpha H_3) + \left\{ \left[em_W A^\mu W^+_\mu - gm_Z s^2_w Z^\mu W^+_\mu\right] G^- + h.c. \right\},$$
(4.2.14a)

From the interaction terms ZZH_1 and ZZH_3 it follows that the states H_1 and H_3 act as CP-even. From the terms $Z\varphi_i\varphi_j$ the following information can be extracted: H_2 and G^0 are CP-odd, states S_1 and S_2 have opposite CP numbers. The two photon state $A_{\mu}A^{\mu}$ couples to a pair of the same species charged particles $\varphi_i^{\pm}\varphi_i^{\mp}$, and thus the state $\varphi_i^{\pm}\varphi_i^{\mp}$ is CP-even.

The Feynman rules for the interactions of the scalars φ_i and the gauge bosons V_j are:

$$\begin{aligned} \varphi_i V_j V_k &= i \, S \, g \left(\varphi_i V_j V_k \right) g^{\mu\nu}, \\ \varphi_i \stackrel{\leftrightarrow}{\partial} \varphi_j V_k &= S \, g \left(\varphi_i \varphi_j V_k \right) (p_j - p_i)^{\mu}, \text{ for all momenta ingoing,} \\ \varphi_i \varphi_j V_k V_l &= i \, S \, g \left(\varphi_i \varphi_j V_k V_l \right) g^{\mu\nu}, \end{aligned} \tag{4.2.15}$$

where S is the symmetry factor: $S = \prod_i n_i!$, for the *i* identical particles of species n, $g(\varphi_i V_j V_k)$, $g(\varphi_i \partial \varphi_j V_k)$, and $g(\varphi_i \varphi_j V_k V_l)$ are the corresponding scalar-gauge bosons coefficients of vertices obtained directly from eqs. (4.2.14a-4.2.14c), and p_i are the incoming four-momenta.

4.3 Scalar-Fermion Interactions

The general Yukawa Lagrangian is as follows:

$$\mathcal{L}_Y = \mathcal{L}_Y^d + \mathcal{L}_Y^u + \mathcal{L}_Y^e + \mathcal{L}_Y^\nu. \tag{4.3.1}$$

We work in the massless neutrino limit and thus the \mathcal{L}_Y^{ν} term vanishes. Since the scalar potential is S_3 -symmetric, we consider that the Yukawa Lagrangian is also S_3 -symmetric. We assume the following S_3 structure $(2+1)_L \otimes (2+1)_R \otimes (2+1)_h$:

$$S_{3} \text{ doublets:} \quad \begin{pmatrix} \overline{Q}_{1} \\ \overline{Q}_{2} \end{pmatrix}_{L}, \quad \begin{pmatrix} u_{1} \\ u_{2} \end{pmatrix}_{R}, \quad \begin{pmatrix} d_{1} \\ d_{2} \end{pmatrix}_{R}, \quad \begin{pmatrix} h_{1} \\ h_{2} \end{pmatrix};$$

$$S_{3} \text{ singlets:} \quad \overline{Q}_{3L}, \quad u_{3R}, \quad d_{3R}, \quad h_{S};$$

$$(4.3.2)$$

where indices $\overline{1,3}$ label quark families.

The Yukawa Lagrangian must be in the invariant singlet. Singlets of S_3 can be obtained from multiplication of two singlets or two doublets, where one factor could arise from the product of two doublets:

$$\begin{array}{l}
\mathbf{1} \otimes \mathrm{any} = \mathrm{any}, \\
\mathbf{2} \otimes \mathbf{2} = \mathbf{1} \oplus \mathbf{1}' \oplus \mathbf{2}
\end{array}$$
(4.3.3)

Consider possible Yukawa terms. As an example, if we want to couple the scalar singlet h_S to fermions, we should couple it to a fermion singlet. Yukawa couplings of *d*-quarks that are singlets under S_3 can thus be constructed in 5 different ways, with independent coefficients $\{y_1^d, \ldots, y_5^d\}$ [35, 53]:

$$y_1^d: \quad \bar{Q}_{1L}^0 h_S d_{1R}^0 + \bar{Q}_{2L}^0 h_S d_{2R}^0, \tag{4.3.4a}$$

$$y_2^d: \ (\bar{Q}_{1L}^0 h_2 + \bar{Q}_{2L}^0 h_1) d_{1R}^0 + (\bar{Q}_{1L}^0 h_1 - \bar{Q}_{2L}^0 h_2) d_{2R}^0, \tag{4.3.4b}$$

$$y_3^d: \ \bar{Q}_{3L}^0 h_S d_{3R}^0, \tag{4.3.4c}$$

$$y_4^d: \ (\bar{Q}_{1L}^0 h_1 + \bar{Q}_{2L}^0 h_2) d_{3R}^0, \tag{4.3.4d}$$

$$y_5^d: \ \bar{Q}_{3L}^0(h_1 d_{1R}^0 + h_2 d_{2R}^0), \tag{4.3.4e}$$

where the Yukawa couplings y_i^d are assumed to be real. The quark sector of the Yukawa Lagrangian is:

$$\mathcal{L}_Y^q = \mathcal{L}_Y^d + \mathcal{L}_Y^u + \text{h.c.}, \qquad (4.3.5)$$

where

$$\begin{aligned} -\mathcal{L}_{Y}^{d} &= y_{1}^{d} \left[\bar{Q}_{1L}^{0} h_{S} d_{1R}^{0} + \bar{Q}_{2L}^{0} h_{S} d_{2R}^{0} \right] + y_{2}^{d} \left[(\bar{Q}_{1L}^{0} h_{2} + \bar{Q}_{2L}^{0} h_{1}) d_{1R}^{0} + (\bar{Q}_{1L}^{0} h_{1} - \bar{Q}_{2L}^{0} h_{2}) d_{2R}^{0} \right] \\ &+ y_{3}^{d} \bar{Q}_{3L}^{0} h_{S} d_{3R}^{0} + y_{4}^{d} \left[(\bar{Q}_{1L}^{0} h_{1} + \bar{Q}_{2L}^{0} h_{2}) d_{3R}^{0} \right] + y_{5}^{d} \bar{Q}_{3L}^{0} (h_{1} d_{1R}^{0} + h_{2} d_{2R}^{0}), \end{aligned}$$

$$-\mathcal{L}_{Y}^{u} = y_{1}^{u} \left[\bar{Q}_{1L}^{0} \tilde{h}_{S} u_{1R}^{0} + \bar{Q}_{2L}^{0} \tilde{h}_{S} u_{2R}^{0} \right] + y_{2}^{u} \left[(\bar{Q}_{1L}^{0} \tilde{h}_{2} + \bar{Q}_{2L}^{0} \tilde{h}_{1}) u_{1R}^{0} + (\bar{Q}_{1L}^{0} \tilde{h}_{1} - \bar{Q}_{2L}^{0} \tilde{h}_{2}) u_{2R}^{0} \right] + y_{3}^{u} \bar{Q}_{3L}^{0} \tilde{h}_{S} u_{3R}^{0} + y_{4}^{u} \left[(\bar{Q}_{1L}^{0} \tilde{h}_{1} + \bar{Q}_{2L}^{0} \tilde{h}_{2}) u_{3R}^{0} \right] + y_{5}^{u} \bar{Q}_{3L}^{0} (\tilde{h}_{1} u_{1R}^{0} + \tilde{h}_{2} u_{2R}^{0}).$$

$$(4.3.6b)$$

As a result, the most general fermion mass matrix is of the following form:

$$\mathcal{M}_{u} = \frac{1}{\sqrt{2}} \begin{pmatrix} y_{1}^{u}w_{S}^{*} + y_{2}^{u}w_{2}^{*} & y_{2}^{u}w_{1}^{*} & y_{4}^{u}w_{1}^{*} \\ y_{2}^{u}w_{1}^{*} & y_{1}^{u}w_{S}^{*} - y_{2}^{u}w_{2}^{*} & y_{4}^{u}w_{2}^{*} \\ y_{5}^{u}w_{1}^{*} & y_{5}^{u}w_{2}^{*} & y_{3}^{u}w_{S}^{*} \end{pmatrix}, \qquad (4.3.7a)$$

$$\mathcal{M}_{d} = \frac{1}{\sqrt{2}} \begin{pmatrix} y_{1}^{d}w_{S} + y_{2}^{d}w_{2} & y_{2}^{d}w_{1} & y_{4}^{d}w_{1} \\ y_{2}^{d}w_{1} & y_{1}^{d}w_{S} - y_{2}^{d}w_{2} & y_{4}^{d}w_{2} \\ y_{5}^{d}w_{1} & y_{5}^{d}w_{2} & y_{3}^{d}w_{S} \end{pmatrix}. \qquad (4.3.7b)$$

Decomposition of the mass matrices into interactions with different scalar doublets yields:

$$\mathcal{M}_{1} \approx \begin{pmatrix} 0 & y_{2} & y_{4} \\ y_{2} & 0 & 0 \\ y_{5} & 0 & 0 \end{pmatrix}, \qquad (4.3.8a)$$
$$\mathcal{M}_{2} \approx \begin{pmatrix} y_{2} & 0 & 0 \\ 0 & -y_{2} & y_{4} \\ 0 & y_{5} & 0 \end{pmatrix}, \qquad (4.3.8b)$$
$$\mathcal{M}_{S} \approx \begin{pmatrix} y_{1} & 0 & 0 \\ 0 & y_{1} & 0 \\ 0 & 0 & y_{3} \end{pmatrix}. \qquad (4.3.8c)$$

It is straightforward to consider which of the vacuum configurations lead to unrealistic cases, e.g., models with a single active SU(2) doublet result in unrealistic eigenvalues. For h_1 (R-I-2a), the

(4.3.6a)

first generation becomes massless, and for h_2 (R-II-2), one of the eigenvalues is negative, and for h_S (R-I-1), two states are mass-degenerate.

Of particular interest are mass-eigenstates of fermion states. Rotation from the weak basis into the mass basis is performed by the following unitary transformations:

$$u_{L}^{0} = V_{u}u_{L}, \qquad u_{R}^{0} = U_{u}u_{R}, d_{L}^{0} = V_{d}d_{L}, \qquad d_{R}^{0} = U_{d}d_{R}.$$
(4.3.9)

In the mass-eigenstates basis we expect to get definite fermion masses. The diagonal entries should therefore correspond to:

diag
$$(m_u, m_c, m_t) \equiv \mathcal{M}_u = V_u^{\dagger} \mathcal{M}_u U_u,$$

diag $(m_d, m_s, m_b) \equiv \hat{\mathcal{M}}_d = V_d^{\dagger} \mathcal{M}_d U_d.$

$$(4.3.10)$$

We also define hermitian mass-squared matrices:

$$\mathcal{H}_{u} = \mathcal{M}_{u} \left(\mathcal{M}_{u} \right)^{\dagger} = V_{u} \, \hat{\mathcal{M}}_{u}^{2} V_{u}^{\dagger},
\mathcal{H}_{d} = \mathcal{M}_{d} \left(\mathcal{M}_{d} \right)^{\dagger} = V_{d} \, \hat{\mathcal{M}}_{d}^{2} V_{d}^{\dagger},$$
(4.3.11)

or equivalently:

$$\mathcal{H}'_{u} = (\mathcal{M}_{u})^{\dagger} \mathcal{M}_{u} = U_{u} \hat{\mathcal{M}}_{u}^{2} U_{u}^{\dagger},$$

$$\mathcal{H}'_{d} = (\mathcal{M}_{d})^{\dagger} \mathcal{M}_{d} = U_{d} \hat{\mathcal{M}}_{d}^{2} U_{d}^{\dagger},$$

(4.3.12)

The left-handed diagonalization matrix of eq. (4.3.9) is defined by solving eq. (4.3.11) and the right-handed diagonalization matrix from:

$$U = \mathcal{M}^{-1} V \hat{\mathcal{M}}.$$
 (4.3.13)

We define hermitian mass-squared matrix invariants:

$$A_{(u,d)} \equiv \operatorname{tr}(\mathcal{H}_{(u,d)}), \tag{4.3.14a}$$

$$B_{(u,d)} \equiv -(\mathcal{H}_{(u,d)})_{11}(\mathcal{H}_{(u,d)})_{22} - (\mathcal{H}_{(u,d)})_{22}(\mathcal{H}_{(u,d)})_{33} - (\mathcal{H}_{(u,d)})_{33}(\mathcal{H}_{(u,d)})_{11} + (\mathcal{H}_{(u,d)})_{12}(\mathcal{H}_{(u,d)})_{21} + (\mathcal{H}_{(u,d)})_{23}(\mathcal{H}_{(u,d)})_{32} + (\mathcal{H}_{(u,d)})_{31}(\mathcal{H}_{(u,d)})_{13},$$

$$(4.3.14b)$$

$$C_{(u,d)} \equiv \det(\mathcal{H}_{(u,d)}). \tag{4.3.14c}$$

CP violation can be inspected by inspecting the determinant [54–56]:

$$J = \text{Det} \left(\mathcal{H}_d \mathcal{H}_u - \mathcal{H}_u \mathcal{H}_d \right).$$
(4.3.15)

The C-III-c- ν^2 vacuum configuration is given by $\{e^{i\sigma}\frac{v}{\sqrt{2}}, \frac{v}{\sqrt{2}}, 0\}$. Substituting the vacuum configuration into eq. (4.3.7) results in the quark mass matrices:

$$\mathcal{M}_{u} = \frac{v}{2} \begin{pmatrix} y_{2}^{u} & e^{-i\sigma}y_{2}^{u} & e^{-i\sigma}y_{4}^{u} \\ e^{-i\sigma}y_{2}^{u} & -y_{2}^{u} & y_{4}^{u} \\ e^{-i\sigma}y_{5}^{u} & y_{5}^{u} & 0 \end{pmatrix}, \qquad (4.3.16a)$$

$$\mathcal{M}_{d} = \frac{v}{2} \begin{pmatrix} y_{2}^{d} & e^{i\sigma}y_{2}^{d} & e^{i\sigma}y_{4}^{d} \\ e^{i\sigma}y_{2}^{d} & -y_{2}^{d} & y_{4}^{d} \\ e^{i\sigma}y_{5}^{d} & y_{5}^{d} & 0 \end{pmatrix}, \qquad (4.3.16b)$$

and hermitian mass-squared matrices are:

$$\mathcal{H}_{u} = \frac{1}{2}v^{2} \begin{pmatrix} (y_{2}^{u})^{2} + \frac{1}{2}(y_{4}^{u})^{2} & is_{\sigma}(y_{2}^{u})^{2} + \frac{1}{2}e^{-i\sigma}(y_{4}^{u})^{2} & c_{\sigma}y_{2}^{u}y_{5}^{u} \\ -is_{\sigma}(y_{2}^{u})^{2} + \frac{1}{2}e^{i\sigma}(y_{4}^{u})^{2} & (y_{2}^{u})^{2} + \frac{1}{2}(y_{4}^{u})^{2} & 0 \\ c_{\sigma}y_{2}^{u}y_{5}^{u} & 0 & (y_{5}^{u})^{2} \end{pmatrix},$$
(4.3.17a)

$$\mathcal{H}_{d} = \frac{1}{2}v^{2} \begin{pmatrix} (y_{2}^{d})^{2} + \frac{1}{2}(y_{4}^{d})^{2} & -is_{\sigma}(y_{2}^{d})^{2} + \frac{1}{2}e^{i\sigma}(y_{4}^{d})^{2} & c_{\sigma}y_{2}^{d}y_{5}^{d} \\ is_{\sigma}(y_{2}^{d})^{2} + \frac{1}{2}e^{-i\sigma}(y_{4}^{d})^{2} & (y_{2}^{d})^{2} + \frac{1}{2}(y_{4}^{d})^{2} & 0 \\ c_{\sigma}y_{2}^{d}y_{5}^{d} & 0 & (y_{5}^{d})^{2} \end{pmatrix}.$$
 (4.3.17b)

Hermitian mass-squared matrix invariants are:

$$A_{u} = \frac{1}{2}v^{2} (y_{5}^{u})^{2} + \frac{1}{2}v^{2} \left[2 (y_{2}^{u})^{2} + (y_{4}^{u})^{2} \right], \qquad (4.3.18a)$$

$$A_d = \frac{1}{2}v^2 \left(y_5^d\right)^2 + \frac{1}{2}v^2 \left[2 \left(y_2^d\right)^2 + \left(y_4^d\right)^2\right], \qquad (4.3.18b)$$

and

$$B_{u} = -\frac{1}{8}v^{4} \left[c_{2\sigma} \left((y_{2}^{u})^{2} - (y_{4}^{u})^{2} - (y_{5}^{u})^{2} \right) (y_{2}^{u})^{2} + (y_{2}^{u})^{4} + 3 \left((y_{4}^{u})^{2} + (y_{5}^{u})^{2} \right) (y_{2}^{u})^{2} + 2 (y_{4}^{u})^{2} (y_{5}^{u})^{2} \right],$$

$$(4.3.19a)$$

$$B_{d} = -\frac{1}{8}v^{4} \left[c_{2\sigma} \left(\left(y_{2}^{d} \right)^{2} - \left(y_{4}^{d} \right)^{2} - \left(y_{5}^{d} \right)^{2} \right) \left(y_{2}^{d} \right)^{2} + \left(y_{2}^{d} \right)^{4} + 3 \left(\left(y_{4}^{d} \right)^{2} + \left(y_{5}^{d} \right)^{2} \right) \left(y_{2}^{d} \right)^{2} + 2 \left(y_{4}^{d} \right)^{2} \left(y_{5}^{d} \right)^{2} \right],$$

$$(4.3.19b)$$

and

$$C_{u} = \frac{1}{32} v^{6} \left(5 - 3c_{2\sigma} \right) \left(y_{2}^{u} \right)^{2} \left(y_{4}^{u} \right)^{2} \left(y_{5}^{u} \right)^{2}, \qquad (4.3.20a)$$

$$C_d = \frac{1}{32} v^6 \left(5 - 3c_{2\sigma} \right) \left(y_2^d \right)^2 \left(y_4^d \right)^2 \left(y_5^d \right)^2.$$
(4.3.20b)

The CP check results in:

$$\begin{split} J &= \frac{1}{1024} \left(e^{-4i\sigma} - 1 \right) v^{12} c_{\sigma}^{2} \left[\left(y_{4}^{d} \right)^{2} \left(y_{2}^{u} \right)^{2} + \left(y_{2}^{d} - y_{4}^{d} \right) \left(y_{2}^{d} + y_{4}^{d} \right) \left(y_{4}^{u} \right)^{2} \right] \\ & \left[\left(y_{2}^{u} \left(y_{2}^{d} y_{5}^{d} y_{2}^{u} + \left(y_{2}^{d} \right)^{2} y_{5}^{u} - \left(y_{4}^{d} \right)^{2} y_{5}^{u} \right) - e^{2i\sigma} y_{2}^{d} \left(y_{2}^{d} y_{2}^{u} y_{5}^{u} + y_{5}^{d} \left(y_{2}^{u} \right)^{2} - y_{5}^{d} \left(y_{4}^{u} \right)^{2} \right) \right) \\ & \left(- y_{2}^{d} \left(y_{2}^{d} y_{2}^{u} y_{5}^{u} + y_{5}^{d} \left(y_{2}^{u} \right)^{2} - y_{5}^{d} \left(y_{4}^{u} \right)^{2} \right) + e^{2i\sigma} y_{2}^{u} \left(y_{2}^{d} y_{5}^{d} y_{2}^{u} + \left(y_{2}^{d} \right)^{2} y_{5}^{u} - \left(y_{4}^{d} \right)^{2} y_{5}^{u} \right) \right) \\ & - e^{2i\sigma} \left(2y_{2}^{d} y_{5}^{d} \left(y_{5}^{u} \right)^{2} - y_{2}^{d} y_{5}^{d} \left(2 \left(y_{2}^{u} \right)^{2} + \left(y_{4}^{u} \right)^{2} \right) + \left(2 \left(y_{2}^{d} \right)^{2} + \left(y_{4}^{d} \right)^{2} - 2 \left(y_{5}^{d} \right)^{2} \right) y_{2}^{u} y_{5}^{u} \right)^{2} \right]. \end{aligned}$$

$$\tag{4.3.21}$$

Both $\mathcal{H}_{(u,d)}$ are hermitian. By using the complex number identities in polar coordinates it is possible to rotate away complex phases. Consider \mathcal{H}_u of eq. (4.3.17a). Redefinition yields:

$$is_{\sigma} (y_2)^2 + \frac{1}{2} e^{-i\sigma} (y_4)^2 \equiv r_1 e^{i\varphi_1},$$

$$is_{\sigma} (y_2)^2 + \frac{1}{2} e^{-i\sigma} (y_5)^2 \equiv r_2 e^{i\varphi_2},$$
(4.3.22)

and thus hermitian mass-squared matrices become:

$$\mathcal{M}\mathcal{M}^{\dagger} = \frac{1}{2}v^{2} \begin{pmatrix} (y_{2})^{2} + \frac{1}{2}(y_{4})^{2} & r_{1}e^{i\varphi_{1}} & c_{\sigma}y_{2}y_{5} \\ r_{1}e^{-i\varphi_{1}} & (y_{2})^{2} + \frac{1}{2}(y_{4})^{2} & 0 \\ c_{\sigma}y_{2}y_{5} & 0 & (y_{5})^{2} \end{pmatrix}, \qquad (4.3.23a)$$
$$\mathcal{M}^{\dagger}\mathcal{M} = \frac{1}{2}v^{2} \begin{pmatrix} (y_{2})^{2} + \frac{1}{2}(y_{5})^{2} & r_{2}e^{-i\varphi_{2}} & c_{\sigma}y_{2}y_{4} \\ r_{2}e^{i\varphi_{2}} & (y_{2})^{2} + \frac{1}{2}(y_{5})^{2} & 0 \\ c_{\sigma}y_{2}y_{4} & 0 & (y_{4})^{2} \end{pmatrix}. \qquad (4.3.23b)$$

After performing rotations $\mathcal{R}_{\varphi_1}^{\dagger}\mathcal{M}\mathcal{M}^{\dagger}\mathcal{R}_{\varphi_1}$ and $\mathcal{R}_{\varphi_2}^{\dagger}\mathcal{M}^{\dagger}\mathcal{M}\mathcal{R}_{\varphi_2}$, where

$$\mathcal{R}_{\varphi_{(1,2)}} = \text{diag}\left(1, \, e^{\mp i\varphi_{(1,2)}}, \, 1\right), \tag{4.3.24}$$

hermitian mass-squared matrices become symmetric:

$$\mathcal{H} \approx \begin{pmatrix} H_{11} & H_{12} & H_{13} \\ H_{12} & H_{11} & 0 \\ H_{13} & 0 & H_{33} \end{pmatrix}.$$
 (4.3.25)

Full diagonalization of hermitian matrices is performed in terms of 4 parameters, three angles of the SO(3) rotation and the phase rotation of eq. (4.3.24). Due to complexity of analytical expressions we perform a numerical fit. We note that it is impossible to get a physically meaningful solution when trying to reduce the number of Yukawa couplings, *i.e.*, setting some of the $y_i = 0$.

In total, we need to fit 7 different parameters of which 6 are Yukawa couplings, counting both up and down Yukawa couplings separately, and also the overall phase σ of h_1 . In the C-III-c model there are only three non-zero Yukawa couplings and this is the minimal number of couplings needed to fit mass values of the three fermion generations. Since σ is a free parameter, we found that the best fit for masses is achieved by considering the following exponential polynomial:

$$y_i = e^{\sum_{j=0}^5 A_j^i \sigma^j}, \tag{4.3.26}$$

where A_j^i are real numbers and in some cases are equal to zero. In principle, we need to fit 3 different Yukawa couplings and thus it makes sense to take a look at different orderings of the Yukawa couplings. For simplicity, we use the following notation:

$$Y_{ijk} \equiv y_i > y_j > y_k. \tag{4.3.27}$$

One of the constraints on the Yukawa couplings comes from the CKM matrix. The absolute value of the CKM matrix

$$V_{\rm CKM} = V_u^{\dagger} V_d \tag{4.3.28}$$

is a well-known quantity. The PDG presents the following CKM matrix [29]:

$$V_{\rm CKM} = \begin{pmatrix} 0.9742 & 0.2243 & 0.00394 \\ 0.218 & 0.997 & 0.0422 \\ 0.0081 & 0.0394 & 1.019 \end{pmatrix}.$$
 (4.3.29)

As stated earlier, we consider that σ is a free parameter and therefore it can be used to fit the value of the CKM matrix. If there was at least one additional free parameter there would arise a possibility to cancel the net effect of the σ value. Therefore the σ phase needs to be fixed so that both experimental and theoretical values of the CKM matrix are in agreement. We found that in most cases diagonal elements of the CKM matrix are close to unity when $\sigma = 0$. It should be noted that based on results of the spectrum generator of section 6.2, the minimum value of σ is close to 0.16π .

Consider different Yukawa couplings orderings (4.3.27). Both orderings Y_{245} and Y_{254} result in unnaturally low second family masses and thus are not considered. Best fit is achieved in the limit

 $\sigma = 0.16\pi$:

$$Y_{425}^{u}Y_{425}^{d}: |V_{\rm CKM}| = \begin{pmatrix} 0.999 & 0.032 & 0.001 \\ 0.028 & 0.876 & 0.482 \\ 0.016 & 0.481 & 0.876 \end{pmatrix},$$

$$Y_{452}^{u}Y_{452}^{d}: |V_{\rm CKM}| = \begin{pmatrix} 0.876 & 0.028 & 0.482 \\ 0.032 & 0.999 & 0.001 \\ 0.481 & 0.016 & 0.876 \end{pmatrix},$$

$$Y_{524}^{u}Y_{524}^{d}: |V_{\rm CKM}| = \begin{pmatrix} 0.624 & 0.782 & 0.011 \\ 0.782 & 0.624 & 0.005 \\ 0.011 & 0.005 & 1 \end{pmatrix},$$

$$Y_{542}^{u}Y_{542}^{d}: |V_{\rm CKM}| = \begin{pmatrix} 0.877 & 0.481 & 0.001 \\ 0.481 & 0.877 & 0.001 \\ 0.481 & 0.877 & 0.001 \\ 0.001 & 0.001 & 1 \end{pmatrix}.$$
(4.3.30)

The only meaningful model is $Y_{542}^u Y_{542}^d$. Another interesting case is to consider non-identical orderings of the up-/down-quarks. For $\sigma = 0.36\pi$ we get something even closer to the PDG value:

$$Y_{524}^{u}Y_{542}^{d}: |V_{\rm CKM}| = \begin{pmatrix} 0.975 & 0.224 & 0.001 \\ 0.224 & 0.975 & 0.001 \\ 0.001 & 0.001 & 1 \end{pmatrix},$$

$$Y_{542}^{u}Y_{524}^{d}: |V_{\rm CKM}| = \begin{pmatrix} 0.975 & 0.223 & 0.005 \\ 0.223 & 0.975 & 0.005 \\ 0.005 & 0.005 & 1 \end{pmatrix}.$$

$$(4.3.31)$$

We assume that the only sensible orderings are $Y_{524}^u Y_{542}^d$ and $Y_{542}^u Y_{524}^d$. Since σ is a free parameter and neither result is in perfect agreement with the experimental CKM value, we consider the following acceptable range:

$$\sigma = [0.34\pi, 0.38\pi]. \tag{4.3.32}$$

In this range we get the following CKM matrix values:

$$Y_{524}^{u}Y_{542}^{d}: |V_{\text{CKM}}| = \begin{pmatrix} 0.966 - 0.982 & 0.257 - 0.191 & 0.002 - 0.001 \\ 0.257 - 0.191 & 0.966 - 0.982 & 0.001 \\ 0.002 - 0.001 & 0.002 - 0.001 & 1 \end{pmatrix},$$

$$Y_{542}^{u}Y_{524}^{d}: |V_{\text{CKM}}| = \begin{pmatrix} 0.967 - 0.982 & 0.256 - 0.19 & 0.006 - 0.004 \\ 0.256 - 0.19 & 0.967 - 0.982 & 0.006 - 0.004 \\ 0.006 - 0.004 & 0.005 - 0.004 & 1 \end{pmatrix}.$$
(4.3.33)

The nearly identity form of the CKM matrix $|V_{\text{CKM}}| = \mathcal{I} + \mathcal{O}(10^{-2})$ is achieved for both cases when $\sigma \geq 0.493\pi$. This choice of σ results in nearly mass degenerate states $m_{S_1} = m_{S_2}$.

Now, as we have two viable models, we move to the scalar-fermion interactions. The scalar-fermion interactions can be extracted from the Yukawa Lagrangian:

$$g(\xi_i \bar{f}_j f_k) = \left(V_f^{\dagger} \mathcal{M}_i^{\text{int}} U_f P_R + U_f^{\dagger} \left(\mathcal{M}_i^{\text{int}} \right)^{\dagger} V_f P_L \right)_{jk}, \qquad (4.3.34)$$

where the interaction matrices $\mathcal{M}_i^{\text{int}}$ are equivalent to the ones of eq. (4.3.7) with substitution of $w_i \to \xi_i$, where ξ_i are fields of interest multiplied by appropriate coefficients. For simplicity, consider the u-quark sector:

$$\mathcal{M}_{H_{1},u}^{\text{int}} = \frac{1}{2} \begin{pmatrix} e^{-i\alpha}y_{2}^{u} & \frac{1}{2}e^{i(\alpha-\sigma)}y_{2}^{u} & e^{i(\alpha-\sigma)}y_{4}^{u} \\ e^{i(\alpha-\sigma)}y_{2}^{u} & -e^{-i\alpha}y_{2}^{u} & e^{-i\alpha}y_{4}^{u} \\ e^{i(\alpha-\sigma)}y_{5}^{u} & e^{-i\alpha}y_{5}^{u} & 0 \end{pmatrix},$$
(4.3.35a)

$$\mathcal{M}_{H_2,u}^{\text{int}} = \frac{1}{2} \begin{pmatrix} y_2^u & -e^{-i\sigma}y_2^u & -e^{-i\sigma}y_4^u \\ -e^{-i\sigma}y_2^u & -y_2^u & y_4^u \\ -e^{-i\sigma}y_5^u & y_5^u & 0 \end{pmatrix},$$
(4.3.35b)

$$\mathcal{M}_{H_{3},u}^{\text{int}} = \frac{1}{2} \begin{pmatrix} -ie^{-i\alpha}y_{2}^{u} & ie^{i(\alpha-\sigma)}y_{2}^{u} & ie^{i(\alpha-\sigma)}y_{4}^{u} \\ ie^{i(\alpha-\sigma)}y_{2}^{u} & iy_{2}^{u}e^{-i\alpha} & -ie^{-i\alpha}y_{4}^{u} \\ ie^{i(\alpha-\sigma)}y_{5}^{u} & -ie^{-i\alpha}y_{5}^{u} & 0 \end{pmatrix},$$
(4.3.35c)

where α is the *CP*-even sector diagonalization angle given by eq. (4.1.13).

We denote

$$\tilde{\mathcal{M}}_{(i,f)}^{\text{int}} \equiv V_f^{\dagger} \mathcal{M}_i^{\text{int}} U_f \tag{4.3.36}$$

and therefore¹

$$g(\xi_i \bar{f} f) = \tilde{\mathcal{M}}_{(i,f)}^{\text{int}} P_R + \left(\tilde{\mathcal{M}}_{(i,f)}^{\text{int}}\right)^{\dagger} P_L.$$
(4.3.37)

The resulting diagonal interactions of the form $\xi_i \bar{f}_j f_j$ are:

$$\tilde{\mathcal{M}}_{jj}\gamma_5 + \mathbb{R}e(\tilde{\mathcal{M}}_{jj})\left(1 - \gamma_5\right),\tag{4.3.38}$$

and the FCNC $\xi_i \bar{f}_j f_k$ are of the form:

$$\frac{1}{2}\left[\left(\tilde{\mathcal{M}}_{jk} + (\tilde{\mathcal{M}}_{kj})^*\right) + \gamma_5\left(\tilde{\mathcal{M}}_{jk} - (\tilde{\mathcal{M}}_{kj})^*\right)\right],\tag{4.3.39}$$

where $\tilde{\mathcal{M}}_{jk}$ are the elements of $\tilde{\mathcal{M}}_{(i,f)}^{\text{int}}$. It follows that there are FCNC present if and only if the off-diagonal elements of the $\tilde{\mathcal{M}}_{(i,f)}^{\text{int}}$ matrix are non-zero.

Consider the C-III-c- ν^2 vacuum configuration $\{e^{i\sigma}\frac{v}{\sqrt{2}}, \frac{v}{\sqrt{2}}, 0\}$. The left-handed and righthanded matrices V and U diagonalize the fermion mass matrices. Therefore it follows that any sort of combination $A\{e^{i\sigma}, 1, 0\}$, where A is either complex or real, results in a diagonal interaction form. Provided that coefficients next to the fields do not obey the aforementioned condition, the resulting off-diagonal elements of the $\tilde{\mathcal{M}}_{(i,f)}^{\text{int}}$ are non-zero and thus result in FCNC.

Due to the fact that the fermion sector diagonalization is performed numerically, it is not obvious how the elements of interaction matrices look like. We consider Yukawa models $Y_{524}^u Y_{542}^d$ and $Y_{542}^u Y_{524}^d$. Of particular interest are interactions involving the SM-like Higgs boson H_1 . In case of the vanishing $\alpha = 0$ we achieve the SM limit:

$$g\left(H_1\bar{f}f\right) = \frac{m_f}{v}.\tag{4.3.40}$$

For $\alpha \neq 0$, the diagonal elements can be expressed systematically as:

$$g_{ii} = \operatorname{diag}\left((\mathbb{R} + \mathbb{I}\gamma_5)_{11}, \, (\mathbb{R} + \mathbb{I}\gamma_5)_{22}, \, \mathbb{R}_{33}\right),\tag{4.3.41}$$

$$(\tilde{\mathcal{M}}_{(i,u)}^{\text{int}})_{\text{Charged}} = \tilde{\mathcal{M}}_{(i,u)}^{\text{int}} V_{\text{CKM}}, \\ (\tilde{\mathcal{M}}_{(i,d)}^{\text{int}})_{\text{Charged}} = V_{\text{CKM}} \tilde{\mathcal{M}}_{(i,d)}^{\text{int}}.$$

¹We do not consider the charged scalar interactions here. The charged flavour currents result in the CKM matrix. The Yukawa Lagrangian should be changed appropriately:

where \mathbb{R} is a real number, \mathbb{I} is an imaginary number. The off-diagonal elements are of the following form:

$$g_{ij} = (\mathbb{R} + \mathbb{I})_{ij} + (\mathbb{R} + \mathbb{I})'_{ij} \gamma_5, \qquad (4.3.42)$$

and are hermitian, $g_{ij} = g_{ji}^{\dagger}$. One of the disastrous consequences of the model are FCNC, see eqs. (6.2.1, 6.2.2). We found that in most cases the off-diagonal elements dominate over the diagonal ones. There is no way to control such processes as the model is numerically fixed, there are no free parameters.

In both Yukawa models, with an increasing value of σ , the value of the diagonal third family coupling g_{33} tends to zero. This indicates that there should exist an upper boundary condition for the angle α so that the coupling of the third family to the SM-like Higgs boson is meaningful. We assume that the SM-like limit is achieved by $c_{\alpha} \ge 0.9$. Another interesting property is that with an increasing α value, values of the diagonal elements g_{ii} decrease while of the off-diagonal elements g_{ij} increase. For $\alpha = 0.14\pi$, the third family quarks couple with a strength $g_{33} \approx 0.9 \frac{m_3}{v}$.

Interactions involving the H_2 scalar do not depend on the angle α . Elements of the interaction matrix are of the same form as given by eqs. (4.3.41, 4.3.42). One would assume that each following fermion generation would couple stronger to the scalar than the previous one. This is not the case for the coupling g_{33} . In principle, both g_{11} and g_{22} couple to H_2 as CP-indefinite, while the third family as CP-odd. However, the value of g_{33} is only one magnitude different from g_{11} , e.g., for $Y_{524}^u Y_{542}^d$ and $\sigma = 0.36\pi$ the diagonal elements for the up-quarks are:

$$g_{11}^{u} \approx -1.4 \times 10^{-6} + 2.2 \times 10^{-6} \, i \, \gamma_{5},$$

$$g_{22}^{u} \approx -2.4 \times 10^{-9} - 2.2 \times 10^{-3} \, i \, \gamma_{5},$$

$$g_{33}^{u} \approx 1.6 \times 10^{-5} \, i \, \gamma_{5}.$$
(4.3.43)

For the down-quarks: $g_{33}^d = \mathcal{O}(10^{-8})$. The same is true for the $Y_{542}^u Y_{524}^d$ model: $g_{33}^u = 0$ and $g_{33}^d = \mathcal{O}(10^{-6})$. From the element g_{33}^u it follows that for σ in the range (4.3.32) there is no interaction $H_2 \bar{t} t$.

Fermion-scalar interactions involving H_3 once again depend on the angle α . The diagonal elements are of the same form as for H_1 (4.3.41). The off-diagonal elements are of the same form as in eq. (4.3.42). In contrast to $H_1\bar{f}f$, with an increasing α value, values of the diagonal elements g_{ii} increase while the off-diagonal elements g_{ij} decrease. This is a totally expected behavior and arises from the fact that both states H_1 and H_3 were diagonalized by the same angle α .

So far we considered only the quark sector. We assume that there are no right-handed neutrinos ν_R and therefore neglect neutrino masses and the PMNS matrix. The PMNS matrix is not fixed in terms of the left-/right-handed diagonalization matrices and therefore this results in a freedom of the Yukawa couplings ordering. Governed by the quarks sector result, we scanned the range $\sigma = [0.34\pi, 0.38\pi]$ and $\alpha = [0, 0.14\pi]$. We found no significant deviations from the general conditions for the quark sector. The only interesting observation involves interactions of the H_2 scalar with the third family:

$$Y_{425}^e = Y_{524}^e = \mathcal{O}\left(10^{-5}\right),$$

$$Y_{452}^e = Y_{542}^e = \mathcal{O}\left(10^{-10}\right).$$
(4.3.44)

It seems unnatural that the scalar H_2 would "not bother" about the third family and thus we consider models with higher g_{33} value. We conclude that models $Y_{524}^u Y_{542}^d Y_{425}^e$ and $Y_{524}^u Y_{542}^d Y_{524}^e$ are of interest and thus those are considered for numerical evaluation in section 6.2.

All in all, the Yukawa sector does not behave as desired: unrealistic CKM matrix, FCNC contribution is too significant, the H_2 field interacts with a really low strength with the third family. The scalar potential is softly broken and therefore additional singlets could be introduced to the Yukawa sector. Additional Yukawa singlets may fix some of the issues. We propose that the following singlets² could be added to (4.3.4):

$$\mu_6^d: \ \bar{Q}_{3L}^0 h_1 d_{3R}^0, \tag{4.3.45a}$$

?

²In case of $w_1 = 0$ and $w_2 = 0$, to generate realistic fermion masses the y_8 coupling could be added.

$$y_7^d: \ \bar{Q}_{3L}^0 h_2 d_{3R}^0,$$
 (4.3.45b)

$$y_8^d: \ Q_{2L}^0 h_S d_{2R}^0, \tag{4.3.45c}$$

so that the decomposition (4.3.8) results in:

$$\mathcal{M}_{1} \approx \begin{pmatrix} 0 & y_{2} & y_{4} \\ y_{2} & 0 & 0 \\ y_{5} & 0 & y_{6} \end{pmatrix}, \qquad (4.3.46a)$$
$$\mathcal{M}_{2} \approx \begin{pmatrix} y_{2} & 0 & 0 \\ 0 & -y_{2} & y_{4} \\ 0 & y_{5} & y_{7} \end{pmatrix}, \qquad (4.3.46b)$$

$$\mathcal{M}_{S} \approx \begin{pmatrix} y_{1} & 0 & 0 \\ 0 & y_{1} + y_{8} & 0 \\ 0 & 0 & y_{3} \end{pmatrix}.$$
 (4.3.46c)

Even a single additional Yukawa coupling, $y_7 = y_6$, should result in a more realistic CKM matrix. This is due to the fact that now there is an additional free parameter y_6 , but the CKM matrix depends on both y_6^u and y_6^d . Additional parameters may also enable control over the FCNC. A full analysis of the broken Yukawa sector is beyond the scope of the thesis.

 $u_7 I$

 u_5

Scalar-Scalar Interactions 4.4

The Feynman rules are obtained by expanding the S_3 -symmetric scalar potential with respect to the mass-eigenstates and multiplying the relevant terms by -iS, where S is the symmetry factor defined in the same way as in section 4.2. We consider that the SU(2) doublets in terms of the mass-eigenstates are given by eq. (4.1.23).

For simplicity, we express the rotation angle of the S_3 singlet in terms of the overall phase $\gamma = \sigma/2$, which we got from the diagonalization procedure of the S₃ singlet neutral sector (4.1.20). We note that the scalar-scalar couplings are written down in a slightly different way. The couplings $g(H_iH_jH_k)$ and $g(H_iH_jH_kH_l)$ are presented with the symmetry factor S. Therefore, the Feynman rules are defined as:

$$H_i H_j H_k H_l = -ig \left(H_i H_j H_k H_l \right).$$
(4.4.1)

The trilinear couplings involving the same species are:

$$g(H_1H_1H_1) = 3v\left[(2\lambda_1 - \lambda_2 + \lambda_3)c_\alpha + (\lambda_2 + \lambda_3)c_{3\alpha - 2\sigma}\right], \qquad (4.4.2a)$$

$$g(H_3H_3H_3) = -3v\left[(2\lambda_1 - \lambda_2 + \lambda_3)s_\alpha - (\lambda_2 + \lambda_3)s_{3\alpha - 2\sigma}\right].$$
(4.4.2b)

The trilinear couplings involving only the neutral fields of the S_3 doublet are:

$$g(H_1H_1H_3) = -v\left[(2\lambda_1 - \lambda_2 + \lambda_3)s_\alpha + 3(\lambda_2 + \lambda_3)s_{3\alpha - 2\sigma}\right], \qquad (4.4.3a)$$

$$g(H_1H_2H_2) = v\left[(2\lambda_1 + \lambda_2 + 3\lambda_3)c_{\alpha} - (\lambda_2 + \lambda_3)c_{\alpha-2\sigma}\right], \qquad (4.4.3b)$$

$$g(H_1H_3H_3) = v\left[\left(2\lambda_1 - \lambda_2 + \lambda_3\right)c_\alpha - 3\left(\lambda_2 + \lambda_3\right)c_{3\alpha - 2\sigma}\right],\tag{4.4.3c}$$

$$g(H_2H_2H_3) = -v[(2\lambda_1 + \lambda_2 + 3\lambda_3)s_{\alpha} - (\lambda_2 + \lambda_3)s_{\alpha-2\sigma}].$$
(4.4.3d)

The trilinear couplings involving both neutral fields of the S_3 doublet and singlet are:

$$g(H_1S_1S_1) = v\left[(\lambda_5 + \lambda_6)c_\alpha + 2\lambda_7c_{\alpha-\sigma}\right], \qquad (4.4.4a)$$

$$g(H_1S_2S_2) = v\left[\left(\lambda_5 + \lambda_6\right)c_\alpha - 2\lambda_7c_{\alpha-\sigma}\right], \qquad (4.4.4b)$$

$$g(H_2S_1S_2) = -2v\lambda_7 \mathbf{s}_\sigma, \tag{4.4.4c}$$

$$g(H_3S_1S_1) = -v\left[\left(\lambda_5 + \lambda_6\right)\mathbf{s}_{\alpha} + 2\lambda_7\mathbf{s}_{\alpha-\sigma}\right], \qquad (4.4.4d)$$

$$g(H_3S_2S_2) = -v\left[(\lambda_5 + \lambda_6)s_\alpha - 2\lambda_7s_{\alpha-\sigma}\right].$$
(4.4.4e)

The trilinear couplings involving the charged fields are:

$$g\left(H_1H^{\pm}H^{\mp}\right) = v\left[\left(2\lambda_1 + \lambda_2 - \lambda_3\right)c_{\alpha} - \left(\lambda_2 + \lambda_3\right)c_{\alpha-2\sigma}\right],\tag{4.4.5a}$$

$$g\left(H_{1}S^{+}S^{+}\right) = v\lambda_{5}c_{\alpha}, \tag{4.4.5b}$$

$$g\left(H_{3}H^{\pm}H^{\pm}\right) = -v\left[\left(2\lambda_{1} + \lambda_{2} - \lambda_{3}\right)\mathbf{s}_{\alpha} - \left(\lambda_{2} + \lambda_{3}\right)\mathbf{s}_{\alpha-2\sigma}\right], \qquad (4.4.5c)$$

$$a\left(H_{2}S^{\pm}S^{\mp}\right) = -v\lambda_{7}\mathbf{s}. \qquad (4.4.5d)$$

$$g\left(\Pi_{35}^{\pm}S^{\pm}\right) = -i\lambda_{5}s_{\alpha}, \qquad (4.4.5d)$$

$$g\left(S_{1}H^{+}S^{+}\right) = \mp ie^{-2} v \lambda_{7} s_{\sigma}, \qquad (4.4.5e)$$

$$g\left(S_2H^{\pm}S^{+}\right) = -e^{\pm \frac{1}{2}}v\lambda_7 s_{\sigma}.$$
(4.4.5f)

The quartic couplings involving the same species are:

$$g(H_1H_1H_1H_1) = g(H_3H_3H_3H_3) = 3[2\lambda_1 - \lambda_2 + \lambda_3 + (\lambda_2 + \lambda_3)c_{4\alpha - 2\sigma}], \qquad (4.4.6a)$$

$$g(H_1H_1H_1H_1) = 6(\lambda_1 - \lambda_2c_2^2 + \lambda_3c_2^2) \qquad (4.4.6b)$$

$$g(H_2H_2H_2H_2) = 6(\lambda_1 - \lambda_2 s_{\sigma}^2 + \lambda_3 c_{\sigma}^2), \qquad (4.4.6b)$$

$$a(S_1 S_2 S_1 S_2) = a(S_2 S_2 S_2 S_2) = 6\lambda_2, \qquad (4.4.6c)$$

$$g(S_1S_1S_1S_1) = g(S_2S_2S_2S_2) = 6\lambda_8,$$
(4.4.0c)
$$(H^{\pm}H^{\pm}H^{\pm}H^{\pm}) = 4(\lambda_1 - \lambda_2^2 + \lambda_3^2)$$
(4.4.6d)

$$g\left(H^{\perp}H^{+}H^{+}H^{+}\right) = 4\left(\lambda_{1} - \lambda_{2}s_{\sigma}^{2} + \lambda_{3}c_{\sigma}^{2}\right), \qquad (4.4.6d)$$

$$g\left(S^{\pm}S^{\mp}S^{\pm}S^{\mp}\right) = 4\lambda_8. \tag{4.4.6e}$$

The quartic couplings involving only the neutral fields of the S_3 doublet are:

$$g(H_1H_1H_1H_3) = -3(\lambda_2 + \lambda_3)s_{4\alpha - 2\sigma}, \qquad (4.4.7a)$$

$$g(H_1H_1H_2H_2) = 2[\lambda_1 + \lambda_3 - (\lambda_2 + \lambda_3)s_{2\alpha - \sigma}s_{\sigma}], \qquad (4.4.7b)$$

$$g(H_1H_1H_3H_3) = 2\lambda_1 - \lambda_2 + \lambda_3 - 3(\lambda_2 + \lambda_3)c_{4\alpha - 2\sigma}, \qquad (4.4.7c)$$

$$g(H_1H_2H_2H_3) = -2(\lambda_2 + \lambda_3)c_{2\alpha - \sigma}s_{\sigma}, \qquad (4.4.7d)$$

$$g(H_1H_3H_3H_3) = 3(\lambda_2 + \lambda_3)s_{4\alpha - 2\sigma}, \qquad (4.4.7e)$$

$$g(H_2H_2H_3H_3) = 2[\lambda_1 + \lambda_3 + (\lambda_2 + \lambda_3)s_{2\alpha - \sigma}s_{\sigma}].$$
(4.4.7f)

The quartic couplings involving only the neutral fields of the S_3 singlet are:

$$g(S_1 S_1 S_2 S_2) = 2\lambda_8. \tag{4.4.8}$$

The quartic couplings involving both neutral fields of the S_3 doublet and singlet are:

$$g(H_1H_1S_1S_1) = g(H_3H_3S_2S_2) = \lambda_5 + \lambda_6 + 2\lambda_7c_{2\alpha-\sigma}, \qquad (4.4.9a)$$

$$g(H_1H_1S_2S_2) = g(H_3H_3S_1S_1) = \lambda_5 + \lambda_6 - 2\lambda_7c_{2\alpha-\sigma}, \qquad (4.4.9b)$$

$$g(H_1H_2S_1S_2) = 2\lambda_7 s_{\alpha-\sigma}, \qquad (4.4.9c)$$

$$g(H_1H_3S_1S_1) = -2\lambda_7 s_{2\alpha-\sigma},$$
 (4.4.9d)

$$g(H_1H_3S_2S_2) = 2\lambda_7 s_{2\alpha-\sigma},$$
 (4.4.9e)

$$g(H_2H_2S_1S_1) = \lambda_5 + \lambda_6 + 2\lambda_7 c_\sigma, \qquad (4.4.9f)$$

$$g(H_2H_2S_2S_2) = \lambda_5 + \lambda_6 - 2\lambda_7c_\sigma, \qquad (4.4.9g)$$

$$g\left(H_2H_3S_1S_2\right) = 2\lambda_7 c_{\alpha-\sigma}.\tag{4.4.9h}$$

The quartic couplings involving either the neutral fields of the S_3 doublet or singlet along with either charged fields of the S_3 doublet or singlet are:

$$g(H_1 H_1 H^{\pm} H^{\mp}) = 2\lambda_1 + (\lambda_2 - \lambda_3) c_{2\alpha} - (\lambda_2 + \lambda_3) c_{2(\alpha - \sigma)}, \qquad (4.4.10a)$$

$$g(H_1H_1S^{\pm}S^{\mp}) = g(H_2H_2S^{\pm}S^{\mp}) = g(H_3H_3S^{\pm}S^{\mp}) = \lambda_5,$$
 (4.4.10b)

$$g\left(H_1H_3H^{\pm}H^{\mp}\right) = -\left(\lambda_2 - \lambda_3\right)s_{2\alpha} + \left(\lambda_2 + \lambda_3\right)s_{2(\alpha-\sigma)},\tag{4.4.10c}$$

$$g\left(H_2H_2H^{\pm}H^{\mp}\right) = 2\left(\lambda_1 - \lambda_2 s_{\sigma}^2 + \lambda_3 c_{\sigma}^2\right), \qquad (4.4.10d)$$

$$g\left(H_3H_3H^{\pm}H^{\mp}\right) = 2\lambda_1 - (\lambda_2 - \lambda_3)\operatorname{c}_{2\alpha} + (\lambda_2 + \lambda_3)\operatorname{c}_{2(\alpha - \sigma)},\tag{4.4.10e}$$

$$g\left(S_{1}S_{1}H^{\pm}H^{\mp}\right) = g\left(S_{2}S_{2}H^{\pm}H^{\mp}\right) = \lambda_{5}, \qquad (4.4.10f)$$

$$g(S_1 S_1 S^{\pm} S^{\mp}) = g(S_2 S_2 S^{\pm} S^{\mp}) = 2\lambda_8.$$
 (4.4.10g)

The quartic couplings involving both neutral fields of the S_3 doublet and singlet along with a pair of the charged fields of the S_3 doublet and singlet are:

$$g\left(H_1S_1H^{\pm}S^{\mp}\right) = \mp \frac{i}{2}e^{\pm \frac{i\sigma}{2}}\left(\lambda_6 \mathbf{s}_{\alpha} - 2\lambda_7 \mathbf{s}_{\alpha-\sigma}\right), \qquad (4.4.11a)$$

$$g\left(H_1 S_2 H^{\pm} S^{\mp}\right) = \frac{1}{2} e^{\pm \frac{i\sigma}{2}} \left(\lambda_6 \mathbf{s}_{\alpha} + 2\lambda_7 \mathbf{s}_{\alpha-\sigma}\right), \qquad (4.4.11b)$$

$$g\left(H_2S_1H^{\pm}S^{\mp}\right) = \frac{1}{2}e^{\pm\frac{i\sigma}{2}}\left(\lambda_6 + 2\lambda_7c_{\sigma}\right), \qquad (4.4.11c)$$

$$g\left(H_2 S_2 H^{\pm} S^{\mp}\right) = \pm \frac{i}{2} e^{\pm \frac{i\sigma}{2}} \left(\lambda_6 - 2\lambda_7 c_{\sigma}\right), \qquad (4.4.11d)$$

$$g\left(H_3S_1H^{\pm}S^{\mp}\right) = \mp \frac{i}{2}e^{\pm \frac{i\sigma}{2}}\left(\lambda_6c_{\alpha} - 2\lambda_7c_{\alpha-\sigma}\right), \qquad (4.4.11e)$$

$$g\left(H_3S_2H^{\pm}S^{\mp}\right) = \frac{1}{2}e^{\pm\frac{i\sigma}{2}}\left(\lambda_6c_{\alpha} + 2\lambda_7c_{\alpha-\sigma}\right).$$
(4.4.11f)

The quartic couplings involving only the charged fields are:

$$g\left(H^{\pm}H^{\mp}S^{\pm}S^{\mp}\right) = \lambda_5 + \lambda_6, \qquad (4.4.12a)$$

$$g\left(H^{\pm}H^{\pm}S^{\mp}S^{\mp}\right) = 4e^{\pm\imath\sigma}\lambda_7 c_{\sigma}.$$
(4.4.12b)

The trilinear couplings involving only the neutral fields and the would-be Goldstone boson are:

$$g\left(G^{0}G^{0}H_{1}\right) = v\left[\left(\lambda_{2} + \lambda_{3}\right)c_{\alpha-2\sigma} + \left(2\lambda_{1} - \lambda_{2} + \lambda_{3}\right)c_{\alpha}\right], \qquad (4.4.13a)$$

$$g\left(G^{0}G^{0}H_{3}\right) = -v\left[\left(\lambda_{2} + \lambda_{3}\right)s_{\alpha-2\sigma} + \left(2\lambda_{1} - \lambda_{2} + \lambda_{3}\right)s_{\alpha}\right], \qquad (4.4.13b)$$

$$g\left(G^{0}H_{1}H_{2}\right) = 2\left(\lambda_{2} + \lambda_{3}\right)vc_{\sigma}s_{\alpha-\sigma},\tag{4.4.13c}$$

$$g\left(G^{0}H_{2}H_{3}\right) = 2\left(\lambda_{2} + \lambda_{3}\right)vc_{\sigma}c_{\alpha-\sigma},\tag{4.4.13d}$$

$$g\left(G^0S_1S_2\right) = 2\lambda_7 v c_{\sigma}.\tag{4.4.13e}$$

The trilinear couplings involving the charged fields and the would-be Goldstone boson are:

$$g\left(H_1G^{\mp}G^{\pm}\right) = v\left[(\lambda_2 + \lambda_3)c_{\alpha-2\sigma} + (2\lambda_1 - \lambda_2 + \lambda_3)c_{\alpha}\right], \qquad (4.4.14a)$$

$$g\left(H_1G^+H^{\perp}\right) = \pm iv\left[\left(\lambda_2 + \lambda_3\right)s_{\alpha-2\sigma} + \left(\lambda_2 - \lambda_3\right)s_{\alpha}\right], \qquad (4.4.14b)$$

$$g\left(H_2G^{\mp}H^{\pm}\right) = 2\lambda_3 v, \qquad (4.4.14c)$$

$$g\left(H_3G^{\mp}G^{\pm}\right) = -v\left[(\lambda_2 + \lambda_3)s_{\alpha-2\sigma} + (2\lambda_1 - \lambda_2 + \lambda_3)s_{\alpha}\right], \qquad (4.4.14d)$$

$$g(H_3G^+G^{\pm}) = -v[(\lambda_2 + \lambda_3)s_{\alpha - 2\sigma} + (2\lambda_1 - \lambda_2 + \lambda_3)s_{\alpha}], \qquad (4.4.14d)$$
$$g(H_3G^{\mp}H^{\pm}) = \pm iv[(\lambda_2 + \lambda_3)c_{\alpha - 2\sigma} + (\lambda_2 - \lambda_3)c_{\alpha}], \qquad (4.4.14e)$$

$$g\left(H_3G^+H^+\right) = \pm iv\left[\left(\lambda_2 + \lambda_3\right)c_{\alpha-2\sigma} + \left(\lambda_2 - \lambda_3\right)c_{\alpha}\right],\tag{4.4.14e}$$

$$g\left(S_1G^{\mp}S^{\pm}\right) = \frac{1}{2}e^{\mp \frac{i\sigma}{2}}v\left(\lambda_6 + 2\lambda_7 c_{\sigma}\right), \qquad (4.4.14f)$$

$$g\left(S_2G^{\mp}S^{\pm}\right) = \mp \frac{1}{2}ie^{\mp \frac{i\sigma}{2}}v\left(\lambda_6 - 2c_{\sigma}\lambda_7\right).$$
(4.4.14g)

The quartic couplings involving neutral states with at least one would-be Goldstone boson are:

$$g(G^{0}G^{0}G^{0}G^{0}) = 3[2\lambda_{1} - \lambda_{2} + \lambda_{3} + (\lambda_{2} + \lambda_{3})c_{2\sigma}], \qquad (4.4.15a)$$

$$g(G^{0}G^{0}G^{0}H_{2}) = -3(\lambda_{2} + \lambda_{3})s_{2\sigma}, \qquad (4.4.15b)$$

$$g\left(G^{0}G^{0}H_{1}H_{1}\right) = 2\left[\lambda_{1} + \lambda_{3} + (\lambda_{2} + \lambda_{3})s_{\sigma}s_{2\alpha-\sigma}\right], \qquad (4.4.15c)$$

$$g\left(G^{0}G^{0}H_{1}H_{3}\right) = 2\left(\lambda_{2} + \lambda_{3}\right)s_{\sigma}c_{2\alpha-\sigma},\tag{4.4.15d}$$

$$g(G^{0}G^{0}H_{2}H_{2}) = 2\lambda_{1} - \lambda_{2} + \lambda_{3} - 3(\lambda_{2} + \lambda_{3})c_{2\sigma}, \qquad (4.4.15e)$$

$$g(G^{0}G^{0}H_{3}H_{3}) = 2[\lambda_{1} + \lambda_{3} - (\lambda_{2} + \lambda_{3})s_{\sigma}s_{2\alpha-\sigma}], \qquad (4.4.15f)$$

$$g\left(G^{0}H_{1}H_{1}H_{2}\right) = 2\left(\lambda_{2} + \lambda_{3}\right)c_{\sigma}s_{2\alpha-\sigma},\tag{4.4.15g}$$

$$g(G^{0}H_{1}H_{2}H_{3}) = 2(\lambda_{2} + \lambda_{3})c_{\sigma}c_{2\alpha-\sigma}, \qquad (4.4.15h)$$

$$g(G^0H_2H_2H_2) = 3(\lambda_2 + \lambda_3)s_{2\sigma},$$
 (4.4.15i)

$$g\left(G^{0}H_{2}H_{3}H_{3}\right) = -2\left(\lambda_{2} + \lambda_{3}\right)c_{\sigma}\mathbf{s}_{2\alpha-\sigma},\tag{4.4.15j}$$

$$g\left(G^0 G^0 S_1 S_1\right) = \lambda_5 + \lambda_6 - 2\lambda_7 c_\sigma, \qquad (4.4.15k)$$

$$g(G^{0}G^{0}S_{1}S_{1}) = \lambda_{5} + \lambda_{6} + 2\lambda_{7}c_{\sigma}, \qquad (4.4.151)$$

$$g(G^{0}H, S, S) = 2\lambda_{5}c_{\sigma}, \qquad (4.4.15m)$$

$$g\left(G^{0}H_{1}S_{1}S_{2}\right) = 2\lambda_{7}c_{\alpha-\sigma}, \qquad (4.4.15m)$$

$$g(G^{0}H_{2}S_{1}S_{1}) = 2\lambda_{7}s_{\sigma}, \qquad (4.4.15n)$$

$$g(G^{0}H_{2}S_{2}S_{2}) = -2\lambda_{7}s_{\sigma}, \qquad (4.4.150)$$

$$g(G^{0}H_{2}S_{2}S_{2}) = -2\lambda_{7}s_{\sigma}, \qquad (4.4.150)$$

$$g\left(G^{\circ}H_{3}S_{1}S_{2}\right) = -2\lambda_{7}s_{\alpha-\sigma}.$$
(4.4.15p)

The quartic couplings involving the charged fields and the neutral fields along with the would-be Goldstone boson are:

$g\left(G^{0}G^{0}G^{\mp}G^{\pm}\right) = 2\lambda_{1} - \lambda_{2} + \lambda_{3} + (\lambda_{2} + \lambda_{3})c_{2\sigma},$	(4.4.16a)
$g\left(G^0G^0G^{\mp}H^{\pm}\right) = \mp i\left(\lambda_2 + \lambda_3\right)s_{2\sigma},$	(4.4.16b)
$g\left(G^0G^0H^{\mp}H^{\pm}\right) = 2\lambda_1 + \lambda_2 - \lambda_3 - (\lambda_2 + \lambda_3)c_{2\sigma},$	(4.4.16c)
$g\left(G^0G^0S^{\mp}S^{\pm}\right) = \lambda_5,$	(4.4.16d)
$g\left(G^0H_1G^{\mp}H^{\pm}\right) = 2\lambda_3 \mathbf{s}_{\alpha},$	(4.4.16e)
$g\left(G^0H_2G^{\mp}G^{\pm}\right) = -\left(\lambda_2 + \lambda_3\right)s_{2\sigma},$	(4.4.16f)
$g\left(G^{0}H_{2}G^{\mp}H^{\pm}\right) = \mp i\left[\lambda_{2} - \lambda_{3} + \left(\lambda_{2} + \lambda_{3}\right)c_{2\sigma}\right],$	(4.4.16g)
$g\left(G^0H_2H^{\mp}H^{\pm}\right) = (\lambda_2 + \lambda_3)\mathbf{s}_{2\sigma},$	(4.4.16h)
$g\left(G^0H_3G^{\mp}H^{\pm}\right) = 2\lambda_3 c_{lpha},$	(4.4.16i)
$g\left(H_2H_2G^{\mp}G^{\pm}\right) = 2\lambda_1 + \lambda_2 - \lambda_3 - (\lambda_2 + \lambda_3) c_{2\sigma},$	(4.4.16j)
$g\left(H_2H_2G^{\mp}H^{\pm}\right) = \pm i\left(\lambda_2 + \lambda_3\right)s_{2\sigma},$	(4.4.16k)
$g\left(H_1H_2G^{\mp}H^{\pm}\right) = 2\lambda_3 c_{\alpha},$	(4.4.16l)
$g\left(H_2H_3G^{\mp}H^{\pm}\right) = -2\lambda_3 \mathbf{s}_{\alpha},$	(4.4.16m)
$g\left(H_1H_1G^{\mp}G^{\pm}\right) = 2\lambda_1 + \left(\lambda_2 + \lambda_3\right)c_{2(\alpha-\sigma)} - \left(\lambda_2 - \lambda_3\right)c_{2\alpha},$	(4.4.16n)
$g\left(H_1H_1G^{\mp}H^{\pm}\right) = \pm i\left[\left(\lambda_2 + \lambda_3\right)s_{2(\alpha-\sigma)} + \left(\lambda_2 - \lambda_3\right)s_{2\alpha}\right],$	(4.4.160)
$g\left(H_1H_3G^{\mp}G^{\pm}\right) = -\left(\lambda_2 + \lambda_3\right)\mathbf{s}_{2(\alpha-\sigma)} + \left(\lambda_2 - \lambda_3\right)\mathbf{s}_{2\alpha},$	(4.4.16p)
$g\left(H_1H_3G^{\mp}H^{\pm}\right) = \pm i\left[\left(\lambda_2 + \lambda_3\right)c_{2(\alpha-\sigma)} + \left(\lambda_2 - \lambda_3\right)c_{2\alpha}\right],$	(4.4.16q)
$g\left(H_3H_3G^{\mp}G^{\pm}\right) = 2\lambda_1 - \left(\lambda_2 + \lambda_3\right)c_{2(\alpha-\sigma)} + \left(\lambda_2 - \lambda_3\right)c_{2\alpha},$	(4.4.16r)
$g\left(H_3H_3G^{\mp}H^{\pm}\right) = \mp i\left[\left(\lambda_2 - \lambda_3\right)s_{2\alpha} + \left(\lambda_2 + \lambda_3\right)s_{2(\alpha-\sigma)}\right],$	(4.4.16s)
$g\left(S_1S_1G^{\mp}G^{\pm}\right) = g\left(S_2S_2G^{\mp}G^{\pm}\right) = \lambda_5,$	(4.4.16t)

and

$$g(G^{0}S_{1}G^{\mp}S^{\pm}) = \pm \frac{1}{2}ie^{\pm \frac{i\sigma}{2}} (\lambda_{6} - 2\lambda_{7}c_{\sigma}), \qquad (4.4.17a)$$

$$g\left(G^{0}S_{2}G^{\mp}S^{\pm}\right) = \frac{1}{2}e^{\pm\frac{i\sigma}{2}}\left(\lambda_{6} + 2\lambda_{7}c_{\sigma}\right), \qquad (4.4.17b)$$

$$g\left(G^{0}S_{1}H^{\mp}S^{\pm}\right) = \lambda_{7}e^{\mp\frac{i\sigma}{2}}s_{\sigma}, \qquad (4.4.17c)$$

$$g\left(G^{0}S_{2}H^{\mp}S^{\pm}\right) = g\left(H_{2}S_{1}G^{\mp}S^{\pm}\right) = \pm i\lambda_{7}e^{\mp\frac{i\sigma}{2}}\mathbf{s}_{\sigma}, \qquad (4.4.17d)$$

$$g\left(H_2 S_2 G^{\mp} S^{\pm}\right) = -\lambda_7 e^{\pm \frac{\omega}{2}} \mathbf{s}_{\sigma}, \qquad (4.4.17e)$$

(, , , , ,)

$$g\left(H_1S_1G^{\mp}S^{\pm}\right) = \frac{1}{2}e^{\mp\frac{i\sigma}{2}}\left(\lambda_6c_{\alpha} + 2\lambda_7c_{\alpha-\sigma}\right), \qquad (4.4.17f)$$

$$g\left(H_1 S_2 G^{\mp} S^{\pm}\right) = \mp \frac{1}{2} i e^{\mp \frac{i\sigma}{2}} \left(\lambda_6 c_{\alpha} - 2\lambda_7 c_{\alpha-\sigma}\right), \qquad (4.4.17g)$$

$$g\left(H_3S_1G^{\mp}S^{\pm}\right) = -\frac{1}{2}e^{\mp\frac{i\sigma}{2}}\left(\lambda_6 \mathbf{s}_{\alpha} + 2\lambda_7 \mathbf{s}_{\alpha-\sigma}\right), \qquad (4.4.17h)$$

$$g\left(H_3S_2G^{\mp}S^{\pm}\right) = \pm \frac{1}{2}ie^{\mp \frac{i\sigma}{2}}\left(\lambda_6 \mathbf{s}_{\alpha} - 2\lambda_7 \mathbf{s}_{\alpha-\sigma}\right).$$
(4.4.17i)

The quartic couplings involving only the charged fields and the would-be Goldstone boson are:

$$g\left(G^{\mp}G^{\pm}G^{\mp}G^{\pm}\right) = 2\left[2\lambda_1 - \lambda_2 + \lambda_3 + (\lambda_2 + \lambda_3)c_{2\sigma}\right], \qquad (4.4.18a)$$

$$g\left(G^{\mp}G^{\pm}G^{\mp}H^{\pm}\right) = \mp 2i\left(\lambda_2 + \lambda_3\right)s_{2\sigma}, \qquad (4.4.18b)$$

$$g\left(G^{\mp}H^{\pm}G^{\mp}H^{\pm}\right) = 4\left(\lambda_2 + \lambda_3\right)c_{\sigma}^2,\tag{4.4.18c}$$

$$g\left(G^{\mp}S^{\pm}G^{\mp}S^{\pm}\right) = 4\lambda_7 e^{\mp i\sigma} c_{\sigma}, \qquad (4.4.18d)$$

$$g\left(G^{\mp}G^{\pm}H^{\mp}G^{\pm}\right) = \pm 2i\left(\lambda_2 + \lambda_3\right)s_{2\sigma},\tag{4.4.18e}$$

$$g\left(G^{\mp}G^{\pm}H^{\mp}H^{\pm}\right) = 2\left[\lambda_{1} + \lambda_{3} - (\lambda_{2} + \lambda_{3})c_{2\sigma}\right], \qquad (4.4.18f)$$

$$g(G^{+}H^{+}H^{+}H^{+}) = \pm 2i(\lambda_{2} + \lambda_{3})s_{2\sigma},$$

$$(4.4.18g)$$

$$= (G^{\mp}S^{\pm}H^{\mp}S^{\pm}) = \pm 4i\lambda e^{\pm i\sigma}r,$$

$$(4.4.18b)$$

$$g\left(G^{\mp}S^{\pm}H^{\mp}S^{\pm}\right) = \pm 4i\lambda_7 e^{\mp i\sigma}\mathbf{s}_{\sigma},\tag{4.4.18h}$$

$$g\left(G^{\mp}G^{\pm}S^{\mp}S^{\pm}\right) = \lambda_5 + \lambda_6. \tag{4.4.18i}$$

From the trilinear couplings involving the same species states it follows that the states H_1 and H_3 are CP-even. From the neutral trilinear couplings we conclude that the H_2 state is CP-odd and the states S_1 and S_2 have an opposite CP quantum numbers. This is in agreement with the extracted information from the scalar-gauge bosons couplings, see section 4.2.

4.5 Constraints

4.5.1 Constraints From the Scalar Masses

We consider constraints from the scalar masses. First of all, we assume that all of the scalars masses squared are positive and non-zero. Secondly, we assume that the H_1 field corresponds to the recently discovered Higgs boson with $m_H = 125$ GeV.

Taking into consideration the first assumption, from the charged sector we get that:

$$\lambda_2 > 0, \tag{4.5.1a}$$

$$\lambda_5 > -\frac{2\mu_0^2}{v^2}.$$
 (4.5.1b)

From the neutral sector of the S_3 doublet it follows that:

$$\lambda_3 > -\min(\lambda_1, \lambda_2), \qquad (4.5.2a)$$

$$\lambda_1 + \lambda_3 > \Delta. \tag{4.5.2b}$$

The determinant of the mass-squared matrix $\mathcal{M}^2_{\text{Neutral}-12}$ of eq. (4.1.9) should be positive definite:

$$\det\left(\mathcal{M}_{\text{Neutral}-12}^{2}\right) = 8\left(\lambda_{1} - \lambda_{2}\right)\left(\lambda_{2} + \lambda_{3}\right)^{2}s_{\sigma}^{2}v^{6} > 0.$$

$$(4.5.3)$$

Therefore it follows that:

$$\lambda_1 > \lambda_2. \tag{4.5.4}$$

Taking into consideration the neutral masses of the S_3 singlet we get the following constraints:

$$\frac{2\mu_0^2}{v^2} + \lambda_5 + \lambda_6 > 0,$$

$$\frac{2\mu_0^2}{v^2} + \lambda_5 + \lambda_6 - |\lambda_7 c_{2\sigma}| > 0.$$
(4.5.5)

4.5.2 The Standard Model Limit

In the SM, the Higgs boson and gauge boson Feynman rules are given by [57]:

$$hW^{\pm}_{\mu}W^{\mp}_{\nu} = igm_W g_{\mu\nu},$$
 (4.5.6a)

$$hZ_{\mu}Z_{\nu} = i\frac{g}{c_w}m_Z g_{\mu\nu}, \qquad (4.5.6b)$$

$$hhW^{\pm}_{\mu}W^{\mp}_{\nu} = \frac{i}{2}g^2 g_{\mu\nu}, \qquad (4.5.6c)$$

$$hhZ_{\mu}Z_{\nu} = \frac{i}{2} \frac{g^2}{c_w^2} g_{\mu\nu},$$
 (4.5.6d)

and the Higgs boson and fermion Feynman rules are:

$$hf\bar{f} = -i\frac{m_f}{v}.\tag{4.5.7}$$

Assuming that the SM-like Higgs boson is H_1 , in the SM limit we find that $c_{\alpha} = 1$ and therefore $\alpha = 0$. The rotation angle α was defined in eq. (4.1.13). It follows that alongside $\alpha = 0$ another condition should be satisfied:

$$(\lambda_2 + \lambda_3) \mathbf{s}_{2\sigma} = 0, \tag{4.5.8}$$

where $\lambda_2 + \lambda_3 = 0$ leads to the exact C-III-c³ case as in this limit we get $\nu^2 = 0$. The other option leads to $\sigma = 0$ or $\sigma = \frac{1}{2}\pi$. The $\sigma = 0$ constraint results in a real vacuum configuration. Moreover, solving for the minimization conditions we get that $\nu^2 = 0$ and this is exactly the R-II-3 vacuum configuration. In the case of $\sigma = \frac{1}{2}\pi$ we get that $\nu^2 = 0$. Therefore we conclude that there is no exact SM limit for the C-III-c- ν^2 vacuum configuration. Nevertheless, we assume that the SM limit is achieved by $c_{\alpha} \geq 0.9$.

In the SM, the Higgs-Higgs boson H trilinear and quartic Feynman rules are:

$$HHH = -3i\frac{m_h^2}{v},$$

$$HHHH = -3i\frac{m_h^2}{v^2}.$$
(4.5.9)

The C-III-c- ν^2 model results in $g(H_1H_1H_1)$ (4.4.2a) and $g(H_1H_1H_1H_1)$ (4.4.6a). In principle, there is some freedom and we do not compare these terms against the SM couplings. For an insight of the non 3HDM see Refs. [58–60].

4.5.3 Potential Stability

The scalar potential needs to be stable, see Refs. [16, 61]. This implies that the scalar potential of eq. (2.1.7) should be positive in all space directions for asymptotically large values of fields, *i.e.*, for $|h_1|$, and $|h_2|$, and $|h_S|$ approaching infinity. This is the most basic constraint as it forces the existence of a stable minimum.

Necessary potential stability conditions were presented in Ref. [37]:

$$\lambda_1 > 0, \tag{4.5.10a}$$

$$\lambda_8 > 0, \tag{4.5.10b}$$

$$\lambda_1 + \lambda_3 > 0, \tag{4.5.10c}$$

$$2\lambda_1 + (\lambda_3 - \lambda_2) > |\lambda_2 + \lambda_3|, \qquad (4.5.10d)$$

$$\lambda_5 + 2\sqrt{\lambda_8 \left(\lambda_1 + \lambda_3\right)} > 0, \tag{4.5.10e}$$

$$\lambda_5 + \lambda_6 + 2\sqrt{\lambda_8} \left(\lambda_1 + \lambda_3\right) > 2|\lambda_7|, \qquad (4.5.10f)$$

$$\lambda_1 + \lambda_3 + \lambda_5 + \lambda_6 + 2\lambda_7 + \lambda_8 > 2|\lambda_4|. \tag{4.5.10g}$$

³We remind that the exact C-III-c vacuum configurations results in only one massive neutral scalar.

In Ref. [34], following the approach of Refs. [62, 63], it was shown that although for the most general scalar potential stability conditions are quite involved, for vacuum configurations with the $\lambda_4 = 0$ constraint there exists an explicit direction in the space of the scalar potential. Another method, in terms of bilinears, was discussed in Ref. [64].

We present the general formulation of Ref. [34]. The SU(2) doublets can be re-expressed as follows:

$$h_i = ||h_i||\tilde{h}_i, \text{ for } i = \{1, 2, S\},$$
(4.5.11)

where $||h_i||$ is the norm of the spinor, and \hat{h}_i is a unit spinor. Assuming that the two different systems have the same origin, the norms can be parameterized in terms of relations between the Cartesian and spherical coordinates:

$$||h_1|| = rc_{\gamma}s_{\theta}, \tag{4.5.12a}$$

$$||h_2|| = r \mathbf{s}_\gamma \mathbf{s}_\theta, \tag{4.5.12b}$$

$$||h_S|| = rc_\theta, \tag{4.5.12c}$$

where $r \in [0, \infty)$, $\gamma \in [0, \pi/2]$, $\theta \in [0, \pi/2]$. The SU(2) invariant products are:

$$\hat{h}_2^{\dagger} \hat{h}_1 = \rho_3 e^{i\theta_3}, \tag{4.5.13a}$$

$$\hat{h}_{S}^{\dagger}\hat{h}_{2} = \rho_{1}e^{i\theta_{1}}, \qquad (4.5.13b)$$

$$\hat{h}_{1}^{\dagger}\hat{h}_{S} = \rho_{2}e^{i\theta_{2}},\tag{4.5.13c}$$

$$\hat{h}_i^{\dagger} \hat{h}_j = \left(\hat{h}_j^{\dagger} \hat{h}_i \right)^*, \qquad (4.5.13d)$$

where $\theta_i \in [0, 2\pi)$ and $\rho_i \in [0, 1]$ due to the fact that solutions lie within a unit sphere.

For asymptotically large field values the main contribution comes from the quartic terms V_4 of the scalar potential. Thus we consider only the relevant quartic couplings. In this case the potential stability condition is given by:

$$V_4 = r^4 \sum_{i=1}^{8} \lambda_i A_i \ge 0, \quad \forall \ \{\rho_i, \ \theta_i, \ \gamma, \ \theta\},$$
(4.5.14)

where

$$A_1 = \mathbf{s}_{\theta}^4, \tag{4.5.15a}$$

$$A_2 = -\rho_3^2 \mathbf{s}_{\theta_3}^2 \mathbf{s}_{2\gamma}^2 \mathbf{s}_{\theta}^4, \tag{4.5.15b}$$

$$A_3 = \left(c_{2\gamma}^2 + \rho_3^2 c_{\theta_3}^2 s_{2\gamma}^2\right) s_{\theta}^4, \tag{4.5.15c}$$

$$A_{4} = 2 \left(\rho_{1} c_{\theta_{1}} c_{2\gamma} + 2\rho_{2} \rho_{3} c_{\theta_{2}} c_{\theta_{3}} c_{\gamma}^{2} \right) s_{\gamma} c_{\theta} s_{\theta}^{3}, \qquad (4.5.15d)$$

$$A_5 = \frac{1}{4} s_{2\theta}^2, \tag{4.5.15e}$$

$$A_6 = \frac{1}{4} \left(\rho_1^2 \mathbf{s}_{\gamma}^2 + \rho_2^2 \mathbf{c}_{\gamma}^2 \right) \mathbf{s}_{2\theta}^2, \tag{4.5.15f}$$

$$A_7 = \frac{1}{2} \left(\rho_1^2 c_{2\theta_1} s_{\gamma}^2 + \rho_2^2 c_{2\theta_2} c_{\gamma}^2 \right) s_{2\theta}^2, \qquad (4.5.15g)$$

$$A_8 = c_{\theta}^4.$$
 (4.5.15h)

Necessary and sufficient conditions, provided that the $\lambda_4 = 0$ constraint is applied, are [49]:

$$\lambda_1 > 0, \tag{4.5.16a}$$

$$\lambda_8 > 0, \tag{4.5.16b}$$

$$\lambda_1 - \lambda_2 > 0, \tag{4.5.16c}$$

 $\lambda_1 + \lambda_3 > 0, \tag{4.5.16d}$

$$\lambda_5 + \min[0, \lambda_6 - 2|\lambda_7|] > -2\sqrt{\lambda_1\lambda_8},$$
(4.5.16e)
$$\lambda_{5} + \min[0, \lambda_{6} - 2|\lambda_{7}|] > -2\sqrt{(\lambda_{1} - \lambda_{2})\lambda_{8}}, \qquad (4.5.16f)$$

$$\lambda_5 + \min[0, \lambda_6 - 2|\lambda_7|] > -2\sqrt{(\lambda_1 + \lambda_3)}\,\lambda_8.$$
(4.5.16g)

In the C-III-c- ν^2 model eqs. (4.5.16a, 4.5.16c, 4.5.16d) are by default satisfied due to the masssquared parameters. Eq. (4.5.16b) puts a lower bound on the free coupling λ_8 of the model. The left side of terms (4.5.16e-4.5.16g) is equivalent and thus only the lowest value of the square root should be considered as it results in the most severe constraint. It turns out that only eq. (4.5.16f) can be considered. For simplicity, it can be split into:

$$\lambda_5 > -2\sqrt{(\lambda_1 - \lambda_2)\,\lambda_8},\tag{4.5.17a}$$

$$\lambda_5 + \lambda_6 - 2|\lambda_7| > -2\sqrt{(\lambda_1 - \lambda_2)}\,\lambda_8. \tag{4.5.17b}$$

4.5.4 Perturbativity

The soft perturbativity limit is given by directly imposing constraints on the quartic couplings:

$$|\lambda_i| \le \lambda_{\max}.\tag{4.5.18}$$

The most conservative choice is to set $\lambda_{\text{max}} = 4\pi$. A smaller value $\lambda_{\text{max}} = 2\pi$ was adopted in Ref. [65]. We adopt a more widely used convention of $\lambda_{\text{max}} = 4\pi$.

For the C-III-c- ν^2 , the number of checks of (4.5.18) can be reduced. We discuss relations between the couplings and masses in section 4.1.1. We suppose that the heaviest states are $m_{\xi} = 1$ TeV. First of all, from $|\lambda_2| \leq 4\pi$ it follows that $m_{H^{\pm}} \leq 1234.36$ GeV. Next, the $|\lambda_5| \leq 4\pi$ is fixed by μ_0^2 :

$$\mu_0^2 \in \left[-2\pi v^2 + m_{S^{\pm}}^2, \, 2\pi v^2 + m_{S^{\pm}}^2\right]. \tag{4.5.19}$$

The constraint $|\lambda_3| \leq 4\pi$ can also be neglected as it requires mass splitting of order 1200 GeV, which in our case makes little sense. From $|\lambda_1| \leq 4\pi$ it follows that:

$$m_{H^{\pm}}^2 < |-m_{H_2}^2 + m_{H_3}^2 + m_{H^{\pm}}^2| \le (1228.01 \,\text{GeV})^2,$$
 (4.5.20)

and the lower boundary follows from the fact that $m_{H_2} \leq m_{H_3}$. From $|\lambda_6| \leq 4\pi$ one can derive:

$$m_{S^{\pm}} \in \left[\operatorname{Max} \left[\operatorname{Re} \left(\sqrt{\frac{-4\pi v^2 + m_{S_1}^2 + m_{S_2}^2}{2}} \right), 100 \right], \operatorname{Min} \left[\operatorname{Re} \left(\sqrt{\frac{4\pi v^2 + m_{S_1}^2 + m_{S_2}^2}{2}} \right), 1000 \right] \right],$$

$$(4.5.21)$$

in GeV units.

The only constraint to be checked is $|\lambda_7| \leq 4\pi$, which turns out to be way too involved to be analytically checked.

We also take into consideration a more severe perturbativity limit in terms of limiting the overall strength of the quartic scalar-scalar interactions⁴:

$$|g\left(\varphi_i\varphi_j\varphi_k\varphi_l\right)| \le 4\pi,\tag{4.5.22}$$

where the quartic couplings were presented in section 4.4. One of the most obvious limits comes from the coupling $g(S_1S_1S_1S_1) = 6\pi$ (4.4.6c) by directly considering the perturbativity limit (4.5.22), $\lambda_8 \leq \frac{2}{3}\pi$. Some of the couplings depend on a single λ_5 , e.g., $g(H_1H_1S^{\pm}S^{\mp}) = \lambda_5$ (4.4.10b), and such checks simplify to (4.5.18). The other simple relation is $g(H^{\pm}H^{\mp}S^{\pm}S^{\mp}) = \lambda_5 + \lambda_6$ (4.4.12a). Most of the quartic scalar-scalar couplings depend on trigonometric functions and thus should be numerically checked if those satisfy the perturbativity condition (4.5.22).

⁴Interactions involving the would-be Goldstone bosons are not considered.

4.5.5 Tree-Level Unitarity

The two-body scattering processes involving longitudinal bosons and the Higgs boson in the SM were pioneered by Lee, Quigg and Thacker [66] and further on analysed in Refs. [67, 68]. The unitarity constraints of the 2HDM are well-known and different methods can be found in Refs. [69, 70]. The general idea behind the unitarity bound is that the Born amplitude for elastic longitudinal vector-boson scattering may not result in a higher than unity amplitude.

The process of finding the unitarity limit is straightforward due to the Goldstone equivalence theorem, which relates the longitudinally polarized vector boson and the Goldstone bosons in the high-energy limit [71, 72]. It is sufficient to take a look at the $2 \rightarrow 2$ scattering processes of the gauge-eigenstates. The number of 2-body states is given by a binomial:

$$m = \binom{n+1}{2} = \frac{1}{2}n(1+n), \qquad (4.5.23)$$

where n is the number of scalar degrees of freedom. The most general case results in a matrix of dimension $\dim(m)$.

Due to computational complexity of the eigenvalues of the most general scattering matrix it is worth a try to find a basis in which the scattering matrix S is block-diagonal. This approach makes sense as not all of the 2-body scattering processes are possible as those are restricted by the S_3 -symmetric potential and by discrete symmetries like CP or \mathbb{Z}_2 .

The eigenvalues of the block-diagonal matrix S is a list of eigenvalues of each sub-block diagonal matrix:

$$\det (S - \lambda \mathcal{I}) = \det (S_1 - \lambda \mathcal{I}) \times \dots \times \det (S_n - \lambda \mathcal{I}).$$
(4.5.24)

The electric charge should be conserved in the scattering processes and thus it is straightforward to split the scattering matrix S based on the total charge. The neutral scattering matrix is denoted by S^0 , and the singly charged scattering matrix is denoted by S^+ , and the doubly charged scattering matrix is given by S^{++} . Therefore the form of the scattering matrix including the channels based on the electrical charge is as follows:

$$S = \text{diag}\left(S^0, S^+, S^{++}\right). \tag{4.5.25}$$

We start with the neutral channel. The neutral scattering matrix can be expressed as:

$$S^{0} = \operatorname{diag}\left(S_{1}^{0}, S_{2}^{0}, S_{3}^{0}, S_{4}^{0}, S_{5}^{0}\right).$$

$$(4.5.26)$$

Each of the matrices S_i^n is obtained by taking a look at the states $\langle \Psi_i^n | \Psi_i^n \rangle$, where the two-particles states Ψ_i^n are given by:

$$\Psi_1^0 = \left\{ \left| h_1^+ h_2^- \right\rangle, \left| h_1^- h_2^+ \right\rangle, \left| \tilde{\eta}_1 \tilde{\eta}_2 \right\rangle, \left| \tilde{\chi}_1 \tilde{\chi}_2 \right\rangle, \left| \tilde{\eta}_1 \tilde{\chi}_2 \right\rangle, \left| \tilde{\eta}_2 \tilde{\chi}_1 \right\rangle \right\},$$
(4.5.27a)

$$\Psi_2^0 = \left\{ \left| h_1^+ h_S^- \right\rangle, \left| h_1^- h_S^+ \right\rangle, \left| \tilde{\eta}_1 \tilde{\eta}_S \right\rangle, \left| \tilde{\chi}_1 \tilde{\chi}_S \right\rangle, \left| \tilde{\eta}_1 \tilde{\chi}_S \right\rangle, \left| \tilde{\eta}_S \tilde{\chi}_1 \right\rangle \right\},$$
(4.5.27b)

$$\Psi_3^0 = \{ |\tilde{\eta}_1 \tilde{\chi}_1 \rangle, |\tilde{\eta}_2 \tilde{\chi}_2 \rangle, |\tilde{\eta}_S \tilde{\chi}_S \rangle \}, \qquad (4.5.27c)$$

$$\Psi_4^0 = \left\{ \left| h_2^+ h_S^- \right\rangle, \left| h_2^- h_S^+ \right\rangle, \left| \tilde{\eta}_2 \tilde{\eta}_S \right\rangle, \left| \tilde{\chi}_2 \tilde{\chi}_S \right\rangle, \left| \tilde{\eta}_2 \tilde{\chi}_S \right\rangle, \left| \tilde{\eta}_S \tilde{\chi}_2 \right\rangle \right\},$$
(4.5.27d)

$$\Psi_{5}^{0} = \left\{ \left| h_{1}^{+} h_{1}^{-} \right\rangle, \left| h_{2}^{-} h_{2}^{+} \right\rangle, \left| h_{S}^{-} h_{S}^{+} \right\rangle, \frac{1}{\sqrt{2}} \left| \tilde{\eta}_{1} \tilde{\eta}_{1} \right\rangle, \frac{1}{\sqrt{2}} \left| \tilde{\eta}_{2} \tilde{\eta}_{2} \right\rangle, \\ \frac{1}{\sqrt{2}} \left| \tilde{\eta}_{S} \tilde{\eta}_{S} \right\rangle, \frac{1}{\sqrt{2}} \left| \tilde{\chi}_{1} \tilde{\chi}_{1} \right\rangle, \frac{1}{\sqrt{2}} \left| \tilde{\chi}_{2} \tilde{\chi}_{2} \right\rangle, \frac{1}{\sqrt{2}} \left| \tilde{\chi}_{S} \tilde{\chi}_{S} \right\rangle \right\},$$

$$(4.5.27e)$$

where the factor of $\frac{1}{\sqrt{2}}$ is due to the Bose-Einstein statistics. States are organized so that each of the separate sets $\langle \Psi_i^n | \Psi_i^n \rangle$ is a block-diagonal component. The elements of the neutral scattering

matrix are:

$$S_{1}^{0} = \begin{pmatrix} 2(\lambda_{1} - \lambda_{2}) & 4(\lambda_{2} + \lambda_{3}) & 2\lambda_{3} & 2\lambda_{3} & -2i\lambda_{2} & 2i\lambda_{2} \\ 4(\lambda_{2} + \lambda_{3}) & 2(\lambda_{1} - \lambda_{2}) & 2\lambda_{3} & 2\lambda_{3} & 2i\lambda_{2} & -2i\lambda_{2} \\ 2\lambda_{3} & 2\lambda_{3} & 2(\lambda_{1} + \lambda_{3}) & 2(\lambda_{2} + \lambda_{3}) & 0 & 0 \\ 2\lambda_{3} & 2\lambda_{3} & 2(\lambda_{2} + \lambda_{3}) & 2(\lambda_{1} + \lambda_{3}) & 0 & 0 \\ 2i\lambda_{2} & -2i\lambda_{2} & 0 & 0 & 2(\lambda_{1} - 2\lambda_{2} - \lambda_{3}) & 2(\lambda_{2} + \lambda_{3}) \\ -2i\lambda_{2} & 2i\lambda_{2} & 0 & 0 & 2(\lambda_{2} + \lambda_{3}) & 2(\lambda_{1} - 2\lambda_{2} - \lambda_{3}) \end{pmatrix}, \qquad (4.5.28a)$$

$$S_{2}^{0} = \begin{pmatrix} \lambda_{5} + \lambda_{6} & 4\lambda_{7} & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & -\frac{i}{2}(\lambda_{6} - 2\lambda_{7}) \\ \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & \lambda_{5} + \lambda_{6} + 2\lambda_{7} & 0 & 0 \\ \frac{1}{2}(\lambda_{6} - 2\lambda_{7}) & \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & 0 & 0 & \lambda_{5} + \lambda_{6} - 2\lambda_{7} & 0 \\ \frac{1}{2}(\lambda_{6} - 2\lambda_{7}) & \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & 0 & 0 & \lambda_{5} + \lambda_{6} - 2\lambda_{7} \\ \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & -\frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & 0 & 0 & \lambda_{5} + \lambda_{6} - 2\lambda_{7} \\ \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & -\frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & 0 & 0 & \lambda_{5} + \lambda_{6} - 2\lambda_{7} \\ \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & -\frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & 0 & 0 & \lambda_{5} + \lambda_{6} - 2\lambda_{7} \\ 2\lambda_{7} & 2\lambda_{7} & 2\lambda_{8} \end{pmatrix}, \qquad (4.5.28b)$$

$$S_5^0 = \begin{pmatrix} S_{511}^0 & S_{512}^0 \\ S_{512}^{0 \text{ T}} & S_{522}^- \end{pmatrix}$$
(4.5.28e)

where

$$S_{5_{11}}^{0} = \begin{pmatrix} 4(\lambda_{1} + \lambda_{3}) & 2(\lambda_{1} - \lambda_{2}) & \lambda_{5} + \lambda_{6} & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \sqrt{2}(\lambda_{1} - \lambda_{3}) \\ 2(\lambda_{1} - \lambda_{2}) & 4(\lambda_{1} + \lambda_{3}) & \lambda_{5} + \lambda_{6} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) \\ \lambda_{5} + \lambda_{6} & \lambda_{5} + \lambda_{6} & 4\lambda_{8} & \frac{\lambda_{5}}{\sqrt{2}} & \frac{\lambda_{5}}{\sqrt{2}} \\ \sqrt{2}(\lambda_{1} + \lambda_{3}) & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} & 3(\lambda_{1} + \lambda_{3}) & \lambda_{1} + \lambda_{3} \\ \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} & \lambda_{1} + \lambda_{3} & 3(\lambda_{1} + \lambda_{3}) \end{pmatrix}, \qquad (4.5.29a)$$

$$S_{5_{12}}^{0} = \begin{pmatrix} \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} & \lambda_{1} + \lambda_{3} & 3(\lambda_{1} + \lambda_{3}) \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \sqrt{2}(\lambda_{1} - \lambda_{3}) & \sqrt{2}(\lambda_{1} + \lambda_{3}) & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \lambda_{5} + \lambda_{6} + 2\lambda_{7} \end{pmatrix} & \lambda_{1} + \lambda_{3} & \frac{\lambda_{5}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \lambda_{5} + \lambda_{6} + 2\lambda_{7} \end{pmatrix} & \lambda_{1} + \lambda_{3} & \frac{\lambda_{1}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \lambda_{6} + 2\lambda_{7} \end{pmatrix} & \lambda_{1} + \lambda_{3} & \frac{\lambda_{1}}{\sqrt{2}} \\ \frac{\lambda_{5}}{\sqrt{2}} & \lambda_{6} + 2\lambda_{7} \end{pmatrix} & \lambda_{1} + \lambda_{3} & \frac{\lambda_{1}}{\sqrt{2}} \\ \lambda_{8} & \frac{1}{2}(\lambda_{5} + \lambda_{6} + 2\lambda_{7}) & \frac{1}{2}(\lambda_{5} + \lambda_{6} + 2\lambda_{7}) & \lambda_{8} \end{pmatrix} \end{pmatrix}$$

$$(4.5.29c)$$

The singly charged scattering matrix is:

$$S^{+} = \operatorname{diag}\left(S_{1}^{+}, S_{2}^{+}, S_{3}^{+}, S_{4}^{+}\right).$$
(4.5.30)

The singly charged two-particles states are:

$$\Psi_{1}^{+} = \left\{ \left| h_{1}^{+} \tilde{\eta}_{1} \right\rangle, \left| h_{2}^{+} \tilde{\eta}_{2} \right\rangle, \left| h_{S}^{+} \tilde{\eta}_{S} \right\rangle, \left| h_{1}^{+} \tilde{\chi}_{1} \right\rangle, \left| h_{2}^{+} \tilde{\chi}_{2} \right\rangle, \left| h_{S}^{+} \tilde{\chi}_{S} \right\rangle \right\},$$

$$\Psi_{1}^{+} = \left\{ \left| h_{1}^{+} \tilde{\eta}_{2} \right\rangle, \left| h_{2}^{+} \tilde{\eta}_{2} \right\rangle, \left| h_{S}^{+} \tilde{\chi}_{2} \right\rangle, \left| h_{S}^{+} \tilde{\chi}_{S} \right\rangle \right\},$$

$$(4.5.31a)$$

$$\Psi_{1}^{+} = \left\{ \left| h_{1}^{+} \tilde{\eta}_{2} \right\rangle, \left| h_{2}^{+} \tilde{\eta}_{2} \right\rangle, \left| h_{S}^{+} \tilde{\chi}_{2} \right\rangle, \left| h_{S}^{+} \tilde{\chi}_{S} \right\rangle \right\},$$

$$(4.5.31a)$$

$$\Psi_2^+ = \left\{ \left| h_1^+ \tilde{\eta}_2 \right\rangle, \left| h_2^+ \tilde{\eta}_1 \right\rangle, \left| h_1^+ \tilde{\chi}_2 \right\rangle, \left| h_2^+ \tilde{\chi}_1 \right\rangle \right\}, \tag{4.5.31b}$$

$$\Psi_2^+ = \left\{ \left| h_1^+ \tilde{\eta}_2 \right\rangle, \left| h_2^+ \tilde{\eta}_1 \right\rangle, \left| h_1^+ \tilde{\chi}_2 \right\rangle, \left| h_2^+ \tilde{\chi}_1 \right\rangle \right\} \tag{4.5.31c}$$

$$\Psi_{3}^{+} = \left\{ \left| h_{1}^{+} \tilde{\eta}_{S} \right\rangle, \left| h_{S}^{+} \tilde{\eta}_{1} \right\rangle, \left| h_{1}^{+} \tilde{\chi}_{S} \right\rangle, \left| h_{S}^{+} \tilde{\chi}_{1} \right\rangle \right\},$$

$$\Psi_{1}^{+} = \left\{ \left| h_{1}^{+} \tilde{\eta}_{S} \right\rangle, \left| h_{S}^{+} \tilde{\chi}_{S} \right\rangle, \left| h_{S}^{+} \tilde{\chi}_{2} \right\rangle \right\}$$

$$(4.5.31c)$$

$$(4.5.31d)$$

$$\Psi_4^+ = \left\{ \left| h_2^+ \tilde{\eta}_S \right\rangle, \left| h_S^+ \tilde{\eta}_2 \right\rangle, \left| h_2^+ \tilde{\chi}_S \right\rangle, \left| h_S^+ \tilde{\chi}_2 \right\rangle \right\}.$$
(4.5.31d)

The sub-matrices of the singly charged scattering matrix are:

$$S_{1}^{+} = \begin{pmatrix} 2(\lambda_{1} + \lambda_{3}) & 2\lambda_{3} & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & 0 & -2i\lambda_{2} & \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) \\ 2\lambda_{3} & 2(\lambda_{1} + \lambda_{3}) & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & -2i\lambda_{2} & 0 & \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) \\ \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & 2\lambda_{8} & \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & \frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & 0 \\ 0 & 2i\lambda_{2} & -\frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & 2(\lambda_{1} + \lambda_{3}) & 2\lambda_{3} & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) \\ 2i\lambda_{2} & 0 & -\frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & 2\lambda_{3} & 2(\lambda_{1} + \lambda_{3}) & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) \\ -\frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & -\frac{i}{2}(\lambda_{6} - 2\lambda_{7}) & 0 & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & \frac{1}{2}(\lambda_{6} + 2\lambda_{7}) & 2\lambda_{8} \end{pmatrix},$$

$$(4.5.32a)$$

$$S_{2}^{+} = \begin{pmatrix} 2(\lambda_{1} - \lambda_{3}) & 2\lambda_{3} & 0 & 2i\lambda_{2} \\ 2\lambda_{3} & 2(\lambda_{1} - \lambda_{3}) & 2i\lambda_{2} & 0 \\ 0 & -2i\lambda_{2} & 2(\lambda_{1} - \lambda_{3}) & 2\lambda_{3} \\ -2i\lambda_{2} & 0 & 2\lambda_{3} & 2(\lambda_{1} - \lambda_{3}) \end{pmatrix},$$
(4.5.32b)

$$S_{3}^{+} = \begin{pmatrix} \lambda_{5} & \frac{1}{2} (\lambda_{6} + 2\lambda_{7}) & 0 & -\frac{i}{2} (\lambda_{6} - 2\lambda_{7}) \\ \frac{1}{2} (\lambda_{6} + 2\lambda_{7}) & \lambda_{5} & -\frac{i}{2} (\lambda_{6} - 2\lambda_{7}) & 0 \\ 0 & \frac{i}{2} (\lambda_{6} - 2\lambda_{7}) & \lambda_{5} & \frac{1}{2} (\lambda_{6} + 2\lambda_{7}) \\ i (\lambda_{6} - 2\lambda_{7}) & \lambda_{5} & \frac{1}{2} (\lambda_{6} + 2\lambda_{7}) \end{pmatrix},$$
(4.5.32c)

The doubly-charged two-particle states are given by:

$$\Psi^{++} = \left\{ \frac{1}{\sqrt{2}} \left| h_1^+ h_1^+ \right\rangle, \, \frac{1}{\sqrt{2}} \left| h_2^+ h_2^+ \right\rangle, \, \frac{1}{\sqrt{2}} \left| h_S^+ h_S^+ \right\rangle \right\}.$$
(4.5.33)

The doubly-charged scattering matrix is:

$$S^{++} = \begin{pmatrix} 2(\lambda_1 + \lambda_3) & 2(\lambda_2 + \lambda_3) & 2\lambda_7 \\ 2(\lambda_2 + \lambda_3) & 2(\lambda_1 + \lambda_3) & 2\lambda_7 \\ 2\lambda_7 & 2\lambda_7 & 2\lambda_8 \end{pmatrix}.$$
 (4.5.34)

After solving for eigenvalues of the scattering matrix S we find that in total there are 18 particular eigenvalues:

$$\Lambda_1 = 2\left(\lambda_1 \pm \lambda_2\right),\tag{4.5.35a}$$

$$\Lambda_2 = \lambda_5 \pm \lambda_6, \tag{4.5.35b}$$

$$\Lambda_3 = \lambda_5 \pm 2\lambda_7, \tag{4.5.35c}$$

$$\Lambda_4 = 2\left(\lambda_1 \pm \lambda_2 - 2\lambda_3\right), \tag{4.5.35d}$$

$$\Lambda_5 = 2\left(\lambda_1 + \lambda_2 + 4\lambda_3\right), \tag{4.5.35e}$$

$$\Lambda_6 = 2\left(\lambda_1 - 5\lambda_2 - 2\lambda_3\right),\tag{4.5.351}$$

$$\Lambda_7 = \lambda_5 + 2\lambda_6 \pm 6\lambda_7, \tag{4.5.35g}$$

$$\Lambda_8 = \lambda_1 - \lambda_2 + 2\lambda_3 + \lambda_8 \pm \sqrt{2\lambda_6^2 + (\lambda_1 - \lambda_2 + 2\lambda_3 - \lambda_8)^2}, \qquad (4.5.35h)$$

$$\Lambda_9 = \lambda_1 + \lambda_2 + 2\lambda_3 + \lambda_8 \pm \sqrt{8\lambda_7^2 + (\lambda_1 + \lambda_2 + 2\lambda_3 - \lambda_8)^2}, \tag{4.5.35i}$$

$$\Lambda_{10} = 5\lambda_1 - \lambda_2 + 2\lambda_3 + 3\lambda_8 \pm \sqrt{2(2\lambda_5 + \lambda_6)^2 + (5\lambda_1 - \lambda_2 + 2\lambda_3 - 3\lambda_8)^2}.$$
 (4.5.35j)

We confirm that the eigenvalues we got are in perfect agreement with those of Ref. [37] in the limit of $\lambda_4 \rightarrow 0$.

In the high-energy limit, the partial wave amplitude takes the simple form:

$$|a_j| \le 1.$$
 (4.5.36)

In Ref. [67] it was suggested that a stronger constraint may be applied based on the reality of the partial-wave amplitude and the Cauchy–Schwarz inequality:

$$|\operatorname{Re}(a_j)| \le \frac{1}{2}.\tag{4.5.37}$$

In principle, for the NHDM models, the unitarity constraint simplifies to a check:

$$|\Lambda_i| \le 16\pi,\tag{4.5.38}$$

where the factor of 16π comes from the Jacob-Wick expansion [73]. The $|\Lambda_i| \leq 8\pi$ corresponds to eq. (4.5.37).

The number of checks (4.5.38) can be reduced considering the soft perturbativity limit (4.5.18), $|\lambda_{\max}| = 4\pi$. It follows that if a specific Λ_i of eq. (4.5.35) depends on less than four λ_i , such eigenvalues can be neglected, namely eqs. (4.5.35a - 4.5.35c).

4.5.6 Electroweak Oblique Parameters

The electroweak oblique parameters are parametrised by the self-energy functions S, and T, and U [74,75]. These parameters are defined in a way that they vanish in the SM. In terms of the extended scalar models these parameters limit how far the electroweak sector can be extended from the SM reference point.

Experimental constraints for the reference values $m_{h_{\text{SM,ref}}} = 125 \text{ GeV}$ and $m_{t,\text{ref}} = 172.5 \text{ GeV}$ were presented by the Gfitter group [76, 77]:

$$S = 0.04 \pm 0.11, \tag{4.5.39a}$$

$$T = 0.09 \pm 0.14, \tag{4.5.39b}$$

$$U = -0.02 \pm 0.11. \tag{4.5.39c}$$

The guideline on how to derive the electroweak-oblique parameters for the NHDM was presented in [78,79]. The SU(2) doublets in the Higgs basis are given by:

$$H_{1} = \begin{pmatrix} G^{+} \\ \frac{1}{\sqrt{2}} \left(v + \frac{1}{\sqrt{2}} \left(\tilde{\eta}_{1} + \tilde{\eta}_{2} \right) + iG_{0} \end{pmatrix} \end{pmatrix}, \qquad (4.5.40a)$$

$$H_2 = \begin{pmatrix} H^+ \\ \frac{1}{\sqrt{2}} \left(\varphi_2 + \frac{i}{\sqrt{2}} \left(-\tilde{\chi}_1 + \tilde{\chi}_2 \right) \right) \end{pmatrix}, \qquad (4.5.40b)$$

$$H_3 = \begin{pmatrix} S^+ \\ \frac{1}{\sqrt{2}} \left(\tilde{\eta}_S + i \tilde{\chi}_S \right) \end{pmatrix}.$$
(4.5.40c)

The V and U rotation matrices of Ref. [78] correspond to⁵:

$$\begin{pmatrix} \frac{1}{\sqrt{2}} (\tilde{\eta}_{1} + \tilde{\eta}_{2}) + iG_{0} \\ \varphi_{2} + \frac{i}{\sqrt{2}} (-\tilde{\chi}_{1} + \tilde{\chi}_{2}) \\ \tilde{\eta}_{S} + i\tilde{\eta}_{S} \end{pmatrix} = V \begin{pmatrix} G_{0} \\ \varphi_{1} \\ \varphi_{2} \\ \varphi_{3} \\ S_{1} \\ S_{2} \end{pmatrix}, \qquad (4.5.41)$$

and

$$\begin{pmatrix} G^+ \\ H^+ \\ S^+ \end{pmatrix} = U \begin{pmatrix} G^+ \\ H^+ \\ S^+ \end{pmatrix}.$$
(4.5.42)

⁵Note that the $\varphi_i = H_i$ in the generic basis, see eq. (4.1.24).

Therefore U is an identity matrix $U\equiv \mathcal{I}_3$ and V is:

$$V = \begin{pmatrix} i & c_{\alpha} & 0 & -s_{\alpha} & 0 & 0\\ 0 & is_{\alpha} & 1 & ic_{\alpha} & 0 & 0\\ 0 & 0 & 0 & e^{i\gamma} & ie^{i\gamma} \end{pmatrix},$$
(4.5.43)

see eq. (4.1.17).

Calculations result in the following values:

$$\begin{split} S &= \frac{m_W^2 s_w^2}{24 \alpha \pi^2 v^2} \bigg[\left(1 - 2 s_w^2 \right)^2 G\left(m_{H^\pm}^2, m_{H^\pm}^2, m_Z^2 \right) + \left(1 - 2 s_w^2 \right)^2 G\left(m_{S^\pm}^2, m_{S^\pm}^2, m_Z^2 \right) \\ &\quad + s_\alpha^2 G\left(m_{H_1}^2, m_{H_2}^2, m_Z^2 \right) + c_\alpha^2 G\left(m_{H_2}^2, m_{H_3}^2, m_Z^2 \right) + G\left(m_{S_1}^2, m_{S_2}^2, m_Z^2 \right) \\ &\quad - 2 \ln\left(m_{H^\pm}^2 \right) - 2 \ln\left(m_{S^\pm}^2 \right) + \ln\left(m_{H_2}^2 \right) + \ln\left(m_{H_3}^2 \right) + \ln\left(m_{S_1}^2 \right) + \ln\left(m_{S_2}^2 \right) \\ &\quad - s_\alpha^2 \left(\hat{G}\left(m_{H_1}^2, m_Z^2 \right) - \hat{G}\left(m_{H_3}^2, m_Z^2 \right) \right) \bigg], \end{split}$$

$$T &= \frac{1}{16 \alpha \pi^2 v^2} \bigg[s_\alpha^2 F\left(m_{H^\pm}^2, m_{H_1}^2 \right) + F\left(m_{H^\pm}^2, m_{H_2}^2 \right) + c_\alpha^2 F\left(m_{H^\pm}^2, m_{H_3}^2 \right) + F\left(m_{S^\pm}^2, m_{S_1}^2 \right) \\ &\quad + F\left(m_{S^\pm}^2, m_{S_2}^2 \right) - s_\alpha^2 F\left(m_{H_1}^2, m_{H_2}^2 \right) - c_\alpha^2 F\left(m_{H_2}^2, m_{H_3}^2 \right) - F\left(m_{S_1}^2, m_{S_2}^2 \right) \\ &\quad + 3 s_\alpha^2 \left(F\left(m_W^2, m_{H_1}^2 \right) - F\left(m_Z^2, m_{H_1}^2 \right) + F\left(m_Z^2, m_{H_3}^2 \right) - F\left(m_W^2, m_{H_3}^2 \right) \right) \bigg], \end{aligned}$$

$$U &= \frac{m_W^2 s_w^2}{24 \alpha \pi^2 v^2} \bigg[s_\alpha^2 G\left(m_{H^\pm}^2, m_{H_1}^2, m_W^2 \right) + G\left(m_{H^\pm}^2, m_{H_2}^2, m_W^2 \right) - \left(1 - 2 s_w^2 \right)^2 G\left(m_{H^\pm}^2, m_{H_3}^2, m_Z^2 \right) \\ &\quad - \left(1 - 2 s_w^2 \right)^2 G\left(m_{S^\pm}^2, m_{S^\pm}^2, m_Z^2 \right) - s_\alpha^2 G\left(m_{H_1}^2, m_{H_2}^2, m_Z^2 \right) - c_\alpha^2 G\left(m_{H^\pm}^2, m_{H_3}^2, m_Z^2 \right) \\ &\quad - \left(1 - 2 s_w^2 \right)^2 G\left(m_{S^\pm}^2, m_{S^\pm}^2, m_Z^2 \right) - s_\alpha^2 G\left(m_{H_1}^2, m_{H_2}^2, m_Z^2 \right) - c_\alpha^2 G\left(m_{H_2}^2, m_{H_3}^2, m_Z^2 \right) \\ &\quad - G\left(m_{S_1}^2, m_{S_2}^2, m_Z^2 \right) + s_\alpha^2 \left(\hat{G}\left(m_{H_1}^2, m_Z^2 \right) - \hat{G}\left(m_{H_1}^2, m_W^2 \right) \right) \\ \\ &\quad + s_\alpha^2 \left(\hat{G}\left(m_{H_3}^2, m_W^2 \right) - \hat{G}\left(m_{H_3}^2, m_Z^2 \right) \right) \bigg].$$

$$(4.5.44c)$$

The functions, which appear in the expressions above, are given by the well-known function [80]:

$$F(I,J) \equiv \begin{cases} \frac{I+J}{2} - \frac{IJ}{I-J} \ln \frac{I}{J} & , I \neq J \\ 0 & , I = J \end{cases},$$
(4.5.45)

and [78, 79]:

$$G(I, J, Q) \equiv -\frac{16}{3} + \frac{5(I+J)}{Q} - \frac{2(I-J)^2}{Q^2}$$
(4.5.46a)

$$+\frac{3}{Q}\left[\frac{I^2+J^2}{I-J}-\frac{I^2-J^2}{Q}+\frac{(I-J)^3}{3Q^2}\right]\ln\frac{I}{J}+\frac{r}{Q^3}f(t,r),$$
(4.5.46b)

$$\tilde{G}(I,J,Q) \equiv -2 + \left(\frac{I-J}{Q} - \frac{I+J}{I-J}\right) \ln \frac{I}{J} + \frac{f(t,r)}{Q}, \qquad (4.5.46c)$$

$$\hat{G}(I,Q) \equiv G(I,Q,Q) + 12\tilde{G}(I,Q,Q),$$
(4.5.46d)

where

$$f(t,r) \equiv \begin{cases} \sqrt{r} \ln \left| \frac{t - \sqrt{r}}{t + \sqrt{r}} \right| & ,r > 0\\ 0 & ,r = 0 ,\\ 2\sqrt{-r} \arctan \left(\frac{\sqrt{-r}}{t} \right) & ,r < 0 \end{cases}$$
(4.5.47)

where

$$t \equiv I + J - Q, \tag{4.5.48a}$$

$$r \equiv Q^2 - 2Q(I+J) + (I-J)^2.$$
(4.5.48b)

Chapter 5 The R-II-1a Model

In this chapter we consider the R-II-1a model. The vacuum configuration is given by¹:

$$\{0, w, w_S\},\tag{5.0.1}$$

and the minimization conditions are:

$$\mu_0^2 = \frac{1}{2}\lambda_4 \frac{w_2^3}{w_S} - \frac{1}{2} \left(\lambda_5 + \lambda_6 + 2\lambda_7\right) w_2^2 - \lambda_8 w_S^2, \tag{5.0.2a}$$

$$\mu_1^2 = -(\lambda_1 + \lambda_3) w_2^2 + \frac{3}{2} \lambda_4 w_2 w_S - \frac{1}{2} (\lambda_5 + \lambda_6 + 2\lambda_7) w_S^2.$$
(5.0.2b)

An interesting property of this model, as mentioned in section 2.4, is that the \mathbb{Z}_2 symmetry is preserved for:

$$h_1 \to -h_1, \quad h_2 \to h_2, \quad h_S \to h_S.$$
 (5.0.3)

Thus the DM candidate resides in the inert SU(2) doublet h_1 , $\langle h_1 \rangle = 0$. This model is, by default, CP conserving.

5.1 The Mass-Squared Matrices

The charged mass-squared matrix is given by:

$$\mathcal{M}_{\text{Charged}}^{2} = \begin{pmatrix} (\mathcal{M}_{\text{Ch}}^{2})_{11} & 0 & 0\\ 0 & (\mathcal{M}_{\text{Ch}}^{2})_{22} & (\mathcal{M}_{\text{Ch}}^{2})_{23}\\ 0 & (\mathcal{M}_{\text{Ch}}^{2})_{23} & (\mathcal{M}_{\text{Ch}}^{2})_{33} \end{pmatrix},$$
(5.1.1)

where

$$(\mathcal{M}_{\rm Ch}^2)_{11} = -2\lambda_3 w_2^2 + \frac{5}{2}\lambda_4 w_2 w_S - \frac{1}{2}(\lambda_6 + 2\lambda_7) w_S^2, \qquad (5.1.2a)$$

$$(\mathcal{M}_{\rm Ch}^2)_{22} = \frac{1}{2} w_S \left[\lambda_4 w_2 - (\lambda_6 + 2\lambda_7) w_S \right], \qquad (5.1.2b)$$

$$(\mathcal{M}_{\rm Ch}^2)_{23} = -\frac{1}{2} w_2 \left[\lambda_4 w_2 - (\lambda_6 + 2\lambda_7) w_S \right], \qquad (5.1.2c)$$

$$(\mathcal{M}_{\rm Ch}^2)_{33} = \frac{1}{2} \frac{w_2^2}{w_S} \left[\lambda_4 w_2 - (\lambda_6 + 2\lambda_7) w_S \right].$$
(5.1.2d)

The lower-right components of the mass-squared matrix are diagonalizable by a rotation matrix:

$$\mathcal{R}_{\beta} = \begin{pmatrix} c_{\beta} & s_{\beta} \\ -s_{\beta} & c_{\beta} \end{pmatrix}, \qquad (5.1.3)$$

¹We note that R-II-1a is a real vacuum configuration and thus VEVs are given by the absolute values. The hatted VEVs \hat{w}_i are no longer used.

where the rotation angle is:

$$\mathbf{t}_{\beta} = \frac{w_S}{w_2} = \frac{\mathbf{s}_{\beta}v}{\mathbf{c}_{\beta}v}.$$
 (5.1.4)

By convention $0 < \beta < \frac{\pi}{2}$. This rotation can be identified as the Higgs basis rotation. The charged scalar states can be expressed as:

$$h^{\pm} = h_1^{\pm}, \tag{5.1.5a}$$

$$G^{\pm} = c_{\beta}h_2^{\pm} + s_{\beta}h_S^{\pm}, \qquad (5.1.5b)$$

$$H^{\pm} = -s_{\beta}h_2^{\pm} + c_{\beta}h_S^{\pm}, \qquad (5.1.5c)$$

with masses:

$$m_{h^{\pm}}^2 = -2\lambda_3 w_2^2 + \frac{5}{2}\lambda_4 w_2 w_S - \frac{1}{2}(\lambda_6 + 2\lambda_7) w_S^2, \qquad (5.1.6a)$$

$$m_{H^{\pm}}^{2} = \frac{v^{2}}{2w_{S}} \left[\lambda_{4}w_{2} - (\lambda_{6} + 2\lambda_{7}) w_{S} \right].$$
 (5.1.6b)

The neutral components of the inert doublet h_1 are already diagonalized. Masses of the two neutral states are given by:

$$m_{\tilde{\eta}}^2 = \frac{9}{2} \lambda_4 w_2 w_S, \tag{5.1.7a}$$

$$m_{\tilde{\chi}}^2 = -2(\lambda_2 + \lambda_3)w_2^2 + \frac{5}{2}\lambda_4 w_2 w_S - 2\lambda_7 w_S^2.$$
(5.1.7b)

The doublets h_2 and h_s acquire a non-zero VEV and thus are active. The neutral mass-squared matrix is block-diagonal in the basis:

$$\{\tilde{\eta}_2, \, \tilde{\eta}_S, \, \tilde{\chi}_2, \, \tilde{\chi}_S\}.$$
 (5.1.8)

Therefore the mass-squared matrix can be split into:

$$\mathcal{M}_{\text{Neutral}-2S}^2 = \text{diag}\left(\mathcal{M}_{\tilde{\eta}}^2, \, \mathcal{M}_{\tilde{\chi}}^2\right).$$
(5.1.9)

The mass-squared matrix of the CP-odd sector is:

$$\mathcal{M}_{\chi}^{2} = \begin{pmatrix} (\mathcal{M}_{\chi}^{2})_{11} & (\mathcal{M}_{\chi}^{2})_{12} \\ (\mathcal{M}_{\chi}^{2})_{12} & (\mathcal{M}_{\chi}^{2})_{22} \end{pmatrix},$$
(5.1.10)

where

$$(\mathcal{M}_{\chi}^2)_{11} = \frac{1}{2} w_S \left(\lambda_4 w_2 - 4\lambda_7 w_S \right), \qquad (5.1.11a)$$

$$(\mathcal{M}_{\chi}^2)_{12} = -\frac{1}{2} w_2 \left(\lambda_4 w_2 - 4\lambda_7 w_S\right), \qquad (5.1.11b)$$

$$(\mathcal{M}_{\chi}^2)_{22} = \frac{w_2^2}{2w_S} \left(\lambda_4 w_2 - 4\lambda_7 w_S\right).$$
(5.1.11c)

It is diagonalizable by performing a rotation \mathcal{R}_{β} of eq. (5.1.3). The two *CP*-odd states are:

$$G^0 = c_\beta \tilde{\chi}_2 + s_\beta \tilde{\chi}_S, \qquad (5.1.12a)$$

$$A = -\mathbf{s}_{\beta}\tilde{\chi}_2 + \mathbf{c}_{\beta}\tilde{\chi}_S, \qquad (5.1.12b)$$

where

$$m_A^2 = \frac{v^2}{2w_S} \left(\lambda_4 w_2 - 4\lambda_7 w_S\right).$$
(5.1.13)

The CP-even mass-squared matrix is:

$$(\mathcal{M}_{\eta}^{2}) = \begin{pmatrix} (\mathcal{M}_{\eta}^{2})_{11} & (\mathcal{M}_{\eta}^{2})_{12} \\ (\mathcal{M}_{\eta}^{2})_{12} & (\mathcal{M}_{\eta}^{2})_{22} \end{pmatrix},$$
(5.1.14)

where

$$(\mathcal{M}_{\eta}^2)_{11} = \frac{1}{2} w_2 \left[4 \left(\lambda_1 + \lambda_3 \right) w_2 - 3\lambda_4 w_S \right], \qquad (5.1.15a)$$

$$(\mathcal{M}_{\eta}^{2})_{12} = -\frac{1}{2}w_{2}\left[3\lambda_{4}w_{2} - 2\left(\lambda_{5} + \lambda_{6} + 2\lambda_{7}\right)w_{S}\right], \qquad (5.1.15b)$$

$$(\mathcal{M}_{\eta}^2)_{22} = \frac{1}{2w_S} \left(\lambda_4 w_2^3 + 4\lambda_8 w_S^3 \right).$$
 (5.1.15c)

It is diagonalizable by a rotation matrix:

$$\mathcal{R}_{\alpha} = \begin{pmatrix} c_{\alpha} & s_{\alpha} \\ -s_{\alpha} & c_{\alpha} \end{pmatrix}, \qquad (5.1.16)$$

where the rotation angle is:

$$t_{2\alpha} = \frac{-2w_2w_S(3\lambda_4w_2 - 2(\lambda_5 + \lambda_6 + 2\lambda_7)w_S)}{4(\lambda_1 + \lambda_3)w_2^2w_S - \lambda_4(w_2^3 + 3w_2w_S^2) - 4\lambda_8w_S^3}.$$
(5.1.17)

The CP-even states are thus:

$$h = c_{\alpha} \tilde{\eta}_2 + s_{\alpha} \tilde{\eta}_S, \tag{5.1.18a}$$

$$H = -\mathbf{s}_{\alpha}\tilde{\eta}_2 + \mathbf{c}_{\alpha}\tilde{\eta}_S,\tag{5.1.18b}$$

with masses:

$$m_h^2 = \frac{1}{4w_S^2} \left[4\left(\lambda_1 + \lambda_3\right) w_2^2 w_S^2 + \lambda_4 w_2 w_S \left(w_2^2 - 3w_S^2\right) + 4\lambda_8 w_S^4 - w_S \Delta \right], \qquad (5.1.19a)$$

$$m_H^2 = \frac{1}{4w_S^2} \left[4\left(\lambda_1 + \lambda_3\right) w_2^2 w_S^2 + \lambda_4 w_2 w_S \left(w_2^2 - 3w_S^2\right) + 4\lambda_8 w_S^4 + w_S \Delta \right], \qquad (5.1.19b)$$

where

$$\Delta^{2} = -8 (\lambda_{1} + \lambda_{3}) \lambda_{4} w_{2}^{5} w_{S} + 2 \left[8 (\lambda_{1} + \lambda_{3})^{2} + 21\lambda_{4}^{2} \right] w_{2}^{4} w_{S}^{2} - 8\lambda_{4} \left[3 (\lambda_{1} + \lambda_{3} + 2 (\lambda_{5} + \lambda_{6} + 2\lambda_{7})) - \lambda_{8} \right] w_{2}^{3} w_{S}^{3} + \left[9\lambda_{4}^{2} + 16 \left((\lambda_{5} + \lambda_{6} + 2\lambda_{7})^{2} - 2 (\lambda_{1} + \lambda_{3}) \lambda_{8} \right) \right] w_{2}^{2} w_{S}^{4} + \lambda_{4} w_{2}^{6} + 24\lambda_{4} \lambda_{8} w_{2} w_{S}^{5} + 16\lambda_{8}^{2} w_{S}^{6}.$$

$$(5.1.20)$$

We identify the lighter h state as the SM-like Higgs boson.

On the other hand, governed by the fact that both the charged $\mathcal{M}^2_{\text{Charged}}$ and the *CP*-odd \mathcal{M}^2_{χ} mass-squared matrices are diagonalizable by going into the Higgs basis², *i.e.*, by the \mathcal{R}_{β} rotation, we could assume that the SU(2) doublets are rotated into a new basis:

$$\begin{pmatrix} h_2'\\ h_S' \end{pmatrix} = \mathcal{R}_\beta \begin{pmatrix} h_2\\ h_S \end{pmatrix}.$$
(5.1.21)

Taking this into consideration, the CP-even mass-squared matrix \mathcal{M}^2_η gets modified:

$$\mathcal{R}_{\beta}\mathcal{M}_{\eta}^{2}\mathcal{R}_{\beta}^{\mathrm{T}} \equiv \tilde{\mathcal{M}}_{\eta}^{2} = \begin{pmatrix} (\tilde{\mathcal{M}}_{\eta}^{2})_{11} & (\tilde{\mathcal{M}}_{\eta}^{2})_{12} \\ (\tilde{\mathcal{M}}_{\eta}^{2})_{12} & (\tilde{\mathcal{M}}_{\eta}^{2})_{22} \end{pmatrix},$$
(5.1.22)

where

$$(\tilde{\mathcal{M}}_{\eta}^{2})_{11} = \frac{2}{v^{2}} \left[(\lambda_{1} + \lambda_{3}) w_{2}^{4} - 2\lambda_{4} w_{2}^{3} w_{S} + (\lambda_{5} + \lambda_{6} + 2\lambda_{7}) w_{2}^{2} w_{S}^{2} + \lambda_{8} w_{S}^{4} \right],$$
(5.1.23a)

²To be more precise, due to the SU(2) doublet h_1 we could determine the rotation matrix as $\mathcal{R}_{\beta-3} \equiv \operatorname{diag}(1,\mathcal{R}_{\beta})$.

$$(\tilde{\mathcal{M}}_{\eta}^{2})_{12} = \frac{w_{2}}{v^{2}} \bigg[\lambda_{4} w_{2} \left(3w_{S}^{2} - w_{2}^{2} \right) - \left(2\lambda_{1} + 2\lambda_{3} - \lambda_{5} - \lambda_{6} - 2\lambda_{7} \right) w_{2}^{2} w_{S} - \left(\lambda_{5} + \lambda_{6} + 2\lambda_{7} - 2\lambda_{8} \right) w_{S}^{3} \bigg],$$

$$(\tilde{\mathcal{M}}_{\eta}^{2}) = \frac{w_{2}}{v^{2}} \bigg[\lambda_{4} (c_{1}^{2} - 2c_{1}^{2} - 2c_{1}^{2} + c_{1}^{4} + c_{1}^{4}) + A(c_{1}^{2} + c_{1}^{2} + c_{1}^{2}$$

$$(\tilde{\mathcal{M}}_{\eta}^2)_{22} = \frac{w_2}{2v^2 w_S} \left[\lambda_4 \left(6w_2^2 w_S^2 - 3w_S^4 + w_2^4 \right) + 4 \left(\lambda_1 + \lambda_3 - \lambda_5 - \lambda_6 - 2\lambda_7 + \lambda_8 \right) w_2 w_S^3 \right].$$
(5.1.23c)

The mass-squared matrix $\tilde{\mathcal{M}}_{\eta}^2$ is diagonalizable by $\mathcal{R}_{\alpha'}$ of eq. (5.1.16). Due to the \mathcal{R}_{β} rotation, α' is defined by

$$t_{2\alpha'} = \frac{4w_2w_S \left[2\left(\lambda_1 + \lambda_3\right)w_2^2w_S + \lambda_4\left(w_2^3 - 3w_2w_S^2\right) + \lambda_a\left(w_S^3 - w_2^2w_S\right) - 2\lambda_8w_S^3\right]}{4\left(\lambda_1 + \lambda_3\right)\left(w_2^2w_S^3 - w_2^4w_S\right) + \lambda_4\left(14w_2^3w_S^2 - 3w_2w_S^4 + w_2^5\right) - 8w_2^2\lambda_aw_S^3 + 4\lambda_8\left(w_2^2w_S^3 - w_S^5\right)},\tag{5.1.24}$$

where

$$\lambda_a = \lambda_5 + \lambda_6 + 2\lambda_7. \tag{5.1.25}$$

The CP-even states are now expressed as:

$$h = c_{\alpha'+\beta} \tilde{\eta}_2 + s_{\alpha'+\beta} \tilde{\eta}_S, \qquad (5.1.26a)$$

$$H = -\mathbf{s}_{\alpha'+\beta}\,\tilde{\eta}_2 + \mathbf{c}_{\alpha'+\beta}\,\tilde{\eta}_S. \tag{5.1.26b}$$

Masses of the above states coincide with eq. (5.1.19), as expected. The relation between the angles is trivial:

$$\beta + \alpha' = \alpha. \tag{5.1.27}$$

Doublets in terms of the mass-eigenstates are:

$$h_1 = \begin{pmatrix} h^{\pm} \\ \frac{1}{\sqrt{2}} \left(\tilde{\eta} + i \tilde{\chi} \right) \end{pmatrix}, \tag{5.1.28a}$$

$$h_2 = \begin{pmatrix} \mathbf{c}_{\beta}G^{\pm} - \mathbf{s}_{\beta}H^{\pm} \\ \frac{1}{\sqrt{2}}\left(\mathbf{c}_{\beta}v + \mathbf{c}_{\alpha}h - \mathbf{s}_{\alpha}H + i\left(\mathbf{c}_{\beta}G^0 - \mathbf{s}_{\beta}A\right)\right) \end{pmatrix},$$
(5.1.28b)

$$h_{S} = \begin{pmatrix} \mathbf{s}_{\beta}G^{\pm} + \mathbf{c}_{\beta}H^{\pm} \\ \frac{1}{\sqrt{2}}\left(\mathbf{s}_{\beta}v + \mathbf{s}_{\alpha}h + \mathbf{c}_{\alpha}H + i\left(\mathbf{s}_{\beta}G^{0} + \mathbf{c}_{\beta}A\right)\right) \end{pmatrix}.$$
 (5.1.28c)

It is not very appealing to deal with VEVs as input parameters and therefore both w_2 and w_s can be traded for t_β and v. In this case, the mass-squared parameters are as follows:

$$\begin{split} m_{h^{\pm}}^{2} &= \frac{1}{2}v^{2} \left[-4\lambda_{3}c_{\beta}^{2} + \frac{5}{2}\lambda_{4}s_{2\beta} - (\lambda_{6} + 2\lambda_{7})s_{\beta}^{2} \right], \\ m_{H^{\pm}}^{2} &= \frac{1}{2}v^{2} \left(\lambda_{4}\frac{1}{t_{\beta}} - \lambda_{6} - 2\lambda_{7} \right), \\ m_{\tilde{\eta}}^{2} &= \frac{9}{4}v^{2}\lambda_{4}s_{2\beta}, \\ m_{\tilde{\chi}}^{2} &= \frac{1}{2}v^{2} \left[-4\left(\lambda_{2} + \lambda_{3}\right)c_{\beta}^{2} + \frac{5}{2}\lambda_{4}s_{2\beta} - 4\lambda_{7}s_{\beta}^{2} \right], \\ m_{A}^{2} &= \frac{1}{2}v^{2} \left(\lambda_{4}\frac{1}{t_{\beta}} - 4\lambda_{7} \right), \\ m_{h}^{2} &= v^{2} \left[\left(\lambda_{1} + \lambda_{3}\right)c_{\beta}^{2} + \frac{1}{4}\lambda_{4}\frac{1}{t_{\beta}}\left(c_{\beta}^{2} - 3s_{\beta}^{2}\right) + \lambda_{8}s_{\beta}^{2} \right] - \frac{v^{2}}{4s_{\beta}}\Delta, \\ m_{H}^{2} &= v^{2} \left[\left(\lambda_{1} + \lambda_{3}\right)c_{\beta}^{2} + \frac{1}{4}\lambda_{4}\frac{1}{t_{\beta}}\left(c_{\beta}^{2} - 3s_{\beta}^{2}\right) + \lambda_{8}s_{\beta}^{2} \right] + \frac{v^{2}}{4s_{\beta}}\Delta, \end{split}$$

where

$$\Delta^{2} = -8 (\lambda_{1} + \lambda_{3}) \lambda_{4} c_{\beta}^{5} s_{\beta} + 2 \left[8 (\lambda_{1} + \lambda_{3})^{2} + 21\lambda_{4}^{2} \right] c_{\beta}^{4} s_{\beta}^{2} + \left[9\lambda_{4}^{2} + 16 \left((\lambda_{5} + \lambda_{6} + 2\lambda_{7})^{2} - 2 (\lambda_{1} + \lambda_{3}) \lambda_{8} \right) \right] c_{\beta}^{2} s_{\beta}^{4} - \lambda_{4} \left[3 (\lambda_{1} + \lambda_{3} + 2 (\lambda_{5} + \lambda_{6} + 2\lambda_{7})) - \lambda_{8} \right] s_{2\beta}^{3} + \lambda_{4}^{2} c_{\beta}^{6} + 24\lambda_{4} \lambda_{8} c_{\beta} s_{\beta}^{5} + 16\lambda_{8}^{2} s_{\beta}^{6}.$$
(5.1.30)

5.1.1 Quartic Couplings in Terms of Masses

The mass-squared parameters cannot be inverted in a simple way, as they were in section 4.1.1, to result in λ_i expressed in terms of the mass-squared parameters. This procedure is not trivial due to the more complicative mass-squared parameters. Therefore, we present the result without specifying how we derived it. The λ couplings in terms of the mass-squared parameters are:

$$\lambda_1 = \frac{v^2 \left[9 \left(m_{h^{\pm}}^2 + \mathbf{s}_{\alpha}^2 m_h^2 + \mathbf{c}_{\alpha}^2 m_H^2\right) - m_{\tilde{\eta}}^2\right] - 9m_{h^{\pm}}^2 w_S^2}{18v^2 w_2^2},$$
(5.1.31a)

$$\lambda_2 = \frac{\left(m_{h^{\pm}}^2 - m_{\tilde{\chi}}^2\right)v^2 + \left(m_A^2 - m_{h^{\pm}}^2\right)w_S^2}{2v^2w_2^2},\tag{5.1.31b}$$

$$\lambda_3 = \frac{\left(4m_{\tilde{\eta}}^2 - 9m_{h^{\pm}}^2\right)v^2 + 9m_{h^{\pm}}^2w_S^2}{18v^2w_2^2},\tag{5.1.31c}$$

$$\lambda_4 = \frac{2m_{\tilde{\eta}}^2}{9w_2 w_S},\tag{5.1.31d}$$

$$\lambda_5 = \frac{2m_{h^{\pm}}^2}{v^2} + \frac{w_2 m_{\tilde{\eta}}^2 - \frac{9}{2} \mathbf{s}_{2\alpha} w_S(m_H^2 - m_h^2)}{9w_2 w_S^2},\tag{5.1.31e}$$

$$\lambda_6 = \frac{m_A^2 - 2m_{h^\pm}^2}{v^2} + \frac{m_{\tilde{\eta}}^2}{9w_S^2},\tag{5.1.31f}$$

$$\lambda_7 = \frac{1}{18} \left(\frac{m_{\tilde{\eta}}^2}{w_S^2} - \frac{9m_A^2}{v^2} \right), \tag{5.1.31g}$$

$$\lambda_8 = \frac{9w_S^2 \left(c_\alpha^2 m_h^2 + s_\alpha^2 m_H^2\right) - w_2^2 m_{\tilde{\eta}}^2}{18w_S^4},\tag{5.1.31h}$$

or in terms of the β parameter:

$$\lambda_1 = \frac{1}{18v^2 c_\beta^2} \left(9m_{h^\pm}^2 - 9s_\beta^2 m_{H^\pm}^2 - m_{\tilde{\eta}}^2 + 9s_\alpha^2 m_h^2 + 9c_\alpha^2 m_H^2 \right),$$
(5.1.32a)

$$\lambda_2 = \frac{1}{2v^2 c_\beta^2} \left[s_\beta^2 (m_A^2 - m_{H^{\pm}}^2) + m_{h^{\pm}}^2 - m_{\tilde{\chi}}^2 \right], \qquad (5.1.32b)$$

$$\lambda_3 = \frac{1}{18v^2 c_\beta^2} \left(-9m_{h^{\pm}}^2 + 9s_\beta^2 m_{H^{\pm}}^2 + 4m_{\tilde{\eta}}^2 \right), \qquad (5.1.32c)$$

$$\lambda_4 = \frac{m_{\tilde{\eta}}^2}{9v^2 s_{2\beta}},\tag{5.1.32d}$$

$$\lambda_5 = \frac{1}{9v^2} \left[18m_{H^{\pm}}^2 + m_{\tilde{\eta}}^2 \frac{1}{s_{\beta}^2} + 9\frac{s_{2\alpha}}{s_{\beta}}(m_h^2 - m_H^2) \right], \qquad (5.1.32e)$$

$$\lambda_6 = \frac{1}{9v^2} \left(-18m_{H^{\pm}}^2 + m_{\tilde{\eta}}^2 \frac{1}{\mathbf{s}_{\beta}^2} + 9m_A^2 \right), \qquad (5.1.32f)$$

$$\lambda_7 = \frac{1}{18v^2} \left(m_{\tilde{\eta}}^2 \frac{1}{\mathbf{s}_{\beta}^2} - 9m_A^2 \right), \tag{5.1.32g}$$

$$\lambda_8 = \frac{1}{18v^2 s_\beta^2} \left(-m_{\tilde{\eta}}^2 \frac{1}{t_\beta^2} + 9m_h^2 c_\alpha^2 + 9m_H^2 s_\alpha^2 \right).$$
(5.1.32h)

5.2 R-II-1a in the Higgs Basis

The Higgs basis transformation is given by:

$$\begin{pmatrix} H_1 \\ H_2 \\ H_3 \end{pmatrix} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & c_\beta & s_\beta \\ 0 & -s_\beta & c_\beta \end{pmatrix} \begin{pmatrix} h_1 \\ h_2 \\ h_S \end{pmatrix}.$$
 (5.2.1)

The SU(2) doublets in terms of the mass-eigenstates (5.1.28) are:

$$H_1 = \begin{pmatrix} h^{\pm} \\ \frac{1}{\sqrt{2}} \left(\tilde{\eta} + i\tilde{\chi} \right) \end{pmatrix}, \tag{5.2.2a}$$

$$H_2 = \begin{pmatrix} G^{\pm} \\ \frac{1}{\sqrt{2}} \left(v + c_{\alpha-\beta}h - s_{\alpha-\beta}H + iG^0 \right) \end{pmatrix}, \qquad (5.2.2b)$$

$$H_3 = \begin{pmatrix} H^{\pm} \\ \frac{1}{\sqrt{2}} \left(\mathbf{s}_{\alpha-\beta}h + \mathbf{c}_{\alpha-\beta}H + iA \right) \end{pmatrix}.$$
 (5.2.2c)

In the basis of eq. (2.5.14), the Euler angles are:

$$\beta_1 = \frac{\pi}{2},\tag{5.2.3a}$$

$$\mathbf{t}_{\beta_2} = \frac{w_S}{w_2}.\tag{5.2.3b}$$

Although $\hat{w}_1 = 0$, we rotate the SU(2) doublets in a such way that $\langle H_1 \rangle = v$. This results in the following quadratic couplings:

$$Y_{11} = c_{\beta_2}^2 \mu_1^2 + s_{\beta_2}^2 \mu_0^2, \tag{5.2.4a}$$

$$Y_{13} = -\frac{1}{2} \mathbf{s}_{2\beta_2} \mu_1^2 + \frac{1}{2} \mathbf{s}_{2\beta_2} \mu_0^2, \qquad (5.2.4b)$$

$$Y_{22} = \mu_1^2, \tag{5.2.4c}$$

$$Y_{33} = c_{\beta_2}^2 \mu_0^2 + s_{\beta_2}^2 \mu_1^2, \qquad (5.2.4d)$$

and the quartic couplings are:

$$Z_{1111} = \lambda_1 c_{\beta_2}^4 + \lambda_3 c_{\beta_2}^4 - 2\lambda_4 c_{\beta_2}^3 s_{\beta_2} + \frac{1}{4} \lambda_5 s_{2\beta_2}^2 + \frac{1}{4} \lambda_6 s_{2\beta_2}^2 + \frac{1}{2} \lambda_7 s_{2\beta_2}^2 + \lambda_8 s_{\beta_2}^4, \tag{5.2.5a}$$

$$Z_{1113} = \lambda_1 c_{\beta_2}^3 s_{\beta_2} - 2\lambda_3 c_{\beta_2}^3 s_{\beta_2} + \lambda_4 \left(1 - 2c_{2\beta_2}\right) c_{\beta_2}^2 + \frac{1}{4} \lambda_5 s_{4\beta_2} + \frac{1}{4} \lambda_6 s_{4\beta_2} + \frac{1}{2} \lambda_7 s_{4\beta_2} + 2\lambda_8 c_{\beta_2} s_{\beta_2}^3,$$
(5.2.5b)

$$Z_{1122} = 2\lambda_1 c_{\beta_2}^2 - 2\lambda_3 c_{\beta_2}^2 + \lambda_4 s_{2\beta_2} + \lambda_5 s_{\beta_2}^2,$$
(5.2.5c)

$$Z_{1133} = \frac{1}{2}\lambda_1 s_{2\beta_2}^2 + \frac{1}{2}\lambda_3 s_{2\beta_2}^2 + \frac{1}{2}\lambda_4 s_{4\beta_2} + \frac{1}{4}\lambda_5 \left(c_{4\beta_2} + 3\right) - \frac{1}{2}\lambda_6 s_{2\beta_2}^2 - \lambda_7 s_{2\beta_2}^2 + \frac{1}{2}\lambda_8 s_{2\beta_2}^2, \qquad (5.2.5d)$$

$$Z_{1212} = \lambda_2 c_{\beta_2}^2 + \lambda_3 c_{\beta_2}^2 + \frac{1}{2} \lambda_4 s_{2\beta_2} + \lambda_7 s_{\beta_2}^2, \qquad (5.2.5e)$$

$$Z_{1221} = -2\lambda_2 c_{\beta_2}^2 + 2\lambda_3 c_{\beta_2}^2 + \lambda_4 s_{2\beta_2} + \lambda_6 s_{\beta_2}^2,$$
(5.2.5f)

$$Z_{1223} = \lambda_2 s_{2\beta_2} - \lambda_3 s_{2\beta_2} + \lambda_4 c_{2\beta_2} + \frac{1}{2} \lambda_6 s_{2\beta_2}, \qquad (5.2.5g)$$

$$Z_{1232} = -\lambda_2 s_{2\beta_2} - \lambda_3 s_{2\beta_2} + \lambda_4 c_{2\beta_2} + \lambda_7 s_{2\beta_2},$$
(5.2.5h)

$$Z_{1313} = \frac{1}{4}\lambda_1 s_{2\beta_2}^2 + \frac{1}{4}\lambda_3 s_{2\beta_2}^2 + \frac{1}{4}\lambda_4 s_{4\beta_2} - \frac{1}{4}\lambda_5 s_{2\beta_2}^2 - \frac{1}{4}\lambda_6 s_{2\beta_2}^2 + \lambda_7 \left(c_{\beta_2}^4 + s_{\beta_2}^4 \right) + \frac{1}{4}\lambda_8 s_{2\beta_2}^2, \quad (5.2.5i)$$

$$Z_{1322} = \lambda_1 s_{2\beta_2} + \lambda_3 s_{2\beta_2} + \lambda_4 c_{2\beta_2} + \frac{1}{2} \lambda_5 s_{2\beta_2}, \qquad (5.2.5j)$$

$$Z_{1331} = \frac{1}{2}\lambda_1 s_{2\beta_2}^2 + \frac{1}{2}\lambda_3 s_{2\beta_2}^2 + \frac{1}{2}\lambda_4 s_{4\beta_2} - \frac{1}{2}\lambda_5 s_{2\beta_2}^2 + \frac{1}{4}\lambda_6 \left(c_{4\beta_2} + 3\right) - \lambda_7 s_{2\beta_2}^2 + \frac{1}{2}\lambda_8 s_{2\beta_2}^2, \qquad (5.2.5k)$$

$$Z_{1333} = -2\lambda_1 c_{\beta_2} s_{\beta_2}^3 - 2\lambda_3 c_{\beta_2} s_{\beta_2}^3 - \lambda_4 \left(2c_{2\beta_2} + 1\right) s_{\beta_2}^2 - \frac{1}{4}\lambda_5 s_{4\beta_2} - \frac{1}{4}\lambda_6 s_{4\beta_2} - \frac{1}{2}\lambda_7 s_{4\beta_2} + 2\lambda_8 c_{\beta_2}^3 s_{\beta_2},$$

$$(5.2.51)$$

$$Z_{2222} = \lambda_1 + \lambda_3, \tag{5.2.5m}$$

$$Z_{2233} = 2\lambda_1 s_{\beta_2}^2 - 2\lambda_3 s_{\beta_2}^2 - \lambda_4 s_{2\beta_2} + \lambda_5 c_{\beta_2}^2,$$
(5.2.5n)

$$Z_{2323} = \lambda_2 s_{\beta_2}^2 + \lambda_3 s_{\beta_2}^2 - \frac{1}{2} \lambda_4 s_{2\beta_2} + \lambda_7 c_{\beta_2}^2, \qquad (5.2.50)$$

$$Z_{2332} = -2\lambda_2 \mathbf{s}_{\beta_2}^2 + 2\lambda_3 \mathbf{s}_{\beta_2}^2 - \lambda_4 \mathbf{s}_{2\beta_2} + \lambda_6 \mathbf{c}_{\beta_2}^2, \tag{5.2.5p}$$

$$Z_{3333} = \lambda_1 s_{\beta_2}^4 + \lambda_3 s_{\beta_2}^4 + 2\lambda_4 c_{\beta_2} s_{\beta_2}^3 + \frac{1}{4} \lambda_5 s_{2\beta_2}^2 + \frac{1}{4} \lambda_6 s_{2\beta_2}^2 + \frac{1}{2} \lambda_7 s_{2\beta_2}^2 + \lambda_8 c_{\beta_2}^4.$$
(5.2.5q)

5.3 Scalar-Gauge Boson Interactions

After substituting the SU(2) doublets in terms of the mass-eigenstates (5.1.28) into the kinetic Lagrangian of eq. (4.2.14), the resulting terms are:

$$\begin{split} \mathcal{L}_{VVH} &= \left[\frac{g}{2c_w}m_Z Z_\mu Z^\mu + gm_W W^+_\mu W^{\mu-1}\right] (c_{\alpha-\beta}h - s_{\alpha-\beta}H) \\ &+ \left\{\left[em_W A^\mu W^+_\mu - gm_Z s^2_w Z^\mu W^+_\mu\right] G^- + h.c.\right\}, \end{split} \tag{5.3.1a} \\ \mathcal{L}_{VHH} &= -\frac{g}{2c_w} Z^\mu \left(\tilde{\eta} \overleftrightarrow{\partial}_\mu \tilde{\chi} + c_{\alpha-\beta}h \overleftrightarrow{\partial}_\mu G^0 + s_{\alpha-\beta}h \overleftrightarrow{\partial}_\mu A - s_{\alpha-\beta}H \overleftrightarrow{\partial}_\mu G^0 + c_{\alpha-\beta}H \overleftrightarrow{\partial}_\mu A\right) \\ &- \frac{g}{2} \left\{iW^+_\mu \left(ih^- \overleftrightarrow{\partial}^\mu \tilde{\chi} + h^- \overleftrightarrow{\partial}^\mu \tilde{\eta} + iG^- \overleftrightarrow{\partial}^\mu G^0 + c_{\alpha-\beta}G^- \overleftrightarrow{\partial}^\mu h + s_{\alpha-\beta}H^- \overleftrightarrow{\partial}^\mu h - s_{\alpha-\beta}G^- \overleftrightarrow{\partial}^\mu H + c_{\alpha-\beta}H^- \overleftrightarrow{\partial}^\mu H + iH^- \overleftrightarrow{\partial}^\mu A\right) + h.c.\right\} \\ &+ \left[ieA^\mu + \frac{ig}{2}\frac{c_{2w}}{c_w} Z^\mu\right] \left(h^+ \overleftrightarrow{\partial}_\mu h^- + G^+ \overleftrightarrow{\partial}_\mu G^- + H^+ \overleftrightarrow{\partial}_\mu H^-\right), \end{aligned} \tag{5.3.1b} \\ \mathcal{L}_{VVHH} &= \left[\frac{g^2}{8c_w^2} Z_\mu Z^\mu + \frac{g^2}{4}W^+_\mu W^{\mu-1}\right] \left(\tilde{\eta}^2 + \tilde{\chi}^2 + h^2 + H^2 + (G^0)^2 + A^2) \right) \\ &+ \left\{\left[\frac{eg}{2}A^\mu W^+_\mu - \frac{g^2}{2}\frac{s^2_w}{c_w} Z^\mu W^+_\mu\right] \left(\tilde{\eta}h^- + i\tilde{\chi}h^- + iG^0G^- + c_{\alpha-\beta}hG^- + s_{\alpha-\beta}hH^- - s_{\alpha-\beta}HG^- + c_{\alpha-\beta}HH^- + iAH^-\right) + h.c.\right\} \\ &+ \left[e^2A_\mu A^\mu + eg\frac{c_{2w}}{c_w} A_\mu Z^\mu + \frac{g^2}{4}\frac{c^2_{2w}}{c^2_w} Z_\mu Z^\mu + \frac{g^2}{2}W^-_\mu W^{\mu+1}\right] \left(h^-h^+ + H^-H^+ + G^-G^+\right). \end{aligned} \tag{5.3.1c}$$

From the interaction terms ZZh and ZZH it follows that the states h and H are CP-even and therefore the state A should be CP-odd.

Provided that the *h* scalar is associated with the SM-like Higgs boson, from the interactions hZZ and $hW^{\pm}W^{\mp}$ it follows that the SM limit is reached for $c_{\alpha-\beta} = 1$.

5.4 Scalar-Fermion Interactions

Formulation of the S_3 -symmetric Yukawa sector was presented in section 4.3. The general result is considered. The R-II-1a vacuum configuration is given by $\{0, c_\beta v, s_\beta v\}$. The inert SU(2) doublet

is h_1 and therefore from eq. (4.3.8) it follows that the fermion mass matrices are block-diagonal:

$$\mathcal{M}_{u} = \frac{v}{\sqrt{2}} \begin{pmatrix} y_{1}^{u} \mathbf{s}_{\beta} + y_{2}^{u} \mathbf{c}_{\beta} & 0 & 0\\ 0 & y_{1}^{u} \mathbf{s}_{\beta} - y_{2}^{u} \mathbf{c}_{\beta} & y_{4}^{u} \mathbf{c}_{\beta}\\ 0 & y_{5}^{u} \mathbf{c}_{\beta} & y_{3}^{u} \mathbf{s}_{\beta} \end{pmatrix},$$
(5.4.1a)

$$\mathcal{M}_{d} = \frac{v}{\sqrt{2}} \begin{pmatrix} y_{1}^{d} \mathbf{s}_{\beta} + y_{2}^{d} \mathbf{c}_{\beta} & 0 & 0\\ 0 & y_{1}^{d} \mathbf{s}_{\beta} - y_{2}^{d} \mathbf{c}_{\beta} & y_{4}^{d} \mathbf{c}_{\beta} \\ 0 & y_{5}^{d} \mathbf{c}_{\beta} & y_{3}^{d} \mathbf{s}_{\beta} \end{pmatrix},$$
(5.4.1b)

and hermitian mass-squared matrices are:

$$\mathcal{H}_{u} = \frac{v^{2}}{2} \begin{pmatrix} (y_{1}^{u}\mathbf{s}_{\beta} + y_{2}^{u}\mathbf{c}_{\beta})^{2} & 0 & 0 \\ 0 & (y_{1}^{u}\mathbf{s}_{\beta} - y_{2}^{u}\mathbf{c}_{\beta})^{2} + (y_{4}^{u})^{2}\mathbf{c}_{\beta}^{2} & \mathbf{c}_{\beta} \left[(y_{1}^{u}y_{5}^{u} + y_{3}^{u}y_{4}^{u})\mathbf{s}_{\beta} - y_{2}^{u}y_{5}^{u}\mathbf{c}_{\beta} \right] \\ 0 & \mathbf{c}_{\beta} \left[(y_{1}^{u}y_{5}^{u} + y_{3}^{u}y_{4}^{u})\mathbf{s}_{\beta} - y_{2}^{u}y_{5}^{u}\mathbf{c}_{\beta} \right] & (y_{3}^{u})^{2}\mathbf{s}_{\beta}^{2} + (y_{5}^{u})^{2}\mathbf{c}_{\beta}^{2} \end{pmatrix},$$

$$(5.4.2a)$$

$$\mathcal{H}_{d} = \frac{v^{2}}{2} \begin{pmatrix} \left(y_{1}^{d} \mathbf{s}_{\beta} + y_{2}^{d} \mathbf{c}_{\beta}\right)^{2} & 0 & 0 \\ 0 & \left(y_{1}^{d} \mathbf{s}_{\beta} - y_{2}^{d} \mathbf{c}_{\beta}\right)^{2} + \left(y_{4}^{d}\right)^{2} \mathbf{c}_{\beta}^{2} & \mathbf{c}_{\beta} \left[\left(y_{1}^{d} y_{5}^{d} + y_{3}^{d} y_{4}^{d}\right) \mathbf{s}_{\beta} - y_{2}^{d} y_{5}^{d} \mathbf{c}_{\beta} \right] \\ 0 & \mathbf{c}_{\beta} \left[\left(y_{1}^{d} y_{5}^{d} + y_{3}^{d} y_{4}^{d}\right) \mathbf{s}_{\beta} - y_{2}^{d} y_{5}^{d} \mathbf{c}_{\beta} \right] & \left(y_{3}^{d}\right)^{2} \mathbf{s}_{\beta}^{2} + \left(y_{5}^{d}\right)^{2} \mathbf{c}_{\beta}^{2} \end{pmatrix} \right).$$

$$(5.4.2b)$$

Hermitian mass-squared matrix invariants are (4.3.14):

$$A_{u} = \frac{1}{2}v^{2} \left(c_{\beta}y_{2}^{u} + s_{\beta}y_{1}^{u}\right)^{2} + \frac{1}{2}v^{2} \left[-s_{2\beta}y_{1}^{u}y_{2}^{u} + c_{\beta}^{2} \left((y_{2}^{u})^{2} + (y_{4}^{u})^{2}\right) + s_{\beta}^{2} \left(y_{1}^{u}\right)^{2}\right] + \frac{1}{2}v^{2} \left[c_{\beta}^{2} \left(y_{5}^{u}\right)^{2} + s_{\beta}^{2} \left(y_{3}^{u}\right)^{2}\right],$$
(5.4.3a)

$$A_{d} = \frac{1}{2}v^{2} \left(c_{\beta}y_{2}^{d} + y_{1}^{d}s_{\beta} \right)^{2} + \frac{1}{2}v^{2} \left[-s_{2\beta}y_{1}^{d}y_{2}^{d} + c_{\beta}^{2} \left(\left(y_{2}^{d} \right)^{2} + \left(y_{4}^{d} \right)^{2} \right) + s_{\beta}^{2} \left(y_{1}^{d} \right)^{2} \right] + \frac{1}{2}v^{2} \left[c_{\beta}^{2} \left(y_{5}^{d} \right)^{2} + s_{\beta}^{2} \left(y_{3}^{d} \right)^{2} \right],$$
(5.4.3b)

and

$$B_{u} = -\frac{1}{4}v^{4} \left[2c_{\beta}^{3}s_{\beta} \left[\left((y_{4}^{u})^{2} + (y_{5}^{u})^{2} \right) y_{1}^{u} + y_{3}^{u}y_{4}^{u}y_{5}^{u} \right] y_{2}^{u} \right. \\ \left. + \frac{1}{4}s_{2\beta}^{2} \left[-2y_{1}^{u}y_{3}^{u}y_{4}^{u}y_{5}^{u} + 2\left(y_{2}^{u} \right)^{2} \left(y_{3}^{u} \right)^{2} - \left(y_{1}^{u} \right)^{2} \left(2\left(y_{2}^{u} \right)^{2} - \left(y_{4}^{u} \right)^{2} - \left(y_{5}^{u} \right)^{2} \right) \right] \right]$$
(5.4.4a)
$$\left. + c_{\beta}^{4} \left((y_{2}^{u})^{2} + \left(y_{4}^{u} \right)^{2} \right) \left((y_{2}^{u})^{2} + \left(y_{5}^{u} \right)^{2} \right) + s_{\beta}^{4} \left(y_{1}^{u} \right)^{2} \left(\left(y_{1}^{u} \right)^{2} + 2\left(y_{3}^{u} \right)^{2} \right) \right] \right] ,$$

$$B_{u} = -\frac{1}{4}v^{4} \left[2c_{\beta}^{3}s_{\beta} \left[\left(\left(y_{4}^{d} \right)^{2} + \left(y_{5}^{d} \right)^{2} \right) y_{1}^{d} + y_{3}^{d}y_{4}^{d}y_{5}^{d} \right] y_{2}^{d} \right. \\ \left. + \frac{1}{4}s_{2\beta}^{2} \left[-2y_{1}^{d}y_{3}^{d}y_{4}^{d}y_{5}^{d} + 2\left(y_{2}^{d} \right)^{2} \left(y_{3}^{d} \right)^{2} - \left(y_{1}^{d} \right)^{2} \left(2\left(y_{2}^{d} \right)^{2} - \left(y_{4}^{d} \right)^{2} - \left(y_{5}^{d} \right)^{2} \right) \right] \right] ,$$

$$\left. + c_{\beta}^{4} \left(\left(y_{2}^{d} \right)^{2} + \left(y_{4}^{d} \right)^{2} \right) \left(\left(y_{2}^{d} \right)^{2} + \left(y_{5}^{d} \right)^{2} \right) + s_{\beta}^{4} \left(y_{1}^{d} \right)^{2} \left(\left(y_{1}^{d} \right)^{2} + 2\left(y_{3}^{d} \right)^{2} \right) \right] \right] ,$$

$$\left. + c_{\beta}^{4} \left(\left(y_{2}^{d} \right)^{2} + \left(y_{4}^{d} \right)^{2} \right) \left(\left(y_{2}^{d} \right)^{2} + \left(y_{5}^{d} \right)^{2} \right) + s_{\beta}^{4} \left(y_{1}^{d} \right)^{2} \left(\left(y_{1}^{d} \right)^{2} + 2\left(y_{3}^{d} \right)^{2} \right) \right] \right] ,$$

and

$$C_{u} = \frac{1}{32} v^{6} \left(c_{\beta} y_{2}^{u} + s_{\beta} y_{1}^{u} \right)^{2} \left[c_{2\beta} \left(y_{1}^{u} y_{3}^{u} + y_{4}^{u} y_{5}^{u} \right) + s_{2\beta} y_{2}^{u} y_{3}^{u} - y_{1}^{u} y_{3}^{u} + y_{4}^{u} y_{5}^{u} \right]^{2},$$
(5.4.5a)

$$C_{d} = \frac{1}{32} v^{6} \left(c_{\beta} y_{2}^{d} + s_{\beta} y_{1}^{d} \right)^{2} \left[c_{2\beta} \left(y_{1}^{d} y_{3}^{d} + y_{4}^{d} y_{5}^{d} \right) + s_{2\beta} y_{2}^{d} y_{3}^{d} - y_{1}^{d} y_{3}^{d} + y_{4}^{d} y_{5}^{d} \right]^{2}.$$
(5.4.5b)

The *CP* check of eq. (4.3.15) results in an expected value J = 0.

Hermitian mass-squared matrices are diagonalizable by the left-handed rotation matrix:

$$V_L^{(u,d)} = \begin{pmatrix} 1 & 0 & 0\\ 0 & c_{\theta_L^{(u,d)}} & s_{\theta_L^{(u,d)}}\\ 0 & -s_{\theta_L^{(u,d)}} & c_{\theta_L^{(u,d)}} \end{pmatrix},$$
(5.4.6)

where

$$\mathbf{t}_{2\theta_{L}^{(u,d)}} = \frac{2\mathbf{c}_{\beta} \left[c_{\beta} y_{2}^{(u,d)} y_{5}^{(u,d)} - \mathbf{s}_{\beta} \left(y_{3}^{(u,d)} y_{4}^{(u,d)} + y_{1}^{(u,d)} y_{5}^{(u,d)} \right) \right]}{-\mathbf{s}_{2\beta} y_{1}^{(u,d)} y_{2}^{(u,d)} + \mathbf{c}_{\beta}^{2} \left[\left(y_{2}^{(u,d)} \right)^{2} + \left(y_{4}^{(u,d)} \right)^{2} - \left(y_{5}^{(u,d)} \right)^{2} \right] + \mathbf{s}_{\beta}^{2} \left[\left(y_{1}^{(u,d)} \right)^{2} - \left(y_{3}^{(u,d)} \right)^{2} \right]} (5.4.7)$$

By the look of the left-handed diagonalization matrix we determine the CKM matrix (4.3.28):

$$V_{\rm CKM} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & c_{\theta_L^d - \theta_L^u} & s_{\theta_L^d - \theta_L^u} \\ 0 & -s_{\theta_L^d - \theta_L^u} & c_{\theta_L^d - \theta_L^u} \end{pmatrix}.$$
 (5.4.8)

Such parameterization can be fitted up to the order $\mathcal{O}(10^{-1})$ and results in unrealistic V_{CKM} .

From eq. (5.4.1) it follows that there are five different non-zero Yukawa couplings present. In order to determine masses, three of those are sufficient. This leaves us with two numerically unconstrained Yukawa couplings. Due to this, there will be a net effect which will contribute to FCNC. By taking a look at the fermion mass matrices eq. (5.4.1) one can notice that those become purely diagonal if we impose an additional \mathbb{Z}_2 symmetry on the Yukawa couplings³:

In this limit, the Yukawa couplings in terms of masses are given by:

$$y_{1}^{(u,d)} = \frac{\left(m_{1}^{(u,d)} + m_{2}^{(u,d)}\right)}{\sqrt{2}vs_{\beta}},$$

$$y_{2}^{(u,d)} = \frac{\left(m_{1}^{(u,d)} - m_{2}^{(u,d)}\right)}{\sqrt{2}vc_{\beta}},$$

$$y_{3}^{(u,d)} = \frac{\sqrt{2}m_{3}^{(u,d)}}{vs_{\beta}},$$

(5.4.10)

where $m_i^{(u,d)}$ stands for a mass of a specific fermion generation. We consider the following ordering of the fermion masses:

$$m_3^{(u,d)} > m_2^{(u,d)} > m_1^{(u,d)}.$$
 (5.4.11)

The fermion diagonalization matrices are simplified to $V_f = U_f = \mathcal{I}_3$. The interaction matrix (4.3.36) of the SM-like Higgs boson with fermions is therefore:

$$\tilde{\mathcal{M}}_{h\bar{f}f} = \begin{pmatrix} \frac{\left(m_{1}^{f}\mathbf{s}_{\alpha+\beta} + m_{2}^{f}\mathbf{s}_{\alpha-\beta}\right)}{v\,\mathbf{s}_{2\beta}} & 0 & 0\\ 0 & \frac{\left(m_{1}^{f}\mathbf{s}_{\alpha-\beta} + m_{2}^{f}\mathbf{s}_{\alpha+\beta}\right)}{v\,\mathbf{s}_{2\beta}} & 0\\ 0 & 0 & \frac{m_{3}^{f}\mathbf{s}_{\alpha}}{v\,\mathbf{s}_{\beta}} \end{pmatrix}.$$
(5.4.12)

 $^{^{3}}$ We consider this limit due to the fact that the CKM matrix is unrealistic and not to introduce additional variables so that FCNC are controlled.

The scalar-fermion couplings are given by eq. (4.3.37) and can be easily extracted from the Yukawa Lagrangian:

$$g\left((h/H)\bar{f}_{i}f_{i}\right) = \operatorname{diag}\left(\frac{\left(m_{1}^{f}C_{(h/H)}^{1} + m_{2}^{f}C_{(h/H)}^{2}\right)}{v\,\mathrm{s}_{2\beta}}, \frac{\left(m_{1}^{f}C_{(h/H)}^{2} + m_{2}^{f}C_{(h/H)}^{1}\right)}{v\,\mathrm{s}_{2\beta}}, \frac{m_{3}^{f}C_{(h/H)}^{3}}{v\,\mathrm{s}_{\beta}}\right),$$
(5.4.13a)
$$g\left(A\bar{f}_{i}f_{i}\right) = \operatorname{diag}\left((-1)^{\delta_{u,f}}\frac{i\gamma_{5}\left(m_{1}^{f}c_{2\beta} + m_{2}^{f}\right)}{v\,\mathrm{s}_{2\beta}}, (-1)^{\delta_{u,f}}\frac{i\gamma_{5}\left(m_{1}^{f} + m_{2}^{f}c_{2\beta}\right)}{v\,\mathrm{s}_{2\beta}}, (-1)^{\delta_{u,f}}\frac{i\gamma_{5}m_{3}^{f}}{v\,\mathrm{s}_{2\beta}}\right),$$
(5.4.13b)

where

$$C_h^i = \{ \mathbf{s}_{\alpha+\beta}, \, \mathbf{s}_{\alpha-\beta}, \, \mathbf{s}_{\alpha} \}, \tag{5.4.14a}$$

$$C_H^i = \{ \mathbf{c}_{\alpha+\beta}, \, \mathbf{c}_{\alpha-\beta}, \, \mathbf{c}_{\alpha} \}, \tag{5.4.14b}$$

and $\delta_{u,f}$ is the Kronecker delta introduced due to the Yukawa Lagrangian term $\bar{u}_L^0 (h_i^0)^* u_R^0$.

The SM limit for h is in agreement with the one in section 5.3, $c_{\alpha-\beta} = 1$, however, additional constraints in terms of the β angle should be considered. We assume that the SM-like limit is reachable for $c_{\alpha-\beta} \ge 0.9$.

5.5 Scalar-Scalar Interactions

We provide general trilinear and quartic scalar-scalar couplings. The Feynman rules are given by eq. (4.4.1). The symmetry factors are accounted for, see section 4.4 for a discussion. For simplicity, we define:

$$\lambda_5 + \lambda_6 + 2\lambda_7 = \lambda_a, \tag{5.5.1a}$$

$$\lambda_5 + \lambda_6 - 2\lambda_7 = \lambda_b. \tag{5.5.1b}$$

The trilinear scalar-scalar couplings involving the same species are:

$$g(hhh) = 3v \left[c_{\alpha}^{3} \left(2 \left(\lambda_{1} + \lambda_{3} \right) c_{\beta} - \lambda_{4} s_{\beta} \right) + c_{\alpha}^{2} s_{\alpha} \left(\lambda_{a} s_{\beta} - 3\lambda_{4} c_{\beta} \right) + \lambda_{a} c_{\alpha} c_{\beta} s_{\alpha}^{2} + 2\lambda_{8} s_{\alpha}^{3} s_{\beta} \right], \quad (5.5.2a)$$

$$g(HHH) = -3v \left[s_{\alpha}^{3} \left(2 \left(\lambda_{1} + \lambda_{3} \right) c_{\beta} - \lambda_{4} s_{\beta} \right) + c_{\alpha} s_{\alpha}^{2} \left(3\lambda_{4} c_{\beta} - \lambda_{a} s_{\beta} \right) + \lambda_{a} c_{\alpha}^{2} c_{\beta} s_{\alpha} - 2\lambda_{8} c_{\alpha}^{3} s_{\beta} \right]. \quad (5.5.2b)$$

The trilinear couplings involving the neutral fields are:

$$g\left(\tilde{\eta}\tilde{\eta}h\right) = v\left[\mathbf{s}_{\alpha}\left(\lambda_{a}\mathbf{s}_{\beta} + 3\lambda_{4}\mathbf{c}_{\beta}\right) + \mathbf{c}_{\alpha}\left(2\left(\lambda_{1} + \lambda_{3}\right)\mathbf{c}_{\beta} + 3\lambda_{4}\mathbf{s}_{\beta}\right)\right],\tag{5.5.3a}$$

$$g\left(\tilde{\eta}\tilde{\eta}H\right) = v\left[c_{\alpha}\left(\lambda_{a}s_{\beta} + 3\lambda_{4}c_{\beta}\right) - s_{\alpha}\left(2\left(\lambda_{1} + \lambda_{3}\right)c_{\beta} + 3\lambda_{4}s_{\beta}\right)\right],\tag{5.5.3b}$$

$$g\left(\tilde{\chi}\tilde{\chi}h\right) = v\left[\mathbf{s}_{\alpha}\left(\lambda_{b}\mathbf{s}_{\beta} + \lambda_{4}\mathbf{c}_{\beta}\right) + \mathbf{c}_{\alpha}\left(2\left(\lambda_{1} - 2\lambda_{2} - \lambda_{3}\right)\mathbf{c}_{\beta} + \lambda_{4}\mathbf{s}_{\beta}\right)\right],\tag{5.5.3c}$$

$$g\left(\tilde{\chi}\tilde{\chi}H\right) = v\left[c_{\alpha}\left(\lambda_{b}s_{\beta} + \lambda_{4}c_{\beta}\right) - s_{\alpha}\left(2\left(\lambda_{1} - 2\lambda_{2} - \lambda_{3}\right)c_{\beta} + \lambda_{4}s_{\beta}\right)\right],\tag{5.5.3d}$$

$$g\left(\tilde{\eta}\tilde{\chi}A\right) = v\left[\lambda_4 c_{2\beta} - \left(\lambda_2 + \lambda_3 - \lambda_7\right) s_{2\beta}\right],$$
(5.5.3e)

$$g(hhH) = -v \left[c_{\alpha}^{3} \left(3\lambda_{4}c_{\beta} - \lambda_{a}s_{\beta} \right) + c_{\alpha}^{2}s_{\alpha} \left(\left(-2\lambda_{a} + 6\lambda_{1} + 6\lambda_{3} \right)c_{\beta} - 3\lambda_{4}s_{\beta} \right) + \lambda_{a}c_{\beta}s_{\alpha}^{3} - 2c_{\alpha}s_{\alpha}^{2} \left(3\lambda_{4}c_{\beta} + \left(3\lambda_{8} - 2\lambda_{7} \right)s_{\beta} \right) + \left(\lambda_{5} + \lambda_{6} \right)s_{\alpha}s_{2\alpha}s_{\beta} \right],$$

$$(5.5.3f)$$

$$g(hHH) = v \left[s_{\alpha}^{3} \left(\lambda_{a} s_{\beta} - 3\lambda_{4} c_{\beta} \right) + c_{\alpha} s_{\alpha}^{2} \left(\left(-2\lambda_{a} + 6\lambda_{1} + 6\lambda_{3} \right) c_{\beta} - 3\lambda_{4} s_{\beta} \right) \right. \\ \left. + \lambda_{a} c_{\alpha}^{3} c_{\beta} + 2 c_{\alpha}^{2} s_{\alpha} \left(3\lambda_{4} c_{\beta} - \left(\lambda_{a} - 3\lambda_{8} \right) s_{\beta} \right) \right],$$

$$(5.5.3g)$$

$$g(AAh) = v \left[c_{\alpha} \left(\lambda_{b} c_{\beta}^{3} + 2\lambda_{4} c_{\beta}^{2} s_{\beta} - \lambda_{4} s_{\beta}^{3} + (\lambda_{1} + \lambda_{3} - 2\lambda_{7}) s_{\beta} s_{2\beta} \right) + s_{\alpha} s_{\beta} \left(\lambda_{b} s_{\beta}^{2} + 2 \left(\lambda_{8} - 2\lambda_{7} \right) c_{\beta}^{2} - \frac{1}{2} \lambda_{4} s_{2\beta} \right) \right],$$

$$(5.5.3h)$$

$$g(AAH) = -v \left[\lambda_b c_\beta^3 s_\alpha - s_\beta^3 \left(\lambda_b c_\alpha + \lambda_4 s_\alpha \right) + 2 c_\beta^2 s_\beta \left(\left(2\lambda_7 - \lambda_8 \right) c_\alpha + \lambda_4 s_\alpha \right) \right. \\ \left. + c_\beta s_\beta^2 \left(\lambda_4 c_\alpha + 2 \left(\lambda_1 + \lambda_3 - 2\lambda_7 \right) s_\alpha \right) \right].$$
(5.5.3i)

The trilinear couplings involving the charged fields are:

_

$$g\left(\tilde{\eta}h^{\pm}H^{\mp}\right) = \frac{1}{4}v\left[4\lambda_{4}c_{2\beta} + \left(-4\lambda_{3} + \lambda_{6} + 2\lambda_{7}\right)s_{2\beta}\right],$$
(5.5.4a)

$$g\left(\tilde{\chi}h^{\pm}H^{\mp}\right) = \mp \frac{1}{4}iv\left(4\lambda_2 + \lambda_6 - 2\lambda_7\right)s_{2\beta},\tag{5.5.4b}$$

$$g\left(hH^{\mp}H^{\pm}\right) = -v\left[c_{\alpha}\left(-\lambda_{5}c_{\beta}^{3} - 2\lambda_{4}c_{\beta}^{2}s_{\beta} + \left(-2\lambda_{1} - 2\lambda_{3} + \lambda_{6} + 2\lambda_{7}\right)c_{\beta}s_{\beta}^{2} + \lambda_{4}s_{\beta}^{3}\right) + s_{\alpha}s_{\beta}\left(\left(\lambda_{6} + 2\lambda_{7} - 2\lambda_{8}\right)c_{\beta}^{2} + \frac{1}{2}\lambda_{4}s_{2\beta} - \lambda_{5}s_{\beta}^{2}\right)\right],$$

$$(5.5.4c)$$

$$g\left(HH^{\mp}H^{\pm}\right) = -v\left[\lambda_{5}c_{\beta}^{3}s_{\alpha} + c_{\beta}^{2}s_{\beta}\left(\left(\lambda_{6} + 2\lambda_{7} - 2\lambda_{8}\right)c_{\alpha} + 2\lambda_{4}s_{\alpha}\right)\right. + c_{\beta}s_{\beta}^{2}\left(\lambda_{4}c_{\alpha} + \left(2\lambda_{1} + 2\lambda_{3} - \lambda_{6} - 2\lambda_{7}\right)s_{\alpha}\right) - s_{\beta}^{3}\left(\lambda_{5}c_{\alpha} + \lambda_{4}s_{\alpha}\right)\right],$$

$$(5.5.4d)$$

$$g\left(hh^{\pm}h^{\mp}\right) = v\left[c_{\alpha}\left(2\left(\lambda_{1}-\lambda_{3}\right)c_{\beta}+\lambda_{4}s_{\beta}\right)+s_{\alpha}\left(\lambda_{4}c_{\beta}+\lambda_{5}s_{\beta}\right)\right],\tag{5.5.4e}$$

$$g\left(Hh^{\pm}h^{\mp}\right) = v\left[-\mathbf{s}_{\alpha}\left(2\left(\lambda_{1}-\lambda_{3}\right)\mathbf{c}_{\beta}+\lambda_{4}\mathbf{s}_{\beta}\right)+\mathbf{c}_{\alpha}\left(\lambda_{4}\mathbf{c}_{\beta}+\lambda_{5}\mathbf{s}_{\beta}\right)\right].$$
(5.5.4f)

The quartic couplings involving the same species are:

$$g\left(\tilde{\eta}\tilde{\eta}\tilde{\eta}\tilde{\eta}\tilde{\eta}\right) = g\left(\tilde{\chi}\tilde{\chi}\tilde{\chi}\tilde{\chi}\right) = 6\left(\lambda_1 + \lambda_3\right),\tag{5.5.5a}$$

$$g(hhhh) = 6\left[\frac{1}{4}\lambda_a s_{2\alpha}^2 + (\lambda_1 + \lambda_3) c_{\alpha}^4 - 2\lambda_4 c_{\alpha}^3 s_{\alpha} + \lambda_8 s_{\alpha}^4\right], \qquad (5.5.5b)$$

$$g(HHHH) = 6\left[\frac{1}{4}\lambda_a s_{2\alpha}^2 + \lambda_8 c_\alpha^4 + 2\lambda_4 c_\alpha s_\alpha^3 + (\lambda_1 + \lambda_3) s_\alpha^4\right], \qquad (5.5.5c)$$

$$g(AAAA) = 6\left[\frac{1}{4}\lambda_a s_{2\beta}^2 + \lambda_8 c_\beta^4 + 2\lambda_4 c_\beta s_\beta^3 + (\lambda_1 + \lambda_3) s_\beta^4\right].$$
 (5.5.5d)

The quartic couplings involving only the neutral fields are:

$$g\left(\tilde{\eta}\tilde{\eta}\tilde{\chi}\tilde{\chi}\right) = 2\left(\lambda_1 + \lambda_3\right),\tag{5.5.6a}$$

$$g\left(\tilde{\eta}\tilde{\eta}AA\right) = \lambda_b c_\beta^2 + 2\left(\lambda_1 - 2\lambda_2 - \lambda_3\right) s_\beta^2 - \lambda_4 s_{2\beta},\tag{5.5.6b}$$

$$\left(\tilde{z}\tilde{z}\tilde{z}AA\right) = \lambda_b c_\beta^2 + 2\left(\lambda_1 - 2\lambda_2 - \lambda_3\right) s_\beta^2 - \lambda_4 s_{2\beta},\tag{5.5.6b}$$

$$g\left(\tilde{\chi}\tilde{\chi}AA\right) = \lambda_a c_\beta^2 + 2\left(\lambda_1 + \lambda_3\right) s_\beta^2 - 3\lambda_4 s_{2\beta},\tag{5.5.6c}$$

$$g\left(\tilde{\alpha}\tilde{\alpha}hh\right) = \lambda_a c_\beta^2 + 2\left(\lambda_1 + \lambda_3\right) s_\beta^2 + 2\lambda_4 s_{2\beta},\tag{5.5.6c}$$

$$g\left(\tilde{\eta}\tilde{\eta}hh\right) = \lambda_a s_{\alpha}^2 + 2\left(\lambda_1 + \lambda_3\right) c_{\alpha}^2 + 3\lambda_4 s_{2\alpha},\tag{5.5.6d}$$

$$g\left(\tilde{\eta}\tilde{\eta}hH\right) = \frac{1}{2}\left(\lambda_a - 2\lambda_1 - 2\lambda_3\right)s_{2\alpha} + 3\lambda_4c_{2\alpha},\tag{5.5.6e}$$

$$g\left(\tilde{\eta}\tilde{\eta}HH\right) = \lambda_a c_\alpha^2 + 2\left(\lambda_1 + \lambda_3\right) s_\alpha^2 - 3\lambda_4 s_{2\alpha},\tag{5.5.6f}$$

$$g\left(\tilde{\chi}\tilde{\chi}hh\right) = \lambda_b \mathbf{s}_{\alpha}^2 + 2\left(\lambda_1 - 2\lambda_2 - \lambda_3\right)\mathbf{c}_{\alpha}^2 + \lambda_4 \mathbf{s}_{2\alpha},\tag{5.5.6g}$$

$$g\left(\tilde{\chi}\tilde{\chi}hh\right) = \left(\lambda_2 - \lambda_3\right)\mathbf{c}_{\alpha}^2 + \lambda_4 \mathbf{s}_{2\alpha},\tag{5.5.6g}$$

$$g(\tilde{\chi}\tilde{\chi}hH) = (\lambda_b - 2\lambda_1 + 4\lambda_2 + 2\lambda_3)c_{\alpha}s_{\alpha} + \lambda_4c_{2\alpha},$$

$$(5.5.6h)$$

$$(5.5.6h)$$

$$g\left(\tilde{\chi}\tilde{\chi}HH\right) = \lambda_b c_{\alpha}^2 + 2\left(\lambda_1 - 2\lambda_2 - \lambda_3\right) s_{\alpha}^2 - \lambda_4 s_{2\alpha},$$

$$(5.5.6i)$$

$$g\left(\tilde{\chi}\tilde{\chi}hA\right) = -c_{\alpha}\left[2\left(\lambda_2 + \lambda_2\right)s_{\alpha} - \lambda_4 c_{\alpha}\right] - s_{\alpha}\left(\lambda_4 s_{\alpha} - 2\lambda_7 c_{\alpha}\right)$$

$$(5.5.6i)$$

$$g(\eta\chi nA) = -c_{\alpha} \left[2(\lambda_2 + \lambda_3) s_{\beta} - \lambda_4 c_{\beta} \right] - s_{\alpha} \left(\lambda_4 s_{\beta} - 2\lambda_7 c_{\beta} \right), \tag{5.5.0}$$

$$g(\eta\chi HA) = s_{\alpha} \left[2(\lambda_2 + \lambda_3) s_{\beta} - \lambda_4 c_{\beta} \right] - c_{\alpha} \left(\lambda_4 s_{\beta} - 2\lambda_7 c_{\beta} \right), \tag{5.5.6k}$$

$$g(hhhH) = -3c_{\alpha} \left[\lambda_4 c_{3\alpha} + s_{\alpha} \left(\left(\lambda_1 + \lambda_3 - \lambda_a + \lambda_8\right) c_{2\alpha} + \lambda_1 + \lambda_3 - \lambda_8 \right) \right],$$
(5.5.6l)

$$g(hhHH) = \frac{1}{4} \left[\lambda_a + 3\lambda_1 + 3\lambda_3 + 3\lambda_8 + 6\lambda_4 s_{4\alpha} - 3(\lambda_1 + \lambda_3 - \lambda_a + \lambda_8) c_{4\alpha}\right],$$
(5.5.6m)

$$g(hHHH) = -\frac{5}{2} \mathbf{s}_{\alpha} \left[(\lambda_a + \lambda_1 + \lambda_3 - 3\lambda_8) \mathbf{c}_{\alpha} - (\lambda_1 + \lambda_3 - \lambda_a + \lambda_8) \mathbf{c}_{3\alpha} + 2\lambda_4 \mathbf{s}_{3\alpha} \right], \qquad (5.5.6n)$$
$$g(AAhh) = \mathbf{c}_{\alpha}^2 \left[\lambda_b \mathbf{c}_{\beta}^2 + 2\left(\lambda_1 + \lambda_3\right) \mathbf{s}_{\beta}^2 + \lambda_4 \mathbf{s}_{2\beta} \right] + \mathbf{s}_{\alpha}^2 \left(\lambda_b \mathbf{s}_{\beta}^2 + 2\lambda_8 \mathbf{c}_{\beta}^2 \right) - \mathbf{s}_{2\alpha} \mathbf{s}_{\beta} \left(\lambda_4 \mathbf{s}_{\beta} + 4\lambda_7 \mathbf{c}_{\beta} \right),$$

$$\int (11160) = c_{\alpha} \left[\lambda_{b} c_{\beta} + 2 \left(\lambda_{1} + \lambda_{3} \right) s_{\beta} + \lambda_{4} s_{2\beta} \right] + s_{\alpha} \left(\lambda_{b} s_{\beta} + 2\lambda_{8} c_{\beta} \right) - s_{2\alpha} s_{\beta} \left(\lambda_{4} s_{\beta} + 4\lambda_{7} c_{\beta} \right),$$
(5.5.60)

$$g(AAhH) = \frac{1}{2} \mathbf{s}_{2\alpha} \left[-\lambda_1 - \lambda_3 + \lambda_8 - \lambda_4 \mathbf{s}_{2\beta} + (\lambda_1 + \lambda_3 - \lambda_b + \lambda_8) \mathbf{c}_{2\beta} \right] + \mathbf{s}_{\alpha}^2 \mathbf{s}_{\beta} \left(4\lambda_7 \mathbf{c}_{\beta} + \lambda_4 \mathbf{s}_{\beta} \right) - \mathbf{c}_{\alpha}^2 \mathbf{s}_{\beta} \left(\lambda_4 \mathbf{s}_{\beta} + 4\lambda_7 \mathbf{c}_{\beta} \right),$$
(5.5.6p)

$$g(AAHH) = c_{\alpha}^{2} \left(\lambda_{b} s_{\beta}^{2} + 2\lambda_{8} c_{\beta}^{2}\right) + s_{\alpha}^{2} \left[\lambda_{b} c_{\beta}^{2} + 2\left(\lambda_{1} + \lambda_{3}\right) s_{\beta}^{2} + \lambda_{4} s_{2\beta}\right] + s_{2\alpha} s_{\beta} \left(\lambda_{4} s_{\beta} + 4\lambda_{7} c_{\beta}\right).$$
(5.5.6q)

The quartic couplings involving both neutral and charged fields are:

$$g\left(\tilde{\eta}\tilde{\eta}h^{\pm}h^{\mp}\right) = g\left(\tilde{\chi}\tilde{\chi}h^{\pm}h^{\mp}\right) = 2\left(\lambda_1 + \lambda_3\right),\tag{5.5.7a}$$

$$g\left(\tilde{\eta}\tilde{\eta}H^{\mp}H^{\pm}\right) = -2\left(\lambda_3 - \lambda_1\right)s_{\beta}^2 - \lambda_4 s_{2\beta} + \lambda_5 c_{\beta}^2, \qquad (5.5.7b)$$

$$g\left(\tilde{\chi}\tilde{\chi}H^{\mp}H^{\pm}\right) = -2\left(\lambda_3 - \lambda_1\right)s_{\beta}^2 - \lambda_4 s_{2\beta} + \lambda_5 c_{\beta}^2, \qquad (5.5.7c)$$

$$g\left(\tilde{\eta}hh^{\mp}H^{\pm}\right) = -c_{\alpha}\left(2\lambda_{3}s_{\beta} - \lambda_{4}c_{\beta}\right) - \lambda_{4}s_{\alpha}s_{\beta} + \frac{1}{2}\left(\lambda_{6} + 2\lambda_{7}\right)s_{\alpha}c_{\beta},\tag{5.5.7d}$$

$$g\left(\tilde{\eta}Hh^{\mp}H^{\pm}\right) = \mathbf{s}_{\alpha}\left(2\lambda_{3}\mathbf{s}_{\beta} - \lambda_{4}\mathbf{c}_{\beta}\right) - \lambda_{4}\mathbf{c}_{\alpha}\mathbf{s}_{\beta} + \frac{1}{2}\left(\lambda_{6} + 2\lambda_{7}\right)\mathbf{c}_{\alpha}\mathbf{c}_{\beta},\tag{5.5.7e}$$

$$g\left(\tilde{\chi}hh^{\mp}H^{\pm}\right) = \pm i \left[2\lambda_2 \mathbf{c}_{\alpha}\mathbf{s}_{\beta} + \frac{1}{2}\left(\lambda_6 - 2\lambda_7\right)\mathbf{s}_{\alpha}\mathbf{c}_{\beta}\right], \qquad (5.5.7f)$$

$$g\left(\tilde{\chi}Hh^{\mp}H^{\pm}\right) = \pm i \left[-2\lambda_2 \mathbf{s}_{\alpha}\mathbf{s}_{\beta} + \frac{1}{2}\left(\lambda_6 - 2\lambda_7\right)\mathbf{c}_{\alpha}\mathbf{c}_{\beta}\right], \qquad (5.5.7g)$$

$$g\left(hhh^{\pm}h^{\mp}\right) = 2\left(\lambda_1 - \lambda_3\right)c_{\alpha}^2 + \lambda_4 s_{2\alpha} + \lambda_5 s_{\alpha}^2, \qquad (5.5.7h)$$

$$g\left(hHh^{\pm}h^{\mp}\right) = -\left(\lambda_1 - \lambda_3 - \frac{1}{2}\lambda_5\right)s_{2\alpha} + \lambda_4c_{2\alpha},\tag{5.5.7i}$$

$$g\left(HHh^{\pm}h^{+}\right) = -2\left(\lambda_{3} - \lambda_{1}\right)s_{\alpha}^{2} - \lambda_{4}s_{2\alpha} + \lambda_{5}c_{\alpha}^{2}, \qquad (5.5.7j)$$

$$g\left(AAh^{\pm}h^{\pm}\right) = -2\left(\lambda_{3} - \lambda_{1}\right)s_{\alpha}^{2} - \lambda_{4}s_{2\alpha} + \lambda_{5}c_{\alpha}^{2}, \qquad (5.5.7j)$$

$$g\left(\tilde{A}An^{-}n^{+}\right) = -2\left(\lambda_{3} - \lambda_{1}\right)s_{\beta} - \lambda_{4}s_{2\beta} + \lambda_{5}c_{\beta}, \qquad (5.5.7k)$$

$$g\left(\tilde{\eta}Ah^{\pm}H^{\mp}\right) = \pm i\left[-2\lambda_{2}s_{\beta}^{2} + \frac{1}{2}\left(\lambda_{6} - 2\lambda_{7}\right)c_{\beta}^{2}\right], \qquad (5.5.7l)$$

$$g\left(\eta A h^{+} H^{+}\right) = \pm i \left[-2\lambda_{2} \mathbf{s}_{\beta} + \frac{1}{2} \left(\lambda_{6} - 2\lambda_{7}\right) \mathbf{c}_{\beta}\right], \qquad (5.5.1)$$

$$g\left(\tilde{\chi}Ah^{\pm}H^{+}\right) = 2\lambda_{3}s_{\beta}^{2} - \lambda_{4}s_{2\beta} + \frac{1}{2}\left(\lambda_{6} + 2\lambda_{7}\right)c_{\beta}^{2},$$

$$g\left(hhH^{\mp}H^{\pm}\right) = c_{\alpha}^{2}\left[\lambda_{5}c_{\beta}^{2} + 2\left(\lambda_{1} + \lambda_{3}\right)s_{\beta}^{2} + \lambda_{4}s_{2\beta}\right]$$
(5.5.7m)

$$-\lambda_4 s_{2\alpha} s_\beta^2 + \lambda_5 s_\alpha^2 s_\beta^2 - \frac{1}{2} \lambda_6 s_{2\alpha} s_{2\beta} - \lambda_7 s_{2\alpha} s_{2\beta} + 2\lambda_8 c_\beta^2 s_\alpha^2, \qquad (5.5.7n)$$

$$g(hHH^{\mp}H^{\pm}) = \frac{1}{2} \mathbf{s}_{2\alpha} \left[(\lambda_1 + \lambda_3 - \lambda_5 + \lambda_8) \mathbf{c}_{2\beta} - \lambda_1 - \lambda_3 + \lambda_8 - \lambda_4 \mathbf{s}_{2\beta} \right] + \mathbf{s}_{\alpha}^2 \mathbf{s}_{\beta} \left[\lambda_4 \mathbf{s}_{\beta} + (\lambda_6 + 2\lambda_7) \mathbf{c}_{\beta} \right] - \mathbf{c}_{\alpha}^2 \mathbf{s}_{\beta} \left[\lambda_4 \mathbf{s}_{\beta} + (\lambda_6 + 2\lambda_7) \mathbf{c}_{\beta} \right],$$
(5.5.70)

$$g\left(HHH^{\mp}H^{\pm}\right) = c_{\alpha}^{2}\left(\lambda_{5}s_{\beta}^{2} + 2\lambda_{8}c_{\beta}^{2}\right) + s_{\alpha}^{2}\left[2\left(\lambda_{1} + \lambda_{3}\right)s_{\beta}^{2} + \lambda_{4}s_{2\beta} + \lambda_{5}c_{\beta}^{2}\right] + s_{2\alpha}s_{\beta}\left[\lambda_{4}s_{\beta} + \left(\lambda_{6} + 2\lambda_{7}\right)c_{\beta}\right],$$
(5.5.7p)

$$g(AAH^{\mp}H^{\pm}) = 2\left[(\lambda_1 + \lambda_3) s_{\beta}^4 + 2\lambda_4 c_{\beta} s_{\beta}^3 + \frac{1}{4} \lambda_a s_{2\beta}^2 + \lambda_8 c_{\beta}^4 \right].$$
(5.5.7q)

The quartic couplings involving only the charged fields are:

$$g\left(h^{\pm}h^{\mp}h^{\pm}h^{\mp}\right) = 4\left(\lambda_1 + \lambda_3\right), \qquad (5.5.8a)$$

$$g\left(H^{\mp}H^{\mp}H^{\pm}H^{\pm}\right) = 4\left[\left(\lambda_{1} + \lambda_{3}\right)s_{\beta}^{4} + 2\lambda_{4}c_{\beta}s_{\beta}^{3} + \frac{1}{4}\lambda_{a}s_{2\beta}^{2} + \lambda_{8}c_{\beta}^{4}\right], \qquad (5.5.8b)$$

$$g\left(h^{\pm}h^{\pm}H^{\mp}H^{\mp}\right) = 4\left[\left(\lambda_{2} + \lambda_{3}\right)s_{\beta}^{2} - \frac{1}{2}\lambda_{4}s_{2\beta} + \lambda_{7}c_{\beta}^{2}\right],$$
(5.5.8c)

$$g\left(h^{\pm}h^{\mp}H^{\mp}H^{\pm}\right) = 2\left(\lambda_1 - \lambda_2\right)s_{\beta}^2 - 2\lambda_4s_{2\beta} + \left(\lambda_5 + \lambda_6\right)c_{\beta}^2$$
(5.5.8d)

The trilinear couplings involving only the neutral fields and the would-be Goldstone boson are:

$$g\left(\tilde{\eta}\tilde{\chi}G^{0}\right) = 2v\left[\left(\lambda_{2} + \lambda_{3}\right)c_{\beta}^{2} + \frac{1}{2}\lambda_{4}s_{2\beta} + \lambda_{7}s_{\beta}^{2}\right],$$
(5.5.9a)

$$g\left(G^{0}G^{0}h\right) = v\left[\mathbf{s}_{\alpha}\left(\lambda_{a}\mathbf{c}_{\beta}^{2}\mathbf{s}_{\beta} - \lambda_{4}\mathbf{c}_{\beta}^{3} + 2\lambda_{8}\mathbf{s}_{\beta}^{3}\right) + \mathbf{c}_{\alpha}\mathbf{c}_{\beta}\left(\lambda_{a}\mathbf{s}_{\beta}^{2} + 2\left(\lambda_{1} + \lambda_{3}\right)\mathbf{c}_{\beta}^{2} - \frac{3}{2}\lambda_{4}\mathbf{s}_{2\beta}\right)\right], \quad (5.5.9b)$$

$$g\left(G^{0}G^{0}H\right) = -v\left[-c_{\alpha}\left(\lambda_{a}c_{\beta}^{2}s_{\beta} - \lambda_{4}c_{\beta}^{3} + 2\lambda_{8}s_{\beta}^{3}\right) + s_{\alpha}\left(2\left(\lambda_{1} + \lambda_{3}\right)c_{\beta}^{3} - 3\lambda_{4}c_{\beta}^{2}s_{\beta} + \left(\lambda_{5} + \lambda_{6}\right)c_{\beta}s_{\beta}^{2} + \lambda_{7}s_{\beta}s_{2\beta}\right)\right],$$

$$(5.5.9c)$$

$$g\left(G^{0}Ah\right) = -v\left[c_{\alpha}\left(\lambda_{4}c_{\beta}^{3} + \left(2\lambda_{1} + 2\lambda_{3} - \lambda_{5} - \lambda_{6}\right)c_{\beta}^{2}s_{\beta} - 2\lambda_{4}c_{\beta}s_{\beta}^{2} + 2\lambda_{7}s_{\beta}^{3}\right) - s_{\alpha}c_{\beta}\left(\frac{1}{2}\lambda_{4}s_{2\beta} + 2\lambda_{7}c_{\beta}^{2} - \left(\lambda_{5} + \lambda_{6} - 2\lambda_{8}\right)s_{\beta}^{2}\right)\right],$$

$$g\left(G^{0}AH\right) = -v\left[-s_{\alpha}\left(\lambda_{4}c_{\beta}^{3} + \left(2\lambda_{1} + 2\lambda_{3} - \lambda_{5} - \lambda_{6}\right)c_{\beta}^{2}s_{\beta} + \lambda_{4}s_{\beta}s_{2\beta} - 2\lambda_{7}s_{\beta}^{3}\right) - c_{\alpha}c_{\beta}\left(\frac{1}{2}\lambda_{4}s_{2\beta} + 2\lambda_{7}c_{\beta}^{2} - \left(\lambda_{5} + \lambda_{6} - 2\lambda_{8}\right)s_{\beta}^{2}\right)\right].$$

$$(5.5.9d)$$

$$(5.5.9d)$$

$$(5.5.9d)$$

$$(5.5.9e)$$

The trilinear couplings involving the charged fields and the would-be Goldstone boson are:

$$g\left(\tilde{\eta}h^{\pm}G^{\mp}\right) = \frac{1}{2}v\left[4\lambda_{3}c_{\beta}^{2} + 2\lambda_{4}s_{2\beta} + (\lambda_{6} + 2\lambda_{7})s_{\beta}^{2}\right],$$
(5.5.10a)

$$g\left(\tilde{\chi}h^{\pm}G^{\mp}\right) = \pm iv\left[2\lambda_2 c_{\beta}^2 - \frac{1}{2}\left(\lambda_6 - 2\lambda_7\right)s_{\beta}^2\right],$$
(5.5.10b)

$$g\left(hG^{\mp}G^{\pm}\right) = v\left[s_{\alpha}\left(\lambda_{a}c_{\beta}^{2}s_{\beta} - \lambda_{4}c_{\beta}^{3} + 2\lambda_{8}s_{\beta}^{3}\right) + c_{\alpha}c_{\beta}\left(\lambda_{a}s_{\beta}^{2} + 2\left(\lambda_{1} + \lambda_{3}\right)c_{\beta}^{2} - \frac{3}{2}\lambda_{4}s_{2\beta}\right)\right],$$
(5.5.10c)

$$g\left(hG^{\mp}H^{\pm}\right) = -\frac{1}{4}v\left[c_{\alpha}\left(\lambda_{4}c_{\beta}+3\lambda_{4}c_{3\beta}+2s_{\beta}\left(\left(2\lambda_{1}+2\lambda_{3}-\lambda_{a}\right)c_{2\beta}+2\left(\lambda_{1}+\lambda_{3}\right)-\lambda_{5}\right)\right)\right.$$

$$\left.-2c_{\beta}s_{\alpha}\left(\left(\lambda_{a}-2\lambda_{8}\right)c_{2\beta}-\lambda_{5}+2\lambda_{8}+\lambda_{4}s_{2\beta}\right)\right],$$
(5.5.10d)

$$g\left(HG^{\mp}G^{\pm}\right) = v \left[c_{\alpha} \left(\lambda_{a} c_{\beta}^{2} s_{\beta} - \lambda_{4} c_{\beta}^{3} + 2\lambda_{8} s_{\beta}^{3}\right) - s_{\alpha} \left(2 \left(\lambda_{1} + \lambda_{3}\right) c_{\beta}^{3} - 3\lambda_{4} c_{\beta}^{2} s_{\beta} + \left(\lambda_{5} + \lambda_{6}\right) c_{\beta} s_{\beta}^{2} + \lambda_{7} s_{\beta} s_{2\beta}\right) \right],$$

$$g\left(HG^{\mp}H^{\pm}\right) = \frac{1}{4} v \left[2c_{\alpha} c_{\beta} \left(\left(\lambda_{a} - 2\lambda_{8}\right) c_{2\beta} - \lambda_{5} + 2\lambda_{8} + \lambda_{4} s_{2\beta}\right) + s_{\alpha} \left(\lambda_{4} c_{\beta} + 3\lambda_{4} c_{3\beta} + \left(2\lambda_{1} + 2\lambda_{3} - \lambda_{5} + \lambda_{6} + 2\lambda_{7}\right) s_{\beta} + \left(2\lambda_{1} + 2\lambda_{3} - \lambda_{a}\right) s_{3\beta}\right) \right],$$

$$(5.5.10f)$$

$$(5.5.10f)$$

$$g\left(AG^{\mp}H^{\pm}\right) = \mp \frac{1}{2}iv\left(\lambda_6 - 2\lambda_7\right).$$
(5.5.10g)

The quartic couplings involving the neutral states with at least one neutral would-be Goldstone boson are:

$$g\left(G^{0}G^{0}G^{0}G^{0}\right) = 6\left[\left(\lambda_{1} + \lambda_{3}\right)c_{\beta}^{4} - 2\lambda_{4}c_{\beta}^{3}s_{\beta} + \frac{1}{4}\lambda_{a}s_{2\beta}^{2} + \lambda_{8}s_{\beta}^{4}\right],$$
(5.5.11a)

$$g\left(G^{0}G^{0}G^{0}A\right) = -3c_{\beta}\left[\lambda_{4}c_{3\beta} + s_{\beta}\left(\lambda_{1} + \lambda_{3} - \lambda_{8} + \left(\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8}\right)c_{2\beta}\right)\right],$$
(5.5.11b)

$$g\left(G^{0}G^{0}AA\right) = \frac{1}{4} \begin{bmatrix} 3\lambda_{1} + 3\lambda_{3} + 3\lambda_{8} + 6\lambda_{4}s_{4\beta} + \lambda_{a} - 3\left(\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8}\right)c_{4\beta} \end{bmatrix},$$
(5.5.11c)

$$g(G^{0}AAA) = -\frac{3}{2}s_{\beta}[2\lambda_{4}s_{3\beta} + (\lambda_{a} + \lambda_{1} + \lambda_{3} - 3\lambda_{8})c_{\beta} - (\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8})c_{3\beta}], \quad (5.5.11d)$$

$$g\left(G^{\circ}G^{\circ}hh\right) = c_{\alpha}^{2}\left[2\left(\lambda_{1}+\lambda_{3}\right)c_{\beta}^{2}-\lambda_{4}s_{2\beta}+\lambda_{b}s_{\beta}^{2}\right]+s_{\alpha}^{2}\left[\lambda_{b}c_{\beta}^{2}+2\lambda_{8}s_{\beta}^{2}\right]-c_{\beta}s_{2\alpha}\left[\lambda_{4}c_{\beta}-4\lambda_{7}s_{\beta}\right],$$

$$(5.5.11e)$$

$$g\left(G^{0}G^{0}hH\right) = -\frac{1}{2}\mathbf{s}_{\alpha}\left[\lambda_{1} + \lambda_{3} - \lambda_{4}\mathbf{s}_{2\beta} - \lambda_{8} + \left(\lambda_{1} + \lambda_{3} - \lambda_{b} + \lambda_{8}\right)\mathbf{c}_{2\beta}\right] + \mathbf{c}_{\beta}\mathbf{s}_{\alpha}^{2}\left(\lambda_{4}\mathbf{c}_{\beta} - 4\lambda_{7}\mathbf{s}_{\beta}\right) - \mathbf{c}_{\alpha}^{2}\mathbf{c}_{\beta}\left(\lambda_{4}\mathbf{c}_{\beta} - 4\lambda_{7}\mathbf{s}_{\beta}\right),$$
(5.5.11f)

$$g\left(G^{0}G^{0}HH\right) = -\mathbf{s}_{\alpha}\left[4\lambda_{7}\mathbf{c}_{\alpha}\mathbf{s}_{2\beta} + \mathbf{s}_{\alpha}\mathbf{s}_{\beta}\left(2\lambda_{4}\mathbf{c}_{\beta} - \lambda_{b}\mathbf{s}_{\beta}\right) - 2\mathbf{c}_{\beta}^{2}\left(\lambda_{4}\mathbf{c}_{\alpha} + \left(\lambda_{1} + \lambda_{3}\right)\mathbf{s}_{\alpha}\right)\right] + \mathbf{c}_{\alpha}^{2}\left(\lambda_{b}\mathbf{c}_{\beta}^{2} + 2\lambda_{8}\mathbf{s}_{\beta}^{2}\right),$$
(5.5.11g)

$$g\left(G^{0}Ahh\right) = -\frac{1}{2} \left[s_{2\beta} \left(\lambda_{1} + \lambda_{3} - \lambda_{8} - \lambda_{4}s_{2\alpha} + \left(\lambda_{1} + \lambda_{3} - \lambda_{b} + \lambda_{8}\right)c_{2\alpha}\right) + 2c_{\alpha}c_{2\beta} \left(\lambda_{4}c_{\alpha} - 4\lambda_{7}s_{\alpha}\right) \right],$$
(5.5.11h)

$$g\left(G^{0}AhH\right) = \frac{1}{4} \left[2\lambda_{4}s_{2(\alpha+\beta)} + 8\lambda_{7}c_{2\alpha}c_{2\beta} + 2\left(\lambda_{1} + \lambda_{3} - \lambda_{b} + \lambda_{8}\right)s_{2\alpha}s_{2\beta}\right], \qquad (5.5.11i)$$

$$g\left(G^{0}AHH\right) = \frac{1}{2} \left[-2\lambda_{4}s_{\alpha}s_{\alpha+2\beta} - 4\lambda_{7}c_{2\beta}s_{2\alpha} - s_{2\beta}\left(\lambda_{1} + \lambda_{3} - \lambda_{8} - \left(\lambda_{1} + \lambda_{3} - \lambda_{b} + \lambda_{8}\right)c_{2\alpha}\right) \right],$$
(5.5.11j)

$$g\left(\tilde{\eta}\tilde{\eta}G^{0}G^{0}\right) = 2\left(\lambda_{1} - 2\lambda_{2} - \lambda_{3}\right)c_{\beta}^{2} + \lambda_{4}s_{2\beta} + \lambda_{b}s_{\beta}^{2},$$
(5.5.11k)

$$g\left(\tilde{\eta}\tilde{\eta}G^{0}A\right) = \frac{1}{2}\left(-2\lambda_{1}+4\lambda_{2}+2\lambda_{3}+\lambda_{b}\right)s_{2\beta}+\lambda_{4}c_{2\beta},$$
(5.5.111)

$$g\left(\tilde{\chi}\tilde{\chi}G^{0}G^{0}\right) = 2\left(\lambda_{1} + \lambda_{3}\right)c_{\beta}^{2} + 3\lambda_{4}s_{2\beta} + \lambda_{a}s_{\beta}^{2},$$
(5.5.11m)

$$g\left(\tilde{\chi}\tilde{\chi}G^{0}A\right) = -\frac{1}{2}\left(2\lambda_{1} + 2\lambda_{3} - \lambda_{a}\right)s_{2\beta} + 3\lambda_{4}c_{2\beta},$$
(5.5.11n)

$$g\left(\tilde{\eta}\tilde{\chi}hG^{0}\right) = c_{\alpha}\left[2\left(\lambda_{2}+\lambda_{3}\right)c_{\beta}+\lambda_{4}s_{\beta}\right] + s_{\alpha}\left(\lambda_{4}c_{\beta}+2\lambda_{7}s_{\beta}\right),$$
(5.5.110)

$$g\left(\tilde{\eta}\tilde{\chi}HG^{0}\right) = -\mathbf{s}_{\alpha}\left[2\left(\lambda_{2}+\lambda_{3}\right)\mathbf{c}_{\beta}+\lambda_{4}\mathbf{s}_{\beta}\right] + \mathbf{c}_{\alpha}\left(\lambda_{4}\mathbf{c}_{\beta}+2\lambda_{7}\mathbf{s}_{\beta}\right).$$
(5.5.11p)

The quartic couplings involving the charged fields and the neutral fields along with the would-be Goldstone boson are:

$$g\left(\tilde{\eta}\tilde{\eta}G^{\mp}G^{\pm}\right) = 2\left(\lambda_1 - \lambda_3\right)c_{\beta}^2 + \lambda_4 s_{2\beta} + \lambda_5 s_{\beta}^2, \qquad (5.5.12a)$$

$$g\left(\tilde{\eta}\tilde{\eta}G^{\mp}H^{\pm}\right) = \lambda_4 c_{2\beta} + \frac{1}{2}\left(-2\lambda_1 + 2\lambda_3 + \lambda_5\right)s_{2\beta},\tag{5.5.12b}$$

$$g\left(\tilde{\chi}\tilde{\chi}G^{\mp}G^{\pm}\right) = 2\left(\lambda_1 - \lambda_3\right)c_{\beta}^2 + \lambda_4 s_{2\beta} + \lambda_5 s_{\beta}^2, \qquad (5.5.12c)$$

$$g\left(\tilde{\chi}\tilde{\chi}G^{\mp}H^{\pm}\right) = \lambda_4 c_{2\beta} + \frac{1}{2}\left(-2\lambda_1 + 2\lambda_3 + \lambda_5\right)s_{2\beta},\tag{5.5.12d}$$

$$g\left(\tilde{\eta}hh^{\mp}G^{\pm}\right) = \frac{1}{2} \left[2c_{\alpha}\left(2\lambda_{3}c_{\beta} + \lambda_{4}s_{\beta}\right) + s_{\alpha}\left(2\lambda_{4}c_{\beta} + (\lambda_{6} + 2\lambda_{7})s_{\beta}\right)\right],\tag{5.5.12e}$$

$$g\left(\tilde{\eta}Hh^{\mp}G^{\pm}\right) = \frac{1}{2}c_{\alpha}\left[2\lambda_{4}c_{\beta} + \left(\lambda_{6} + 2\lambda_{7}\right)s_{\beta}\right] - s_{\alpha}\left(2\lambda_{3}c_{\beta} + \lambda_{4}s_{\beta}\right),\tag{5.5.12f}$$

$$g\left(\tilde{\chi}hh^{\mp}G^{\pm}\right) = \pm i \left[-2\lambda_2 c_{\alpha}c_{\beta} + \frac{1}{2}\left(\lambda_6 - 2\lambda_7\right)s_{\alpha}s_{\beta}\right], \qquad (5.5.12g)$$

$$g\left(\tilde{\chi}Hh^{\mp}G^{\pm}\right) = \pm i \left[2\lambda_2 c_\beta s_\alpha + \frac{1}{2}\left(\lambda_6 - 2\lambda_7\right) c_\alpha s_\beta\right], \qquad (5.5.12h)$$

$$g\left(G^{0}G^{0}h^{\pm}h^{+}\right) = 2\left(\lambda_{1} - \lambda_{3}\right)c_{\beta}^{2} + \lambda_{4}s_{2\beta} + \lambda_{5}s_{\beta}^{2},\tag{5.5.12i}$$

$$g\left(G^{0}Ah^{\pm}h^{\mp}\right) = \lambda_{4}c_{2\beta} + \frac{1}{2}\left(-2\lambda_{1} + 2\lambda_{3} + \lambda_{5}\right)s_{2\beta},\tag{5.5.12j}$$

$$g\left(G^{0}\tilde{\eta}h^{\pm}G^{\mp}\right) = \pm i\left[-2\lambda_{2}c_{\beta}^{2} + \frac{1}{2}\left(\lambda_{6} - 2\lambda_{7}\right)s_{\beta}^{2}\right],$$
(5.5.12k)

$$g\left(G^{0}\tilde{\eta}h^{\pm}H^{\mp}\right) = g\left(A\tilde{\eta}h^{\pm}G^{\pm}\right) = \pm \frac{1}{4}i\left(4\lambda_{2} + \lambda_{6} - 2\lambda_{7}\right)s_{2\beta},\tag{5.5.12l}$$

$$g\left(G^{0}\tilde{\chi}h^{\pm}G^{\pm}\right) = \frac{1}{2}\left[4\lambda_{3}c_{\beta}^{2} + 2\lambda_{4}s_{2\beta} + (\lambda_{6} + 2\lambda_{7})s_{\beta}^{2}\right],$$
(5.5.12m)

$$g\left(G^{0}\tilde{\chi}h^{\pm}H^{\pm}\right) = g\left(A\tilde{\chi}h^{\pm}G^{\pm}\right) = \frac{1}{4}\left[4\lambda_{4}c_{2\beta} + \left(-4\lambda_{3} + \lambda_{6} + 2\lambda_{7}\right)s_{2\beta}\right],$$
(5.5.12n)

$$g\left(G^{0}hG^{\mp}H^{\pm}\right) = g\left(AHG^{\mp}H^{\pm}\right) = \pm \frac{1}{2}i\left(\lambda_{6} - 2\lambda_{7}\right)s_{\alpha-\beta},$$
(5.5.12o)

$$g\left(G^{0}HG^{\mp}H^{\pm}\right) = g\left(AhG^{\pm}H^{\mp}\right) = \pm \frac{1}{2}i\left(\lambda_{6} - 2\lambda_{7}\right)c_{\alpha-\beta},$$
(5.5.12p)

$$g\left(hhG^{\mp}G^{\pm}\right) = c_{\alpha}^{2}\left[2\left(\lambda_{1}+\lambda_{3}\right)c_{\beta}^{2}-\lambda_{4}s_{2\beta}+\lambda_{5}s_{\beta}^{2}\right]-c_{\beta}s_{2\alpha}\left(\lambda_{4}c_{\beta}-\left(\lambda_{6}+2\lambda_{7}\right)s_{\beta}\right)\right.$$

+
$$s_{\alpha}^{2}\left(\lambda_{5}c_{\beta}^{2}+2\lambda_{8}s_{\beta}^{2}\right),$$
(5.5.12q)

$$g\left(hhG^{\mp}H^{\pm}\right) = \frac{1}{2} \left[-2c_{\alpha}c_{2\beta}\left(\lambda_{4}c_{\alpha}-\left(\lambda_{6}+2\lambda_{7}\right)s_{\alpha}\right) - s_{2\beta}\left(\lambda_{1}+\lambda_{3}-\lambda_{4}s_{2\alpha}-\lambda_{8}+\left(\lambda_{1}+\lambda_{3}-\lambda_{5}+\lambda_{8}\right)c_{2\alpha}\right)\right],$$
(5.5.12r)

$$g\left(hHG^{\mp}G^{\pm}\right) = -\frac{1}{2}\mathbf{s}_{2\alpha}\left[\lambda_{1} + \lambda_{3} - \lambda_{4}\mathbf{s}_{2\beta} - \lambda_{8} + (\lambda_{1} + \lambda_{3} - \lambda_{5} + \lambda_{8})\mathbf{c}_{2\beta}\right] + \mathbf{c}_{\beta}\mathbf{s}_{\alpha}^{2}\left[\lambda_{4}\mathbf{c}_{\beta} - (\lambda_{6} + 2\lambda_{7})\mathbf{s}_{\beta}\right] - \mathbf{c}_{\alpha}^{2}\mathbf{c}_{\beta}\left[\lambda_{4}\mathbf{c}_{\beta} - (\lambda_{6} + 2\lambda_{7})\mathbf{s}_{\beta}\right],$$
(5.5.12s)

$$g(hHG^{\mp}H^{\pm}) = \frac{1}{4} \left[2\lambda_4 s_{2(\alpha+\beta)} + 2(\lambda_6 + 2\lambda_7) c_{2\alpha} c_{2\beta} + 2(\lambda_1 + \lambda_3 - \lambda_5 + \lambda_8) s_{2\alpha} s_{2\beta} \right], \quad (5.5.12t)$$

$$g\left(HHG^{\mp}G^{\pm}\right) = -\lambda_{4}s_{\alpha}^{2}s_{2\beta} + \lambda_{5}s_{\alpha}^{2}s_{\beta}^{2} - \frac{1}{2}\lambda_{6}s_{2\alpha}s_{2\beta} - \lambda_{7}s_{2\alpha}s_{2\beta} + c_{\alpha}^{2}\left(\lambda_{5}c_{\beta}^{2} + 2\lambda_{8}s_{\beta}^{2}\right) + 2c_{\beta}^{2}s_{\alpha}\left[(\lambda_{1} + \lambda_{3})s_{\alpha} + \lambda_{4}c_{\alpha}\right],$$

$$(5.5.12u)$$

$$g\left(HHG^{\mp}H^{\pm}\right) = -\frac{1}{2}\left(\lambda_{5} - 2\lambda_{8}\right)c_{\alpha}^{2}s_{2\beta} - \frac{1}{2}s_{2\alpha}\left[\lambda_{4}s_{2\beta} + (\lambda_{6} + 2\lambda_{7})c_{2\beta}\right] - s_{\alpha}^{2}\left[\lambda_{4}c_{2\beta} + \frac{1}{2}\left(2\left(\lambda_{1} + \lambda_{3}\right) - \lambda_{5}\right)s_{2\beta}\right],$$
(5.5.12v)

and

$$g\left(G^{0}G^{0}G^{\mp}G^{\pm}\right) = 2\left[\left(\lambda_{1} + \lambda_{3}\right)c_{\beta}^{4} - 2\lambda_{4}c_{\beta}^{3}s_{\beta} + \lambda_{a}c_{\beta}^{2}s_{\beta}^{2} + \lambda_{8}s_{\beta}^{4}\right], \qquad (5.5.13a)$$

$$g\left(G^{0}G^{0}G^{\mp}H^{\pm}\right) = -c_{\beta}\left[\lambda_{4}c_{\beta\beta} + s_{\beta}\left(\lambda_{1} + \lambda_{2} - \lambda_{2} + (\lambda_{1} + \lambda_{2} - \lambda_{3} + \lambda_{3})c_{\beta\beta}\right)\right] \qquad (5.5.13b)$$

$$g(G^{0}G^{0}H^{\mp}H^{\pm}) = -c_{\beta} \left[\lambda_{4}c_{3\beta} + s_{\beta} \left(\lambda_{1} + \lambda_{3} - \lambda_{8} + (\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8})c_{2\beta}\right)\right], \qquad (3.3.13b)$$

$$g(G^{0}G^{0}H^{\mp}H^{\pm}) = \frac{1}{4} \left[\lambda_{1} + \lambda_{3} + 3\lambda_{5} - \lambda_{6} - 2\lambda_{7} + \lambda_{8} + 2\lambda_{4}s_{4\beta} - (\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8})c_{4\beta}\right],$$

$$g\left(G^{0}AG^{\mp}G^{\pm}\right) = -c_{\beta}\left[\lambda_{1} + \lambda_{3} - \lambda_{8} + \lambda_{4}c_{3\beta} + s_{\beta}\left(\left(\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8}\right)c_{2\beta}\right)\right], \qquad (5.5.13d)$$

$$g\left(G^{0}AG^{\mp}H^{\pm}\right) = \frac{1}{4}\left[\lambda_{1} + \lambda_{3} - \lambda_{5} + \lambda_{6} + 2\lambda_{7} + \lambda_{8} + 2\lambda_{4}s_{4\beta} - \left(\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8}\right)c_{4\beta}\right],$$

$$(5.5.13e)$$

$$g\left(G^{0}AH^{\mp}H^{\pm}\right) = -\frac{1}{2}\mathbf{s}_{\beta}\left[2\lambda_{4}\mathbf{s}_{3\beta} + \left(\lambda_{a} + \lambda_{1} + \lambda_{3} - 3\lambda_{8}\right)\mathbf{c}_{\beta} - \left(\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8}\right)\mathbf{c}_{3\beta}\right], \quad (5.5.13f)$$
$$g\left(AAG^{\mp}G^{\pm}\right) = \frac{1}{4}\left[\lambda_{1} + \lambda_{3} + 2\lambda_{4}\mathbf{s}_{4\beta} + 3\lambda_{5} - \lambda_{6} - 2\lambda_{7} + \lambda_{8} - \left(\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8}\right)\mathbf{c}_{4\beta}\right], \quad (5.5.13g)$$

$$g\left(AAH^{\mp}G^{\pm}\right) = -\frac{1}{2}\mathbf{s}_{\beta}\left[2\lambda_{4}\mathbf{s}_{3\beta} + (\lambda_{a} + \lambda_{1} + \lambda_{3} - 3\lambda_{8})\mathbf{c}_{\beta} - (\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8})\mathbf{c}_{3\beta}\right].$$
 (5.5.13h)

The quartic couplings involving only the charged fields and the would-be Goldstone boson are:

$$g\left(G^{\mp}G^{\mp}G^{\pm}G^{\pm}\right) = 4\left[\left(\lambda_1 + \lambda_3\right)c_{\beta}^4 - 2\lambda_4c_{\beta}^3s_{\beta} + \lambda_ac_{\beta}^2s_{\beta}^2 + \lambda_8s_{\beta}^4\right],\tag{5.5.14a}$$

$$g\left(G^{+}G^{+}G^{\pm}H^{\pm}\right) = -c_{\beta}\left[2\lambda_{4}c_{3\beta} + 2s_{\beta}\left(\lambda_{1} + \lambda_{3} - \lambda_{8} + \left(\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8}\right)c_{2\beta}\right)\right], \qquad (5.5.14b)$$

$$g\left(G^{\mp}G^{\mp}H^{\pm}H^{\pm}\right) = \frac{1}{2}\left[\lambda_1 + \lambda_3 - \lambda_5 - \lambda_6 + 6\lambda_7 + \lambda_8 + 2\lambda_4 \mathbf{s}_{4\beta} - (\lambda_1 + \lambda_3 - \lambda_a + \lambda_8)\mathbf{c}_{4\beta}\right],$$
(5.5.14c)

$$g\left(G^{\mp}H^{\mp}G^{\pm}H^{\pm}\right) = \frac{1}{2}\left[\lambda_1 + \lambda_3 + \lambda_8 + 2\lambda_4 \mathbf{s}_{4\beta} + \lambda_b - (\lambda_1 + \lambda_3 - \lambda_a + \lambda_8)\mathbf{c}_{4\beta}\right],\tag{5.5.14d}$$

$$g\left(G^{\mp}H^{\mp}H^{\pm}H^{\pm}\right) = -\mathbf{s}_{\beta}\left[2\lambda_{4}\mathbf{s}_{3\beta} + (\lambda_{a} + \lambda_{1} + \lambda_{3} - 3\lambda_{8})\mathbf{c}_{\beta} - (\lambda_{1} + \lambda_{3} - \lambda_{a} + \lambda_{8})\mathbf{c}_{3\beta}\right], \quad (5.5.14e)$$

$$g\left(h^{\pm}h^{\pm}G^{\mp}G^{\mp}\right) = 4\left[\left(\lambda_{2} + \lambda_{3}\right)c_{\beta}^{2} + \frac{1}{2}\lambda_{4}s_{2\beta} + \lambda_{7}s_{\beta}^{2}\right],$$
(5.5.14f)

$$g(h^{\pm}h^{\pm}G^{+}H^{+}) = 2\lambda_{4}c_{2\beta} - 2(\lambda_{2} + \lambda_{3} - \lambda_{7})s_{2\beta}, \qquad (5.5.14g)$$

$$g(h^{\pm}h^{\mp}G^{\pm}G^{\mp}) = 2(\lambda_1 - \lambda_2)c_{\beta}^2 + 2\lambda_4 s_{2\beta} + (\lambda_5 + \lambda_6)s_{\beta}^2, \qquad (5.5.14h)$$

$$g(h^{\pm}h^{\mp}H^{\pm}G^{\mp}) = 2\lambda_4 c_{2\beta} + \frac{1}{2}(-2\lambda_1 + 2\lambda_2 + \lambda_5 + \lambda_6)s_{2\beta}.$$
 (5.5.14i)

From the trilinear couplings involving the same species particles it follows that the states h and H are CP-even. From the trilinear couplings involving only the neutral states we get that A is CP-odd. Although the states $\tilde{\eta}$ and $\tilde{\chi}$ do not mix, it is impossible to determine their CP properties from the couplings.

5.6 Constraints

Necessary⁴ stability constraints of Ref. [37] are considered, see eq. (4.5.10). Alongside, if the necessary stability conditions are satisfied, we perform an additional numerical minimization of the potential using the Mathematica function NMinimize.

In section 4.5.5 we verified that the unitarity constraints for $\lambda_4 = 0$ are in agreement with those of Ref. [37]. We also verify that we get the same eigenvalues of the S matrix when $\lambda_4 \neq 0$. For convenience, we present the eigenvalues of Ref. [37]:

$$a_{1}^{\pm} = \left(\lambda_{1} - \lambda_{2} + \frac{\lambda_{5} + \lambda_{6}}{2}\right) \pm \sqrt{\left(\lambda_{1} - \lambda_{2} + \frac{\lambda_{5} + \lambda_{6}}{2}\right)^{2} - 4\left\{\left(\lambda_{1} - \lambda_{2}\right)\left(\frac{\lambda_{5} + \lambda_{6}}{2}\right) - \lambda_{4}^{2}\right\}},$$

$$(5.6.1a)$$

$$a_{2}^{\pm} = (\lambda_{1} + \lambda_{2} + 2\lambda_{3} + \lambda_{8}) \pm \sqrt{(\lambda_{1} + \lambda_{2} + 2\lambda_{3} + \lambda_{8})^{2} - 4\{\lambda_{8}(\lambda_{1} + \lambda_{2} + 2\lambda_{3}) - 2\lambda_{7}^{2}\}}, \quad (5.6.1b)$$

$$a_{3}^{\pm} = (\lambda_{1} - \lambda_{2} + 2\lambda_{3} + \lambda_{8}) \pm \sqrt{(\lambda_{1} - \lambda_{2} + 2\lambda_{3} + \lambda_{8})^{2} - 4\left\{\lambda_{8}(\lambda_{1} - \lambda_{2} + 2\lambda_{3}) - \frac{\lambda_{6}^{2}}{2}\right\}}, \quad (5.6.1c)$$

$$a_4^{\pm} = \left(\lambda_1 + \lambda_2 + \frac{\lambda_5}{2} + \lambda_7\right) \pm \sqrt{\left(\lambda_1 + \lambda_2 + \frac{\lambda_5}{2} + \lambda_7\right)^2 - 4\left\{\left(\lambda_1 + \lambda_2\right)\left(\frac{\lambda_5}{2} + \lambda_7\right) - \lambda_4^2\right\}},\tag{5.6.1d}$$

$$a_{5}^{\pm} = (5\lambda_{1} - \lambda_{2} + 2\lambda_{3} + 3\lambda_{8}) \\ \pm \sqrt{(5\lambda_{1} - \lambda_{2} + 2\lambda_{3} + 3\lambda_{8})^{2} - 4\left\{3\lambda_{8}(5\lambda_{1} - \lambda_{2} + 2\lambda_{3}) - \frac{1}{2}(2\lambda_{5} + \lambda_{6})^{2}\right\}},$$
(5.6.1e)

$$a_{6}^{\pm} = \left(\lambda_{1} + \lambda_{2} + 4\lambda_{3} + \frac{\lambda_{5}}{2} + \lambda_{6} + 3\lambda_{7}\right)$$
$$\pm \sqrt{\left(\lambda_{1} + \lambda_{2} + 4\lambda_{3} + \frac{\lambda_{5}}{2} + \lambda_{6} + 3\lambda_{7}\right)^{2} - 4\left\{\left(\lambda_{1} + \lambda_{2} + 4\lambda_{3}\right)\left(\frac{\lambda_{5}}{2} + \lambda_{6} + 3\lambda_{7}\right) - 9\lambda_{4}^{2}\right\}},$$
(5.6.1f)

$$b_1 = \lambda_5 + 2\lambda_6 - 6\lambda_7, \tag{5.6.1g}$$

$$b_2 = \lambda_5 - 2\lambda_7, \tag{5.6.1h}$$

$$b_{3} = 2(\lambda_{1} - 5\lambda_{2} - 2\lambda_{3}), \tag{5.6.1i}$$

$$b_{4,5} = 2(\lambda_1 \pm \lambda_2 - 2\lambda_3), \qquad (5.6.1j)$$

⁴The stability conditions are necessary and not sufficient as $\lambda_4 \neq 0$, see Ref. [34].

$$b_6 = \lambda_5 - \lambda_6$$

The unitarity constraint is thus:

$$|a_i^{\pm}| \le 16\pi, \quad |b_i| \le 16\pi.$$
 (5.6.2)

The soft perturbativity condition (4.5.18) along with a more severe perturbativity condition in terms of the quartic couplings eq. (4.5.22) are considered. Most of the quartic couplings depend on trigonometric functions and therefore it is not a trivial task to extract limits in terms of λ_i from such terms. By inspecting the coupling $g(\tilde{\eta}\tilde{\eta}\tilde{\eta}\tilde{\eta})$ (5.5.5a) the following relation can be extracted: $|\lambda_1 + \lambda_3| \leq \frac{2}{3}\pi$.

The λ_i couplings can be constrained by the mass-squared parameters. From $m_{\tilde{\eta}}^2 > 0$ we get that $\lambda_4 > 0$. We consider that $m_{\tilde{\eta}}^2 < m_{\tilde{\chi}}^2$ and thus:

$$\lambda_2 + \lambda_3 + \lambda_4 \mathbf{t}_\beta + \lambda_7 \mathbf{t}_\beta^2 < 0. \tag{5.6.3}$$

From the m_A^2 mass-squared parameter it follows that:

$$\lambda_4 - 4\lambda_7 \mathbf{t}_\beta > 0. \tag{5.6.4}$$

The inert scalar sector is bounded by the necessary stability conditions, mainly that the terms $\sqrt{\dots}$ (4.5.10) should be positive definite:

$$9t_{\beta} \left(m_h^2 c_{\alpha}^2 + m_H^2 s_{\alpha}^2 \right) > m_{\tilde{\eta}}^2, \qquad (5.6.5a)$$

$$m_h^2 \mathbf{s}_{\alpha}^2 + m_H^2 \mathbf{c}_{\alpha}^2 + m_A^2 \mathbf{s}_{\beta}^2 - 2m_{H^+}^2 \mathbf{s}_{\beta}^2 > \frac{1}{9} m_{\tilde{\eta}}^2 + m_{\tilde{\chi}}^2.$$
(5.6.5b)

We use these constraints to numerically bound the mass terms $m_{\tilde{\eta}}$ and $m_{\tilde{\chi}}$ from above.

5.6.1 Electroweak Oblique Parameters

The SU(2) doublets in the Higgs basis were presented in eq. (5.2.2). The rotation matrix for the charged sector is:

$$\begin{pmatrix} G^{\pm} \\ H^{\pm} \\ h^{\pm} \end{pmatrix} = U \begin{pmatrix} G^{\pm} \\ H^{\pm} \\ h^{\pm} \end{pmatrix}, \qquad (5.6.6)$$

and thus U is simply an identity matrix \mathcal{I}_3 . For the neutral sector we get:

$$\begin{pmatrix} \left(h \operatorname{c}_{\beta-\alpha} + H \operatorname{s}_{\beta-\alpha} + iG^{0}\right) \\ \left(-h \operatorname{s}_{\beta-\alpha} + H \operatorname{c}_{\beta-\alpha} + iA\right) \\ \left(\tilde{\eta} + i\tilde{\chi}\right) \end{pmatrix} = V \begin{pmatrix} G^{0} \\ A \\ h \\ H \\ \tilde{\eta} \\ \tilde{\chi} \end{pmatrix},$$
(5.6.7)

where

$$V = \begin{pmatrix} i & 0 & c_{\beta-\alpha} & s_{\beta-\alpha} & 0 & 0\\ 0 & i & -s_{\beta-\alpha} & c_{\beta-\alpha} & 0 & 0\\ 0 & 0 & 0 & 1 & i \end{pmatrix}.$$
 (5.6.8)

The expressions for the S, T, U parameters are as follows:

$$T = \frac{g^2}{64m_W^2 \pi^2 \alpha} \left[F\left(m_{h^{\pm}}^2, m_{\tilde{\eta}}^2\right) + F\left(m_{h^{\pm}}^2, m_{\tilde{\chi}}^2\right) + s_{\beta-\alpha}^2 F\left(m_{H^{\pm}}^2, m_h^2\right) + c_{\beta-\alpha}^2 F\left(m_{H^{\pm}}^2, m_H^2\right) \right. \\ \left. + F\left(m_{H^{\pm}}^2, m_A^2\right) - F\left(m_{\tilde{\eta}}^2, m_{\tilde{\chi}}^2\right) - s_{\beta-\alpha}^2 F\left(m_h^2, m_A^2\right) - c_{\beta-\alpha}^2 F\left(m_H^2, m_A^2\right) \right. \\ \left. + 3s_{\beta-\alpha}^2 \left(F\left(m_{W^{\pm}}^2, m_h^2\right) - F\left(m_Z^2, m_h^2\right) + F\left(m_Z^2, m_H^2\right) - F\left(m_{W^{\pm}}^2, m_H^2\right)\right) \right],$$

(5.6.1k)

$$\begin{split} S &= \frac{g^2 s_w^2}{96\pi^2 \alpha} \left[c_{2w}^2 \left(G\left(m_{h^{\pm}}^2, m_{h^{\pm}}^2, m_Z^2 \right) + G\left(m_{H^{\pm}}^2, m_{H^{\pm}}^2, m_Z^2 \right) \right) + G\left(m_{\tilde{\eta}}^2, m_{\tilde{\chi}}^2, m_Z^2 \right) \\ &+ s_{\beta - \alpha}^2 G\left(m_h^2, m_A^2, m_Z^2 \right) + c_{\beta - \alpha}^2 G\left(m_H^2, m_A^2, m_Z^2 \right) - 2\ln(m_{h^{\pm}}^2) - 2\ln(m_{H^{\pm}}^2) \\ &+ \ln(m_{\tilde{\eta}}^2) + \ln(m_{\tilde{\chi}}^2) + \ln(m_H^2) + \ln(m_A^2) + s_{\beta - \alpha}^2 \left(\hat{G}\left(m_H^2, m_Z^2 \right) - \hat{G}\left(m_h^2, m_Z^2 \right) \right) \right], \\ (5.6.9b) \end{split}$$

$$U &= \frac{g^2 s_w^2}{96\pi^2 \alpha} \left[G\left(m_{h^{\pm}}^2, m_{\tilde{\eta}}^2, m_{W^{\pm}}^2 \right) + G\left(m_{h^{\pm}}^2, m_{\tilde{\chi}}^2, m_{W^{\pm}}^2 \right) + s_{\beta - \alpha}^2 G\left(m_{H^{\pm}}^2, m_h^2, m_{W^{\pm}}^2 \right) \right] + c_{\beta - \alpha}^2 G\left(m_{H^{\pm}}^2, m_{H^{\pm}}^2, m_{W^{\pm}}^2 \right) + G\left(m_{H^{\pm}}^2, m_{A^{\pm}}^2, m_{W^{\pm}}^2 \right) \\ &+ c_{\beta - \alpha}^2 G\left(m_{H^{\pm}}^2, m_H^2, m_{W^{\pm}}^2 \right) + G\left(m_{H^{\pm}}^2, m_A^2, m_W^2 \right) - c_{2w}^2 G\left(m_{h^{\pm}}^2, m_{H^{\pm}}^2, m_Z^2 \right) \\ &- c_{2w}^2 G\left(m_{H^{\pm}}^2, m_{H^{\pm}}^2, m_Z^2 \right) - G\left(m_{\tilde{\eta}}^2, m_{\tilde{\chi}}^2, m_Z^2 \right) - s_{\beta - \alpha}^2 G\left(m_h^2, m_A^2, m_Z^2 \right) \\ &- c_{\beta - \alpha}^2 G\left(m_H^2, m_A^2, m_Z^2 \right) + s_{\beta - \alpha}^2 \left(\hat{G}\left(m_h^2, m_Z^2 \right) - \hat{G}\left(m_h^2, m_W^2 \right) \right) \\ \\ &+ s_{\beta - \alpha}^2 \left(\hat{G}\left(m_H^2, m_{M^{\pm}}^2, m_Z^2 \right) - \hat{G}\left(m_H^2, m_Z^2 \right) \right) \right], \end{aligned}$$

$$(5.6.9c)$$

see section 4.5.6 for the functions. The electroweak oblique parameters of the R-II-1a model resemble the case of the 2HDM model [81].

Chapter 6

Numerical Analysis

6.1 General Approach

After expressing interactions via physical scalar states in previous chapters, we proceed with numerical evaluation of the C-III-c- ν^2 and the R-II-1a models. We use Mathematica for the spectrum generator. The main input is specified in terms of the physical scalar masses and additional parameters based on the considered model. For the C-III-c- ν^2 model the following input is used:

$$\left\{ m_{H^{\pm}}, m_{S^{\pm}}, m_{H_2}, m_{H_3}, m_{S_1}, m_{S_2}, \mu_0^2, \lambda_8 \right\},$$
(6.1.1)

and for the R-II-1a model we use:

$$\{m_{H^{\pm}}, m_{h^{\pm}}, m_{H}, m_{A}, m_{\tilde{n}}, m_{\tilde{\chi}}, \beta, \alpha\}.$$
 (6.1.2)

Both cases result in an \mathbb{R}^8 surface. We do not treat the SM-like Higgs boson mass as a free parameter, it is fixed at the value $m_{H_1} = m_h = 125.09$ GeV. The mass parameters are assumed to be in the range $m_{\xi} = [0.1, 1]$ TeV.

The spectrum generator outputs data of the scalar and fermion sectors based on several checks:

- Stability of the scalar potential;
- Tree-level unitarity;
- Soft perturbativity and the more severe perturbativity conditions based on the quartic scalarscalar interactions;
- The electroweak oblique parameters;
- Limitations from the CKM matrix;

The C-III-c- ν^2 model constraints were discussed in section 4.5 and the R-II-1a model constraints in section 5.6. For the electroweak oblique parameters we apply direct constraints from the Gfitter group (4.5.39) without the correlation coefficients. Although the absolute value of $V_{\rm CKM}$ is a wellknown quantity, neither the C-III-c- ν^2 nor the R-II-1a models result in realistic cases. Since we are solely interested in a possible DM candidate, and not a truly realistic fermionic sector, the off-diagonal couplings should not play a significant role. For the charged scalar decays we assume that the CKM matrix is approximated by the identity matrix, $V_{\rm CKM} = \mathcal{I}_3$, and for the decays involving W^{\pm} the standard $V_{\rm CKM}$ [29] is used. The general approach is presented in Figure 6.1.

For the relic density evaluation we use micrOMEGAs [82–84]. The freeze-out scenario is considered, the three-body final states are computed for annihilation processes only, VW/VZdecay = 1, the effective vertices $\varphi_i gg$ and $\varphi_i \gamma \gamma$ are not considered. We do not focus on the details of the decays involving the DM. In order to use micrOMEGAs, all of the vertices should be specified. This is not a trivial task. We use SARAH [85, 86] to produce CalcHEP [87] model files that can be subsequently used by micrOMEGAs. The relic density is compared against the Planck [88] results $\Omega_{\rm CDM}h^2 = 0.120 \pm 0.001$. The PDG [29] provides a value $\Omega_{\rm CDM}h^2 = 0.1186 \pm 0.0020$ based



Figure 6.1: The general algorithm of the spectrum generator and additional checks. The "(0/1)" of the spectrum generator indicates the fail/pass switch.

on Planck TT+lowP+lensing and $\Omega_{\rm CDM}h^2 = 0.1184 \pm 0.0012$ based on Planck TT+lowP+lensing+ext. Earlier observations from WMAP [89] resulted in $\Omega_{\rm CDM}h^2 = 0.1147 \pm 0.0051$. We consider a broader acceptable relic density range, $\Omega_{\rm CDM}h^2 = 0.12 \pm 0.01$.

We compare our models against some of the available Higgs boson experimental results. The micrOMEGAs code enables additional comparison against experimental constraints using the HiggsBounds [90-92] code.

The following versions of the codes are used:

- SARAH 4.14.1¹;
- micrOMEGAs 5.0.8;
- HiggsBounds 5.3.2;

¹We note that as of the current version there is a bug in the function CalcHepVertex of the file Package/Outputs/calchep.m when dealing with fields with complex phases and exporting a model using the MakeCHep function.

The full analysis of the results is beyond the scope of the thesis and therefore preliminary results are presented in Figures 6.2 to 6.8. The data should be further constrained.



6.2 The C-III-c- ν^2 Model

Figure 6.2: General output of the spectrum generator for the C-III-c- ν^2 model. Scatter plots of different parameters are presented: a) masses of the neutral states $H_2 - H_3$, b) masses of the neutral states $S_1 - S_2$, c) masses of the charged states $H^{\pm} - S^{\pm}$, d) angles $\sigma - \alpha$. The boxplot with whiskers indicates where 1/4 - 3/4 of the data points are situated along with the medians.

We present the general output of the spectrum generator in Figure 6.2. From the spectrum generator we can extract the following information on masses:

$$\max (m_{H_2}) = 688 \text{ GeV},$$
$$\max (m_{H_3}) = 697 \text{ GeV},$$
$$\max (m_{H^{\pm}}) = 758 \text{ GeV},$$
$$\max (\{m_{S_1}, m_{S_2}, m_{S^{\pm}}\}) = 1000 \text{ GeV},$$

and the angles in radians lie in:

$$\begin{array}{l} 0.159\pi <\!\! \sigma < 0.498\pi, \\ 0 <\!\! \alpha < 0.496\pi. \end{array}$$

The quartic couplings lie in the following ranges:

0.007

$$0.006 < <_{1}/\pi < 1.590,$$

$$0.026 <_{2}/\pi < 1.510,$$

$$-0.744 <_{3}/\pi < 1.011,$$

$$-1.271 <_{5}/\pi < 4,$$

$$-3.715 <_{6}/\pi < 4,$$

$$0 <_{7}/\pi < 1.918,$$

$$0 <_{8}/\pi < \frac{2}{3}.$$

Figure 6.3: Output of the spectrum generator for the C-III-c- ν^2 model based on the SM-like limit and the V_{CKM} absolute value. Scatter plots of different parameters are presented: a) masses of the neutral states $H_2 - H_3$, b) masses of the neutral states $S_1 - S_2$, c) masses of the charged states $H^{\pm} - S^{\pm}$, d) angles $\sigma - \alpha$. The boxplot with whiskers indicates where 1/4 - 3/4 of the data points are situated along with the medians.

Not all of the points of Figure 6.2 are within an acceptable range. The α angle is fixed by the SM-like limit, see section 4.5.2, and the overall phase σ by V_{CKM} (4.3.32). After fixing the angles in the following range: $c_{\alpha} \geq 0.9$ and $\sigma = [0.34\pi, 0.38\pi]$, the m_{H_2} - m_{H_3} distribution is changed drastically, see Figure 6.3.

We assume that the DM candidate is the scalar S_1 . After scanning for an acceptable range of the relic density $\Omega_{\rm CDM}h^2$ with micrOMEGAs, we found that the annihilation channels are too efficient and neither of the Yukawa models resulted in an acceptable $\Omega_{\rm CDM}h^2$ value. Moreover, the numerical value of the relic density is several orders of magnitudes lower than the acceptable one, $(\Omega_{\rm CDM}h^2)_{\rm C-III-c-\nu^2} < 10^{-3}$. As mentioned in section 4.3, the FCNC are also way too high. The branching ratios of the SM-like Higgs boson H_1 and of the off-diagonal fermion interactions are:

$$Y_{524}^{u}Y_{542}^{d}: \operatorname{Br}(H_{1} \to \bar{f}_{i}f_{j}) \approx 10^{-3},$$

$$Y_{542}^{u}Y_{524}^{d}: \operatorname{Br}(H_{1} \to \bar{f}_{i}f_{j}) \approx 10^{-2},$$
(6.2.1)

and the leading decay channels of the other neutral active scalars are^2 :

$$\begin{array}{ll}
Y_{524}^{u}Y_{542}^{d}: & Y_{542}^{u}Y_{524}^{d}: \\
H_{2} \to (db, \, cc, \, \mu\tau) & H_{2} \to (sb, \, uc, \, \mu\tau) \\
H_{3} \to (H^{\pm}W^{\mp}, \, H_{2}Z, \, ct) & H_{3} \to (H^{\pm}W^{\mp}, \, H_{2}Z, \, ut)
\end{array}$$
(6.2.2)



Figure 6.4: Output of the spectrum generator for the C-III-c- ν^2 model based on the SM-like limit and the $V_{\rm CKM}$ absolute value when the masses of the inert doublet h_S are allowed to be lower than 100 GeV. Scatter plots of different parameters are presented: a) masses of the neutral states $H_2 - H_3$, b) masses of the neutral states $S_1 - S_2$, c) masses of the charged states $H^{\pm} - S^{\pm}$, d) angles $\sigma - \alpha$. The boxplot with whiskers indicates where 1/4 - 3/4 of the data points are situated along with the medians.

We performed an additional scan of the data presented in Figure 6.2, when the angles α and β are not fixed. This did not result in a positive result from micrOMEGAs. Therefore, an additional scan of the area $m_{S_1} = [10, 100]$ GeV was performed, the corresponding masses of the inert doublet h_S were also allowed in the sub-100 GeV region. The scanned area can be seen in Figure 6.4. This resulted in an acceptable $\Omega_{\rm CDM}h^2$ value. Nevertheless, this brings another issue: in many cases the primary decay channel of the SM-like Higgs boson H_1 is ${\rm Br}(H_1 \to S_1S_1) \approx (8-10) \times 10^{-1}$.

²The mass parameters of the scalars should be considered as not all of the decay channels are on-shell at a given scalar mass. The "bars" of the fermions \bar{f}_i are dropped. Branching ratios are within one order of magnitude.

This area corresponds to the region $m_{S_1} = [10, 60)$ of Figure 6.5b. Only the right-side blob of Figure 6.5b may result in more acceptable SM-like Higgs boson decays.

We were not able to discriminate the scanned area based on different Yukawa models, *i.e.*, models $Y_{524}^u Y_{542}^d Y_{425}^e$ and $Y_{524}^u Y_{542}^d Y_{524}^e$, and therefore both models are incorporated in a single scatter plot of Figure 6.5. The HiggsBounds result is not considered as the model did not result in acceptable decay channels. Several possible DM candidates are presented in Table 6.1.



Figure 6.5: Constrained benchmark points of Figure 6.4 based on the acceptable $\Omega_{\text{CDM}}h^2$ parameter range after performing a scan with micrOMEGAs. Yukawa models $Y_{524}^u Y_{542}^d Y_{425}^e$ and $Y_{524}^u Y_{542}^d Y_{524}^e$ are considered. Scatter plots of different parameters are presented: a) masses of the neutral states $H_2 - H_3$, b) masses of the neutral states $S_1 - S_2$, c) masses of the charged states $H^{\pm} - S^{\pm}$, d) angles $\sigma - \alpha$. The boxplot with whiskers indicates where 1/4 - 3/4 of the data points are situated along with the medians.

Benchmark point	Ωh^2	m_{H_2}	m_{H_3}	$m_{H^{\pm}}$	m_{S_1}	m_{S_2}	$m_{S^{\pm}}$	σ	α
A ₁	0.1207	168.39	218.56	270.10	82.30	121.49	179.26	1.0493	0.4336
A ₂	0.1202	149.42	427.74	376.22	90.82	180.14	265.60	1.0214	0.0190
A ₃	0.1199	153.51	320.50	186.42	73.61	132.40	181.51	1.0126	0.1228
B ₁	0.1208	149.42	427.74	376.22	90.82	180.44	265.60	1.0214	0.0596
B_2	0.1198	139.18	493.33	135.22	86.46	230.50	270.44	1.1331	0.0327
B ₃	0.1999	160.47	213.48	291.10	79.80	131.95	100.43	1.0835	0.3962

Benchmark point	$\Gamma(H_1)$	$\Gamma(H_2)$	$\Gamma(H_3)$	$\Gamma(H^{\pm})$	$\Gamma(S_2)$	$\Gamma(S^{\pm})$
A ₁	3.2×10^{-3}	3.23×10^{-3}	1.18×10^0	6.43×10^{-1}	6.80×10^{-5}	1.45×10^{-1}
A ₂	3.35×10^{-3}	$3.00 imes 10^{-3}$	$2.40 imes 10^1$	8.55×10^{0}	8.22×10^{-3}	$2.87 imes 10^0$
A ₃	3.36×10^{-3}	3.13×10^{-3}	$1.13 imes 10^1$	1.86×10^{-4}	4.83×10^{-4}	$3.00 imes 10^{-1}$
B ₁	3.52×10^{-3}	3.00×10^{-3}	2.40×10^1	8.55×10^{0}	8.22×10^{-3}	2.87×10^0
B ₂	3.04×10^{-3}	2.22×10^{-3}	$9.54 imes 10^1$	7.59×10^{-5}	$1.07 imes 10^0$	$3.27 imes 10^0$
B ₃	3.05×10^{-3}	2.87×10^{-3}	9.70×10^{-1}	1.75×10^0	3.33×10^{-4}	3.55×10^{-6}

Table 6.1: Some benchmark points. The mass parameters m_{ξ} and the total decay width $\Gamma(\xi)$ are given in GeV. The σ and α are given in radians. The benchmark points A_i indicate the $Y_{524}^u Y_{542}^d Y_{425}^e$ Yukawa model and B_i - $Y_{524}^u Y_{542}^d Y_{524}^e$. In the SM, the Higgs boson total width is $\Gamma(m_{h_{\rm SM}}) = 4.2 \times 10^{-3}$ GeV. The SM-like Higgs boson particle is the scalar H_1 .

Relative annihilation channel contributions to the relic density $\Omega_{\rm CDM} h^2$ in per cents are:

 $A_1:$ $51\% S_1 S_1 \to W^+ W^ 48\% S_1 S_1 \to W^+ W^ 30\% S_1 S_1 \rightarrow \overline{b} b$ $29\% S_1 S_1 \rightarrow \overline{b} b$ $22\% S_1 S_1 \to \bar{b} d$ $16\% S_1 S_1 \rightarrow Z Z$ $4\% S_1 S_1 \to \bar{t} c$ $6\% S_1 S_1 \rightarrow Z Z$ $3\% S_1 S_1 \rightarrow \bar{c} c$ $6\% S_1 S_1 \rightarrow \bar{t} c$ $2\% S_1 S_1 \to \tau \tau$ $2\% S_1 S_1 \to \tau \tau$ $B_2:$ $A_2:$ $48\% S_1 S_1 \to W^+ W^ 67\% S_1 S_1 \to W^+ W^ 29\% S_1 S_1 \rightarrow \overline{b} b$ $23\% S_1 S_1 \rightarrow \overline{b} b$ $7\% S_1 S_1 \rightarrow Z Z$ $15\% S_1 S_1 \rightarrow Z Z$ $2\% S_1 S_1 \rightarrow \bar{c} c$ $4\% S_1 S_1 \rightarrow \bar{t} c$ $3\% S_1 S_1 \rightarrow \bar{c} c$ $1\% S_1 S_1 \to \tau \tau$ $2\% S_1 S_1 \rightarrow \tau \tau$ $B_3:$ $A_3:$ $63\% S_1 S_1 \to W^+ W^ 80\% S_1 S_1 \to W^+ W^ 21\% S_1 S_1 \rightarrow \overline{b} b$ $\begin{array}{c} 12\% \ S_1 \ S_1 \ S_1 \rightarrow Z \ Z \\ 7\% \ S_1 \ S_1 \ S_1 \rightarrow \bar{b} \ b \end{array}$ $10\% S_1 S_1 \rightarrow \bar{b} d$ $2\% S_1 S_1 \rightarrow Z Z$

 $1\% S_1 S_1 \rightarrow \tau \tau$

 $B_1:$

Figure 6.6: General output of the spectrum generator for the R-II-1a model. Scatter plots of different input parameters are presented: a) masses of the neutral states $\tilde{\eta} - \tilde{\chi}$, b) masses of the neutral states H-A, c) masses of the charged states $h^{\pm} - H^{\pm}$, d) angles $\beta - \alpha$. The boxplot with whiskers indicates where 1/4 - 3/4 of the data points are situated along with the medians.

We present the general output of the spectrum generator in Figure 6.6. From the spectrum generator we can extract the following information on masses:

$$\begin{array}{l} \max{(m_{H})} = 652 \ {\rm GeV}, \\ \max{(m_{A})} = 687 \ {\rm GeV}, \\ \max{(m_{H^{\pm}})} = 704 \ {\rm GeV}, \\ \max{(m_{\tilde{\eta}})} = 526 \ {\rm GeV}, \\ \max{(m_{\tilde{\chi}})} = 790 \ {\rm GeV}, \\ \max{(m_{h^{\pm}})} = 792 \ {\rm GeV}, \end{array}$$

and the angles in radians lie in:

Figure 6.7: Output of the spectrum generator for the R-II-1a model based on additional criteria from the $\Omega_{\text{CDM}}h^2$ parameter and the SM-like limit. Scatter plots of different input parameters are presented: a) masses of the neutral states $\tilde{\eta} - \tilde{\chi}$, b) masses of the neutral states H - A, c) masses of the charged states $h^{\pm} - H^{\pm}$, d) angles $\beta - \alpha$. The boxplot with whiskers indicates where 1/4 - 3/4 of the data points are situated along with the medians.

The quartic couplings lie in the following ranges:

$$\begin{split} 0 &< \lambda_1/\pi < 2.254, \\ -1.082 &< \lambda_2/\pi < 1.875, \\ -1.859 &< \lambda_3/\pi < 0.623, \\ 0 &< \lambda_4/\pi < 0.685, \\ -1.651 &< \lambda_5/\pi < 4, \\ -3.850 &< \lambda_6/\pi < 2.326, \\ -1.159 &< \lambda_7/\pi < 0.328, \\ 0 &< \lambda_8/\pi < 1.911. \end{split}$$

We assume that the DM candidate is the scalar $\tilde{\eta}$. The relic density $\Omega_{\rm CDM}h^2$ of the benchmark points represented in Figure 6.6 was evaluated using micrOMEGAs. Only a specific range of points resulted in acceptable values of $\Omega_{\rm CDM}h^2$. Based on the result from micrOMEGAs and the SM-limit, the spectrum generator was tweaked appropriately. The new benchmark points are presented in Figure 6.7. Those were further studied using the micrOMEGAs and HiggsBounds codes. As it turned out, the R-II-1a model with an additional \mathbb{Z}_2 symmetry of the Yukawa couplings resulted in a higher total width of the SM-like Higgs boson. The SM predicts the total width of the Higgs boson $\Gamma(m_{h_{\rm SM}}) = 4.2 \times 10^{-3}$ GeV, while the typical value of R-II-1a model in the SM-like limit is $\Gamma(m_h) \approx 6 \times 10^{-3}$ GeV. We did not manage to find a single point consistent with the HiggsBounds check. Therefore, we present the result based on just the relic density $\Omega_{\rm CDM}h^2$ scan in Figure 6.8.

Benchmark point	Ωh^2	m_H	m_A	$m_{H^{\pm}}$	$m_{ ilde\eta}$	$m_{ ilde{\chi}}$	$m_{h^{\pm}}$	β	α
A ₁	0.1200	389.74	344.11	299.77	104.02	348.97	376.15	1.0169	1.0370
A ₂	0.1202	427.46	371.65	372.9	106.433	344.80	370.03	0.9862	0.9940
A ₃	0.1192	389.91	347.39	196.66	100.97	312.66	469.88	0.9152	1.0350
A_4	0.1205	407.49	515.36	510.42	115.45	363.19	408.44	0.9511	1.0793
A_5	0.1200	445.71	266.09	349.77	115.03	376.74	432.53	0.9312	1.0130

Several benchmark points are presented in Table 6.2.

Benchmark point	$\Gamma(h)$	$\Gamma(H)$	$\Gamma(A)$	$\Gamma(H^{\pm})$	$\Gamma(ilde{\chi})$	$\Gamma(h^{\pm})$
A ₁	6.27×10^{-3}	1.92×10^1	9.05×10^{-3}	3.02×10^0	8.19×10^0	$1.19 imes 10^1$
A_2	6.68×10^{-3}	2.44×10^1	$3.51 imes 10^0$	$5.94 imes 10^0$	$7.63 imes 10^0$	$1.10 imes 10^1$
A_3	6.71×10^{-3}	3.04×10^1	6.41×10^0	3.49×10^{-1}	3.70×10^1	7.09×10^0
A_4	6.49×10^{-3}	2.33×10^1	2.04×10^1	1.31×10^1	8.97×10^{0}	1.54×10^{1}
A_5	6.46×10^{-3}	3.45×10^1	1.46×10^{-2}	6.60×10^0	1.06×10^1	1.93×10^1

Table 6.2: Some benchmark points. The mass parameters m_{ξ} and the total decay width $\Gamma(\xi)$ are given in GeV. The β and α are given in radians.

Relative annihilation channel contributions to the relic density $\Omega_{\rm CDM}h^2$ in per cents are:

 $A_1:$ $A_4:$ $\begin{array}{l} 51\% \hspace{0.1cm} \tilde{\eta} \hspace{0.1cm} \tilde{\eta} \rightarrow W^{+} \hspace{0.1cm} W^{-} \\ 17\% \hspace{0.1cm} \tilde{\eta} \hspace{0.1cm} \tilde{\eta} \rightarrow \bar{c} \hspace{0.1cm} c \\ 15\% \hspace{0.1cm} \tilde{\eta} \hspace{0.1cm} \tilde{\eta} \rightarrow \bar{u} \hspace{0.1cm} u \\ 15\% \hspace{0.1cm} \tilde{\eta} \hspace{0.1cm} \tilde{\eta} \rightarrow Z \hspace{0.1cm} Z \end{array}$

 $| 43\% \ \tilde{\eta} \ \tilde{\eta} \rightarrow W^+ W^ 25\% \ \tilde{\eta} \ \tilde{\eta} \to Z Z$ 11% $\tilde{\eta}\,\tilde{\eta} \to \bar{c}\,c$ $11\% \ \tilde{\eta} \ \tilde{\eta} \to h h$ $9\% \ \tilde{\eta} \ \tilde{\eta} \to \bar{u} \ u$

 $10\% \ \tilde{\eta} \ \tilde{\eta} \to \bar{u} u$

 A_2 :

$52\% \ \tilde{\eta} \ \tilde{\eta} \to W^+ W^-$	$A_5:$	
19% $\tilde{\eta} \tilde{\eta} \to Z Z$	$43\% \ \tilde{\eta} \ \tilde{\eta} \to W^+ W$	r —
$14\%~\tilde\eta\tilde\eta\to\bar cc$	$25\%~\tilde{\eta}\tilde{\eta} \to ZZ$	
14% $\tilde{\eta} \tilde{\eta} \to \bar{u} u$	$12\%~\tilde{\eta}\tilde{\eta}\to hh$	
	$10\% \; \tilde{\eta} \tilde{\eta} \to \bar{c} c$	

 $A_3:$

 $\begin{array}{c} 72\% \hspace{0.1cm} \tilde{\eta} \hspace{0.1cm} \tilde{\eta} \rightarrow W^{+} \hspace{0.1cm} W^{-} \\ 10\% \hspace{0.1cm} \tilde{\eta} \hspace{0.1cm} \tilde{\eta} \rightarrow \bar{c} \hspace{0.1cm} c \\ 9\% \hspace{0.1cm} \tilde{\eta} \hspace{0.1cm} \tilde{\eta} \rightarrow Z \hspace{0.1cm} Z \\ 8\% \hspace{0.1cm} \tilde{\eta} \hspace{0.1cm} \tilde{\eta} \rightarrow \bar{u} \hspace{0.1cm} u \\ 1\% \hspace{0.1cm} \tilde{\eta} \hspace{0.1cm} \tilde{\eta} \rightarrow \bar{b} \hspace{0.1cm} b \end{array}$

Figure 6.8: Constrained benchmark points of Figure 6.7 based on the acceptable $\Omega_{\rm CDM}h^2$ parameter range after performing a scan with micrOMEGAs. Scatter plots of different input parameters are presented: a) masses of the neutral states $\tilde{\eta} - \tilde{\chi}$, b) masses of the neutral states H - A, c) masses of the charged states $h^{\pm} - H^{\pm}$, d) angles $\beta - \alpha$. The boxplot with whiskers indicates where 1/4 - 3/4 of the data points are situated along with the medians.
Chapter 7

Summary and Conclusions

7.1 Results

The motivation for this work was to gain a better understanding of the S_3 -symmetric 3HDM and test if a scalar DM is possible within the framework of the model. The results obtained are summarized as follows:

- The hidden symmetries of the S_3 -symmetric scalar potential were analysed. We were able to identify the massless states using the Goldstone theorem. Also, based on the \mathbb{Z}_2 symmetry and the mass-squared matrices mixing, the vacuum configurations, which could accommodate DM, were identified.
- One of the models, C-III-c, with massless states was studied. The principle of soft symmetry breaking was applied to promote the massless states to massive ones. The softly broken model C-III-c- ν^2 was further analysed; tree-level interactions and constraints considered. It was shown that the C-III-c- ν^2 model is CP conserving. The S_3 -symmetric Yukawa sector resulted in unrealistic $V_{\rm CKM}$ values, the FCNC are way too high. Therefore, the S_3 -symmetric Yukawa sector should be further broken. Numerical evaluation of the model resulted in realistic relic density $\Omega_{\rm CDM}h^2$ values, but the experimental constraints of the SM-like Higgs boson are violated.
- The real vacuum configuration R-II-1a has been analysed. This is the only real vacuum configuration with the \mathbb{Z}_2 symmetry preserved by default. The S_3 -symmetric Yukawa sector resulted in unrealistic V_{CKM} and thus the Yukawa couplings were further constrained by an additional \mathbb{Z}_2 symmetry. Such model resulted in no FCNC, but the decays of the SM-like Higgs boson are violated. Nevertheless, the model resulted in possible relic density $\Omega_{\text{CDM}}h^2$ values.

7.2 Future Research

As this work has shown, the S_3 -symmetric 3HDMs have several interesting properties, which have to be addressed further. A number of interesting problems were considered and some possible solutions provided. Nevertheless, only a tiny amount was covered and many questions are still open. Governed by this fact, we mention proposal for future research:

- It should be checked what happens with the additional massless states after renormalization. Of particular interest is how they act in the high-energy limit.
- Some of the models with the zero VEV components do not result in a possible DM candidate. The R-I-2b, and R-I-2c, and C-I-a models involve zero VEV components but there is mixing present between the states. It should be checked more thoroughly if there is a possible direction of the potential, which would result in inert SU(2) doublets. There are two models R-I-1 and C-I-a, which result in mass-degenerate states. The degenerate states should be further analysed to see what that implies.

- The general conditions for CP violation in the Higgs basis should be derived. This might result in several possible solutions based on the $\mathcal{R}_{\beta_i}\mathcal{R}_{\beta_j}$ orderings and if an additional U(2)transformation is considered.
- The C-III-c- ν^2 and R-II-1a models resulted in unrealistic V_{CKM}. Minimal conditions for realistic values should be further considered. Also, it was assumed that neutrinos are massless. Those should be promoted to massive particles. Moreover, the other vacuum configurations should be checked in terms of the Yukawa sector.
- Although, as shown, both the C-III-c- ν^2 and R-II-1a models are capable of producing realistic relic density $\Omega_{\rm CDM}h^2$ values and thus are viable candidates, the decay rates of the scalars were not realistic. The total width of the SM-like scalars could be enhanced when loops are considered and the effective vertices of scalars-gluons $\varphi_i gg$ and scalars-photons $\varphi_i \gamma \gamma$ are introduced. The DM decays should be analysed at the further leading orders.

Appendix A

Different Forms of the Scalar Potential

A.1 The Derman Potential

The scalar potential in terms of the S_3 -reducible-triplet fields was derived by Derman [33, 38]. He showed that the most general $SU(2) \otimes U(1) \otimes S_3$ Higgs triplet scalar potential can be written as:

$$\begin{split} V &= \sum_{i} \left[-\lambda \left(\phi_{i}^{\dagger} \phi_{i} \right) + A \left(\phi_{i}^{\dagger} \phi_{i} \right)^{2} \right] \\ &+ \sum_{i < j} \left\{ \frac{1}{2} \gamma \left(\phi_{i}^{\dagger} \phi_{j} + \text{h.c.} \right) + C \left(\phi_{i}^{\dagger} \phi_{i} \right) \left(\phi_{j}^{\dagger} \phi_{j} \right) + \bar{C} \left(\phi_{i}^{\dagger} \phi_{j} \right) \left(\phi_{j}^{\dagger} \phi_{i} \right) + \frac{1}{2} D \left[\left(\phi_{i}^{\dagger} \phi_{j} \right)^{2} + \text{h.c.} \right] \right\} \\ &+ \frac{1}{2} E_{1} \sum_{i \neq j} \left[\left(\phi_{i}^{\dagger} \phi_{i} \right) \left(\phi_{i}^{\dagger} \phi_{j} \right) + \text{h.c.} \right] \\ &+ \sum_{i \neq j \neq k \neq i, j < k} \left\{ \frac{1}{2} E_{2} \left[\left(\phi_{i}^{\dagger} \phi_{j} \right) \left(\phi_{k}^{\dagger} \phi_{i} \right) + \text{h.c.} \right] + \frac{1}{2} E_{3} \left[\left(\phi_{i}^{\dagger} \phi_{i} \right) \left(\phi_{k}^{\dagger} \phi_{j} + \text{h.c.} \right) \right] \\ &+ \frac{1}{2} E_{4} \left[\left(\phi_{i}^{\dagger} \phi_{j} \right) \left(\phi_{i}^{\dagger} \phi_{k} \right) + \text{h.c.} \right] \right\}, \end{split}$$

$$(A.1.1)$$

where all of the couplings are assumed to be real. As noted, spontaneous symmetry breaking happens provided that $\lambda > 0$.

It is also worth mentioning that Derman analysed vacua of the form

$$\langle \phi_i \rangle_{\min} = \frac{1}{\sqrt{2}} \begin{pmatrix} \sigma_i e^{i\alpha_i} \\ \rho_i e^{i\theta_i} \end{pmatrix}, \ i = \overline{1,3},$$
 (A.1.2)

where all of the parameters are real. It was noted that the scalar potential has a local minimum at $\theta_1 - \theta_2 = \theta_2 - \theta_3 = 0$, provided that the couplings γ , $\bar{C} + D$, D, E_i are all negative and thus T-invariance is obtained. Charge conservation is also ensured by choosing couplings in this way.

The Derman potential is related to the potential of eq. (2.1.7) by the following transformations:

$$\begin{pmatrix} \mu_0^2 \\ \mu_1^2 \end{pmatrix} = \frac{1}{2} \begin{pmatrix} -2 & 2 \\ -2 & -1 \end{pmatrix} \begin{pmatrix} \lambda \\ \gamma \end{pmatrix}, \tag{A.1.3a}$$

$$\begin{pmatrix} \lambda_1 \\ \lambda_2 \\ \lambda_3 \\ \lambda_4 \\ \lambda_5 \\ \lambda_6 \\ \lambda_7 \\ \lambda_8 \end{pmatrix} = \frac{1}{12} \begin{pmatrix} 4 & 4 & 1 & 1 & -4 & 1 & -2 & 1 \\ 0 & 0 & -3 & 3 & 0 & 3 & 0 & -3 \\ 2 & -1 & 2 & 2 & -2 & -1 & 2 & -1 \\ 4\sqrt{2} & -2\sqrt{2} & -2\sqrt{2} & -2\sqrt{2} & -\sqrt{2} & \sqrt{2} & \sqrt{2} \\ 8 & 8 & -4 & -4 & 4 & -4 & 2 & -4 \\ 8 & -4 & 8 & -4 & 4 & 2 & -4 & -4 \\ 4 & -2 & -2 & 4 & 2 & -2 & -2 & 1 \\ 4 & 4 & 4 & 4 & 8 & 4 & 4 & 4 \end{pmatrix} \begin{pmatrix} A \\ C \\ \bar{C} \\ D \\ E_1 \\ E_2 \\ E_3 \\ E_4 \end{pmatrix}.$$
(A.1.3b)

Taking into consideration the previously mentioned conditions by Derman, one can get the following constraints in terms of μ_i^2 and λ_i :

$$(\mu_0^2 - \mu_1^2) < 0,$$
 (A.1.4a)

$$\left(2\lambda_1 + 8\lambda_3 - 4\sqrt{2}\lambda_4 - 2\lambda_5 + \lambda_6 + 2\lambda_7 + 2\lambda_8\right) < 0, \tag{A.1.4b}$$

$$\left(\lambda_1 + 3\lambda_2 + 4\lambda_3 - 2\sqrt{2}\lambda_4 - \lambda_5 - \lambda_6 + 4\lambda_7 + \lambda_8\right) < 0, \tag{A.1.4c}$$

$$-\left(4\lambda_1 + 4\lambda_3 + \sqrt{2}\lambda_4 - \lambda_5 - \lambda_6 - 2\lambda_7 - 2\lambda_8\right) < 0, \tag{A.1.4d}$$

$$\left(2\lambda_1 + 6\lambda_2 - 4\lambda_3 + 2\sqrt{2}\lambda_4 - 2\lambda_5 + \lambda_6 - 4\lambda_7 + 2\lambda_8\right) < 0, \tag{A.1.4e}$$

$$\left(-4\lambda_1 + 8\lambda_3 + 2\sqrt{2}\lambda_4 + \lambda_5 - 2\lambda_6 - 4\lambda_7 + 2\lambda_8\right) < 0, \tag{A.1.4f}$$

$$\left(\lambda_1 - 3\lambda_2 - 2\lambda_3 + \sqrt{2}\lambda_4 - \lambda_5 - \lambda_6 + \lambda_7 + \lambda_8\right) < 0.$$
(A.1.4g)

A.2 Matrix Form

One of the possibilities is to write down the scalar potential in a matrix form, from which elements for the SU(2)-covariant form of the potential (2.5.7) can be easily extracted. By directly inspecting different combinations of the SU(2) singlets, $h_{ij} \equiv h_i^{\dagger} h_j$, we can write down the S_3 -symmetric scalar potential as:

$$V = \mathcal{H}_2 M + \mathcal{H}_4 \Lambda \mathcal{H}_4^{\mathrm{T}},\tag{A.2.1}$$

where the basis vectors are given by:

$$\mathcal{H}_4 = (h_{11} \ h_{22} \ h_{SS} \ h_{12} \ h_{1S} \ h_{21} \ h_{2S} \ h_{S1} \ h_{S2}),$$

$$\mathcal{H}_2 = \{(\mathcal{H}_4)_i : i \in [1,3]\}.$$
(A.2.2a)
(A.2.2b)

The matrices of eq. (A.2.1) can then be expressed in terms of the couplings as:

$$M = \operatorname{diag} \left(\mu_1^2, \, \mu_1^2, \, \mu_0^2 \right),$$

$$A = \begin{pmatrix} \lambda_1 + \lambda_3 & \lambda_1 - \lambda_3 & \frac{1}{2}\lambda_5 & 0 & 0 & 0 & \frac{1}{2}\lambda_4 & 0 & \frac{1}{2}\lambda_4 \\ \lambda_1 - \lambda_3 & \lambda_1 + \lambda_3 & \frac{1}{2}\lambda_5 & 0 & 0 & 0 & -\frac{1}{2}\lambda_4 & 0 & -\frac{1}{2}\lambda_4 \\ \frac{1}{2}\lambda_5 & \frac{1}{2}\lambda_5 & \lambda_8 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & \lambda_2 + \lambda_3 & \frac{1}{2}\lambda_4 & -\lambda_2 + \lambda_3 & 0 & \frac{1}{2}\lambda_4 & 0 \\ 0 & 0 & 0 & \frac{1}{2}\lambda_4 & \lambda_7 & \frac{1}{2}\lambda_4 & 0 & \frac{1}{2}\lambda_6 & 0 \\ 0 & 0 & 0 & -\lambda_2 + \lambda_3 & \frac{1}{2}\lambda_4 & \lambda_2 + \lambda_3 & 0 & \frac{1}{2}\lambda_4 & 0 \\ \frac{1}{2}\lambda_4 & -\frac{1}{2}\lambda_4 & 0 & 0 & 0 & 0 & \lambda_7 & 0 & \frac{1}{2}\lambda_6 \\ 0 & 0 & 0 & \frac{1}{2}\lambda_4 & \frac{1}{2}\lambda_6 & \frac{1}{2}\lambda_4 & 0 & \lambda_7 & 0 \\ \frac{1}{2}\lambda_4 & -\frac{1}{2}\lambda_4 & 0 & 0 & 0 & 0 & \frac{1}{2}\lambda_6 & 0 & \lambda_7 \end{pmatrix}.$$
(A.2.3)

Appendix B

Derivatives of the Potential With Respect to the Fields

B.1 The First Derivatives

In order to identify minimization conditions we consider first derivatives of the potential with respect to different fields at vacuum. One should always be careful as the set of all derivatives is not independent, but all of the derivatives must vanish simultaneously after applying the minimization conditions. We consider the following vacuum configuration:

$$\{w_1, w_2, w_S\},$$
 (B.1.1)

where, in general, w_i are complex values. The derivatives are as follows:

$$\frac{\partial V}{\partial h_i^+}\Big|_{\langle v\rangle} = \frac{\partial V}{\partial h_S^+}\Big|_{\langle v\rangle} = 0, \tag{B.1.2a}$$

$$\frac{\partial V}{\partial h_i^-}\Big|_{\langle v\rangle} = \frac{\partial V}{\partial h_S^-}\Big|_{\langle v\rangle} = 0, \tag{B.1.2b}$$

$$\begin{split} \frac{\partial V}{\partial w_1}\Big|_{\langle v \rangle} &= \frac{1}{2} \mu_1^2 w_1^* + \frac{1}{2} \lambda_1 w_1^* \left(|w_1|^2 + |w_2|^2 \right) + \frac{1}{2} \lambda_2 \left(w_1 w_2^{*2} - w_1^* |w_2|^2 \right) \\ &\quad + \frac{1}{2} \lambda_3 \left(w_1^* |w_1|^2 + w_1 w_2^{*2} \right) + \frac{1}{2} \lambda_4 \left(w_1 w_2^* w_S^* + w_1^* w_2 w_S^* + w_1^* w_2^* w_S \right) \\ &\quad + \frac{1}{4} \left(\lambda_5 + \lambda_6 \right) w_1^* |w_S|^2 + \frac{1}{2} \lambda_7 w_1 w_S^{*2} = 0, \end{split} \tag{B.1.3a} \\ &\quad + \frac{1}{4} \left(\lambda_5 + \lambda_6 \right) w_1^* |w_S|^2 + \frac{1}{2} \lambda_7 w_1 w_S^{*2} = 0, \\ &\quad + \frac{1}{2} \lambda_3 \left(w_1^{*2} w_2 + w_2^* |w_2|^2 \right) + \frac{1}{4} \lambda_4 \left[2 w_S^* \left(|w_1|^2 - |w_2|^2 \right) + w_S \left(w_1^{*2} - w_2^{*2} \right) \right] \\ &\quad + \frac{1}{4} \left(\lambda_5 + \lambda_6 \right) w_2^* |w_S|^2 + \frac{1}{2} \lambda_7 w_2 w_S^{*2} = 0, \\ &\quad + \frac{1}{4} \left(\lambda_5 + \lambda_6 \right) w_2^* |w_S|^2 + \frac{1}{2} \lambda_7 w_2 w_S^{*2} = 0, \\ &\quad + \frac{1}{2} \lambda_7 \left(w_1^{*2} + w_2^{*2} \right) w_S + \frac{1}{2} \lambda_8 w_S^* |w_S|^2 = 0, \\ &\quad + \frac{1}{2} \lambda_7 \left(w_1^{*2} + w_2^{*2} \right) w_S + \frac{1}{2} \lambda_8 w_S^* |w_S|^2 = 0, \\ &\quad + \frac{1}{2} \lambda_7 \left(w_1^{*2} + w_2^{*2} \right) w_S + \frac{1}{2} \lambda_8 w_S^* |w_S|^2 = 0, \\ &\quad + \frac{1}{2} \lambda_7 \left(w_1^{*2} + w_2^{*2} \right) w_S + \frac{1}{2} \lambda_4 \left(w_1^* w_2 w_S + w_1 w_2^* w_S + w_1 w_2 w_S^* \right) \\ &\quad + \frac{1}{4} \left(\lambda_5 + \lambda_6 \right) w_1 |w_S|^2 + \frac{1}{2} \lambda_7 w_1^* w_S^2 = 0, \end{aligned} \tag{B.1.3d} \\ &\quad + \frac{1}{4} \left(\lambda_5 + \lambda_6 \right) w_1 |w_S|^2 + \frac{1}{2} \lambda_7 w_1^* w_S^2 = 0, \end{aligned}$$

$$\frac{\partial V}{\partial w_2^*}\Big|_{\langle v\rangle} = \frac{1}{2}\mu_1^2 w_2 + \frac{1}{2}\lambda_1 w_2 \left(|w_1|^2 + |w_2|^2\right) + \frac{1}{2}\lambda_2 \left(w_1^2 w_2^* - |w_1|^2 w_2\right) \\
+ \frac{1}{2}\lambda_3 w_2^* \left(w_1^2 + w_2^2\right) + \frac{1}{4}\lambda_4 \left[2 \left(|w_1|^2 - |w_2|^2\right) w_S + \left(w_1^2 - w_2^2\right) w_S^*\right] \\
+ \frac{1}{4} \left(\lambda_5 + \lambda_6\right) w_2 |w_S|^2 + \frac{1}{2}\lambda_7 w_2^* w_S^2 = 0, \\
\frac{\partial V}{\partial w_S^*}\Big|_{\langle v\rangle} = \frac{1}{2}\mu_0^2 w_S + \frac{1}{4}\lambda_4 \left[2|w_1|^2 w_2 + w_2^* \left(w_1^2 - w_2^2\right)\right] + \frac{1}{4} \left(\lambda_5 + \lambda_6\right) w_S \left(|w_1|^2 + |w_2|^2\right) \\
+ \frac{1}{2}\lambda_7 w_S^* \left(w_1^2 + w_2^2\right) + \frac{1}{2}\lambda_8 |w_S|^2 w_S = 0,
\end{aligned}$$
(B.1.3e)
(B.1.3f)

$$\begin{aligned} \frac{\partial V}{\partial \bar{\eta}_{1}}\Big|_{\langle v \rangle} &= \frac{1}{2} \mu_{1}^{2} (w_{1}^{*} + w_{1}) + \frac{1}{2} \lambda_{1} \left[\left(|w_{1}|^{2} + |w_{2}|^{2} \right) (w_{1}^{*} + w_{1}) \right] \\ &+ \frac{1}{2} \lambda_{2} \left[w_{1} w_{2}^{*2} + w_{1}^{*} w_{2}^{2} - |w_{2}|^{2} (w_{1}^{*} + w_{1}) \right] \\ &+ \frac{1}{2} \lambda_{3} \left[w_{1}^{*} |w_{1}|^{2} + w_{1} w_{2}^{*2} + w_{1}^{*} (w_{1}^{2} + w_{2}^{2}) \right] \end{aligned} \tag{B.1.4a} \\ &+ \frac{1}{2} \lambda_{4} \left(w_{1} w_{2}^{*} w_{S}^{*} + w_{1}^{*} w_{2} w_{S}^{*} + w_{1}^{*} w_{2}^{*} w_{S} + w_{1}^{*} w_{2} w_{S} + w_{1} w_{2}^{*} w_{S} + w_{1} w_{2} w_{S} + w_{1} w_{2} w_{S} \right) \\ &+ \frac{1}{2} \lambda_{4} \left(w_{1} w_{2}^{*} w_{S}^{*} + w_{1}^{*} w_{2} w_{S}^{*} + w_{1}^{*} w_{2}^{*} w_{S} + w_{1}^{*} w_{2} w_{S} + w_{1} w_{2}^{*} w_{S} + w_{1} w_{2} w_{S}^{*} \right) \\ &+ \frac{1}{2} \lambda_{4} \left(w_{1} w_{2}^{*} w_{S}^{*} + w_{1}^{*} w_{2} w_{S}^{*} + w_{1}^{*} w_{2}^{*} w_{S} + w_{1}^{*} w_{2} w_{S} + w_{1} w_{2} w_{S}^{*} \right) \\ &+ \frac{1}{2} \lambda_{4} \left(w_{1} w_{2}^{*} w_{S}^{*} + w_{1}^{*} w_{2} w_{S}^{*} + w_{1}^{*} w_{2} w_{S} + w_{1}^{*} w_{2} w_{S} + w_{1} w_{2} w_{S}^{*} \right) \\ &+ \frac{1}{4} \left(\lambda_{5} + \lambda_{6} \right) |w_{S}|^{2} \left(w_{1}^{*} + w_{1} \right) + \frac{1}{2} \lambda_{7} \left(w_{1} w_{S}^{*}^{*} + w_{2} \right) \right] \\ &+ \frac{1}{2} \lambda_{2} \left[w_{1}^{2} w_{2}^{*} + w_{1}^{*}^{2} w_{2} - |w_{1}|^{2} \left(w_{2}^{*} + w_{2} \right) \right] \\ &+ \frac{1}{2} \lambda_{3} \left[w_{2}^{*} |w_{2}|^{2} + w_{1}^{*}^{2} w_{2} + w_{2}^{*} (w_{1}^{2} + w_{2}^{*}) \right] \\ &+ \frac{1}{4} \lambda_{4} \left[2|w_{1}|^{2} \left(w_{S}^{*} + w_{S} \right) + w_{S} \left(w_{1}^{*2} - w_{2}^{*2} - 2|w_{2}|^{2} \right) + w_{S}^{*} \left(w_{1}^{2} - w_{2}^{*} - 2|w_{2}|^{2} \right) \right] \\ &+ \frac{1}{4} \left(\lambda_{5} + \lambda_{6} \right) |w_{S}|^{2} \left(w_{2}^{*} + w_{2} \right) + \frac{1}{2} \lambda_{7} \left(w_{2} w_{S}^{*2} + w_{2}^{*} w_{S}^{*} \right) = 0, \end{aligned} \tag{B.1.4b}$$

$$\frac{\partial V}{\partial \tilde{\eta}_S}\Big|_{\langle v \rangle} = \frac{1}{2}\mu_0^2 (w_S^* + w_S) + \frac{1}{4}\lambda_4 \left(w_1^2 w_2^* + w_1^{*2} w_2 + \left(2|w_1|^2 - |w_2|^2\right) (w_2^* + w_2)\right) \\
+ \frac{1}{4} \left(\lambda_5 + \lambda_6\right) \left(|w_1|^2 + |w_2|^2\right) (w_S^* + w_S) + \frac{1}{2}\lambda_7 \left[w_S^* \left(w_1^2 + w_2^2\right) + w_S \left(w_1^{*2} + w_2^{*2}\right)\right] \\
+ \frac{1}{2}\lambda_8 |w_S|^2 \left(w_S^* + w_S\right) = 0,$$
(B.1.4c)

$$\frac{\partial V}{\partial \tilde{\chi}_{1}}\Big|_{\langle v \rangle} = \frac{i}{2} \mu_{1}^{2} (w_{1}^{*} - w_{1}) + \frac{i}{2} \lambda_{1} \left[\left(|w_{1}|^{2} + |w_{2}|^{2} \right) (w_{1}^{*} - w_{1}) \right] \\
+ \frac{i}{2} \lambda_{2} \left[w_{1} w_{2}^{*2} - w_{1}^{*} w_{2}^{2} - |w_{2}|^{2} (w_{1}^{*} - w_{1}) \right] \\
+ \frac{i}{2} \lambda_{3} \left[w_{1}^{*} |w_{1}|^{2} + w_{1} w_{2}^{*2} - w_{1}^{*} (w_{1}^{2} + w_{2}^{2}) \right] \\
+ \frac{i}{2} \lambda_{4} \left(w_{1} w_{2}^{*} w_{S}^{*} + w_{1}^{*} w_{2} w_{S}^{*} + w_{1}^{*} w_{2}^{*} w_{S} - w_{1}^{*} w_{2} w_{S} - w_{1} w_{2} w_{S}^{*} - w_{1} w_{2} w_{S}^{*} \right) \\
+ \frac{i}{4} \left(\lambda_{5} + \lambda_{6} \right) |w_{S}|^{2} \left(w_{1}^{*} - w_{1} \right) + \frac{i}{2} \lambda_{7} \left(w_{1} w_{S}^{*2} - w_{1}^{*} w_{S}^{2} \right) = 0,$$
(B.1.5a)

$$\frac{\partial V}{\partial \tilde{\chi}_{2}}\Big|_{\langle v \rangle} = \frac{i}{2} \mu_{1}^{2} (w_{2}^{*} - w_{2}) + \frac{i}{2} \lambda_{1} \left[\left(|w_{1}|^{2} + |w_{2}|^{2} \right) (w_{2}^{*} - w_{2}) \right] \\
+ \frac{i}{2} \lambda_{2} \left[-w_{1}^{2} w_{2}^{*} + w_{1}^{*2} w_{2} - |w_{1}|^{2} (w_{2}^{*} - w_{2}) \right] \\
+ \frac{i}{2} \lambda_{3} \left[w_{2}^{*} |w_{2}|^{2} + w_{1}^{*2} w_{2} - w_{2}^{*} (w_{1}^{2} + w_{2}^{2}) \right] \\
- \frac{i}{4} \lambda_{4} \left[2|w_{1}|^{2} (w_{S}^{*} - w_{S}) + w_{S} \left(w_{1}^{*2} - w_{2}^{*2} + 2|w_{2}|^{2} \right) + w_{S}^{*} \left(w_{1}^{2} - w_{2}^{2} + 2|w_{2}|^{2} \right) \right] \\
+ \frac{i}{4} \left(\lambda_{5} + \lambda_{6} \right) |w_{S}|^{2} \left(w_{2}^{*} - w_{2} \right) + \frac{1}{2} \lambda_{7} \left(w_{2} w_{S}^{*2} - w_{2}^{*} w_{S}^{2} \right) = 0,$$
(B.1.5b)

$$\frac{\partial V}{\partial \tilde{\chi}_S}\Big|_{\langle v \rangle} = \frac{i}{2} \mu_0^2 (w_S^* - w_S) + \frac{i}{4} \lambda_4 \left[-w_1^2 w_2^* + w_1^{*2} w_2 + \left(2|w_1|^2 - |w_2|^2 \right) (w_2^* - w_2) \right] \\
+ \frac{i}{4} \left(\lambda_5 + \lambda_6 \right) \left(|w_1|^2 + |w_2|^2 \right) (w_S^* - w_S) + \frac{i}{2} \lambda_7 \left[-w_S^* \left(w_1^2 + w_2^2 \right) + w_S \left(w_1^{*2} + w_2^{*2} \right) \right] \\
+ \frac{i}{2} \lambda_8 |w_S|^2 \left(w_S^* - w_S \right) = 0.$$
(B.1.5c)

In case of the real vacuum configuration the derivatives are of a different form. Nevertheless, one should realize that it is just a matter of the prefactor, which is a constant. Thus derivatives can be divided by it without leading to another solution.

Next, we consider another possible description of the vacuum configuration by explicitly splitting VEVs into a real part \hat{w}_i and a complex phase σ_i :

$$\{\hat{w}_1 e^{i\sigma_1}, \hat{w}_2 e^{i\sigma_2}, \hat{w}_S\}.$$
 (B.1.6)

The only significant change comes from eq. (B.1.3):

$$\frac{\partial V}{\partial \hat{w}_{1}}\Big|_{\langle v \rangle} = \mu_{1}^{2} \hat{w}_{1} + \lambda_{1} \hat{w}_{1} \left(\hat{w}_{1}^{2} + \hat{w}_{2}^{2} \right) - 2\lambda_{2} \hat{w}_{1} \hat{w}_{2}^{2} s_{\sigma_{1}-\sigma_{2}}^{2} + \lambda_{3} \hat{w}_{1} \left(\hat{w}_{1}^{2} + \hat{w}_{2}^{2} c_{2(\sigma_{1}-\sigma_{2})} \right)
+ \lambda_{4} \hat{w}_{1} \hat{w}_{2} \hat{w}_{S} \left(c_{(2\sigma_{1}-\sigma_{2})} + 2c_{\sigma_{2}} \right) + \frac{1}{2} \left(\lambda_{5} + \lambda_{6} \right) \hat{w}_{1} \hat{w}_{S}^{2} + \lambda_{7} \hat{w}_{1} \hat{w}_{S}^{2} c_{2\sigma_{1}} = 0,$$

$$\frac{\partial V}{\partial \hat{w}_{2}} \Big|_{\langle v \rangle} = \mu_{1}^{2} \hat{w}_{2} + \lambda_{1} \hat{w}_{2} \left(\hat{w}_{1}^{2} + \hat{w}_{2}^{2} \right) - 2\lambda_{2} \hat{w}_{1}^{2} \hat{w}_{2} s_{\sigma_{1}-\sigma_{2}}^{2} + \lambda_{3} \hat{w}_{2} \left(\hat{w}_{1}^{2} c_{2(\sigma_{1}-\sigma_{2})} + \hat{w}_{2}^{2} \right)
+ \frac{1}{2} \lambda_{4} \hat{w}_{S} \left[\hat{w}_{1}^{2} c_{(2\sigma_{1}-\sigma_{2})} + \left(2 \hat{w}_{1}^{2} - 3 \hat{w}_{2}^{2} \right) c_{\sigma_{2}} \right] + \frac{1}{2} \left(\lambda_{5} + \lambda_{6} \right) \hat{w}_{2} \hat{w}_{S}^{2} + \lambda_{7} \hat{w}_{2} \hat{w}_{S}^{2} c_{2\sigma_{2}} = 0,$$

$$(B.1.7b)$$

$$\frac{\partial V}{\partial \hat{w}_S}\Big|_{\langle v \rangle} = \mu_0^2 \hat{w}_S + \frac{1}{2} \lambda_4 \hat{w}_2 \left[\hat{w}_1^2 c_{(2\sigma_1 - \sigma_2)} + \left(2\hat{w}_1^2 - \hat{w}_2^2 \right) c_{\sigma_2} \right] \\
+ \frac{1}{2} \left(\lambda_5 + \lambda_6 \right) \left(\hat{w}_1^2 + \hat{w}_2^2 \right) \hat{w}_S + \lambda_7 \hat{w}_S \left[\hat{w}_1^2 c_{2\sigma_1} + \hat{w}_2^2 c_{2\sigma_2} \right] + \lambda_8 \hat{w}_S^3 = 0,$$
(B.1.7c)

$$\frac{\partial V}{\partial \sigma_1}\Big|_{\langle v\rangle} = -(\lambda_2 + \lambda_3) \,\hat{w}_1^2 \hat{w}_2^2 \mathbf{s}_{2(\sigma_1 - \sigma_2)} - \lambda_4 \hat{w}_1^2 \hat{w}_2 \hat{w}_S \mathbf{s}_{(2\sigma_1 - \sigma_2)} - \lambda_7 \hat{w}_1^2 \hat{w}_S^2 \mathbf{s}_{\sigma_1} = 0, \tag{B.1.7d}$$

$$\frac{\partial V}{\partial \sigma_2}\Big|_{\langle v\rangle} = \left(\lambda_2 + \lambda_3\right) \hat{w}_1^2 \hat{w}_2^2 \mathbf{s}_{2(\sigma_1 - \sigma_2)} + \frac{1}{2} \lambda_4 \hat{w}_2 \hat{w}_S \left[\hat{w}_1^2 \mathbf{s}_{(2\sigma_1 - \sigma_2)} - \left(2\hat{w}_1^2 - \hat{w}_2^2\right) \mathbf{s}_{\sigma_2} \right] - \lambda_7 \hat{w}_2^2 \hat{w}_S^2 \mathbf{s}_{2\sigma_2} = 0.$$
(B.1.7e)

All the other derivatives are of the same form except for the fact that they are not expanded in terms of the absolute value \hat{w}_i and the overall phase $e^{i\sigma_i}$.

Appendix C

Potential in the Higgs Basis

C.1 $\mathcal{R}_{\beta_z} \mathcal{R}_{\beta_x}$ Higgs Basis Rotation

We present relations between the couplings in the Higgs basis and the generic basis of eqs. (2.5.9, 2.5.10) using transformations givens by eq. (2.5.14). The quadratic couplings are:

$$Y_{11} = \mu_0^2 \mathbf{s}_{\beta_1}^2 \mathbf{s}_{\beta_2}^2 + \mu_1^2 \left(\mathbf{c}_{\beta_1}^2 + \mathbf{s}_{\beta_1}^2 \mathbf{c}_{\beta_2}^2 \right), \qquad (C.1.1a)$$

$$Y_{12} = \left(\mu_0^2 - \mu_1^2\right) \mathbf{s}_{\beta_1} \mathbf{c}_{\beta_1} \mathbf{s}_{\beta_2}^2, \tag{C.1.1b}$$

$$Y_{13} = \left(\mu_0^2 - \mu_1^2\right) \mathbf{s}_{\beta_1} \mathbf{c}_{\beta_2} \mathbf{s}_{\beta_2},\tag{C.1.1c}$$

$$Y_{22} = \mu_0^2 c_{\beta_1}^2 s_{\beta_2}^2 + \mu_1^2 \left(s_{\beta_1}^2 + c_{\beta_1}^2 c_{\beta_2}^2 \right), \qquad (C.1.1d)$$

$$Y_{23} = \left(\mu_0^2 - \mu_1^2\right) c_{\beta_1} c_{\beta_2} s_{\beta_2}, \tag{C.1.1e}$$

$$Y_{33} = \mu_0^2 c_{\beta_2}^2 + \mu_1^2 s_{\beta_2}^2. \tag{C.1.1f}$$

The soft symmetry breaking terms of eq. (3.2.2) result in a change of parameters in the Higgs basis:

$$Y_{11}' = Y_{11} + \frac{1}{2}\nu^2 c_{\beta_2} c_{\sigma_1 - \sigma_2} s_{2\beta_1} + \mu_2^2 \left(c_{\beta_1}^2 - c_{\beta_2}^2 s_{\beta_1}^2 \right) + \mu_3^2 c_{\beta_1} c_{\sigma_1} s_{\beta_1} s_{\beta_2} + \mu_4^2 c_{\beta_2} c_{\sigma_2} s_{\beta_1}^2 s_{\beta_2}, \quad (C.1.2a)$$

$$Y_{12}' = Y_{12} + \frac{1}{4}\nu^2 c_{\beta_2} \left(c_{2\beta_1 + \sigma_1 - \sigma_2} + c_{2\beta_1 - \sigma_1 + \sigma_2} - 2is_{\sigma_1 - \sigma_2} \right) - \frac{1}{4}\mu_2^2 \left(c_{2\beta_2} + 3 \right) s_{2\beta_1} + \frac{1}{2}\mu_3^2 s_{\beta_2} \left(c_{2\beta_1} c_{\sigma_1} - is_{\sigma_1} \right) + \mu_4^2 c_{\beta_1} c_{\beta_2} c_{\sigma_2} s_{\beta_1} s_{\beta_2},$$
(C.1.2b)

$$Y_{13}' = Y_{13} - \frac{1}{2}\nu^2 e^{-i(\sigma_1 - \sigma_2)} c_{\beta_1} s_{\beta_2} + \mu_2^2 c_{\beta_2} s_{\beta_1} s_{\beta_2} + \frac{1}{2}\mu_3^2 e^{-i\sigma_1} c_{\beta_1} c_{\beta_2} + \frac{1}{2}\mu_4^2 s_{\beta_1} \left(c_{2\beta_2} c_{\sigma_2} - i s_{\sigma_2} \right),$$
(C.1.2c)

$$Y_{22}' = Y_{22} - \frac{1}{2}\nu^2 c_{\beta_2} c_{\sigma_1 - \sigma_2} s_{2\beta_1} + \mu_2^2 \left(s_{\beta_1}^2 - c_{\beta_1}^2 c_{\beta_2}^2\right) - \mu_3^2 c_{\beta_1} c_{\sigma_1} s_{\beta_1} s_{\beta_2} + \mu_4^2 c_{\beta_1}^2 c_{\beta_2} c_{\sigma_2} s_{\beta_2}, \quad (C.1.2d)$$

$$Y_{23}' = Y_{23} + \frac{1}{2}\nu^2 e^{-i(\sigma_1 - \sigma_2)} \mathbf{s}_{\beta_1} \mathbf{s}_{\beta_2} + \mu_2^2 \mathbf{c}_{\beta_1} \mathbf{c}_{\beta_2} \mathbf{s}_{\beta_2} - \frac{1}{2}\mu_3^2 e^{-i\sigma_1} \mathbf{c}_{\beta_2} \mathbf{s}_{\beta_1}$$
(C.1.2e)

$$+\frac{1}{2}\mu_{4}^{2}c_{\beta_{1}}\left(c_{2\beta_{2}}c_{\sigma_{2}}-is_{\sigma_{2}}\right),$$

$$Y'_{33} = Y_{33} - \mu_2^2 \mathbf{s}_{\beta_2}^2 - \mu_4^2 \mathbf{c}_{\beta_2} \mathbf{c}_{\sigma_2} \mathbf{s}_{\beta_2}.$$
 (C.1.2f)

The real quartic couplings are:

$$Z_{1111} = \lambda_1 \left(c_{\beta_1}^2 + c_{\beta_2}^2 s_{\beta_1}^2 \right)^2 - 4\lambda_2 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_1}^2 s_{\sigma_1 - \sigma_2}^2 + \lambda_3 \left[c_{\beta_1}^4 + 2c_{\beta_2}^2 c_{\beta_1}^2 c_{2(\sigma_1 - \sigma_2)} s_{\beta_1}^2 + c_{\beta_2}^4 s_{\beta_1}^4 \right] + \lambda_4 \left[c_{\beta_1}^2 \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) s_{\beta_1}^2 s_{2\beta_2} - 2c_{\beta_2}^3 c_{\sigma_2} s_{\beta_1}^4 s_{\beta_2} \right] + \lambda_5 s_{\beta_1}^2 s_{\beta_2}^2 \left(c_{\beta_1}^2 + c_{\beta_2}^2 s_{\beta_1}^2 \right) + \lambda_6 s_{\beta_1}^2 s_{\beta_2}^2 \left(c_{\beta_1}^2 + c_{\beta_2}^2 s_{\beta_1}^2 \right) + 2\lambda_7 s_{\beta_1}^2 s_{\beta_2}^2 \left(c_{\beta_1}^2 c_{2\sigma_1} + c_{\beta_2}^2 c_{2\sigma_2} s_{\beta_1}^2 \right) + \lambda_8 s_{\beta_1}^4 s_{\beta_2}^4, \qquad (C.1.3a)$$

$$\begin{split} Z_{1122} &= 2\lambda_1 \left(c_{\beta_1}^2 c_{\beta_2}^2 + s_{\beta_1}^2 \right) \left(c_{\beta_1}^2 + c_{\beta_2}^2 s_{\beta_1}^2 \right) + 8\lambda_2 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_1}^2 s_{\sigma_1 - \sigma_2}^2 \\ &\quad - 2\lambda_3 \left[c_{\beta_2}^2 c_{\beta_1}^4 - c_{\beta_1}^2 s_{\beta_1}^2 \left(-4c_{\beta_2}^2 c_{\sigma_1 - \sigma_2}^2 + c_{\beta_2}^4 + 1 \right) + c_{\beta_2}^2 s_{\beta_1}^4 \right] \\ &\quad + \lambda_4 \left[\frac{1}{4} s_{2\beta_2} \left((c_{4\beta_1} + 3) c_{\sigma_2} - 4c_{\sigma_1} c_{\sigma_1 - \sigma_2} s_{2\beta_1}^2 \right) - c_{\beta_2}^3 c_{\sigma_2} s_{2\beta_1}^2 s_{\beta_2}^2 \right] \\ &\quad + \frac{1}{8} \lambda_5 \left[2 \left(c_{4\beta_1} + 3 \right) s_{\beta_2}^2 + s_{2\beta_1}^2 s_{2\beta_2}^2 \right] - 2\lambda_6 c_{\beta_1}^2 s_{\beta_1}^2 s_{\beta_2}^4 \\ &\quad - 4\lambda_7 c_{\beta_1}^2 s_{\beta_1}^2 s_{\beta_2}^2 \left(c_{2\sigma_1} - c_{\beta_2}^2 c_{2\sigma_2} \right) + 2\lambda_8 c_{\beta_1}^2 s_{\beta_1}^2 s_{\beta_2}^4 \\ &\quad - 4\lambda_7 c_{\beta_1}^2 s_{\beta_1}^2 s_{\beta_2}^2 \left(c_{\beta_1}^2 - c_{\beta_2}^2 c_{2\sigma_2} \right) + 2\lambda_8 c_{\beta_1}^2 s_{\beta_1}^2 s_{\beta_2}^4 \\ Z_{1133} &= 2\lambda_1 s_{\beta_2}^2 \left(c_{\beta_1}^2 + c_{\beta_2}^2 s_{\beta_1}^2 \right) - 2\lambda_3 s_{\beta_2}^2 \left(c_{\beta_1}^2 - c_{\beta_2}^2 s_{\beta_1}^2 \right) - 2\lambda_4 c_{\beta_2} c_{\sigma_2} s_{\beta_2} \left(c_{\beta_1}^2 - c_{2\beta_2} s_{\beta_1}^2 \right) \\ &\quad + \lambda_5 \left[c_{\beta_1}^2 c_{\beta_2}^2 + \frac{1}{4} \left(c_{4\beta_2} + 3 \right) s_{\beta_1}^2 \right] - 2\lambda_6 c_{\beta_2}^2 s_{\beta_1}^2 s_{\beta_2}^2 - 4\lambda_7 c_{\beta_2}^2 c_{2\sigma_2} s_{\beta_1}^2 s_{\beta_2}^2 + 2\lambda_8 c_{\beta_2}^2 s_{\beta_1}^2 s_{\beta_2}^2 , \\ (C.1.3c) \end{aligned}$$

$$Z_{1221} = 2\lambda_{1}c_{\beta_{1}}^{2}s_{\beta_{1}}^{2}s_{\beta_{2}}^{4} - \frac{1}{2}\lambda_{2}c_{\beta_{2}}^{2}\left(c_{4\beta_{1}} + 2c_{2(\sigma_{1}-\sigma_{2})}s_{2\beta_{1}}^{2} + 3\right) + 2\lambda_{3}\left[c_{\beta_{2}}^{2}c_{\beta_{1}}^{4} + c_{\beta_{1}}^{2}s_{\beta_{1}}^{2}\left(c_{\beta_{2}}^{4} + 4c_{\beta_{2}}^{2}s_{\sigma_{1}-\sigma_{2}}^{2} + 1\right) + c_{\beta_{2}}^{2}s_{\beta_{1}}^{4}\right] + \frac{1}{4}\lambda_{4}c_{\beta_{2}}s_{\beta_{2}}\left[\left(5c_{4\beta_{1}} + 3\right)c_{\sigma_{2}} - 2s_{2\beta_{1}}^{2}\left(c_{2\beta_{2}}c_{\sigma_{2}} + 2c_{2\sigma_{1}-\sigma_{2}}\right)\right] - 2\lambda_{5}c_{\beta_{1}}^{2}s_{\beta_{1}}^{2}s_{\beta_{2}}^{4} + \lambda_{6}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{4} + 2c_{\beta_{2}}^{2}c_{\beta_{1}}^{2}s_{\beta_{1}}^{2} + s_{\beta_{1}}^{4}\right) - 4\lambda_{7}c_{\beta_{1}}^{2}s_{\beta_{1}}^{2}s_{\beta_{2}}^{2} - (2\lambda_{2}c_{\beta_{1}}^{2}c_{2\sigma_{2}}) + 2\lambda_{8}c_{\beta_{1}}^{2}s_{\beta_{1}}^{2}s_{\beta_{2}}^{4}, Z_{1331} = 2\lambda_{1}c_{\beta_{2}}^{2}s_{\beta_{1}}^{2}s_{\beta_{2}}^{2} - 2\lambda_{2}c_{\beta_{1}}^{2}s_{\beta_{2}}^{2} + 2\lambda_{3}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{2} + c_{\beta_{2}}^{2}s_{\beta_{1}}^{2}\right) - 2\lambda_{4}c_{\beta_{2}}c_{\sigma_{2}}s_{\beta_{2}}\left(c_{\beta_{1}}^{2} - c_{2\beta_{2}}s_{\beta_{1}}^{2}\right) + 2\lambda_{5}c_{\beta_{2}}^{2}s_{\beta_{1}}^{2}s_{\beta_{2}}^{2} + \lambda_{6}\left[c_{\beta_{1}}^{2}c_{\beta_{2}}^{2} + \frac{1}{4}\left(c_{4\beta_{2}} + 3\right)s_{\beta_{1}}^{2}\right] - 4\lambda_{7}c_{\beta_{2}}^{2}c_{2\sigma_{2}}s_{\beta_{1}}^{2}s_{\beta_{2}}^{2} + 2\lambda_{8}c_{\beta_{2}}^{2}s_{\beta_{1}}^{2}s_{\beta_{2}}^{2},$$
(C.1.3e)

$$Z_{2222} = \lambda_1 \left(c_{\beta_1}^2 c_{\beta_2}^2 + s_{\beta_1}^2 \right)^2 - 4\lambda_2 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_1}^2 s_{\sigma_1 - \sigma_2}^2 + \lambda_3 \left[c_{\beta_1}^4 c_{\beta_2}^4 + 2c_{\beta_1}^2 c_{\beta_2}^2 c_{2(\sigma_1 - \sigma_2)} s_{\beta_1}^2 + s_{\beta_1}^4 \right] \\ + \lambda_4 \left[c_{\beta_1}^2 \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) s_{\beta_1}^2 s_{2\beta_2} - 2c_{\beta_1}^4 c_{\beta_2}^3 c_{\sigma_2} s_{\beta_2} \right] + \lambda_5 c_{\beta_1}^2 s_{\beta_2}^2 \left(c_{\beta_1}^2 c_{\beta_2}^2 + s_{\beta_1}^2 \right)$$
(C.1.3f)
$$+ \lambda_6 c_{\beta_1}^2 s_{\beta_2}^2 \left(c_{\beta_1}^2 c_{\beta_2}^2 + s_{\beta_1}^2 \right) + 2\lambda_7 c_{\beta_1}^2 s_{\beta_2}^2 \left(c_{\beta_1}^2 c_{\beta_2}^2 c_{2\sigma_2} + c_{2\sigma_1} s_{\beta_1}^2 \right) + \lambda_8 c_{\beta_1}^4 s_{\beta_2}^4,$$

$$Z_{2233} = 2\lambda_1 s_{\beta_2}^2 \left(c_{\beta_1}^2 c_{\beta_2}^2 + s_{\beta_1}^2 \right) + 2\lambda_3 s_{\beta_2}^2 \left(c_{\beta_1}^2 c_{\beta_2}^2 - s_{\beta_1}^2 \right) + 2\lambda_4 c_{\beta_2} c_{\sigma_2} s_{\beta_2} \left(c_{\beta_1}^2 c_{2\beta_2} - s_{\beta_1}^2 \right) \\ + \lambda_5 \left[\frac{1}{4} \left(c_{4\beta_2} + 3 \right) c_{\beta_1}^2 + c_{\beta_2}^2 s_{\beta_1}^2 \right] - 2\lambda_6 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_2}^2 - 4\lambda_7 c_{\beta_1}^2 c_{\beta_2}^2 c_{2\sigma_2} s_{\beta_2}^2 + 2\lambda_8 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_2}^2,$$

(C.1.3g)

$$Z_{2332} = 2\lambda_1 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_2}^2 - 2\lambda_2 s_{\beta_1}^2 s_{\beta_2}^2 + 2\lambda_3 s_{\beta_2}^2 \left(c_{\beta_1}^2 c_{\beta_2}^2 + s_{\beta_1}^2 \right) + 2\lambda_4 c_{\beta_2} c_{\sigma_2} s_{\beta_2} \left(c_{\beta_1}^2 c_{2\beta_2} - s_{\beta_1}^2 \right) - 2\lambda_5 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_2}^2 + \lambda_6 \left[\frac{1}{4} \left(c_{4\beta_2} + 3 \right) c_{\beta_1}^2 + c_{\beta_2}^2 s_{\beta_1}^2 \right] - 4\lambda_7 c_{\beta_1}^2 c_{\beta_2}^2 c_{2\sigma_2} s_{\beta_2}^2 + 2\lambda_8 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_2}^2,$$
(C.1.3h)

$$Z_{3333} = \lambda_1 s_{\beta_2}^4 + \lambda_3 s_{\beta_2}^4 + 2\lambda_4 c_{\beta_2} c_{\sigma_2} s_{\beta_2}^3 + \lambda_5 c_{\beta_2}^2 s_{\beta_2}^2 + \lambda_6 c_{\beta_2}^2 s_{\beta_2}^2 + 2\lambda_7 c_{\beta_2}^2 c_{2\sigma_2} s_{\beta_2}^2 + \lambda_8 c_{\beta_2}^4, \qquad (C.1.3i)$$

$$Z_{1233} = -2\lambda_1 c_{\beta_1} s_{\beta_1} s_{\beta_2}^4 + 2\lambda_3 c_{\beta_1} \left(c_{\beta_2}^2 + 1\right) s_{\beta_1} s_{\beta_2}^2 + 2\lambda_4 c_{\beta_2}^3 c_{\sigma_2} s_{2\beta_1} s_{\beta_2}$$

$$Z_{1233} = -2\lambda_{1}c_{\beta_{1}}s_{\beta_{1}}s_{\beta_{2}} + 2\lambda_{3}c_{\beta_{1}}(c_{\beta_{2}} + 1)s_{\beta_{1}}s_{\beta_{2}} + 2\lambda_{4}c_{\beta_{2}}c_{\sigma_{2}}s_{\beta_{2}}s_{\beta_{2}}s_{\beta_{2}} - \frac{1}{2\lambda_{5}}c_{2\beta_{2}}s_{2\beta_{1}}s_{\beta_{2}}^{2} - 2\lambda_{6}c_{\beta_{1}}c_{\beta_{2}}^{2}s_{\beta_{1}}s_{\beta_{2}}^{2} - 2\lambda_{7}c_{\beta_{2}}^{2}c_{2\sigma_{2}}s_{2\beta_{1}}s_{\beta_{2}}^{2} + \lambda_{8}c_{\beta_{2}}^{2}s_{2\beta_{1}}s_{\beta_{2}}^{2}, \qquad (C.1.3j)$$
$$Z_{1332} = \lambda_{1}c_{\beta_{2}}^{2}s_{2\beta_{1}}s_{\beta_{2}}^{2} + \lambda_{2}s_{2\beta_{1}}s_{\beta_{2}}^{2} - 2\lambda_{3}c_{\beta_{1}}s_{\beta_{1}}s_{\beta_{2}}^{4} + 2\lambda_{4}c_{\beta_{2}}^{3}c_{\sigma_{2}}s_{2\beta_{1}}s_{\beta_{2}}$$

$$332 = \lambda_1 c_{\beta_2}^2 s_{2\beta_1} s_{\beta_2}^2 + \lambda_2 s_{2\beta_1} s_{\beta_2}^2 - 2\lambda_3 c_{\beta_1} s_{\beta_1} s_{\beta_2}^2 + 2\lambda_4 c_{\beta_2}^2 c_{\sigma_2} s_{2\beta_1} s_{\beta_2} - 2\lambda_5 c_{\beta_1} c_{\beta_2}^2 s_{\beta_1} s_{\beta_2}^2 - \frac{1}{2} \lambda_6 c_{2\beta_2} s_{2\beta_1} s_{\beta_2}^2 - 2\lambda_7 c_{\beta_2}^2 c_{2\sigma_2} s_{2\beta_1} s_{\beta_2}^2 + \lambda_8 c_{\beta_2}^2 s_{2\beta_1} s_{\beta_2}^2,$$
(C.1.3k)

and the complex quartic couplings are:

$$\begin{split} & Z_{1112} = -2\lambda_{1}c_{\beta_{1}}c_{\beta_{3}}s_{\beta_{4}}^{2}\left(c_{\beta_{1}}^{2} + c_{\beta_{2}}^{2}s_{\beta_{1}}^{2}\right) - 2\lambda_{2}c_{\beta_{2}}^{2}c_{\beta_{1}}c_{\alpha_{2}}c_{\beta_{2}}^{2} + e^{2i\sigma_{1}}c_{\beta_{3}}^{2}s_{\beta_{1}}^{2}\right) \\ &\quad -2\lambda_{3}e^{-2i(\sigma_{1}+\sigma_{2})}c_{\beta_{1}}s_{\beta_{1}}s_{2}s_{2}\left[2\left(e^{2i\sigma_{1}} + e^{2i\sigma_{2}}+e^{2i(\sigma_{1}+\sigma_{2})}\right)c_{\beta_{1}}^{2}\right) \\ &\quad -e^{2i\sigma_{1}}s_{\beta_{1}}^{2}\left(e^{2i\sigma_{2}}\left(c_{2}g_{3}+3\right) + c_{2}g_{2}+2e^{2i\sigma_{1}}+3\right)\right] \\ &\quad +\lambda_{5}c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{2} + c_{2}g_{2}s_{\beta_{1}}^{2}\right) + \lambda_{6}c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{2} + c_{2}g_{2}s_{\beta_{1}}^{2}\right) \\ &\quad +2\lambda_{7}c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{2} + c_{2}g_{2}s_{\beta_{1}}^{2}\right) + \lambda_{6}c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{2} + c_{2}g_{\alpha_{3}}s_{\beta_{3}}^{2}\right) \\ &\quad +2\lambda_{7}c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{3}}^{2}\left(c_{\beta_{1}}^{2} + c_{\beta_{3}}^{2}s_{\beta_{3}}^{2}\right) + \lambda_{6}c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{3}}^{2}\left(c_{\beta_{1}}^{2} + c_{2}g_{\alpha_{3}}s_{\beta_{3}}^{2}\right) \\ &\quad +2\lambda_{7}c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{3}}^{2}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{2} + c_{\beta_{3}}s_{\beta_{3}}^{2}s_{\beta_{3}}^{2}\right) + \lambda_{6}c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{3}}^{2}\left(c_{\beta_{1}}^{2} + c_{2}g_{\alpha_{3}}s_{\beta_{3}}^{2}\right) \\ &\quad +2\lambda_{7}c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{3}}s_{\beta_{2}}^{2}\left[e^{-2i\sigma_{1}}c_{\beta_{1}}^{2}-2i\sigma_{1}s_{\beta_{3}}^{2}s_{\beta_{3}}s_{\beta_{3}}^{2}\right] + \lambda_{2}c^{-2i\sigma_{1}}\left(e^{2i\sigma_{1}} - e^{2i\sigma_{2}}\right) \\ &\quad +2\lambda_{7}\left[c_{\beta_{1}}s_{\beta_{3}}s_{\beta_{3}}s_{\beta_{3}}^{2}\left(c_{\beta_{1}}^{2} + s_{\beta_{1}}^{2}s_{\beta_{3}}s_{\beta_{3}}s_{\beta_{3}}^{2}\right) + \lambda_{6}c_{\beta_{2}}s_{\beta_{3}}s_{\beta_{3}}^{2}\right] \\ &\quad +\lambda_{7}\left[e^{2i\sigma_{1}}(2s_{\beta_{1}}s_{\beta_{3}}^{2}\right] + 2e^{-2i\sigma_{1}}\left(c_{\beta_{2}}^{2} + c_{\beta_{3}}s_{\beta_{3}}^{2}\right) \\ &\quad -e^{2i\sigma_{1}}c_{\beta_{3}}s_{\beta_{3}}^{2}\right) + \lambda_{6}c_{\beta_{3}}s_{\beta_{3}}s_{\beta_{3}}^{2} - 2e^{2i\sigma_{1}}c_{\beta_{3}}s_{\beta_{3}}^{2}\right) \\ &\quad +\lambda_{7}\left[e^{2i\sigma_{1}}s_{\beta_{3}}s_{\beta_{3}}^{2} + e^{2i\sigma_{1}}(s_{1}-2)\right] \\ &\quad +\lambda_{7}\left[e^{2i\sigma_{1}}s_{\beta_{3}}s_{\beta_{3}}^{2}\right] + \lambda_{6}c_{\beta_{1}}s_{\beta_{3}}^{2}s_{\beta_{3}}^{2}\right] \\ &\quad +\lambda_{7}\left[e^{2i\sigma_{1}}s_{\beta_{3}}s_{\beta_{3}}^{2} - 2e^{2i\sigma_{2}}c_{\beta_{3}}s_{\beta_{3}}^{2} - e^{2i\sigma_{1}}s_{\beta_{3}}s_{\beta_{3}}^{2}\right] \\ &\quad +\lambda_{7}\left[e^{2i\sigma_{1}}s_{\beta_{3}}s_{\beta_{3}}^{2} - 2\lambda_{6}c_{\alpha}c_{$$

$$\begin{split} \mathcal{Z}_{1222} &= -2\lambda_{1}c_{\beta_{1}}s_{\beta_{1}}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{2}c_{\beta_{2}}^{2}+s_{\beta_{1}}^{2}\right) + 2\lambda_{2}c_{\beta_{2}}^{2}s_{2}s_{1}s_{4}-a_{2}\left(c_{2}s_{1}s_{6}-a_{2}+ic_{4}-a_{2}\right) \\ &\quad -2\lambda_{3}e^{-2i(\sigma_{1}+\sigma_{2})}c_{\beta_{1}}s_{\beta_{1}}\left[e^{2i\sigma_{2}}-e^{2i\sigma_{1}}c_{\beta_{2}}^{2}\right)\left(c_{2}^{2i\sigma_{2}}c_{\beta_{1}}^{2}c_{\beta_{2}}^{2}+e^{2i\sigma_{1}}s_{\beta_{1}}^{2}\right) \\ &\quad +2\lambda_{4}e^{-i(2\sigma_{1}+\sigma_{2})}c_{\beta_{1}}s_{\beta_{2}}s_{\beta_{2}}s_{\beta_{2}}\left[-e^{2i\sigma_{2}}c_{\beta_{1}}^{2}\right] \\ &\quad -e^{2i\sigma_{1}}\left(1+e^{2i\sigma_{2}}\right)\left(c_{\beta_{1}}^{2}\left(c_{\beta_{2}}^{2}+1\right)-s_{\beta_{1}}^{2}\right)+e^{4i\sigma_{1}}s_{\beta_{1}}^{2}\right] \\ &\quad +\lambda_{5}c_{\beta_{1}}s_{\beta_{1}}s_{\beta_{2}}^{2}\left[c_{\beta_{2}}c_{\beta_{1}}^{2}+s_{\beta_{1}}^{2}\right)+\lambda_{6}c_{\beta_{1}}s_{\beta_{1}}s_{\beta_{2}}^{2}\left(c_{2}c_{2}c_{\beta_{1}}^{2}+s_{\beta_{1}}^{2}\right) \\ &\quad +\frac{1}{2}\lambda_{7}s_{\beta_{2}}^{2}\left[s_{2}^{2}c_{\beta_{1}}^{2}+s_{\beta_{1}}^{2}+\lambda_{6}c_{\beta_{1}}s_{\beta_{1}}s_{\beta_{2}}^{2}\left(c_{2}c_{2}c_{\beta_{1}}^{2}+s_{\beta_{1}}^{2}\right) \\ &\quad +2\lambda_{5}c_{\beta_{1}}s_{\beta_{1}}s_{\beta_{2}}^{2}+22c_{2}s_{3}s_{4}s_{4}+2is_{2}s_{5}s_{2}s_{1}\right] + 2\lambda_{5}c_{\beta_{1}}s_{5}s_{4}s_{4}, \\ Z_{1223} = 2\lambda_{1}c_{\beta_{1}}^{2}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}^{2}+2c_{\beta_{1}}c_{\beta_{1}}\left(-c_{\beta_{2}}^{2}+e^{-2i(\sigma_{1}-\sigma_{2})}-1\right)\right] \\ &\quad +\frac{1}{8}\lambda_{4}e^{-i(2\sigma_{1}+\sigma_{2})}s_{\beta_{1}}\left[-2e^{2i\sigma_{1}}\left(1+e^{2i\sigma_{2}}\right)c_{4j_{2}}c_{\beta_{1}}^{2}+e^{2i\sigma_{1}}\left(-1+e^{2i\sigma_{2}}\right)\left(5c_{2\beta_{1}}+1\right) \\ &\quad +2c_{2\beta_{5}}\left(-e^{2i\sigma_{1}}\left(3c_{2\beta_{1}}+1\right)-2e^{2i\sigma_{2}}\left(1+(1+e^{2i\sigma_{1}})c_{2\beta_{1}}\right)\right)\right)\right] \\ &\quad -2\lambda_{5}c_{\beta_{1}}^{2}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{3}}^{2}+2\lambda_{2}c_{\beta_{1}}s_{\beta_{2}}s_{\beta_{2}}\left(c_{\beta_{1}}^{2}+e^{2i(\sigma_{1}-\sigma_{2})}-1\right)\right] \\ &\quad +2\lambda_{2}e^{-2i(\sigma_{1}-\sigma_{2})}s_{\beta_{1}}^{2}s_{\beta_{2}}^{2}+2c_{2}c_{\beta_{1}}s_{\beta_{2}}^{2}}\left(c_{\beta_{1}}^{2}+e^{2i(\sigma_{1}-\sigma_{2})}s_{\beta_{1}}^{2}\right) \\ &\quad -2\lambda_{5}c_{\beta_{1}}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}^{2}+2c_{2}c_{\beta_{2}}s_{\beta_{1}}\right) \\ &\quad +2\lambda_{2}e^{-2i\sigma_{2}}c_{\beta_{1}}s_{\beta_{2}}s_{\beta_{2}}^{2}+2c_{2}c_{\beta_{2}}s_{\beta_{1}}\right) \\ &\quad +2\lambda_{2}e^{-2i\sigma_{2}}c_{\beta_{1}}s_{\beta_{2}}s_{\beta_{2}}^{2}+2c_{2}c_{\beta_{2}}s_{\beta_{1}}\right) \\ &\quad +2\lambda_{5}c_{\beta_{1}}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{2}+e^{2i(\sigma_{1}-\sigma_{2})}s_{\beta_{1}}^{2}\right) \\ &\quad +2\lambda_{5}c_{\beta_{1}}^{2}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}^{2}-2c_{$$

$$\begin{split} Z_{1322} &= -2\lambda_{1}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}\left(c_{\beta_{1}}^{2}c_{\beta_{2}}^{2}+s_{\beta_{1}}^{2}\right) + 2\lambda_{2}\left(-1+e^{-2i(\sigma_{1}-\sigma_{2})}\right)c_{\beta_{1}}^{2}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}} \\ &+ 2\lambda_{3}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}\left[s_{\beta_{1}}^{2}+c_{\beta_{1}}^{2}\left(s_{\beta_{2}}^{2}+e^{-2i(\sigma_{1}-\sigma_{2})}\right)\right] \\ &+ \frac{1}{8}\lambda_{4}e^{-i(2\sigma_{1}+\sigma_{2})}s_{\beta_{1}}\left[-2e^{2i\sigma_{1}}\left(1+e^{2i\sigma_{2}}\right)c_{4\beta_{2}}c_{\beta_{1}}^{2} \\ &+ e^{2i\sigma_{1}}\left(-1+e^{2i\sigma_{2}}\right)\left(5c_{2\beta_{1}}+1\right) \\ &+ 2c_{2\beta_{2}}\left(-e^{2i\sigma_{1}}\left(3c_{2\beta_{1}}+1\right)-2e^{2i\sigma_{2}}\left(1+\left(1+e^{2i\sigma_{1}}\right)c_{2\beta_{1}}\right)\right)\right] \\ &+ \lambda_{5}c_{\beta_{2}}s_{\beta_{3}}s_{\beta_{2}}\left(c_{2\beta_{2}}c_{\beta_{1}}^{2}+s_{\beta_{1}}^{2}\right)-2\lambda_{6}c_{\beta_{1}}^{2}c_{\beta_{2}}s_{\beta_{3}}s_{\beta_{2}}^{2} \\ &+ 2\lambda_{7}e^{-2i(\sigma_{1}+\sigma_{2})}c_{\beta_{1}}c_{\beta_{2}}s_{\beta_{3}}s_{\beta_{2}}\left[e^{2i\sigma_{1}}\left(-c_{\beta_{2}}^{2}+e^{4i\sigma_{2}}s_{\beta_{2}}^{2}\right)+e^{2i\sigma_{2}}\right]+2\lambda_{8}c_{\beta_{1}}^{2}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}^{2}, \\ Z_{1323} &= \lambda_{1}c_{\beta_{2}}^{2}s_{2\beta_{1}}s_{\beta_{2}}^{2}-2\lambda_{2}e^{-2i(\sigma_{1}-\sigma_{2})}c_{\beta_{1}}s_{\beta_{1}}s_{\beta_{2}}^{2}+2\lambda_{3}c_{\beta_{1}}s_{\beta_{1}}s_{\beta_{2}}^{2}\left(c_{\beta_{2}}^{2}-e^{-2i(\sigma_{1}-\sigma_{2})}\right) \\ &+ 2\lambda_{4}e^{-i(2\sigma_{1}+\sigma_{2})}c_{\beta_{1}}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}\left[e^{2i\sigma_{1}}\left(c_{\beta_{2}}^{2}-e^{2i\sigma_{2}}s_{\beta_{2}}^{2}\right)+e^{2i\sigma_{2}}\right]-2\lambda_{5}c_{\beta_{1}}c_{\beta_{2}}^{2}s_{\beta_{1}}s_{\beta_{2}}^{2}, \\ Z_{1333} &= -2\lambda_{1}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}^{2}-2\lambda_{3}c_{\beta_{1}}s_{\beta_{2}}^{2}-4e^{-2i\sigma_{2}}s_{\beta_{2}}^{2}s_{\beta_{1}}s_{\beta_{2}}^{2}\left[1+\left(1+e^{2i\sigma_{2}}\right)c_{2\beta_{2}}\right] \\ &+ \frac{1}{4}\lambda_{5}s_{\beta_{1}}s_{4\beta_{2}}-\frac{1}{4}\lambda_{6}s_{\beta_{1}}s_{4\beta_{2}}+2\lambda_{7}e^{-2i\sigma_{2}}c_{\beta_{2}}s_{\beta_{1}}s_{\beta_{2}}^{2}\left[1+\left(1+e^{2i\sigma_{2}}s_{\beta_{2}}^{2}\right)+2\lambda_{8}c_{\beta_{2}}^{2}s_{\beta_{1}}s_{\beta_{2}}, \\ &-\lambda_{3}\left(c_{\beta_{1}}c_{\beta_{2}}s_{\beta_{2}}+2e^{-2i(\sigma_{1}-\sigma_{2})}c_{\beta_{1}}c_{\beta_{2}}s_{\beta_{1}}^{2}s_{\beta_{2}}\right) \\ &+\lambda_{4}e^{-i(2\sigma_{1}+\sigma_{2})}c_{\beta_{1}}\left[e^{2i\sigma_{2}}\left(c_{2\beta_{2}}s_{\beta_{1}}^{2}+2e^{2i\sigma_{1}}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}^{2}c_{\beta_{2}}^{2}-s_{\beta_{1}}^{2}\right)\right] \\ &+\lambda_{4}e^{-i(2\sigma_{1}+\sigma_{2})}c_{\beta_{1}}\left[e^{2i\sigma_{2}}\left(c_{2\beta_{2}}s_{\beta_{1}}^{2}+2e^{2i\sigma_{1}}s_{\beta_{2}}^{2}\left(c_{\beta_{1}}c_{\beta_{2}}^{2}-s_{\beta_{1}}^{2}\right)\right] \\ &+\lambda_{4}e^{-i(2\sigma_{1}+\sigma_{2})}c_{\beta_{1}}\left[e^{2i\sigma_{2}}\left(c_{2\beta_{2}}$$

$$Z_{2323} = \lambda_1 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_2}^2 + \lambda_2 e^{-2i(\sigma_1 - \sigma_2)} s_{\beta_1}^2 s_{\beta_2}^2 + \lambda_3 s_{\beta_2}^2 \left(c_{\beta_1}^2 c_{\beta_2}^2 + e^{-2i(\sigma_1 - \sigma_2)} s_{\beta_1}^2 \right) + \lambda_4 e^{-i\sigma_2} c_{\beta_2} s_{\beta_2} \left[c_{\beta_1}^2 \left(c_{\beta_2}^2 - e^{2i\sigma_2} s_{\beta_2}^2 \right) - e^{-2i(\sigma_1 - \sigma_2)} s_{\beta_1}^2 \right] + \lambda_5 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_2}^2$$
(C.1.40)
$$- \lambda_6 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_2}^2 + \lambda_7 \left[e^{-2i\sigma_2} c_{\beta_1}^2 \left(c_{\beta_2}^4 + e^{4i\sigma_2} s_{\beta_2}^4 \right) + e^{-2i\sigma_1} c_{\beta_2}^2 s_{\beta_1}^2 \right] + \lambda_8 c_{\beta_1}^2 c_{\beta_2}^2 s_{\beta_2}^2,$$

$$Z_{2333} = -2\lambda_1 c_{\beta_1} c_{\beta_2} s_{\beta_2}^3 - 2\lambda_3 c_{\beta_1} c_{\beta_2} s_{\beta_2}^3 - \lambda_4 e^{-i\sigma_2} c_{\beta_1} s_{\beta_2}^2 \left[1 + \left(1 + e^{2i\sigma_2} \right) c_{2\beta_2} \right] \right] + \frac{1}{4} \lambda_5 c_{\beta_1} s_{4\beta_2} - \frac{1}{4} \lambda_6 c_{\beta_1} s_{4\beta_2} + 2\lambda_7 e^{-2i\sigma_2} c_{\beta_1} c_{\beta_2} s_{\beta_2} \left(-c_{\beta_2}^2 + e^{4i\sigma_2} s_{\beta_2}^2 \right) + 2\lambda_8 c_{\beta_1} c_{\beta_2}^3 s_{\beta_2}.$$

(C.1.4p)

Couplings which potentially lead to CP violation are presented in eq. (C.1.4). In terms of the generic basis, only terms multiplied by λ_2 , or λ_3 , or λ_4 , or λ_7 may result in CP violation in the Higgs basis¹. If the soft symmetry breaking terms of the generic basis (3.2.2) are added, the bilinear terms in the Higgs basis may also lead to CP violation. In this case the couplings Y'_{12} , or Y'_{13} , or Y'_{23} may result in CP violation. Not all of the Z couplings should be checked as:

$$\operatorname{Im}(Z_{1112}) = \operatorname{Im}(Z_{1222}^*), \qquad (C.1.5a)$$

$$\begin{array}{l}
\text{Im} (Z_{1123}) = \text{Im} (Z_{1231}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}), \\
\text{Im} (Z_{1123}) = \text{Im} (Z_{1123}),$$

$$Im(Z_{1223}) = Im(Z_{1322}).$$
(C.1.5c)

It should be noted that this is only true for transformation given by eq. (2.5.14).

¹This is only true when an additional U(2) transformation is not considered.

C.2 $\mathcal{R}_{\beta_y}\mathcal{R}_{\beta_z}$ Higgs Basis Rotation

We present relations between the couplings in the Higgs basis and the generic basis of eqs. (2.5.9, 2.5.10) using transformations given by eq. (2.5.26).

Quadratic couplings in the Higgs basis are:

$$Y_{11} = \frac{1}{v_1^2} \left(\mu_0^2 \hat{w}_S^2 + \mu_1^2 w^2 \right), \qquad (C.2.1a)$$

$$Y_{22} = \mu_1^2, \tag{C.2.1b}$$

$$Y_{13} = \frac{w^2}{N_3 v} \left(-\mu_0^2 + \mu_1^2\right), \qquad (C.2.1c)$$

$$Y_{33} = \frac{1}{v^2} \left(\mu_0^2 w^2 + \mu_1^2 \hat{w}_S^2 \right), \qquad (C.2.1d)$$

(C.2.1e)

and the soft symmetry breaking terms of eq. (3.2.2) result in:

$$Y_{11}' = Y_{11} + \frac{1}{v^2} \left[\nu^2 \hat{w}_1 \hat{w}_2 c_{\sigma_1 - \sigma_2} + \left(\hat{w}_1^2 - \hat{w}_2^2 \right) \mu_2^2 + \hat{w}_1 \hat{w}_S c_{\sigma_1} \mu_3^2 + \hat{w}_2 \hat{w}_S c_{\sigma_2} \mu_4^2 \right],$$
(C.2.2a)

$$Y_{12}' = \frac{1}{N_2 v^2} \left[\frac{1}{2} \nu^2 e^{-i(\sigma_1 + \sigma_2)} v \left(e^{2i\sigma_1} \hat{w}_2^2 - e^{2i\sigma_2} \hat{w}_1^2 \right) + 2v \hat{w}_1 \hat{w}_2 \mu_2^2 + \frac{1}{2} e^{i\sigma_1} v \hat{w}_2 \hat{w}_3 \mu_3^2 - \frac{1}{2} e^{i\sigma_2} v \hat{w}_1 \hat{w}_3 \mu_4^2 \right],$$
(C.2.2b)

$$Y_{13}' = Y_{13} + \frac{1}{N_3 v} \bigg[\nu^2 \hat{w}_1 \hat{w}_2 c_{\sigma_1 - \sigma_2} + \left(\hat{w}_1^2 - \hat{w}_2^2 \right) \mu_2^2 + \frac{1}{2} e^{-i\sigma_1} \hat{w}_1 \mu_3^2 \left(X + e^{2i\sigma_1} \hat{w}_S \right) \\ + \frac{1}{2} e^{-i\sigma_2} \hat{w}_2 \mu_4^2 \left(X + e^{2i\sigma_2} \hat{w}_S \right) \bigg],$$
(C.2.2c)

$$Y_{22}' = Y_{22} + \frac{1}{N_2^2} \left[-\nu^2 \hat{w}_1 \hat{w}_2 c_{\sigma_1 - \sigma_2} + \left(\hat{w}_2^2 - \hat{w}_1^2 \right) \mu_2^2 \right],$$
(C.2.2d)

$$Y_{23}' = \frac{1}{N_2 N_3} \left[\frac{1}{2} \nu^2 e^{-i(\sigma_1 + \sigma_2)} \left(e^{2i\sigma_2} \hat{w}_2^2 - e^{2i\sigma_1} \hat{w}_1^2 \right) + 2\hat{w}_1 \hat{w}_2 \mu_2^2 + \frac{1}{2} e^{-i\sigma_1} \hat{w}_2 X \mu_3^2 - \frac{1}{2} e^{-i\sigma_2} \hat{w}_1 X \mu_4^2 \right],$$
(C.2.2e)

$$Y'_{33} = Y_{33} + \frac{1}{N_2 N_3^2} \left[\nu^2 \hat{w}_1 \hat{w}_2 c_{\sigma_1 - \sigma_2} + \left(\hat{w}_1^2 - \hat{w}_2^2 \right) \mu_2^2 + \hat{w}_1 X c_{\sigma_1} \mu_3^2 + \hat{w}_2 X c_{\sigma_2} \mu_4^2 \right].$$
(C.2.2f)

The real quartic couplings are:

$$Z_{1111} = \frac{1}{v^4} \Biggl\{ \lambda_1 w^4 - 4\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 + \lambda_3 \left[2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] \\ + 2\lambda_4 \hat{w}_2 \hat{w}_S \left[\hat{w}_1^2 \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) - \hat{w}_2^2 c_{\sigma_2} \right] + \lambda_5 w^2 \hat{w}_S^2 \\ + \lambda_6 w^2 \hat{w}_S^2 + 2\lambda_7 \hat{w}_S^2 \left(\hat{w}_1^2 c_{2\sigma_1} + \hat{w}_2^2 c_{2\sigma_2} \right) + \lambda_8 \hat{w}_S^4 \Biggr\},$$
(C.2.3a)

$$Z_{1122} = \frac{1}{N_2^2 v^2} \Biggl\{ 2\lambda_1 w^4 + 8\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 - 2\lambda_3 \left[2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] + 2\lambda_4 \hat{w}_2 \hat{w}_S \left[\hat{w}_2^2 c_{\sigma_2} - \hat{w}_1^2 \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) \right] + \lambda_5 w^2 \hat{w}_S^2 \Biggr\},$$
(C.2.3b)

$$Z_{1133} = \frac{1}{N_3^2 v^2} \Biggl\{ 2\lambda_1 w^4 - 8\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 + 2\lambda_3 \left[2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] \\ + 2\lambda_4 \hat{w}_2 \left[\hat{w}_1^2 \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) - \hat{w}_2^2 c_{\sigma_2} \right] \left(\hat{w}_S + X \right) + \lambda_5 w^2 \left(\hat{w}_S^2 + X^2 \right) \\ - 2\lambda_6 w^4 - 4\lambda_7 w^2 \left(\hat{w}_1^2 c_{2\sigma_1} + \hat{w}_2^2 c_{2\sigma_2} \right) + 2\lambda_8 w^4 \Biggr\},$$
(C.2.3c)

$$Z_{1221} = \frac{1}{N_2^2 v^2} \Biggl\{ -2\lambda_2 \left[2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] + 2\lambda_3 \left[-2\hat{w}_2^2 \hat{w}_1^2 \left(c_{2(\sigma_1 - \sigma_2)} - 2 \right) + \hat{w}_1^4 + \hat{w}_2^4 \right] + 2\lambda_4 \hat{w}_2 \hat{w}_S \left[\hat{w}_2^2 c_{\sigma_2} - \hat{w}_1^2 \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) \right] + \lambda_6 w^2 \hat{w}_S^2 \Biggr\},$$
(C.2.3d)

$$Z_{1331} = \frac{1}{N_3^2 v^2} \Biggl\{ 2\lambda_1 w^4 - 8\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 + 2\lambda_3 \left[2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] \\ + 2\lambda_4 \hat{w}_2 \left[\hat{w}_1^2 \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) - \hat{w}_2^2 c_{\sigma_2} \right] \left(\hat{w}_S + X \right) \\ - 2\lambda_5 w^4 + \lambda_6 w^2 \left(\hat{w}_S^2 + X^2 \right) - 4\lambda_7 w^2 \left(\hat{w}_1^2 c_{2\sigma_1} + \hat{w}_2^2 c_{2\sigma_2} \right) + 2\lambda_8 w^4 \Biggr\},$$
(C.2.3e)

$$Z_{2222} = \frac{1}{N_2^4} \left\{ \lambda_1 w^4 - 4\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 + \lambda_3 \left[2 \hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] \right\},$$
(C.2.3f)

$$Z_{2233} = \frac{1}{N_2^2 N_3^2} \left\{ 2\lambda_1 w^4 + 8\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 - 2\lambda_3 \left[2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] + \lambda_4 \left[2\hat{w}_2^3 X c_{\sigma_2} - 2\hat{w}_1^2 \hat{w}_2 X \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) \right] + \lambda_5 w^2 X^2 \right\},$$
(C.2.3g)

$$Z_{2332} = \frac{1}{N_2^2 N_3^2} \Biggl\{ -2\lambda_2 \left[2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] + 2\lambda_3 \left[-2\hat{w}_2^2 \hat{w}_1^2 \left(c_{2(\sigma_1 - \sigma_2)} - 2 \right) + \hat{w}_1^4 + \hat{w}_2^4 \right] + \lambda_4 \left(2\hat{w}_2^3 X c_{\sigma_2} - 2\hat{w}_1^2 \hat{w}_2 X \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) \right] + \lambda_6 w^2 X^2 \Biggr\},$$

$$(C.2.3h)$$

$$Z_{3333} = \frac{1}{N_3^4} \Biggl\{ \lambda_1 w^4 - 4\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 + \lambda_3 \left[2 \hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] \\ + \lambda_4 \left[2 \hat{w}_1^2 \hat{w}_2 X \left(c_{2\sigma_1 - \sigma_2} + 2c_{\sigma_2} \right) - 2 \hat{w}_2^3 X c_{\sigma_2} \right] + \lambda_5 w^2 X^2 + \lambda_6 w^2 X^2 \\ + 2\lambda_7 X^2 \left(\hat{w}_1^2 c_{2\sigma_1} + \hat{w}_2^2 c_{2\sigma_2} \right) + \lambda_8 X^4 \Biggr\},$$
(C.2.3i)

and the complex ones are:

$$Z_{1112} = \frac{1}{N_2 v^3} \Biggl\{ 4\lambda_2 \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left[\left(\hat{w}_1^2 - \hat{w}_2^2 \right) s_{\sigma_1 - \sigma_2} + i w^2 c_{\sigma_1 - \sigma_2} \right] + 4\lambda_3 \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left[\left(\hat{w}_1^2 - \hat{w}_2^2 \right) s_{\sigma_1 - \sigma_2} + i w^2 c_{\sigma_1 - \sigma_2} \right] + \lambda_4 e^{-i\sigma_2} \hat{w}_1 \hat{w}_S \left[\left(3 + 2e^{2i\sigma_1} \right) \hat{w}_2^2 + e^{2i\sigma_2} \left(4\hat{w}_2^2 - \left(2 + e^{-2i\sigma_1} \right) \hat{w}_1^2 \right) \right] + 2\lambda_7 \left(e^{2i\sigma_1} - e^{2i\sigma_2} \right) \hat{w}_1 \hat{w}_2 \hat{w}_S^2 \Biggr\},$$
(C.2.4a)

$$Z_{1113} = \frac{1}{N_3 v^3} \Biggl\{ 2\lambda_1 w^4 - 8\lambda_2 w_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 + 2\lambda_3 \left[2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] \\ + \lambda_4 e^{-i\sigma_2} \hat{w}_2 \Biggl[2e^{2i\sigma_1} \hat{w}_1^2 \hat{w}_S + e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_1^2 \left(\hat{w}_S + X \right) \\ + \left(2\hat{w}_1^2 - \hat{w}_2^2 \right) \left(2e^{2i\sigma_2} \hat{w}_S + \hat{w}_S + X \right) \Biggr] \\ + \lambda_5 w^2 \hat{w}_S \left(\hat{w}_S + X \right) + \lambda_6 w^2 \hat{w}_S \left(\hat{w}_S + X \right) \\ + 2\lambda_7 \hat{w}_S \left[e^{2i\sigma_2} \hat{w}_2^2 \hat{w}_S + e^{-2i\sigma_1} \hat{w}_1^2 \left(X + e^{4i\sigma_1} \hat{w}_S \right) + e^{-2i\sigma_2} \hat{w}_2^2 X \right] - 2\lambda_8 w^2 \hat{w}_S^2 \Biggr\},$$
(C.2.4b)

$$\begin{split} Z_{1123} &= \frac{1}{N_2 N_3 v^2} \Biggl\{ -4i\lambda_2 e^{-i(\sigma_1 + \sigma_2)} \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right) \\ &\quad -4i\lambda_3 e^{-i(\sigma_1 + \sigma_2)} \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right) \\ &\quad -\lambda_4 e^{-i\sigma_2} \hat{w}_1 \left[e^{2i\sigma_1} \hat{w}_1^2 \hat{w}_S - e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_2^2 \left(3e^{2i\sigma_1} \hat{w}_S + \hat{w}_S + X \right) \right. \\ &\quad + \left(\hat{w}_1^2 - 2 \hat{w}_2^2 \right) \left(\hat{w}_S + X \right) \Biggr] \\ &\quad -2\lambda_7 \left(e^{-2i\sigma_1} - e^{-2i\sigma_2} \right) w^2 \hat{w}_1 \hat{w}_2 \Biggr\}, \\ Z_{1212} &= \frac{1}{N_2^2 v^2} \Biggl\{ \lambda_2 e^{-2i(\sigma_1 + \sigma_2)} \left(e^{2i\sigma_2} \hat{w}_1^2 + e^{2i\sigma_1} \hat{w}_2^2 \right)^2 + \lambda_3 e^{-2i(\sigma_1 + \sigma_2)} \left(e^{2i\sigma_2} \hat{w}_1^2 + e^{2i\sigma_1} \hat{w}_2^2 \right)^2 \\ &\quad + \lambda_4 e^{-i\sigma_2} \hat{w}_2 \hat{w}_S \left(e^{2i\sigma_1} \hat{w}_2^2 - 3e^{2i\sigma_2} \hat{w}_1^2 \right) + \lambda_7 \hat{w}_S^2 \left(e^{2i\sigma_2} \hat{w}_1^2 + e^{2i\sigma_1} \hat{w}_2^2 \right) \Biggr\}, \end{aligned}$$

$$Z_{1213} &= \frac{1}{N_2 N_3 v^2} \Biggl\{ 4\lambda_2 \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left[\left(\hat{w}_1^2 - \hat{w}_2^2 \right) s_{\sigma_1 - \sigma_2} + i w^2 c_{\sigma_1 - \sigma_2} \right] \\ &\quad + \lambda_4 e^{-i\sigma_2} \hat{w}_1 \left[2e^{2i\sigma_1} \hat{w}_2^2 \hat{w}_S - 2e^{2i\sigma_2} \left(\hat{w}_1^2 - 2\hat{w}_2^2 \right) \hat{w}_S - e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_1^2 X + 3\hat{w}_2^2 X \Biggr] \\ &\quad + 2\lambda_7 \left(e^{2i\sigma_1} - e^{2i\sigma_2} \right) \hat{w}_1 \hat{w}_2 \hat{w}_S^2 \Biggr\}, \tag{C.2.4e}$$

$$Z_{1222} = \frac{1}{N_2^3 v} \Biggl\{ -4\lambda_2 \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left[\left(\hat{w}_1^2 - \hat{w}_2^2 \right) s_{\sigma_1 - \sigma_2} + i w^2 c_{\sigma_1 - \sigma_2} \right] -4\lambda_3 \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left[\left(\hat{w}_1^2 - \hat{w}_2^2 \right) s_{\sigma_1 - \sigma_2} + i w^2 c_{\sigma_1 - \sigma_2} \right] +\lambda_4 e^{-i\sigma_2} \hat{w}_1 \hat{w}_S \left[e^{2i\sigma_2} \left(\hat{w}_1^2 - 2 \hat{w}_2^2 \right) - e^{2i\sigma_1} \hat{w}_2^2 \right] \Biggr\},$$

$$Z_{1223} = \frac{1}{N_2^2 N_3 v} \Biggl\{ -2\lambda_2 \left[2 \hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] + 2\lambda_3 \left[-2 \hat{w}_2^2 \hat{w}_1^2 \left(c_{2(\sigma_1 - \sigma_2)} - 2 \right) + \hat{w}_1^4 + \hat{w}_2^4 \right] +\lambda_4 e^{-i\sigma_2} \hat{w}_2 \left[-e^{2i\sigma_1} \hat{w}_1^2 \hat{w}_S + \left(\hat{w}_2^2 - 2 \hat{w}_1^2 \right) \left(X + e^{2i\sigma_2} \hat{w}_S \right) - e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_1^2 X \right] - \lambda_6 w^4 \Biggr\},$$
(C.2.4g)

$$Z_{1231} = \frac{1}{N_2 N_3 v^2} \Biggl\{ 4\lambda_2 \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left[\left(\hat{w}_1^2 - \hat{w}_2^2 \right) s_{\sigma_1 - \sigma_2} + i w^2 c_{\sigma_1 - \sigma_2} \right] \\ + 4\lambda_3 \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left[\left(\hat{w}_1^2 - \hat{w}_2^2 \right) s_{\sigma_1 - \sigma_2} + i w^2 c_{\sigma_1 - \sigma_2} \right] \\ + \lambda_4 e^{-i\sigma_2} \hat{w}_1 \Biggl[- e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_1^2 \hat{w}_S + e^{2i\sigma_1} \hat{w}_2^2 \left(\hat{w}_S + X \right) \\ - e^{2i\sigma_2} \left(\hat{w}_1^2 - 2 \hat{w}_2^2 \right) \left(\hat{w}_S + X \right) + 3 \hat{w}_2^2 \hat{w}_S \Biggr] \Biggr]$$

$$Z_{1232} = \frac{1}{N_2^2 N_3 v} \Biggl\{ 2\lambda_2 e^{-2i(\sigma_1 + \sigma_2)} \left(e^{2i\sigma_2} \hat{w}_1^2 + e^{2i\sigma_1} \hat{w}_2^2 \right)^2 + 2\lambda_3 e^{-2i(\sigma_1 + \sigma_2)} \left(e^{2i\sigma_2} \hat{w}_1^2 + e^{2i\sigma_1} \hat{w}_2^2 \right)^2 + \lambda_4 \hat{w}_2 \left(e^{2i\sigma_1 - i\sigma_2} \hat{w}_2^2 - 3e^{i\sigma_2} \hat{w}_1^2 \right) \left(\hat{w}_S + X \right) - 2\lambda_7 w^2 \left(e^{2i\sigma_2} \hat{w}_1^2 + e^{2i\sigma_1} \hat{w}_2^2 \right) \Biggr\},$$
(C.2.4i)

$$\begin{split} Z_{1233} &= Z_{1332} = \frac{1}{N_2 N_3^2 v} \Biggl\{ 4\lambda_2 \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left[\left(\hat{w}_1^2 - \hat{w}_2^2 \right) s_{\sigma_1 - \sigma_2} + i w^2 c_{\sigma_1 - \sigma_2} \right] \\ &\quad + 4\lambda_3 \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left[\left(\hat{w}_1^2 - \hat{w}_2^2 \right) s_{\sigma_1 - \sigma_2} + i w^2 c_{\sigma_1 - \sigma_2} \right] \\ &\quad + \lambda_4 e^{-i\sigma_2} \hat{w}_1 \Biggl[e^{2i\sigma_1} \hat{w}_2^2 \left(\hat{w}_S + X \right) - e^{2i\sigma_2} \left(\hat{w}_1^2 - 2 \hat{w}_2^2 \right) \left(\hat{w}_S + X \right) \right] \\ &\quad - e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_1^2 X + 3 \hat{w}_2^2 X \Biggr] \\ &\quad - e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_1^2 X + 3 \hat{w}_2^2 X \Biggr] \\ &\quad - 2\lambda_7 \left(e^{2i\sigma_1} - e^{2i\sigma_2} \right) w^2 \hat{w}_1 \hat{w}_2 \Biggr\}, \\ Z_{1313} &= \frac{1}{N_3^2 v^2} \Biggl\{ \lambda_1 w^4 - 4\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 + \lambda_3 \left(2 \hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right) \\ &\quad + \lambda_4 e^{-i\sigma_2} \hat{w}_2 \left[e^{2i\sigma_1} \hat{w}_1^2 \hat{w}_S + \left(2 \hat{w}_1^2 - \hat{w}_2^2 \right) \left(X + e^{2i\sigma_2} \hat{w}_S \right) + e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_1^2 X \Biggr] - \lambda_5 w^4 \\ &\quad - \lambda_6 w^4 \hat{w}_S + \lambda_7 \left[e^{2i\sigma_2} \hat{w}_2^2 \hat{w}_S^2 + e^{-2i\sigma_1} \hat{w}_1^2 \left(X^2 + e^{4i\sigma_1} \hat{w}_S^2 \right) + e^{-2i\sigma_2} \hat{w}_2^2 X^2 \Biggr] + \lambda_8 w^4 \Biggr\},$$
 (C.2.4k)

$$Z_{1322} = \frac{1}{N_2^2 N_3 v} \Biggl\{ 2\lambda_1 w^4 + 8\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 - 2\lambda_3 \left[2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right] \\ + \lambda_4 e^{-i\sigma_2} \hat{w}_2 \left[-e^{2i\sigma_1} \hat{w}_1^2 \hat{w}_S + \left(\hat{w}_2^2 - 2\hat{w}_1^2 \right) \left(X + e^{2i\sigma_2} \hat{w}_S \right) - e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_1^2 X \right] - \lambda_5 w^4 \Biggr\},$$
(C.2.4l)

$$Z_{1323} = \frac{1}{N_2 N_3^2 v} \Biggl\{ -4i\lambda_2 e^{-i(\sigma_1+\sigma_2)} \hat{w}_1 \hat{w}_2 s_{\sigma_1-\sigma_2} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right) -4i\lambda_3 e^{-i(\sigma_1+\sigma_2)} \hat{w}_1 \hat{w}_2 s_{\sigma_1-\sigma_2} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right) +\lambda_4 e^{-i\sigma_2} \hat{w}_1 \left[\hat{w}_S \left(3e^{2i\sigma_2} \hat{w}_2^2 - e^{2i\sigma_1} \hat{w}_1^2 \right) + 2 \left(2 + e^{-2i(\sigma_1-\sigma_2)} \right) \hat{w}_2^2 X - 2 \hat{w}_1^2 X \Biggr] +2\lambda_7 \left(e^{-2i\sigma_1} - e^{-2i\sigma_2} \right) \hat{w}_1 \hat{w}_2 X^2 \Biggr\},$$
(C.2.4m)

$$Z_{1333} = \frac{1}{N_3^3 v} \Biggl\{ 2\lambda_1 w^4 - 8\lambda_2 \hat{w}_1^2 \hat{w}_2^2 s_{\sigma_1 - \sigma_2}^2 + 2\lambda_3 \left(2\hat{w}_2^2 \hat{w}_1^2 c_{2(\sigma_1 - \sigma_2)} + \hat{w}_1^4 + \hat{w}_2^4 \right) \\ + \lambda_4 e^{-i\sigma_2} \hat{w}_2 \Biggl[e^{2i\sigma_1} \hat{w}_1^2 \hat{w}_S + \left(2\hat{w}_1^2 - \hat{w}_2^2 \right) \left(2X + e^{2i\sigma_2} \left(\hat{w}_S + X \right) \right) \\ + e^{2i\sigma_1} \hat{w}_1^2 X + 2e^{-2i(\sigma_1 - \sigma_2)} \hat{w}_1^2 X \Biggr] \\ + \lambda_5 w^2 X \left(\hat{w}_S + X \right) + \lambda_6 w^2 X \left(\hat{w}_S + X \right) \\ + 2\lambda_7 X \Biggl[e^{2i\sigma_2} \hat{w}_2^2 \hat{w}_S + e^{-2i\sigma_1} \hat{w}_1^2 \left(X + e^{4i\sigma_1} \hat{w}_S \right) + e^{-2i\sigma_2} \hat{w}_2^2 X \Biggr] - 2\lambda_8 X^2 w^2 \Biggr\},$$
(C.2.4n)

$$Z_{2223} = \frac{1}{N_2^3 N_3} \Biggl\{ 4i\lambda_2 e^{-i(\sigma_1 + \sigma_2)} \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right) + 4i\lambda_3 e^{-i(\sigma_1 + \sigma_2)} \hat{w}_1 \hat{w}_2 s_{\sigma_1 - \sigma_2} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right) + \lambda_4 e^{-i\sigma_2} \hat{w}_1 X \left[\hat{w}_1^2 + \left(-2 - e^{-2i(\sigma_1 - \sigma_2)} \right) \hat{w}_2^2 \right] \Biggr\},$$

$$Z_{2323} = \frac{1}{N_2^2 N_3^2} \Biggl\{ \lambda_2 e^{-2i(\sigma_1 + \sigma_2)} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right)^2 + \lambda_3 e^{-2i(\sigma_1 + \sigma_2)} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right)^2 + \lambda_4 \left(e^{i\sigma_2 - 2i\sigma_1} \hat{w}_2^3 X - 3e^{-i\sigma_2} \hat{w}_1^2 \hat{w}_2 X \right) + \lambda_7 X^2 \left(e^{-2i\sigma_2} \hat{w}_1^2 + e^{-2i\sigma_1} \hat{w}_2^2 \right) \Biggr\},$$
(C.2.4p)

$$Z_{2333} = \frac{1}{N_2 N_3^3} \Biggl\{ -4i\lambda_2 e^{-i(\sigma_1+\sigma_2)} \hat{w}_1 \hat{w}_2 s_{\sigma_1-\sigma_2} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right) -4i\lambda_3 e^{-i(\sigma_1+\sigma_2)} \hat{w}_1 \hat{w}_2 s_{\sigma_1-\sigma_2} \left(e^{2i\sigma_1} \hat{w}_1^2 + e^{2i\sigma_2} \hat{w}_2^2 \right) +\lambda_4 \hat{w}_1 X \Biggl[\left(\hat{w}_1^2 - 2\hat{w}_2^2 \right) c_{2\sigma_1-\sigma_2} + \left(2\hat{w}_1^2 - 7\hat{w}_2^2 \right) c_{\sigma_2} +i \left(\left(\hat{w}_2^2 - 2\hat{w}_1^2 \right) s_{\sigma_2} + \left(\hat{w}_1^2 + 2\hat{w}_2^2 \right) s_{2\sigma_1-\sigma_2} \right) \Biggr] +2\lambda_7 \left(e^{-2i\sigma_1} - e^{-2i\sigma_2} \right) \hat{w}_1 \hat{w}_2 X^2 \Biggr\}.$$
(C.2.4q)

Bibliography

- [1] G. Aad <u>et al.</u>, "Observation of a new particle in the search for the Standard Model Higgs boson with the ATLAS detector at the LHC," Phys. Lett., vol. B716, pp. 1–29, 2012.
- [2] S. Chatrchyan <u>et al.</u>, "Observation of a new boson at a mass of 125 GeV with the CMS experiment at the LHC," Phys. Lett., vol. B716, pp. 30–61, 2012.
- [3] G. Aad <u>et al.</u>, "Combined Measurement of the Higgs Boson Mass in pp Collisions at $\sqrt{s} = 7$ and 8 TeV with the ATLAS and CMS Experiments," <u>Phys. Rev. Lett.</u>, vol. 114, p. 191803, 2015.
- [4] G. Aad <u>et al.</u>, "Measurements of the Higgs boson production and decay rates and constraints on its couplings from a combined ATLAS and CMS analysis of the LHC pp collision data at $\sqrt{s} = 7$ and 8 TeV," JHEP, vol. 08, p. 045, 2016.
- [5] F. Englert and R. Brout, "Broken Symmetry and the Mass of Gauge Vector Mesons," <u>Phys.</u> Rev. Lett., vol. 13, pp. 321–323, 1964.
- [6] P. W. Higgs, "Broken symmetries, massless particles and gauge fields," <u>Phys. Lett.</u>, vol. 12, pp. 132–133, 1964.
- [7] P. W. Higgs, "Broken Symmetries and the Masses of Gauge Bosons," <u>Phys. Rev. Lett.</u>, vol. 13, pp. 508–509, 1964.
- [8] G. S. Guralnik, C. R. Hagen, and T. W. B. Kibble, "Global Conservation Laws and Massless Particles," Phys. Rev. Lett., vol. 13, pp. 585–587, 1964.
- T. D. Lee, "CP Nonconservation and Spontaneous Symmetry Breaking," <u>Phys. Rept.</u>, vol. 9, pp. 143–177, 1974.
- [10] R. D. Peccei and H. R. Quinn, "CP Conservation in the Presence of Instantons," <u>Phys. Rev.</u> Lett., vol. 38, pp. 1440–1443, 1977.
- [11] K. Inoue, A. Kakuto, H. Komatsu, and S. Takeshita, "Low-Energy Parameters and Particle Masses in a Supersymmetric Grand Unified Model," <u>Prog. Theor. Phys.</u>, vol. 67, p. 1889, 1982.
- [12] J. F. Gunion, S. Dawson, H. E. Haber, and G. L. Kane, <u>The Higgs hunter's guide</u>, vol. 80. Upton, NY: Brookhaven Nat. Lab., 1989.
- [13] G.C. Branco, L. Lavoura and J.P. Silva, CP Violation (Oxford University Press, Oxford, England, 1999)
- [14] H. E. Haber and D. O'Neil, "Basis-independent methods for the two-Higgs-doublet model. II. The Significance of $\tan\beta$," <u>Phys. Rev.</u>, vol. D74, p. 015018, 2006. [Erratum: Phys. Rev.D74,no.5,059905(2006)].
- [15] G. C. Branco, P. M. Ferreira, L. Lavoura, M. N. Rebelo, M. Sher, and J. P. Silva, "Theory and phenomenology of two-Higgs-doublet models," Phys. Rept., vol. 516, pp. 1–102, 2012.
- [16] N. G. Deshpande and E. Ma, "Pattern of Symmetry Breaking with Two Higgs Doublets," Phys. Rev., vol. D18, p. 2574, 1978.

- [17] R. Barbieri, L. J. Hall, and V. S. Rychkov, "Improved naturalness with a heavy Higgs: An Alternative road to LHC physics," Phys. Rev., vol. D74, p. 015007, 2006.
- [18] L. Lopez Honorez, E. Nezri, J. F. Oliver, and M. H. G. Tytgat, "The Inert Doublet Model: An Archetype for Dark Matter," JCAP, vol. 0702, p. 028, 2007.
- [19] I. F. Ginzburg, K. A. Kanishev, M. Krawczyk, and D. Sokolowska, "Evolution of Universe to the present inert phase," Phys. Rev., vol. D82, p. 123533, 2010.
- [20] B. Grzadkowski, O. M. Ogreid, P. Osland, A. Pukhov, and M. Purmohammadi, "Exploring the CP-Violating Inert-Doublet Model," JHEP, vol. 06, p. 003, 2011.
- [21] V. Keus, S. F. King, S. Moretti, and D. Sokolowska, "Dark Matter with Two Inert Doublets plus One Higgs Doublet," JHEP, vol. 11, p. 016, 2014.
- [22] A. Cordero, J. Hernandez-Sanchez, V. Keus, S. F. King, S. Moretti, D. Rojas, and D. Sokolowska, "Dark Matter Signals at the LHC from a 3HDM," <u>JHEP</u>, vol. 05, p. 030, 2018.
- [23] A. Cordero-Cid, J. Hernández-Sánchez, Keus, S. Moretti, D. Rojas, and D. Sokołowska, "Lepton collider signatures of dark CP violation," 2018.
- [24] V. Silveira and A. Zee, "Scalar Phantoms," Phys. Lett., vol. 161B, pp. 136–140, 1985.
- [25] J. McDonald, "Gauge singlet scalars as cold dark matter," <u>Phys. Rev.</u>, vol. D50, pp. 3637–3649, 1994.
- [26] C. P. Burgess, M. Pospelov, and T. ter Veldhuis, "The Minimal model of nonbaryonic dark matter: A Singlet scalar," Nucl. Phys., vol. B619, pp. 709–728, 2001.
- [27] P. Athron <u>et al.</u>, "Status of the scalar singlet dark matter model," <u>Eur. Phys. J.</u>, vol. C77, no. 8, p. 568, 2017.
- [28] D. A. Ross and M. J. G. Veltman, "Neutral Currents in Neutrino Experiments," <u>Nucl. Phys.</u>, vol. B95, pp. 135–147, 1975.
- [29] M. Tanabashi et al., "Review of Particle Physics," Phys. Rev., vol. D98, no. 3, p. 030001, 2018.
- [30] K. Olaussen, P. Osland, and M. A. Solberg, "Symmetry and Mass Degeneration in Multi-Higgs-Doublet Models," JHEP, vol. 07, p. 020, 2011.
- [31] I. P. Ivanov and C. C. Nishi, "Symmetry breaking patterns in 3HDM," <u>JHEP</u>, vol. 01, p. 021, 2015.
- [32] S. Pakvasa and H. Sugawara, "Discrete Symmetry and Cabibbo Angle," <u>Phys. Lett.</u>, vol. 73B, pp. 61–64, 1978.
- [33] E. Derman, "Flavor Unification, τ Decay and b Decay Within the Six Quark Six Lepton Weinberg-Salam Model," Phys. Rev., vol. D19, pp. 317–329, 1979.
- [34] D. Emmanuel-Costa, O. M. Ogreid, P. Osland, and M. N. Rebelo, "Spontaneous symmetry breaking in the S_3 -symmetric scalar sector," <u>JHEP</u>, vol. 02, p. 154, 2016. [Erratum: JHEP08,169(2016)].
- [35] J. Kubo, H. Okada, and F. Sakamaki, "Higgs potential in minimal S_3 invariant extension of the standard model," Phys. Rev., vol. D70, p. 036007, 2004.
- [36] T. Teshima, "Higgs potential in S₃ invariant model for quark/lepton mass and mixing," <u>Phys.</u> Rev., vol. D85, p. 105013, 2012.

- [37] D. Das and U. K. Dey, "Analysis of an extended scalar sector with S_3 symmetry," <u>Phys. Rev.</u>, vol. D89, no. 9, p. 095025, 2014. [Erratum: Phys. Rev.D91,no.3,039905(2015)].
- [38] E. Derman and H.-S. Tsao, "SU(2) X U(1) X S(n) Flavor Dynamics and a Bound on the Number of Flavors," Phys. Rev., vol. D20, p. 1207, 1979.
- [39] J. L. Lavoie, "The mth power of an n × n matrix and the bell polynomials," <u>SIAM Journal</u> on Applied Mathematics, vol. 29, no. 3, pp. 511–514, 1975.
- [40] H.-H. Zhang, W.-B. Yan, and X.-S. Li, "Trace Formulae of Characteristic Polynomial and Cayley-Hamilton's Theorem, and Applications to Chiral Perturbation Theory and General Relativity," Commun. Theor. Phys., vol. 49, p. 801, 2008.
- [41] Y. Nambu, "Quasiparticles and Gauge Invariance in the Theory of Superconductivity," <u>Phys.</u> Rev., vol. 117, pp. 648–663, 1960.
- [42] J. Goldstone, "Field Theories with Superconductor Solutions," <u>Nuovo Cim.</u>, vol. 19, pp. 154–164, 1961.
- [43] J. Goldstone, A. Salam, and S. Weinberg, "Broken Symmetries," <u>Phys. Rev.</u>, vol. 127, pp. 965–970, 1962.
- [44] J. F. Donoghue and L. F. Li, "Properties of Charged Higgs Bosons," <u>Phys. Rev.</u>, vol. D19, p. 945, 1979.
- [45] H. Georgi and D. V. Nanopoulos, "Suppression of Flavor Changing Effects From Neutral Spinless Meson Exchange in Gauge Theories," Phys. Lett., vol. 82B, pp. 95–96, 1979.
- [46] F. J. Botella and J. P. Silva, "Jarlskog like invariants for theories with scalars and fermions," Phys. Rev., vol. D51, pp. 3870–3875, 1995.
- [47] M. P. Bento, H. E. Haber, J. C. Romão, and J. P. Silva, "Multi-Higgs doublet models: physical parametrization, sum rules and unitarity bounds," JHEP, vol. 11, p. 095, 2017.
- [48] O. M. Ogreid, P. Osland, and M. N. Rebelo, "A Simple Method to detect spontaneous CP Violation in multi-Higgs models," JHEP, vol. 08, p. 005, 2017.
- [49] D. Emmanuel-Costa, O. M. Ogreid, P. Osland, and M. N. Rebelo, "Spontaneous symmetry breaking in three-Higgs-doublet S₃-symmetric models," <u>J. Phys. Conf. Ser.</u>, vol. 873, no. 1, p. 012007, 2017.
- [50] T. D. Lee, "A Theory of Spontaneous T Violation," Phys. Rev., vol. D8, pp. 1226–1239, 1973.
- [51] G. C. Branco, J. M. Gerard, and W. Grimus, "Geometrical T Violation," <u>Phys. Lett.</u>, vol. 136B, pp. 383–386, 1984.
- [52] G. C. Branco and M. N. Rebelo, "The Higgs Mass in a Model With Two Scalar Doublets and Spontaneous CP Violation," Phys. Lett., vol. 160B, pp. 117–120, 1985.
- [53] D. Cogollo and J. P. Silva, "Two Higgs doublet models with an S_3 symmetry," <u>Phys. Rev.</u>, vol. D93, no. 9, p. 095024, 2016.
- [54] C. Jarlskog, "Commutator of the Quark Mass Matrices in the Standard Electroweak Model and a Measure of Maximal CP Violation," Phys. Rev. Lett., vol. 55, p. 1039, 1985.
- [55] I. Dunietz, O. W. Greenberg, and D.-d. Wu, "A Priori Definition of Maximal CP Violation," Phys. Rev. Lett., vol. 55, p. 2935, 1985.
- [56] J. Bernabeu, G. C. Branco, and M. Gronau, "CP Restrictions on Quark Mass Matrices," <u>Phys.</u> <u>Lett.</u>, vol. 169B, pp. 243–247, 1986.

- [57] J. C. Romao and J. P. Silva, "A resource for signs and Feynman diagrams of the Standard Model," Int. J. Mod. Phys., vol. A27, p. 1230025, 2012.
- [58] C.-Y. Chen, Q.-S. Yan, X. Zhao, Y.-M. Zhong, and Z. Zhao, "Probing triple-Higgs productions via 4b2γ decay channel at a 100 TeV hadron collider," <u>Phys. Rev.</u>, vol. D93, no. 1, p. 013007, 2016.
- [59] G. Degrassi, M. Fedele, and P. P. Giardino, "Constraints on the trilinear Higgs self coupling from precision observables," JHEP, vol. 04, p. 155, 2017.
- [60] D. Jurčiukonis and L. Lavoura, "The three- and four-Higgs couplings in the general two-Higgsdoublet model," JHEP, vol. 12, p. 004, 2018.
- [61] M. Sher, "Electroweak Higgs Potentials and Vacuum Stability," <u>Phys. Rept.</u>, vol. 179, pp. 273–418, 1989.
- [62] A. W. El Kaffas, W. Khater, O. M. Ogreid, and P. Osland, "Consistency of the two Higgs doublet model and CP violation in top production at the LHC," <u>Nucl. Phys.</u>, vol. B775, pp. 45–77, 2007.
- [63] B. Grzadkowski, O. M. Ogreid, and P. Osland, "Natural Multi-Higgs Model with Dark Matter and CP Violation," Phys. Rev., vol. D80, p. 055013, 2009.
- [64] M. Maniatis and O. Nachtmann, "Stability and symmetry breaking in the general three-Higgsdoublet model," JHEP, vol. 02, p. 058, 2015. [Erratum: JHEP10,149(2015)].
- [65] U. Nierste and K. Riesselmann, "Higgs sector renormalization group in the MS and OMS scheme: The Breakdown of perturbation theory for a heavy Higgs," <u>Phys. Rev.</u>, vol. D53, pp. 6638–6652, 1996.
- [66] B. W. Lee, C. Quigg, and H. B. Thacker, "Weak Interactions at Very High-Energies: The Role of the Higgs Boson Mass," Phys. Rev., vol. D16, p. 1519, 1977.
- [67] M. Luscher and P. Weisz, "Is There a Strong Interaction Sector in the Standard Lattice Higgs Model?," Phys. Lett., vol. B212, pp. 472–478, 1988.
- [68] W. J. Marciano, G. Valencia, and S. Willenbrock, "Renormalization Group Improved Unitarity Bounds on the Higgs Boson and Top Quark Masses," Phys. Rev., vol. D40, p. 1725, 1989.
- [69] I. F. Ginzburg and I. P. Ivanov, "Tree-level unitarity constraints in the most general 2HDM," Phys. Rev., vol. D72, p. 115010, 2005.
- [70] S. Kanemura and K. Yagyu, "Unitarity bound in the most general two Higgs doublet model," Phys. Lett., vol. B751, pp. 289–296, 2015.
- [71] J. M. Cornwall, D. N. Levin, and G. Tiktopoulos, "Derivation of Gauge Invariance from High-Energy Unitarity Bounds on the s Matrix," <u>Phys. Rev.</u>, vol. D10, p. 1145, 1974. [Erratum: Phys. Rev.D11,972(1975)].
- [72] C. E. Vayonakis, "Born Helicity Amplitudes and Cross-Sections in Nonabelian Gauge Theories," Lett. Nuovo Cim., vol. 17, p. 383, 1976.
- [73] M. Jacob and G. C. Wick, "On the general theory of collisions for particles with spin," <u>Annals</u> Phys., vol. 7, pp. 404–428, 1959. [Annals Phys.281,774(2000)].
- [74] M. E. Peskin and T. Takeuchi, "A New constraint on a strongly interacting Higgs sector," Phys. Rev. Lett., vol. 65, pp. 964–967, 1990.
- [75] M. E. Peskin and T. Takeuchi, "Estimation of oblique electroweak corrections," <u>Phys. Rev.</u>, vol. D46, pp. 381–409, 1992.

- [76] M. Baak, J. Cúth, J. Haller, A. Hoecker, R. Kogler, K. Mönig, M. Schott, and J. Stelzer, "The global electroweak fit at NNLO and prospects for the LHC and ILC," <u>Eur. Phys. J.</u>, vol. C74, p. 3046, 2014.
- [77] J. Haller, A. Hoecker, R. Kogler, K. Mönig, T. Peiffer, and J. Stelzer, "Update of the global electroweak fit and constraints on two-Higgs-doublet models," <u>Eur. Phys. J.</u>, vol. C78, no. 8, p. 675, 2018.
- [78] W. Grimus, L. Lavoura, O. M. Ogreid, and P. Osland, "A Precision constraint on multi-Higgsdoublet models," J. Phys., vol. G35, p. 075001, 2008.
- [79] W. Grimus, L. Lavoura, O. M. Ogreid, and P. Osland, "The Oblique parameters in multi-Higgs-doublet models," Nucl. Phys., vol. B801, pp. 81–96, 2008.
- [80] M. J. G. Veltman, "Limit on Mass Differences in the Weinberg Model," <u>Nucl. Phys.</u>, vol. B123, pp. 89–99, 1977.
- [81] H. E. Haber and D. O'Neil, "Basis-independent methods for the two-Higgs-doublet model III: The CP-conserving limit, custodial symmetry, and the oblique parameters S, T, U," <u>Phys.</u> Rev., vol. D83, p. 055017, 2011.
- [82] G. Belanger, F. Boudjema, A. Pukhov, and A. Semenov, "Dark matter direct detection rate in a generic model with micrOMEGAs 2.2," <u>Comput. Phys. Commun.</u>, vol. 180, pp. 747–767, 2009.
- [83] G. Belanger, F. Boudjema, A. Pukhov, and A. Semenov, "micrOMEGAs 3: A program for calculating dark matter observables," Comput. Phys. Commun., vol. 185, pp. 960–985, 2014.
- [84] D. Barducci, G. Belanger, J. Bernon, F. Boudjema, J. Da Silva, S. Kraml, U. Laa, and A. Pukhov, "Collider limits on new physics within micrOMEGAs 4.3," <u>Comput. Phys.</u> Commun., vol. 222, pp. 327–338, 2018.
- [85] F. Staub, "From Superpotential to Model Files for FeynArts and CalcHep/CompHep," Comput. Phys. Commun., vol. 181, pp. 1077–1086, 2010.
- [86] F. Staub, "SARAH 4 : A tool for (not only SUSY) model builders," <u>Comput. Phys. Commun.</u>, vol. 185, pp. 1773–1790, 2014.
- [87] A. Belyaev, N. D. Christensen, and A. Pukhov, "CalcHEP 3.4 for collider physics within and beyond the Standard Model," Comput. Phys. Commun., vol. 184, pp. 1729–1769, 2013.
- [88] N. Aghanim et al., "Planck 2018 results. VI. Cosmological parameters," 2018.
- [89] G. Hinshaw et al., "Nine-Year Wilkinson Microwave Anisotropy Probe (WMAP) Observations: Cosmological Parameter Results," Astrophys. J. Suppl., vol. 208, p. 19, 2013.
- [90] P. Bechtle, O. Brein, S. Heinemeyer, G. Weiglein, and K. E. Williams, "HiggsBounds 2.0.0: Confronting Neutral and Charged Higgs Sector Predictions with Exclusion Bounds from LEP and the Tevatron," Comput. Phys. Commun., vol. 182, pp. 2605–2631, 2011.
- [91] P. Bechtle, O. Brein, S. Heinemeyer, O. Stål, T. Stefaniak, G. Weiglein, and K. E. Williams, "HiggsBounds – 4: Improved Tests of Extended Higgs Sectors against Exclusion Bounds from LEP, the Tevatron and the LHC," Eur. Phys. J., vol. C74, no. 3, p. 2693, 2014.
- [92] P. Bechtle, S. Heinemeyer, O. Stal, T. Stefaniak, and G. Weiglein, "Applying Exclusion Likelihoods from LHC Searches to Extended Higgs Sectors," <u>Eur. Phys. J.</u>, vol. C75, no. 9, p. 421, 2015.